

Title	Superconformal Quantum Mechanics from M2-branes
Author(s)	岡崎, 匡志
Citation	大阪大学, 2015, 博士論文
Version Type	VoR
URL	https://doi.org/10.18910/52316
rights	
Note	

The University of Osaka Institutional Knowledge Archive : OUKA

https://ir.library.osaka-u.ac.jp/

The University of Osaka

Superconformal Quantum Mechanics from M2-branes

Tadashi Okazaki

Department of Physics, Graduate School of Science Osaka University 1-1 Machikaneyama-cho Toyonaka, Osaka 560-0043, Japan

Abstract

In this thesis we discuss the superconformal quantum mechanics arising from the M2-branes. We begin with a comprehensive review of the superconformal quantum mechanics and emphasize that conformal symmetry and supersymmetry in quantum mechanics contain a number of exotic and enlightening properties which do not occur in higher dimensional field theories. We see that superfield and superspace formalism is available for $\mathcal{N} \leq 8$ superconformal mechanical models. We then discuss the M2-branes with a focus on the world-volume descriptions of the multiple M2-branes which are superconformal three-dimensional Chern-Simons matter theories. Finally we argue that the two topics are connected in M-theoretical construction by considering the multiple M2-branes wrapped around a compact Riemann surface and study the emerging IR quantum mechanics. We establish that the resulting quantum mechanics realizes a set of novel $\mathcal{N} \geq 8$ superconformal quantum mechanical models which have not been reached so far. Also we discuss possible applications of the superconformal quantum mechanics to mathematical physics.

Contents

1	Intr	oduction and Overview	6		
	1.1	Historical background	6		
	1.2	What I did	13		
Ι	Su	perconformal Mechanics	15		
2	Con	Conformal Mechanics			
	2.1	$SL(2,\mathbb{R})$ conformal symmetry	16		
	2.2	Spectrum	21		
	2.3	Time evolution	30		
	2.4	Operator	35		
	2.5	Gauged conformal mechanics	40		
	2.6	Black hole	46		
	2.7	Non-linear realization	51		
	2.8	Multi-particle conformal mechanics	57		
	2.9	Calogero model	59		
3	Sup	Superconformal Mechanics			
	3.1	Superalgebra and supergroup	61		
		3.1.1 Lie superalgebra	62		
		3.1.2 Lie supergroup	63		
		3.1.3 Superconformal algebra	66		
3.2 One-dimensional supersymmetry		One-dimensional supersymmetry	70		
		3.2.1 Supermultiplet	70		
		3.2.2 Automorphic duality	73		
	3.3	$\mathcal{N} = 1$ Superconformal mechanics $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	75		
		3.3.1 One particle free action	75		
		3.3.2 Multi-particle model	77		
		3.3.3 Gauged superconformal mechanics	79		

3.4	$\mathcal{N}=2$	2 Superconformal mechanics	82
	3.4.1	One particle model	82
	3.4.2	Multi-particle model	92
	3.4.3	Freedman-Mende model	95
	3.4.4	Gauged superconformal mechanics	96
3.5	$\mathcal{N}=4$	Superconformal mechanics	99
	3.5.1	(4, 4, 0) supermultiplet	104
	3.5.2	(3, 4, 1) supermultiplet	106
	3.5.3	(2, 4, 2) supermultiplet	108
	3.5.4	(1,4,3) supermultiplet	109
	3.5.5	(0,4,4) supermultiplet	111
	3.5.6	Multi-particle model	112
	3.5.7	Gauged superconformal mechanics	119
3.6	$\mathcal{N}=8$	Superconformal mechanics	122
	3.6.1	On-shell $SU(1,1 \frac{N}{2})$ action	123
	3.6.2	Superspace and supermultiplet	124
	3.6.3	Multi-particle model	138
II N	/12-brar	nes	139

4	BLC	G-mode	4	140
	4.1	Const	ruction	140
		4.1.1	Lie 3-algebra	142
		4.1.2	Lagrangian	144
		4.1.3	Gauge transformation	145
		4.1.4	Supersymmetry transformation	146
		4.1.5	M2-brane algebra	147
	4.2	\mathcal{A}_4 BL	G-theory	150
		4.2.1	Quiver gauge structure	151
		4.2.2	Superconformal symmetry	153
		4.2.3	Parity invariance	154
		4.2.4	Moduli space	154
5	ABJM-model 1			
	5.1	Const	ruction	159
		5.1.1	Lagrangian	160
		5.1.2	Supersymmetry transformation	161

	5.2	Modu	li space	161
	5.3	Dualit	y between BLG and ABJ(M)	163
II	I S	CQM	from M2-branes	164
6	$\mathcal{N} =$	16 Su	perconformal Mechanics	165
	6.1	$\mathcal{N}=1$	6 gauged quantum mechanics	165
	6.2	Reduc	tion	169
	6.3	OSp(1	.6 2) superconformal mechanics	174
7	$\mathcal{N} =$	12 Su	perconformal Mechanics	178
	7.1	$\mathcal{N}=1$	2 gauged quantum mechanics	178
	7.2	Reduc	tion	181
	7.3	<i>SU</i> (1,	1 6) superconformal mechanics	184
8	Curv	ved Bra	nes and Topological Twisting	188
	8.1	Topolo	ogical twisting	188
		8.1.1	$d = 4$, $\mathcal{N} = 2$ SYM theories $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	188
		8.1.2	$d = 4$, $\mathcal{N} = 2$ SCFT on $C \times \Sigma$	191
		8.1.3	$d = 4$, $\mathcal{N} = 4$ SYM theories $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	199
		8.1.4	$d = 4$, $\mathcal{N} = 4$ SYM theories on $C \times \Sigma$	208
		8.1.5	$d = 3, \mathcal{N} = 4$ SYM theories $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	210
		8.1.6	$d = 3, \mathcal{N} = 8$ SYM theories $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	211
		8.1.7	$d = 3, \mathcal{N} = 8$ SYM theories on $\mathbb{R} \times \Sigma$	213
		8.1.8	$d = 2, \mathcal{N} = 8$ SYM theories	215
		8.1.9	$d = 3, \mathcal{N} = 4$ Chern-Simons matter theories $\ldots \ldots \ldots$	217
		8.1.10	$d = 3, \mathcal{N} = 5$ Chern-Simons matter theories	223
		8.1.11	$d = 3, \mathcal{N} = 6$ Chern-Simons matter theories	224
		8.1.12	$d = 3, \mathcal{N} = 8$ Chern-Simons matter theories	224
	_	8.1.13	$d = 2, \mathcal{N} = (2, 2)$ non-linear sigma-model	228
	8.2	Curve	d branes and twisted theories	231
		8.2.1	D ₃ -branes and twisted $d = 4$, $\mathcal{N} = 4$ SYM theories	234
		8.2.2	D2-branes and twisted $d = 3$, $\mathcal{N} = 8$ SYM theories	235
		8.2.3	D2-branes and twisted $d = 3$, $\mathcal{N} = 8$ SYM theories on $\mathbb{R} \times \Sigma$	236
		8.2.4	Relationship between $d = 4$ and $d = 3$ twists	237
		8.2.5	D1-branes and twisted $d = 2$, $\mathcal{N} = 8$ SYM theories	239
		8.2.6	M2-branes and twisted BLG theory	240

9	Curved M2-branes and Topological Twisting			
	9.1	M2-branes wrapping a holomorphic curve	242	
	9.2	Supersymmetry in Calabi-Yau space	243	
	9.3	Calibration and supersymmetric cycle	245	
	9.4	Topological twisting	249	
10	10 SCQM from M2-branes in a K3 surface			
	10.1	K ₃ twisting	252	
	10.2	Twisted theory	256	
	10.3	Derivation of quantum mechanics	259	
11	Con	clusion and Discussion	266	
	11.1 Conclusion		266	
	11.2 Future directions			
		11.2.1 AdS_2/CFT_1 correspondence	268	
		11.2.2 Indices and the reduced Gromov-Witten invariants	269	
		11.2.3 1d-2d relation	270	

Acknowledgments

I would like to acknowledge kind support and help with deep appreciation and gratitude to many people. I owe a debt of appreciation to Hirosi Ooguri for continuous guidance during my days at Caltech over the past two and a half years. It goes without saying that my study and experiences at Caltech would not be possible if were not for his arrangement. Moreover this thesis would not have been completed without his suggestion and assistance. I cannot thank him enough for all that he has done for me. I was really happy to study at Caltech guided by whom I respect. I would like to express my gratitude to Yutaka Hosotani for always helping me to study at Caltech and consulting me when I am in trouble. His advice was always a useful resource for me. I am profoundly grateful to Satoshi Yamaguchi for continuing encouragement and innumerable discussions. Over the past five years, particularly in the first two and a half years I could learn many things through collaboration works and countless times of long hours of discussions with him. I am glad that I was able to start my study of string theory with him at Osaka University. It was my indelible memory to talk about physics and life with brilliant members and alumni at Caltech including, Hee-Jong Chung, Abhijit Gadde, Siqi He, Koji Ishiwata, Hyungrok Kim, Kazunobu Maruyoshi, Elena Murchikova, Yu Nakayama, Chan Youn Park, Ke Ye, Du Pei, Pavel Putrov and Wenbin Yan. I extend my attitude to my colleagues at Osaka University including Tetsuya Enomoto, Tomohiro Horita, Takuya Shimotani, Akinori Tanaka, Akio Tomiya. The study meeting continuing for a long stretch of time with them is a good memory now. I wish to thank Greg Marlowe, the house owner for sharing with me a nice place to live at Pasadena and for always encouraging me. I was often relaxed myself by talking with him. I would like to thank Carol Silberstein, the secretary at Caltech and Kumiko Nakayama, Keiko Takeda, the secretaries at Osaka University for arranging a lot of procedures so that I could do my research at Caltech. They are always kind and I was helped many times. Finally I am grateful to my parents for giving constant support to me.

Chapter 1

Introduction and Overview

1.1 Historical background

The pursue of Theory of Everything that describes our nature and achieves unification of all fundamental interactions, i.e. electroweak, strong and gravitational interactions has been a mission agitating theoretical physicists. Historically much of significant developments in theoretical physics were achieved by overcoming the inconsistency between existing concepts. The special relativity was established from the crisis between classical mechanics and electrodynamics, the general relativity was proposed by reconciling the special relativity and Newtonian gravity and quantum field theory was acquired by combining quantum mechanics and the special relativity. However, we are now confronting another contradiction between quantum field theory and the general relativity. Although a quantum field theory as the standard model successfully describes and predicts almost all phenomena controlled by electroweak and strong forces, the general relativity describing the gravity that is the remaining fundamental interaction seriously disagrees with the quantum field theory. This indicates that quantum field theory cannot lead to the correct quantization of gravity. Therefore it is expected that the standard model is regarded as the low-energy effective theory of a more fundamental theory.

String theory has been proposed as the promising candidate for such a fundamental theory since it can naturally describe all fundamental interactions. In string theory all particles are recognized as various vibrational modes of only two different types of fundamental strings; the open strings which have two endpoints and the the closed strings which have no endpoint. One of the most beautiful structures in string theory is that Yang-Mills gauge theories which govern the electroweak and strong interactions as the standard model arise from the vibrations of the open strings while the general relativity that describes the gravitational interaction appears from the vibrations of the closed string. Among the massless states of the open string there are spin-one particles which behave as gauge bosons while among the massless states of the closed string there is a spin-two particle which can be identified with graviton [1, 2].

However, the bosonic string theory is not realistic since the vibrations of the bosonic strings yield only bosonic particles. The lack of fermionic particles can be resolved by introducing supersymmetry in string theory, i.e. the superstring theory. The spectrums of superstrings contain both bosonic and fermionic particles. Hence string theory supports the existence of the supersymmetry.

One of the most fascinating predictions in superstring theory is the existence of the extra dimension in space-time. It turns out that the unitarity and the Lorentz invariance of space-time in which the superstrings live are guaranteed only for tendimensional space-time. In other words, flat space superstrings can only exist in ten dimensions. In order to reconcile the difference between the ten-dimensional space-time in string theory and the four-dimensional space-time in our instinctive knowing physics, the notion of compactification has been proposed. The idea is that since the six extra dimensional compact spatial directions are much smaller than the four-dimensional space-time, the original (1+9)-dimensional space-time effectively looks like (1+3)-dimensional space-time. For the six-dimensional spaces Calabi-Yau manifolds are known to possess phenomenologically promising properties.

Ten-dimensional superstring theory is not a single theory but rather a set of possible five theories; (i) type IIA, (ii) type IIB, (iii) type I (iv) SO(32) heterotic (v) $E_8 \times E_8$ heterotic. When the both left-moving and right-moving modes are taken as superstrings, there are two possibilities; opposite handedness or the same handedness. The former theory is called type IIA while the latter is called type IIB. Type I superstring theory is obtained by the orientifold projection that mods out left-right symmetry of type IIB superstring theory. When the left-moving mode is chosen as the bosonic string and the right-moving mode is taken as the superstring, consistency allows only two different theories; SO(32) heterotic and $E_8 \times E_8$ heterotic superstring theories. The type II superstring theories have d = 10, $\mathcal{N} = 2$ supersymmetry and the type I and heterotic superstring theories possess d = 10, $\mathcal{N} = 1$ supersymmetry ¹.

It has been argued that the five superstring theories are connected with each other. T-duality relates a pair of the two type II superstring theories and also a pair

¹For d = 10, $\mathcal{N} = 1$ supersymmetry a consistent local gauge symmetry group is characterized by the Lie algebras $\mathfrak{so}(32)$ and $E_8 \times E_8$.

of the two heterotic superstring theories. S-duality relates the type I superstring theory to the SO(32) heterotic superstring theory and the type IIB superstring theory to itself. T-dualities and S-dualities generates a discrete non-abelian group, the so-called U-duality group [3, 4]. Remarkably the U-duality groups are recognized as discretization of global symmetry groups of supergravity. In fact it is known that type IIA and IIB superstring theories are the ultraviolet (UV) completions of d = 10 type IIA and IIB supergravities ² whereas type I and heterotic superstring theories are the UV completions of d = 10 type I supergravities.

From the supergravity point of view, it is interesting to note that d = 10 type IIA supergravity arises by dimensional reduction of d = 11 supergravity [8, 9, 10]. d = 11 supergravity [11] is furnished with a particular interest since d = 11 is the highest space-time dimension which realizes a consistent supersymmetric theory containing particles with spins ≥ 2 [12]. d = 11 supergravity possesses a single 32-component spinor supercharge and its Lagrangian is unique if we require that the theory contains at most two-derivative interactions. Therefore the relation between superstring theory and d = 10 supergravity indicates the existence of the UV completion of d = 11 supergravity. It has been argued that in the strong string coupling limit an eleventh direction arises in type IIA superstring theory and the resulting eleven-dimensional theory is referred to as M-theory [13, 4, 14]³. Conversely M-theory reduces to type IIA superstring theory upon the compactification on a spatial circle. Up until now M-theory is the most prospective candidate for the fundamental theory in that it may explain the origin of strings and unify the five superstring theory. However, a familiar perturbative method in string theory is not applicable because M-theory describes the strong coupling region of string theory.

As the fundamental string is a fundamental object in ten-dimensional superstring theory, the membrane appears to play a fundamental role in M-theory. This membrane is called M2-brane. Indeed d = 11 supergravity contains a three-form gauge field, which leads to two stable extended objects as solitonic solutions; electric membrane and magnetic five-brane. Moreover it has been pointed out [15] that the M2-brane is identified with the fundamental string when M-theory is compactified on a circle and reduces to type IIA superstring theory. In spite of the prospective importance for the membranes in M-theory a number of attempts for the quantization of the membranes does not work well hitherto.

²Originally type IIB supergravity was discovered [5] and constructed [6, 7] as the low-energy limit of type IIB superstring theory.

³The letter "M" proposed by E. Witten embodies several possible meanings; membrane, matrix, magic and mother.

Although M-theory is much less understood than string theory due to the difficulty of the quantization of the membranes, we can still obtain several insights and clues from string theory and supergravity. In addition to the fundamental string, string theory also contains extended objects, the so-called D-branes on which open strings can end [16]. In fact ten-dimensional supergravities possess the solutions describing the geometries around such branes. There is a remarkable conjecture, the so-called AdS/CFT correspondence [17, 18, 19] which states that there is the correspondence between string/M-theory on certain supergravity geometries with anti-de-Sitter (AdS) factors and certain conformally invariant quantum field theories. The most basic evidence for the AdS/CFT correspondence is the equivalence between type IIB superstring theory on the $AdS_5 \times S^5$ supergravity geometry constructed as a set of N coincident D₃-branes and the d = 4, $\mathcal{N} = 4$ superconformal U(N) Yang-Mills gauge theory. Namely the low-energy dynamics for the worldvolume of the planar N D₃-branes in flat space-time can be effectively described by (1+3)-dimensional maximally supersymmetric U(N) Yang-Mills gauge theory [20].

Similarly, in the near-horizon limit d = 11 supergravity solutions describing planar M2-branes in flat space-time contain the AdS₄ factors and the low-energy dynamics of the M2-branes are expected to be described by the (1 + 2)-dimensional conformal field theories. As the eleven-dimensional flat background geometry can possess 32 space-time supercharges, the world-volume effective field theory of planar M2-branes should preserve half of the supersymmetry. Also the gauge degrees of freedom are needed to describe the internal degrees of freedom of multiple M2-branes. Thus the low-energy effective field theories of planar M2-branes are d = 3, $\mathcal{N} = 8$ superconformal gauge theories. The candidates for such effective field theories of world-volume dynamics of multiple M2-branes have been proposed as three-dimensional superconformal Chern-Simons matter theories, the so-called BLG-model [21, 22, 23, 24, 25] and the ABJM model [26].

In order to obtain new AdS/CFT examples it is desirable to consider more general supergravity solutions describing the wrapped branes around certain cycles which may be curved. However, in the generic setup where the branes are wrapping an arbitrary manifold, all of the supersymmetries are destined to break down. In other words, specific background geometries of branes and specific cycles wrapped by branes must be chosen to preserve supersymmetry. Mathematically the supersymmetric cycles are characterized by the calibration [27]. There is a remarkable observation [28] that topologically twisted field theories may give rise to the world-volume theories of wrapped branes. For the Euclidean D3-branes wrapped was a specific observation [27].

ping four-manifold there are three calibrated cycles embedded in special holonomy manifolds; (i) special Lagrangian submanifold in Calabi-Yau four-fold, (ii) coassociative submanifold in G_2 manifold and (iii) Cayley submanifold in Spin(7)manifold. Each of them corresponds to three distinct topological twisting procedures ⁴; (i) geometric Langlands (GL) twist [33, 34, 35, 36], (ii) Vafa-Witten twist [37] and (iii) Donaldson-Witten twist [38]. The world-volume of D3-branes can be put on the product of two Riemann surfaces $C \times \Sigma_g$. For the compact Riemann surface Σ_g of genus g > 1 the field theories on the D3-branes are partially twisted on the curved Riemann surface to preserve supersymmetry. Since the compact manifold wrapped by branes introduces into the theory the typical energy scale as its volume, one can consider an additional limit where the energy is much smaller than the inverse size of the cycles. The resulting effective field theories then reduce to the two-dimensional topological sigma-models whose target space is specified by the BPS equations [28, 39, 40, 36, 41, 42, 43, 44].

When the Euclidean M2-branes wrap a compact curved three-manifold, the three-dimensional effective theories on the branes are fully twisted [45]. The SO(3)Euclidean symmetry on the world-volume is topologically twisted in terms of the SO(3) subgroup of the R-symmetry. For the M2-branes wrapping compact Riemann surface Σ_g of genus g supersymmetry is unbroken if the Riemann surface is chosen as holomorphic curve in Calabi-Yau manifold, which are the only known supersymmetric two-cycles. From the supergravity point of view, the solutions which describe the M2-branes wrapping compact Riemann surfaces have been studied [46, 47, 48] by using the gauged supergravity method [40]. The basic observation [49, 50] is that the dimensional reduction of d = 11 supergravity on a seven-sphere can be truncated to give rise to the four-dimensional SO(8)gauged supergravity where SO(8) gauge symmetry corresponds to the isometry of the seven-sphere. Since the planar M2-branes take the form of $AdS_4 \times S^7$, the non-trivial coupling of the external SO(8) gauge field which is nothing but an R-symmetry of the world-volume theory of the planar M2-branes may realize the curved geometries of the form $AdS_2 \times \Sigma_g$ instead of AdS_4 . Thus the uplift of the four-dimensional SO(8) supergravity solutions can be used to construct the d = 11supergravity solutions describing the M2-branes wrapped on holomorphic curves. Correspondingly the three-dimensional effective superconformal field theories are partially twisted for $g \neq 0$ [51]. The SO(2) Euclidean symmetry on the curved Riemann surface is topologically twisted in terms of the SO(2) subgroup of the

⁴Also see [29, 30, 31, 32] for interesting applications of the topological twisted $\mathcal{N} = 4$ super Yang-Mills theories.

R-symmetry. Furthermore it is shown [51] that in the limit where the size of the Riemann surface goes to zero, superconformal quantum mechanical models arise as the low-energy effective theories. This thesis explores the new connection between the M2-branes and the superconformal quantum mechanics.

We should note that conformal symmetry and supersymmetry in one-dimensional field theory, i.e. quantum mechanics, contain a bunch of intriguing properties which do not appear in higher dimensional field theories, as we will discuss in this thesis.

Although supersymmetric quantum mechanics was originally studied as the simple testing model for non-perturbative breaking of supersymmetry [52, 53], supersymmetric quantum mechanics is much more interesting itself. Supersymmetry is closely related to the translational symmetry as the square of the supercharges generates the momentum. However, in one dimension there are no spatial directions and the translational symmetry generator is just the Hamiltonian, which reflects the reduced Poincaré symmetry in one-dimension. The reduced Poincaré symmetry looses the constraints for supersymmetry in one dimension and leads to richer structures than higher dimensional field theories. Indeed there exist supersymmetric quantum mechanical models which cannot reached via naive dimensional reductions from higher dimensional field theories. In parallel with that, there may be a large number of supermultiplets in one-dimension (see Table 3.4) and there is no relationship between the physical bosonic degrees of freedom and fermionic degrees of freedom in supersymmetric quantum mechanics. These properties are special in one dimension.

Conformal symmetry in one-dimension also exhibits unique features. The reduced Poincaré symmetry identifies the generator of a translation with the Hamiltonian *H* and does not allow for the generator of a rotational symmetry. Therefore the one-dimensional conformal symmetry is generated by the Hamiltonian *H*, the dilatation generator *D* and the conformal boost generator *K*, all of which together form the $\mathfrak{sl}(2,\mathbb{R})$ algebra. Therefore the one-dimensional conformal group is $SL(2,\mathbb{R}) \cong SO(1,2)$. The first detailed analysis of conformal mechanics appeared in [54]. The conformal mechanical models are typically characterized by the inverse-square potential ⁵. Inverse-square potential in quantum mechanics is a jewellery box in theoretical physics and mathematics containing black hole physics [65, 66, 67, 68, 69, 70, 71, 72, 73, 74, 75, 76, 77], AdS₂/CFT₁ correspondence [78, 79, 80, 81, 82, 83, 84, 85, 86, 87], QCD [88, 89], quantum Hall effect [90, 91],

⁵The treatment of the inverse-square potential in quantum mechanics was discussed in [55, 56, 57, 58, 59, 60, 61, 62, 63, 64].

Tomonaga-Luttinger liquid [92], string theory [93, 94, 95, 96, 97, 98], spin chains [99, 100, 101, 102, 103, 104, 105, 106], Efimov effect [107, 108, 109, 110], mesoscopic physics [111, 112], quantum dot [113], quantum chaos [114], fractional exclusion statistics [115, 116], random matrix model [117, 118, 119, 120], Seiberg-Witten theory [121, 122], Jack polynomial [123, 124, 125, 126, 127, 128, 129], and relevant algebraic and integrable structures [130, 131, 132, 133]. One of the well-known such quantum mechanical models is the Calogero model [134, 135] which is the multi-particle system with the pairwise inverse-square interaction ⁶. It was firstly proven in [148, 149] that the Calogero model has the $SL(2, \mathbb{R})$ conformal symmetry. The Calogero model and its generalizations can be viewed as a system of free indistinguishable particles [145]. The indistinguishableness implies that the permutation group acting on the configuration of the particles is treated as a discrete gauge symmetry in the system and therefore the Calogero model and its generalized models can be obtained from gauged matrix models [150, 151]. This observation is used to find new conformal mechanical systems by starting gauged matrix models or gauged quantum mechanical models and reducing the systems via Hamiltonian reduction [152, 137].

Since the appearance of the seminal works [153, 154] on superconformal quantum mechanics (SCQM), there has been a great deal of efforts to construct superconformal quantum mechanics. The superconformal quantum mechanical models are characterized by the superconformal group, i.e. the Lie supergroup which contains one-dimensional conformal group $SL(2, \mathbb{R})$ and R-symmetry group as factored bosonic subgroups. One of the most powerful methods to build up superconformal mechanical systems is the superspace and superfield formalism. In fact for $\mathcal{N} \leq 8$ supersymmetric cases it does work and several superconformal quantum mechanical systems are constructed. For $\mathcal{N} = 1$ supersymmetric case, the superconformal group is OSp(1|2) and there is no non-trivial one particle superconformal quantum mechanics. For $\mathcal{N} = 2$ supersymmetric case, the superconformal group is $OSp(2|2) \cong SU(1,1|1)$ and the simplest one particle model is the pioneering work of [153, 154]. For $\mathcal{N} = 4$ supersymmetric case, the generic superconformal group is $D(2,1;\alpha)$ which is a one-parameter family of supergroup. The superspace and superfield formalism keeping track of the exceptional supergroup can be derived by the non-linearlizations technique [155, 156, 157] and several models have been constructed. In the case of $\mathcal{N} = 8$ there exist four different superconformal groups; SU(1,1|4), OSp(8|2), $OSp(4^*|4)$ and F(4). Such several

⁶See [136, 137, 138] for the enlightening reviews on (super)conformal mechanics and also see [139, 140, 141, 142, 143, 144, 145, 146, 147] for excellent reviews on the Calogero model.

choices of superconformal group cannot occur in higher dimensional field theories and thus present a various families of $\mathcal{N} = 8$ superconformal quantum mechanics. However, for $\mathcal{N} > 8$ the superspace and superfield formalism is not unrealistic and unsuccessful. One of the signals for such difficulty is that the number of bosonic and fermionic component fields in the supermultiplets typically becomes greater than the number of supersymmetry when \mathcal{N} is larger than eight (see Table 3.4). In spite of the depression of the superspace and superfield formalism, several $\mathcal{N} > 8$ superconformal quantum mechanical models have been constructed via reduction of the three-dimensional quiver type superconformal Chern-Simons matter theories [51]. As mentioned before, these superconformal quantum mechanical models may capture the low-energy dynamics of the multiple M2-braens wrapped on a compact Riemann surface. We will spell out the details of these superconformal quantum mechanical models in this thesis.

1.2 What I did

The organization of this thesis consists of three parts. In part I and II we will review two main subjects; the superconformal quantum mechanics and the M2-branes. The original part of the author's work based on [51] is part III, in which we will discuss the new connection between the two subjects, that is the superconformal quantum mechanics emerging from M2-branes.

Part I contains two chapters; chapter 2 and 3. In chapter 2 we will discuss various aspects of conformal quantum mechanics. We will start with section 2.1 by studying the DFF-model [54] and its $SL(2, \mathbb{R})$ conformal symmetry and then in section 2.2 we will explore the quantum properties of the system. In section 2.5 we will see that the conformal mechanical models can be derived from the gauged mechanical system via Hamiltonian reduction or Routh reduction. In section 2.6 we will review the observation [65] that in the near horizon of the extreme Reissner-Nordström black hole the motion of the charged particle can be described by the conformal mechanics (2.6.10). In section 2.7 we will present the non-linear realization technique which is useful to construct (super)conformal quantum mechanical models and then review the statement in [74] that DFF-model (2.1.2) is equivalent to the black hole conformal mechanics (2.6.10) in [65]. We will extend the analysis to the multi-particle models, i.e. the sigma-models in section 2.8. We will review the discussion in [69] that the target space of the conformal sigma-model possesses a homothety vector field whose associated one-form is closed. We will argue that the gauging procedure for the multi-particle model, the matrix model

yield the Calogero model in section 2.9. In chapter 3 we will turn to the discussion on the superconformal quantum mechanics. We will recall the Lie superalgebra and Lie supergroup and then discuss the one-dimensional superconformal group (see in Table 3.2) in section 3.1. In section 3.2 we will explain the exisotic structures of supersymmetry in one-dimension, which allows us to construct various supermultiplets.

Part II, which is comprised of two chapters; chapter4 and 5, is devoted to the low-energy effective field theories of the M2-branes. We will review the BLG-model in chapter 4 and the ABJM-model in chapter 5. We will present our notations and conventions and also the several conjectural statements for the BLG-model and the ABJM-model in these chapters.

Part III is the most important part of this thesis. It is based on the author's work of [51], in which we will engage in the superconformal quantum mechanical models arising from the M2-branes. We consider the multiple membranes on a compact Riemann surface and study the IR quantum mechanics by taking the limit where the energy scale is much lower than the inverse size of the Riemann surface. In chapter 6 we will demonstrate that the resulting quantum mechanics from the BLG-model compactified on a torus is the $\mathcal{N} = 16$ superconformal gauged quantum mechanics. Furthermore we will find the OSp(16|2) superconformal quantum mechanics from the reduced system. Similarly in chapter 7 we will investigate the IR quantum mechanics from the ABJM-model compactified on a torus, which turns out to be the $\mathcal{N} = 12$ superconformal gauged quantum mechanics. By the Hamiltonian reduction, or the Routh reduction we will also find the SU(1,1|6) superconformal quantum mechanics from the gauged quantum mechanics. In chapter 8 we will present various examples of the topological twisting, which is the important concept to describe curved branes in string theory and Mtheory. In chapter 9 we will survey the M2-branes wrapped on a curved Riemann surface which is taken as a holomorphic curve in a Calabi-Yau manifold to preserve supersymmetry. We will present a prescription of the topological twisting for the case where the Calabi-Yau space is constructed as the direct sum of the line bundles over the Riemann surface. In chapter 10 we will complete the analysis of the M2-branes wrapped around the holomorphic Riemann surface in a K3 surface. We will find the $\mathcal{N} = 8$ superconformal gauged quantum mechanics which may describe the motion of the two M2-branes wrapping holomorphic curve in a K3 surface. Finally in chapter 11 we will present conclusion and discuss the future directions.

Part I

Superconformal Mechanics

Chapter 2

Conformal Mechanics

In this chapter we will review the conformal quantum mechanics. The simplest model is the so-called DFF model [54]. In section 2.1, 2.2, 2.3 and 2.4 we will learn from the DFF model several remarkable features of the conformal symmetry in one-dimension, which cannot occur in higher dimensional field theories. Then in section 2.5 we will argue the alternative formulation of the conformal mechanical models as the gauged mechanical models. As an interesting application of the conformal mechanics we will discuss the relationship between the conformal mechanics and black hole in section 2.6 and introduce the non-linear realization method to construct (super)conformal quantum mechanics in section 2.7. Finally we will extend the analysis to the multi-particle conformal mechanical models in section 2.8 and 2.9.

2.1 $SL(2,\mathbb{R})$ conformal symmetry

In *d*-dimensions a scale invariant Lagrangian for a scalar field ϕ has the form

$$L = \frac{1}{2} \partial_{\mu} \phi \partial^{\mu} \phi - \gamma \phi^{\frac{2d}{d-2}}$$
(2.1.1)

where γ is a dimensionless coupling constant. In one-dimensional case we get the Lagrangian

$$L = \frac{1}{2} \left(\dot{x}^2 - \frac{\gamma}{x^2} \right). \tag{2.1.2}$$

This simple quantum mechanical model is the so-called DFF-model [54]. To keep particles from falling into the origin, the coupling constant γ should be positive classically. As we will see in the following discussion, quantum mechanically the uncertain principle gives rise to the minimum value $\gamma = -\frac{1}{4}$, however, the

normalizability of the wavefunction of the ground state requires that γ is positive. So we will denote $\gamma = g^2$ for convenience. The Lagrangian (2.1.2) leads to the equation of motion

$$\ddot{x} = \frac{g^2}{x^3}.$$
 (2.1.3)

The action

$$S = \int L \, dt = \frac{1}{2} \int dt \left(\dot{x}^2 - \frac{g^2}{x^2} \right) \tag{2.1.4}$$

is invariant under the transformations

$$t' = \frac{\alpha t + \beta}{\gamma t + \delta'},\tag{2.1.5}$$

$$x'(t') = \frac{1}{\gamma t + \delta} x(t) \tag{2.1.6}$$

where the real numbers α , β , γ and δ form a real 2 × 2 matrix with determinant one

$$A = \begin{pmatrix} \alpha & \gamma \\ \beta & \delta \end{pmatrix}, \quad \det A = 1.$$
 (2.1.7)

1. translation

The subgroup of the matrix (2.1.7)

$$\left(\begin{array}{cc}1&0\\a&1\end{array}\right) \tag{2.1.8}$$

with $\alpha = 1$, $\beta = a$, $\gamma = 0$, $\delta = 1$ yields

$$t' = t + a,$$

 $x'(t') = x(t).$ (2.1.9)

This corresponds to the translation.

2. dilatation

The subgroup of the matrix (2.1.7)

$$\left(\begin{array}{cc}
e^{\frac{b}{2}} & 0\\
0 & e^{-\frac{b}{2}}
\end{array}\right)$$
(2.1.10)

with $\alpha = e^{\frac{b}{2}}$, $\beta = 0$, $\gamma = 0$, $\delta = e^{-\frac{b}{2}}$ generates the transformations

$$t' = e^{b}t,$$

 $x'(t') = e^{\frac{b}{2}}x(t).$ (2.1.11)

This is the dilatation.

3. conformal boost

The subgroup of the matrix (2.1.7)

$$\left(\begin{array}{cc}
1 & -c \\
0 & 1
\end{array}\right)$$
(2.1.12)

with $\alpha = 1$, $\beta = 0$, $\gamma = -c$, $\delta = 1$ corresponds to the transformations

$$t' = \frac{t}{-ct+1},$$

$$x'(t') = \frac{x(t)}{-ct+1}.$$
 (2.1.13)

This is the conformal boost transformation.

From a set of finite transformations (2.1.9), (2.1.11) and (2.1.13) we see that the action (2.1.4) is invariant under the infinitesimal one-dimensional conformal transformations

$$\delta t = f(t) = a + bt + ct^2,$$
 (2.1.14)

$$\delta x = \frac{1}{2}\dot{f}x = \frac{1}{2}(b + 2ct)x. \tag{2.1.15}$$

The passive transformations (2.1.14) and (2.1.15) lead to the active transformations

$$\delta t = 0, \tag{2.1.16}$$

$$\delta x = \frac{1}{2}\dot{f}x - f\dot{x}.$$
 (2.1.17)

Noting that a, b and c are the infinitesimal parameters of the translation, the dilatation and the conformal boost, we can compute the Noether charges, i.e. the Hamiltonian H, the dilatation operator D and the conformal boost operator K

$$H = \frac{p^2}{2} + \frac{g^2}{2x^2},\tag{2.1.18}$$

$$D = tH - \frac{1}{4}(xp + px), \qquad (2.1.19)$$

$$K = t^{2}H - \frac{1}{2}t(xp + px) + \frac{1}{2}x^{2}$$
(2.1.20)

where $p = \dot{x}$ is the canonical momentum. The operators *D* and *K* are the constants of motion in the sense that

$$\frac{\partial D}{\partial t} + [H, D] = 0, \qquad \qquad \frac{\partial K}{\partial t} + [H, K] = 0. \qquad (2.1.21)$$

One can carry out the canonical quantization by establishing the equal time commutation relation

$$[x, p] = i. (2.1.22)$$

Using the commutation relation (2.1.22), we can show that

$$[H, D] = iH, (2.1.23)$$

$$[K, D] = -iK, (2.1.24)$$

$$[H,K] = 2iD \tag{2.1.25}$$

and

$$i[H, x(t)] = \dot{x}(t),$$
 (2.1.26)

$$i[D, x(t)] = t\dot{x}(t) - \frac{1}{2}x(t), \qquad (2.1.27)$$

$$i[K, x(t)] = t^2 \dot{x}(t) - tx(t).$$
(2.1.28)

If we express the time independent part of *D* and *K* as

$$D_0 := -\frac{1}{4} \left(xp + px \right), \tag{2.1.29}$$

$$K_0 := \frac{1}{2}x^2, \tag{2.1.30}$$

then the equations (2.1.26), (2.1.27) and (2.1.28) are rewritten as

$$i[H, x(t)] = \dot{x}(t),$$
 (2.1.31)

$$i[D_0, x(t)] = -\frac{1}{2}x(t),$$
 (2.1.32)

$$i[K_0, x(t)] = 0.$$
 (2.1.33)

These equations are regarded as the Heisenberg equations. The equation (2.1.31) is familiar for general quantum mechanical systems and yields the variation of the operator with respect to time while the equation (2.1.32) gives rise to the scale dimension of the operator.

Note that the explicit time dependence of D and K can be absorbed into the similarity transformations

$$D = e^{itH} D_0 e^{-itH}, K = e^{itH} K_0 e^{-itH} (2.1.34)$$

So we will use the time independent parts as the explicit expressions for *D* and *K* and drop off the subscripts.

Defining

$$T_0 = \frac{1}{2} \left(\frac{K}{a} + aH \right), \qquad (2.1.35)$$

$$T_1 = D,$$
 (2.1.36)

$$T_2 = \frac{1}{2} \left(\frac{K}{a} - aH \right) \tag{2.1.37}$$

where *a* is a constant with dimension of length, we find from (2.1.23)-(2.1.25) the explicit representation of the $\mathfrak{so}(1,2)$ algebra

$$[T_i, T_j] = i\epsilon_{ijk}T^k \tag{2.1.38}$$

where ϵ_{ijk} is a three-index anti-symmetric tensor with $\epsilon_{012} = 1$ and $g_{ij} = \text{diag}(1, -1, -1)$. If we introduce

$$L_0 = \frac{1}{2} \left(\frac{K}{a} + aH \right) = T_0, \tag{2.1.39}$$

$$L_{\pm} = \frac{1}{2} \left(\frac{K}{a} - aH \pm 2iD \right) = T_2 \pm iT_1, \qquad (2.1.40)$$

then we get the explicit representation of the $\mathfrak{sl}(2,\mathbb{R})$ algebra in the Virasoro form

$$[L_n, L_m] = (m - n)L_{m+n}$$
(2.1.41)

with $m, n = 0, \pm$. Note that

$$H = \frac{1}{a} \left[L_0 - \frac{1}{2} \left(L_+ + L_- \right) \right], \qquad (2.1.42)$$

$$D = \frac{1}{2i} (L_{+} - L_{-}), \qquad (2.1.43)$$

$$K = a \left[L_0 + \frac{1}{2} \left(L_+ + L_- \right) \right].$$
(2.1.44)

Recall that in the representation theory the Casimir invariants play an important role since their eigenvalues may characterize the representations. The onedimensional conformal group $SL(2, \mathbb{R})$ is of rank one and therefore possesses one independent second-order Casimir invariant. The second-order Casimir operator C_2 of the $\mathfrak{sl}(2, \mathbb{R})$ algebra is given by

$$C_{2} = T_{0}^{2} - T_{1}^{2} - T_{2}^{2}$$

$$= L_{0}(L_{0} - 1) - L_{+}L_{-}$$

$$= \frac{1}{2}(HK + KH) - D^{2}$$

$$= \frac{g^{2}}{4} - \frac{3}{16}.$$
(2.1.45)

2.2 Spectrum

It is known that the quantum formalism based on the Hamiltonian *H* is awkward to describe the conformal quantum mechanics. The spectrum of *H* is continuous due to the existence of D^{-1} . This is because if $|E\rangle$ is an eigenstate of energy *E*, then $e^{i\alpha D}|E\rangle$ is that of energy $e^{2\alpha}E$ with α being an arbitrary real parameter. Thus the spectrum contains all E > 0 eigenvalues of *H*.

The corresponding wave functions are given by

$$\psi_E(x) = C\sqrt{x} J_{\sqrt{g^2 + \frac{1}{4}}} \left(\sqrt{2E}x\right)$$
(2.2.1)

where *C* is a normalization factor and J_{α} is the Bessel function of the first kind. For each of the eigenstates with the eigenvalues E > 0 there exists a normalizable plane wave.

On the other hand, the wavefunction of the zero energy state is given by ²

$$\psi_0(x) = Cx^{-\frac{1}{2} + \frac{\sqrt{1+4g^2}}{2}}$$
 (2.2.2)

where *C* is a constant value. To make matters worse, this eigenfunction is not even plane wave normalizable and this makes it difficult for us to regard the state with E = 0 as the ground state.

However. it is important to note that [54] any combination

$$G = uH + vD + wK \tag{2.2.3}$$

of the three conformal generators is a constant of motion in the sense that

$$\frac{\partial G}{\partial t} + i[H,G] = 0. \tag{2.2.4}$$

This implies that the transformations generated by G leave the action invariant. Hence we may use the operator G as the new Hamiltonian to study the evolution of the system.

The switching from H to the new evolution operator G can be interpreted as a redefinition of the time and the coordinate. Let us introduce a new time parameter

$$d\tau = \frac{dt}{u + vt + wt^2} \tag{2.2.5}$$

¹ For the DFF-model (2.1.2) this can be readily seen from the behavior of the inverse-square potential as $x \to \infty$ (Figure 2.2).

² Due to the infinite repulsive potential at the origin, we here consider the solution $\psi_0(x)$ satisfying the boundary condition $\psi_0(x)|_{x=0} = 0$.

and a new variable

$$q(\tau) = \frac{x(t)}{\sqrt{u + vt + wt^2}}.$$
(2.2.6)

Then we find the action of *G* on the operator and on the state given by

$$\frac{dq(\tau)}{d\tau} = i[G,q(\tau)], \qquad (2.2.7)$$

$$G|\Psi(\tau)\rangle = i\frac{d}{d\tau}|\Psi(\tau)\rangle$$
 (2.2.8)

as required. Although the operator *G* may describe the evolution in the new time τ , it is not yet complete to justify the passing to the new description. We need to examine whether the new Hamiltonian and the new coordinates cover the whole evolution in time from $t = -\infty$ to $+\infty$. From (2.2.5) we can express the new time parameter as

$$\tau = \int_{t_0}^t \frac{dt'}{u + vt' + wt'^2} + \tau_0.$$
(2.2.9)

This integral depends on the zeros of the denominater and the result is classified by the discriminant

$$\Delta = v^2 - 4uw. \tag{2.2.10}$$

We find

$$\tau = \begin{cases} \frac{1}{\sqrt{\Delta}} \left(\ln \left| \frac{2wt + v - \sqrt{\Delta}}{2wt + v + \sqrt{\Delta}} \right| - \ln \left| \frac{v - \sqrt{\Delta}}{v + \sqrt{\Delta}} \right| \right) & \text{for } \Delta > 0 \\\\ \frac{2}{\sqrt{|\Delta|}} \left(\tan^{-1} \frac{2wt + v}{\sqrt{|\Delta|}} - \tan^{-1} \frac{v}{\sqrt{|\Delta|}} \right) & \text{for } \Delta < 0 \\\\ - \left(\frac{2}{2wt + v} - \frac{2}{v} \right) & \text{for } \Delta = 0 \end{cases}$$
(2.2.11)

where we normalize as $\tau_0 = 0$. For $\Delta > 0$, the parameter τ cannot sweep the whole time region $-\infty \le t \le \infty$. This unpleasant feature is associated with the fact that the corresponding operators are non-compact and their spectrums are physically unacceptable. The dilatation operator D belongs to this class. When $\Delta < 0$, τ can be defined over the whole time interval $-\infty \le t \le \infty$. The corresponding operators in this case generate a compact rotation and their spectrums have physically satisfactory characteristics. In the case of $\Delta = 0$, the whole time interval $-\infty \le t \le \infty$ can be described over $-\infty \le \tau \le \infty$, however, there exists



Figure 2.1: The new time parameter τ as a function of the original time t. The red curve represents the non-compact case $\Delta > 0$, which cannot sweep over the whole time t. The blue curve corresponds to the compact case $\Delta < 0$ covering all the time region. Green curve denotes the case $\Delta = 0$, which contains one singular point in τ .

one singular point in τ at $t = -\frac{v}{2w}$. This is the case for the Hamiltonian *H* and the conformal boost operator *K*. These three cases are illustrated in Figure 2.1.

In terms of the new set of coordinates (2.2.5) and (2.2.6), we can rewrite the action (2.1.4) as

$$S = \int d\tau \left[\frac{1}{2} \dot{q}^2 - \frac{g^2}{2q^2} + \frac{1}{2} \left(\frac{v}{2} + wt \right)^2 q^2 + \frac{1}{2} (v + 2wt) q\dot{q} \right]$$

$$= \int d\tau \left[\frac{1}{2} \dot{q}^2 - \frac{g^2}{2q^2} + \frac{\Delta}{8} q^2 + \frac{1}{2} \frac{d}{d\tau} \left\{ \left(\frac{v}{2} + wt \right) q^2 \right\} \right]$$

$$= \int d\tau \left[\frac{1}{2} \dot{q}^2 - \frac{g^2}{2q^2} + \frac{\Delta}{8} q^2 \right]$$

$$= \int d\tau L_{\tau} \qquad (2.2.12)$$

up to the total τ derivative. Note that the dot denotes τ derivative in (2.2.12). The new Lagrangian L_{τ} leads to the new Hamiltonian

$$H_{\tau} = \dot{q} \frac{\partial L_{\tau}}{\partial \dot{q}} - L_{\tau}$$
$$= \frac{1}{2} \left(\dot{q}^2 + \frac{g^2}{q^2} - \frac{\Delta}{4} q^2 \right)$$
(2.2.13)

with

$$G(x(t), \dot{x}(t)) = H_{\tau}(q(\tau,)\dot{q}(\tau)).$$
(2.2.14)

Note that $L_0 = T_0$ is the compact generator satisfying $\Delta = -1 < 0$. Qualitatively one can see that the potential energy of this new Hamiltonian L_0 acquires the minimum and asymptotes to infinity (Figure 2.2). The new time coordinate τ and variable $q(\tau)$ associated with the generator L_0 are given by

$$\tau = 2\tan^{-1}\left(\frac{t}{a}\right),\tag{2.2.15}$$

$$q(\tau) = \sqrt{\frac{2}{a}} \frac{x(t)}{\sqrt{1 + \left(\frac{t}{a}\right)^2}}.$$
(2.2.16)

As we will discuss in section 2.6, in the black hole interpretation τ can be identified with the proper time of the test particle near the horizon of the extremal black hole [65].

The fact that the operator L_0 is regarded as the new Hamiltonian of the system can be paraphrased as the group theoretical statement that infinite dimensional unitary representations in terms of Hermitian operators of the non-compact group



Figure 2.2: The potentials for the original Hamiltonian H and the new Hamiltonian L_0 . The red line is the potential for H and the blue one is for L_0 .

 $SL(2, \mathbb{R})$ are characterized by the discrete eigenvalues of the Casimir operator C_2 and of the compact generator L_0 ³.

We now look for the eigenvalues and eigenstates of L_0 . Let us denote the eigenvalues and eigenstates of L_0 by r_n and $|n\rangle$

$$L_0|n\rangle = r_n|n\rangle. \tag{2.2.17}$$

From the $\mathfrak{sl}(2,\mathbb{R})$ algebra (2.1.41) one can show that

$$L_0 L_- |r_n\rangle = (r_n - 1) L_- |r_n\rangle,$$
 (2.2.18)

$$L_0L_+|r_n\rangle = (r_n+1)L_+|r_n\rangle,$$
 (2.2.19)

$$L_{-}L_{+} = -\mathcal{C}_{2} + L_{0}(L_{0} + 1), \qquad (2.2.20)$$

$$L_{+}L_{-} = -\mathcal{C}_{2} + L_{0}(L_{0} - 1).$$
(2.2.21)

The relations (2.2.18) and (2.2.19) imply that the operators L_{-} and L_{+} play the role of the annihilation and creation operators respectively. Since the norm of the states $L_{\pm}|r_n\rangle$ must be positive or zero, we require that

$$0 \le |L_{\pm}|r_n\rangle|^2 = -C_2 + r_n(r_n \pm 1).$$
(2.2.22)

Assuming that there exists one positive eigenvalue r_n among the allowed eigenvalues, the creation operator L_+ yields the infinite chain of states

$$|r_n\rangle, |r_n+1\rangle, |r_n+2\rangle, \cdots$$
 (2.2.23)

³The diagonalization of the non-compact operator requires the continuous basis [158, 159, 160, 161].



Figure 2.3: The L_0 spectrum. The ground state has eigenvalue r_0 and the excited states generated by L_+ are equally spaced with unit one.

If we require the existence of the ground state, the spectrum need to be bounded below and the chain must terminate. This means that

$$L_{-}|r_{0}\rangle = 0 \tag{2.2.24}$$

and

$$L_{+}L_{-}|r_{0}\rangle = \left[-\mathcal{C}_{2} + r_{0}(r_{0} - 1)\right]|r_{0}\rangle = 0.$$
(2.2.25)

Therefore the eigenvalues of L_0 are given by a discrete series (see Figure 2.3)

$$r_n = r_0 + n, \quad n = 0, 1, 2, \cdots$$
 (2.2.26)

$$C_2 = r_0(r_0 - 1). \tag{2.2.27}$$

In the following discussion we will thus simplify the expression as $|n\rangle = |r_n\rangle$.

Combining the relation (2.2.27) and the expression (2.1.45), we find two possible values for r_0 as the choice of the positive or negative signs for the square root. However, it turns out that only the positive sign for the square root should be selected.

$$r_0 = \frac{1}{2} \left(1 + \sqrt{g^2 + \frac{1}{4}} \right). \tag{2.2.28}$$

To see this let us determine the lowest eigenfunction $\psi_0(x)$. From the equation (2.2.24) and the explicit expressions for L_{\pm}

$$L_{-} = L_{0} - \frac{x^{2}}{2a} - \frac{1}{2}x\frac{d}{dx} - \frac{1}{4},$$
(2.2.29)

$$L_{+} = L_{0} - \frac{x^{2}}{2a} + \frac{1}{2}x\frac{d}{dx} + \frac{1}{4},$$
 (2.2.30)

we see that $\psi_0(x)$ satisfies the equation

$$\left[x\frac{d}{dx} + \frac{x^2}{a} - \left(2r_0 - \frac{1}{2}\right)\right]\psi_0(x) = 0.$$
 (2.2.31)

Let us choose our units so that a = 1. The generic solution is given by

$$\psi_0(x) = Ce^{-\frac{x^2}{2}}x^{2r_0 - \frac{1}{2}}$$
(2.2.32)

where *C* is the constant of integration. The presence of the infinitely repulsive potential barrier at the origin and the confinement property of the wavefunction requires that

$$\lim_{x \to 0} \psi_0(x) = 0, \tag{2.2.33}$$

$$\lim_{x \to 0} \psi_0'(x) = 0. \tag{2.2.34}$$

These conditions lead to

$$r_0 > \frac{3}{4}$$
 (2.2.35)

which is only satisfied by the positive root solution. Note that (2.2.35) is equivalent to the condition $\gamma > 0$ for the coupling constant as we mentioned. Also one can determine *C* by the normalization condition $\int_0^\infty |\psi_0(x)|^2 dx = 1$ as

$$C = \sqrt{\frac{2}{\Gamma(2r_0)}}.$$
(2.2.36)

Therefore the eigenfunction of the ground state is given by

$$\psi_0(x) = \sqrt{\frac{2}{\Gamma(2r_0)}} e^{-\frac{x^2}{2}} x^{\frac{1}{2} + \sqrt{g^2 + \frac{1}{4}}}.$$
(2.2.37)

This is illustrated in Figure 2.4. Curiously a particle in the L_0 ground state has zero probability of existing at x = 0.

From (2.2.22) and (2.2.27) one can see that the raising and lowering operators L_{\pm} act as

$$L_{\pm}|n\rangle = \sqrt{r_n(r_n \pm 1) - r_0(r_0 - 1)}|n \pm 1\rangle, \qquad (2.2.38)$$

which leads to the relation

$$|n\rangle = \sqrt{\frac{\Gamma(2r_0)}{n!\Gamma(2r_0+n)}} (L_+)^n |0\rangle.$$
 (2.2.39)



Figure 2.4: Wavefunction $\psi_0(x)$ of the L_0 ground state.

Upon the repeated application of the creation operator L_+ on the ground state $|0\rangle$, the eigenfunctions of the excited states found to be [54]

$$\psi_n(x) = \sqrt{\frac{\Gamma(n+1)}{2\Gamma(n+2r_0)}} x^{-\frac{1}{2}} \left(\frac{x^2}{a}\right)^{r_0} e^{-\frac{x^2}{2a}} L_n^{2r_0-1} \left(\frac{x^2}{a}\right)$$
(2.2.40)

where $L_n^{2r_0-1}$ is the associated Laguerre polynomial.

Now consider the thermodynamical aspect of the DFF-model. As we have been discussing, it has been proposed that $L_0 = \frac{1}{2}(aH + \frac{K}{a})$ is treated as the new Hamiltonian instead of the original Hamiltonian *H* in the DFF-model. The surface of the constant value of L_0 in the classical phase space is given by

$$p = \pm \sqrt{2L_0 - \frac{g}{x^2} - a^2 x^2}$$
(2.2.41)

and illustrated in Figure 2.5⁴.

Thus the volume of the phase space with the "energy" below L_0 can be evaluated to be

$$\Gamma(L_0) = 2 \int_0^\infty dx \sqrt{2L_0 - \frac{g}{x^2} - a^2 x^2} = \pi \left(\frac{L_0}{a} - g\right).$$
(2.2.42)

⁴ Note that the phase space is restricted to either x > 0 or x < 0 region due to the infinite potential at the origin.



Figure 2.5: The surface of the constant value of L_0 in the classical phase space. The horizontal axis denotes the canonical variable x while the vertical axis represents the canonical momentum p. The volume of the phase space with "the energy" below L_0 decrease with increase in the coupling constant g and the deformation parameter a. Qualitatively g keeps a particle away from the origin whereas a sucks it into the origin.

According to the additional term $-\pi g$, the result is slightly modified from a simple harmonic oscillator. This corresponds to the fact that the L_0 -ground state of the DFF-model is raised by the increase of the coupling constant g as in (2.2.28). As seen form Figure 2.5, the volume of the phase space with "the energy" below L_0 decrease with increase in the coupling constant g and the deformation parameter a. Therefore qualitatively g keeps a particle away from the origin whereas a sucks it into the origin. These features are in accord with the behavior of the wavefunction $\psi_0(x)$ of the ground state given in (2.2.37) (see also Figure 2.4).

In quantum mechanics the L_0 -spectrum is the discrete value given in (2.2.26). By summing over the spectrum one obtains the partition function

$$Z = \sum_{n=0}^{\infty} e^{-\beta L_0} = \frac{e^{-\beta r_0}}{1 - e^{-\beta}}.$$
(2.2.43)

2.3 Time evolution

So far the DFF-model (2.1.2) has been studied in the *x* space, i.e. the stationary problem at t = 0. Now let us consider the state $|t\rangle$ which is characterized by the time *t*. Let us define the time-dependent function

$$\beta_n := \langle t | n \rangle, \tag{2.3.1}$$

on which the action of the Hamiltonian is realized with the time derivative

$$H = i\frac{d}{dt}.$$
 (2.3.2)

Combining the expression (2.3.2) with the $\mathfrak{sl}(2,\mathbb{R})$ algebra (2.1.23) and the form of the Casimir operator (2.1.45), one finds the action of the dilatation operator *D* and the conformal boost operator *K* on β_n as

$$D = \left(it\frac{d}{dt} + ir_0\right),\tag{2.3.3}$$

$$K = \left(it^2\frac{d}{dt} + 2ir_0t\right). \tag{2.3.4}$$

Thus the compact operator L_0 acts on $\beta_n(t)$ as

$$L_0 = \frac{i}{2} \left[\left(a + \frac{t^2}{a} \right) \frac{d}{dt} + 2r_0 \frac{t}{a} \right].$$
(2.3.5)

From the expressions (2.3.2)-(2.3.5) we can write the actions of the corresponding operators on the state $|t\rangle$ as

$$H|t\rangle = -i\frac{d}{dt}|t\rangle, \qquad (2.3.6)$$

$$D|t\rangle = -i\left(t\frac{d}{dt} + r_0\right)|t\rangle, \qquad (2.3.7)$$

$$K|t\rangle = -i\left(t^2\frac{d}{dt} + 2r_0t\right)|t\rangle, \qquad (2.3.8)$$

$$L_0|t\rangle = -\frac{i}{2} \left[\left(a + \frac{t^2}{a} \right) \frac{d}{dt} + 2r_0 \frac{t}{a} \right] |t\rangle.$$
(2.3.9)

Then the explicit expression (2.3.5) for the operator L_0 leads to the differential equation

$$\frac{i}{2}\left[\left(a+\frac{t^2}{a}\right)\frac{d}{dt}+2r_0\frac{t}{a}\right]\beta_n=r_n\beta_n.$$
(2.3.10)

and its solution is given by

$$\beta_n(t) = (-1)^n \left[\frac{\Gamma(2r_0 + n)}{n!} \right]^{\frac{1}{2}} \left(\frac{a - it}{a + it} \right)^{r_n} \frac{1}{\left(1 + \frac{t^2}{a^2} \right)^{r_0}}.$$
 (2.3.11)

Using the above solution (2.3.11) one finds 2-point function [54]

$$F_{2}(t_{1}, t_{2}) = \langle t_{1} | t_{2} \rangle$$

= $\sum_{n} \beta_{n}(t_{1})\beta_{n}^{*}(t_{2})$
= $\frac{\Gamma(2r_{0})a^{2r_{0}}}{[2i(t_{1} - t_{2})]^{2r_{0}}} \propto \frac{1}{(t_{1} - t_{2})^{2r_{0}}}.$ (2.3.12)

The expression (2.3.12) indicates that the 2-point function is the value of two operators whose effective dimensions are r_0 . Note that the 2-point function satisfies the set of conditions

$$\left(\frac{\partial}{\partial t_1} + \frac{\partial}{\partial t_2}\right) F_2(t_1, t_2) = 0, \qquad (2.3.13)$$

$$\left(t_1\frac{\partial}{\partial t_1} + t_2\frac{\partial}{\partial t_2} + 2r_0\right)F_2(t_1, t_2) = 0, \qquad (2.3.14)$$

$$\left(t_1^2 \frac{\partial}{\partial t_1} + t_2^2 \frac{\partial}{\partial t_2} + 2r_0(t_1 + t_2)\right) F_2(t_1, t_2) = 0.$$
(2.3.15)

Now we want to consider the *E* space. The eigenstate $|E\rangle$ is defined by

$$H|E\rangle = E|E\rangle \tag{2.3.16}$$

and we can expand the eigenstate $|n\rangle$ of the compact operator L_0 as

$$|n\rangle = \int dE C_n(E)|E\rangle \tag{2.3.17}$$

where we have defined

$$C_n(E) := \langle E | n \rangle. \tag{2.3.18}$$

Note that the eigenvalue *E* of the original Hamiltonian *H* is continuous as we have already mentioned. Requiring the $\mathfrak{sl}(2,\mathbb{R})$ algebra (2.1.23) and the realization of the Casimir operator (2.1.45), we get

$$D|E\rangle = -i\left(E\frac{d}{dE} + \frac{1}{2}\right)|E\rangle, \qquad (2.3.19)$$

$$K|E\rangle = \left[-E\frac{d^2}{dE^2} - \frac{d}{dE} + \left(r_0 - \frac{1}{2}\right)^2 \frac{1}{E}\right]|E\rangle.$$
 (2.3.20)

Then we can write the compact operator L_0 as

$$L_0 = \frac{1}{2} \left[-E \frac{d^2}{dE^2} - \frac{d}{dE} + E + \left(r_0 - \frac{1}{2} \right)^2 \frac{1}{E} \right]$$
(2.3.21)

and the explicit expression for $C_n(E)$ can be found by solving the equation

$$\frac{1}{2}\left[-E\frac{d^2}{dE^2} - \frac{d}{dE} + E + \left(r_0 - \frac{1}{2}\right)^2 \frac{1}{E}\right] C_n(E) = r_n C_n(E).$$
(2.3.22)

If we set

$$C_n = 2^{r_0} E^{r_0 - \frac{1}{2}} e^{-E} \varphi_n(E), \qquad (2.3.23)$$

then we see that the function φ_n satisfies the differential equation ⁵

$$\eta \varphi_n'' + (2r_0 - \eta)\varphi_n' + n\varphi_n = 0$$
(2.3.24)

of the associated Laguerre polynomial $L_n^{2r_0-1}$. Putting all together we obtain

$$C_n(E) = 2^{r_0} \sqrt{\frac{\Gamma(n+1)}{\Gamma(n+2r_0)}} \frac{(aE)^{r_0}}{\sqrt{E}} e^{-aE} L_n^{2r_0-1}(2aE).$$
(2.3.25)

⁵The associated Laguerre polynomials $L_n^k(x)$ are defined by the solution of the differential equation

$$\left[x\frac{d^{k+2}}{dx^{k+2}} + (k+1-x)\frac{d^{k+1}}{dx^{k+1}} + n\frac{d^k}{dx^k}\right]L_n^k(x) = 0$$

with $0 \le k \le n$.



Figure 2.6: The probability density $\rho_0(E)$ of the ground state in *E* space. The blue, red and brown lines denote the cases with $r_0 = \frac{3}{2}$, 3 and 10. They have the maximum values at $E = r_0 - \frac{1}{2}$.

Note that the function $C_n(E)$ has the following properties:

$$\sum_{n} C_{n}(E)C_{n}^{*}(E') = \delta(E - E'), \qquad (2.3.26)$$

$$\int_0^\infty dE C_n(E) C_{n'}^*(E) = \delta_{nn'},$$
(2.3.27)

$$C_n(E) = 2^{r_0} E^{\frac{1}{2} - r_0} \int_{-\infty}^{\infty} \frac{dt}{2\pi} e^{iEt} \beta_n(t).$$
 (2.3.28)

Let us discuss the probability density $\rho_n(E)$ in *E* space defined by

$$\rho_n(E) := |C_n(E)|^2.$$
(2.3.29)

For n = 0, i.e. for the ground state of L_0 , the probability density $\rho_0(E)$ is given by

$$\rho_0(E) = \frac{4r^{r_0}}{\Gamma(2r_0)} E^{2r_0 - 1} e^{-2E} \left[L^{2r_0 - 1}(2E) \right]^2$$
(2.3.30)

with a = 1. This is shown in Figure 2.6.

The distribution of *E* of the ground state is peaked at

$$E_0 = r_0 - \frac{1}{2} \tag{2.3.31}$$


Figure 2.7: The probability densities $\rho_n(E)$ of the excited states for $r_0 = \frac{3}{2}$ in *E* space. The red, blue and green lines correspond to the densities with n = 10, 20 and 30. The number of peaks increases with the increase of the excited levels.

and its effective width is ⁶

$$\Gamma = 2\sqrt{E_0} = 2\sqrt{r_0 - \frac{1}{2}}.$$
(2.3.32)

(2.3.31) shows that the peak of the distribution of *E* increases with the increase of r_0 or *g*. (2.3.32) implies that the width grow and the probability dense spread in *E* space with r_0 .

The expectation values for the ground state $|0\rangle$ can be evaluated and we find

$$\langle H \rangle_0 = \langle 0 | H | 0 \rangle = r_0 \tag{2.3.33}$$

$$(\Delta E)^2 := \langle H^2 \rangle_0 - \langle H \rangle_0^2 = \frac{r_0^2}{2}.$$
 (2.3.34)

For n > 0 the probability density $\rho_n(E)$ with $r_0 = \frac{3}{2}$ is illustrated in Figure 2.7. The red, blue and green lines represent the case with n = 10, 20 and 30 respectively. In this case several peaks appear due to the presence of the *n*-th order polynomial. The expectation value of *H* in the excited state $|n\rangle$ is calculated to be [54]

$$\langle H \rangle_n = \langle n | H | n \rangle = r_n.$$
 (2.3.35)

The state $|E\rangle$ provides us with the further properties of the state $|t\rangle$. From the expression (2.3.12) and the relation (2.3.28) with the use of the Hankel integral

⁶The effective width Γ is defined by $\frac{8}{\Gamma} := -\frac{\rho_0(E_0)''}{\rho_0(E_0)}$.

representation of the gamma function

$$\frac{1}{\Gamma(z)} = \frac{i}{2\pi} \int_C dt (-t)^{-z} e^{-t},$$
(2.3.36)

we can show that

$$\langle t|E\rangle = 2^{-r_0} \frac{(aE)^{r_0}}{\sqrt{E}} e^{-iEt}.$$
 (2.3.37)

Using the two relations (2.3.28) and (2.3.37) we see that

$$\int_{-\infty}^{\infty} \frac{dt}{2\pi} |t\rangle \langle t| = \frac{1}{H} \left(\frac{aH}{2}\right)^{2r_0}.$$
(2.3.38)

This indicates the incompleteness of the state $|t\rangle$.

2.4 Operator

Now consider the tensor operator B(t) with the mass dimension Δ . As seen from the set of the Heisenberg equations (2.1.31)-(2.1.33), the $\mathfrak{sl}(2,\mathbb{R})$ invariance of the operator implies that

$$i[H, B(t)] = \frac{d}{dt}B(t),$$
 (2.4.1)

$$i[D, B(t)] = t\frac{d}{dt}B(t) + \Delta B(t), \qquad (2.4.2)$$

$$i[K, B(t)] = t^2 \frac{d}{dt} B(t) + 2t\Delta B(t)$$
(2.4.3)

where $D = tH + D_0$ and $K = t^2H - \frac{1}{2}\{x, p\} + K_0$. This is the *SO*(1,2) Wigner-Eckart theorem. We now want to compute the 3-point function

$$F_{3}(t;t_{2},t_{1}) = \langle t_{2}|B(t)|t_{1}\rangle$$

= $\sum_{n_{1},n_{2}} \beta_{n_{2}}(t_{2})\beta_{n_{1}}^{*}(t_{1})\langle n_{2}|B(t)|n_{1}\rangle.$ (2.4.4)

In analogy with (2.2.15) and (2.2.16) it is convenient to introduce the new variables

$$\tau = 2 \tan^{-1} t, \tag{2.4.5}$$

$$b(\tau) = B(t)(1+t^2)^{\Delta}$$
(2.4.6)

with the relations

$$\frac{db(\tau)}{d\tau} = i[L_0, b(\tau)],$$
(2.4.7)

$$b(\tau) = e^{iL_0\tau}b(0)e^{-iL_0\tau}.$$
(2.4.8)

By using the above expressions we can show that

$$\langle n_2|B(t)|n_1\rangle = \frac{1}{(1+t^2)^{\Delta}} \left(\frac{1-it}{1+it}\right)^{n_1-n_2} \langle n_2|B(0)|n_1\rangle.$$
 (2.4.9)

The equation (2.4.9) enables us to rewrite the 3-point function (2.4.4) as

$$F_{3}(t;t_{2},t_{1}) = \sum_{n_{1},n_{2}} (-1)^{n_{1}+n_{2}} \sqrt{\frac{\Gamma(n_{1}+2r_{0})\Gamma(n_{2}+2r_{0})}{n_{1}!n_{2}!}}$$

$$\times z_{1}^{-n_{1}} z_{2}^{n_{2}} \frac{(1+z_{1})^{2r_{0}}(1+z_{2})^{2r_{0}}}{2^{4r_{0}} z_{1}^{2r_{0}}}$$

$$\times 2^{-2\Delta} \left|\frac{1+z}{z}\right|^{2\Delta} z^{n_{1}-n_{2}} \langle n_{2}|B(0)|n_{1}\rangle$$
(2.4.10)

where we have defined

$$z := \frac{1 - it}{1 + it} = e^{-i\tau}, \qquad \qquad z_i := \frac{1 - it_i}{1 + it_i} = e^{-i\tau_i}. \qquad (2.4.11)$$

On the other hand, the one-dimensional conformal $\mathfrak{sl}(2,\mathbb{R})$ covariance of the 3-point function implies that

$$\left(\frac{\partial}{\partial t} + \frac{\partial}{\partial t_1} + \frac{\partial}{\partial t_2}\right) F_3(t; t_2, t_1) = 0, \qquad (2.4.12)$$

$$\left(t\frac{\partial}{\partial t} + t_1\frac{\partial}{\partial t_1} + t_2\frac{\partial}{\partial t_2} + 2\Delta t + 2r_0(t_1 + t_2)\right)F_3(t;t_2,t_1) = 0, \qquad (2.4.13)$$

$$\left(t^2\frac{\partial}{\partial t} + t_1^2\frac{\partial}{\partial t_1} + t_2^2\frac{\partial}{\partial t_2} + 2\Delta t + 2r_0(t_1 + t_2)\right)F_3(t;t_2,t_1) = 0.$$
(2.4.14)

As the first condition (2.4.12) says that $F_3(t;t_2,t_1) = F_3(t - t_1, t - t_2, t_1 - t_2)$, we impose the ansatz $F_3 = f(t - t_1)^{\alpha_1}(t - t_2)^{\alpha_2}(t_1 - t_2)^{\alpha_3}$ with f being an arbitrary constant value. Then the remaining two conditions (2.4.13) and (2.4.14) restrict to the form of the 3-point function as

$$F_{3}(t;t_{2},t_{1}) = fi^{2r_{0}+\Delta} \frac{1}{(t-t_{1})^{\Delta}(t_{2}-t)^{\Delta}(t_{1}-t_{2})^{-\Delta+2r_{0}}}$$

= $2^{-\Delta-2r_{0}}f \frac{(1+z)^{2\Delta}(1+z_{1})^{2r_{0}}(1+z_{2})^{2r_{0}}}{(z-z_{1})^{\Delta}(z_{2}-z)^{\Delta}(z_{1}-z_{2})^{-\Delta+2r_{0}}}.$ (2.4.15)

Combining the two expressions (2.4.10) and (2.4.15) for the 3-point function, we obtain the relation

$$\sum_{n_1,n_2} (-1)^{n_1+n_2} \sqrt{\frac{\Gamma(n_1+2r_0)\Gamma(n_2+2r_0)}{n_1!n_2!}} z_1^{-n_1} z_2^{n_2} 2^{-\Delta} \left| \frac{1+z}{z} \right|^{2\Delta} z^{n_1-n_2} \langle n_2 | B(0) | n_1 \rangle$$

= $2^{r_0} f z_1^{2r_0} \frac{(1+z)^{2\Delta}}{(z-z_1)^{\Delta} (z_2-z)^{\Delta} (z_1-z_2)^{-\Delta+2r_0}}.$
(2.4.16)

By redefining the variables $z_i \rightarrow zz_i$, we can factor out the time *t* dependence and thus we shall take t = 0 or equivalently z = 1 in the following discussion. Then the left hand side can be regarded as the double Taylor series expansions and one can write the quantity $\langle n_2 | B(0) | n_1 \rangle$ as the coefficients of the expansions as

$$\langle n_2 | B(0) | n_1 \rangle = \frac{f}{(2\pi i)^2} \sqrt{\frac{n_1! n_2!}{\Gamma(n_1 + 2r_0)\Gamma(n_2 + 2r_0)}} (-1)^{n_1 + n_2} 2^{2r_0 + \Delta}$$

$$\times \oint_{C_1} dz_1 \oint_{C_2} dz_2 \frac{z_1^{2r_0 + n_1 - 1} z_2^{-(n_2 + 1)}}{(1 - z_1)^{\Delta} (z_2 - 1)^{\Delta} (z_1 - z_2)^{-\Delta + 2r_0}}$$

$$(2.4.17)$$

where C_1 is the suitable anti-clockwise contour for the coordinates $z_1 = \infty$ and C_0 is for z_2 . Changing the pair of the variables $z_1 = \frac{1}{w_1}$ and $z_2 = w_2$, we find the expression

$$\langle n_2 | B(0) | n_1 \rangle = \frac{f}{(2\pi i)^2} \sqrt{\frac{n_1! n_2!}{\Gamma(n_1 + 2r_0)\Gamma(n_2 + 2r_0)}} (-1)^{n_1 + n_2} 2^{2r_0 + \Delta}$$

$$\times \oint_{C_2} \frac{dw_2}{w_2^{n_2 + 1}} \oint_{C_1} \frac{dw_1}{w_1^{n_1 + 1}} \frac{(1 - w_1 w_2)^{\Delta - 2r_0}}{(1 - w_1)^{\Delta} (w_2 - 1)^{\Delta}}$$

$$(2.4.18)$$

where the integral are carried out around the contours C_1 for w_1 and C_2 for w_2 . Applying the Cauchy theorem, the integration (2.4.18) can be calculated to be [54]

$$\langle n_2 | B(0) | n_1 \rangle = 2^{2r_0 + \Delta} f \sqrt{\frac{n_1! n_2!}{\Gamma(2r_0 + n_1)\Gamma(2r_0 + n_2)}} \\ \times \sum_{k=0}^{\min[n_1, n_2]} (-1)^k \left(\begin{array}{c} \Delta - 2r_0 \\ k \end{array} \right) \left(\begin{array}{c} -\Delta \\ n_1 - k \end{array} \right) \left(\begin{array}{c} -\Delta \\ n_2 - k \end{array} \right).$$
(2.4.19)

For $n_1 = n_2 = 0$ we can read off the explicit formula for the constant *f* as

$$f = 2^{-\Delta - 2r_0} \Gamma(2r_0) \langle 0|B(0)|0\rangle.$$
(2.4.20)

Inserting (2.4.20) into (2.4.15) and reviving the constant factor *a*, we finally get the formula for the 3-point function

$$F_{3}(t,;t_{2},t_{1}) = \langle 0|B(0)|0\rangle \left(\frac{i}{2}\right)^{2r_{0}+\Delta} \frac{\Gamma(2r_{0})a^{2r_{0}}}{(t-t_{1})^{\Delta}(t_{2}-t)^{\Delta}(t_{1}-t_{2})^{-\Delta+2r_{0}}} \\ \propto \frac{1}{(t-t_{1})^{\Delta}(t_{2}-t)^{\Delta}(t_{1}-t_{2})^{-\Delta+2r_{0}}}.$$
(2.4.21)

From the above form (2.4.21) we see that the 3-point function F_3 consists of the two operators with the same mass dimension r_0 and the third operator *B* with the mass dimension Δ .

As seen from (2.3.12) and (2.4.21), the structures of the 2- and 3-point functions suggest that there exists the averaging state and the corresponding operator with the mass dimension r_0 . Let us define the operator O(t) which acts on the L_0 -vacuum to create the state $|t\rangle$:

$$|t\rangle = \mathcal{O}(t)|0\rangle. \tag{2.4.22}$$

Making use of the formulae (2.3.11) and (2.2.39), we can write ⁷

$$\mathcal{O}(t) = N(t) \exp\left[-\overline{z}(t)L_{+}\right]$$
(2.4.23)

where

$$N(t) = \sqrt{\Gamma(2r_0)} \left(\frac{\bar{z}(t) + 1}{2}\right)^{2r_0}.$$
 (2.4.24)

Then the 2- and 3-point functions are given by

$$F_2(t_1, t_2) = \langle t_1 | t_2 \rangle = \langle 0 | \mathcal{O}^{\dagger}(t_1) \mathcal{O}(t_2) | 0 \rangle, \qquad (2.4.25)$$

$$F_3(t;t_2,t_1) = \langle t_2 | B(t) | t_1 \rangle = \langle 0 | \mathcal{O}^{\dagger}(t_1) B(t) \mathcal{O}(t_2) | 0 \rangle.$$
(2.4.26)

Therefore the averaging state is the L_0 ground state $|0\rangle$ and the corresponding operators are $\mathcal{O}^+(t)$ and $\mathcal{O}(t)$. We should note that the conformal invariant correlation functions can be built up although the averaging state $|0\rangle$ is not conformally invariant and the operators $\mathcal{O}(t)$ and $\mathcal{O}^+(t)$ are not primary operators. This is the significant difference from other higher dimensional conformal field theories where one can assume the existence of the normalizable and invariant vacuum states. For quantum field theories we generally treat with the Fock spaces which are underlying on the empty no-particle vacuum states. However, in quantum mechanics we deal with the Hilbert space which is the subspace of the Fock space with the fixed number of the particle. This fact prevents us from constructing the normalizable and invariant empty vacuum state in conformal quantum mechanics.

Noting that

$$[L_{-}, e^{-\overline{z}(t)L_{+}}] = e^{-\overline{z}(t)L_{+}} \left(-2\overline{z}(t)L_{0} + \overline{z}^{2}(t)L_{+}\right), \qquad (2.4.27)$$

$$[L_0, e^{-\overline{z}(t)L_+}] = -e^{-\overline{z}L_+}\overline{z}(t)L_+, \qquad (2.4.28)$$

we see that the state $|t\rangle$ is the eigenstate of the operator $L_{-} + \overline{z}(t)L_{0}$

$$(L_{-} + \overline{z}(t)L_{0}) |t\rangle = -r_{0}\overline{z}(t)|t\rangle$$
(2.4.29)

⁷Such construction of the state $|t\rangle$ has also been considered in [162, 163, 164]

with the eigenvalue $-r_0\overline{z}(t)$. We see that the state $|t\rangle$ is similar to the coherent state $|a\rangle$ which satisfies $L_{-}|a\rangle = a|a\rangle$, however, the additional term $\overline{z}(t)L_0$ deviates from it. In fact the coherent state $|a\rangle$ can be constructed as

$$|a\rangle = \sqrt{\Gamma(2r_0)} \sum_{n} \frac{a^n}{\sqrt{n!\Gamma(2r_0+n)}} |n\rangle$$

= $\Gamma(2r_0) \sum_{n} \frac{a^n}{n!\Gamma(2r_0+n)} (L_+)^n |0\rangle.$ (2.4.30)

Let us define the state

$$|\Psi\rangle := e^{-Ha}|t=0\rangle = e^{-Ha}e^{-L_{+}}|0\rangle.$$
 (2.4.31)

By using the relations

$$L_0 e^{-Ha} = e^{-Ha} \left(\frac{K}{2a} + iD\right),$$
 (2.4.32)

$$\left(\frac{K}{2a}+iD\right)e^{-L_{+}}=e^{-L_{+}}\left(L_{0}-\frac{1}{4}L_{-}\right),$$
 (2.4.33)

one finds that

$$L_0|\Psi\rangle = r_0|\Psi\rangle. \tag{2.4.34}$$

Thus the state $|\Psi\rangle$ defined by (2.4.31) is proportional to the L_0 vacuum state $|0\rangle$

$$|\Psi\rangle = C|0\rangle \tag{2.4.35}$$

where the proportional constant *C* can be determined by noting the relation (2.3.37) as

$$C = \frac{\sqrt{\Gamma(2r_0)}}{2^{2r_0}} \tag{2.4.36}$$

up to a phase factor. Then we obtain the alternative description for the state $|t\rangle$

$$\begin{aligned} |t\rangle &= e^{iHt}|t=0\rangle \\ &= e^{iHt}e^{Ha}\left(C|0\rangle\right) \\ &= \sqrt{\Gamma(2r_0)}2^{-2r_0}e^{(a+it)H}|0\rangle. \end{aligned}$$
(2.4.37)

Let us consider the 4-point function

$$F_{4}(t_{1}, t_{2}, t_{3}, t_{4}) = \langle t_{1} | B(t_{2}) \tilde{B}(t_{3}) | t_{4} \rangle$$

= $\langle 0 | \mathcal{O}^{\dagger}(t_{1}) B(t_{2}) B(t_{3}) \mathcal{O}(t_{4}) | 0 \rangle$ (2.4.38)

where the two different fields B(t) and $\tilde{B}(t)$ carry the mass dimension Δ and $\tilde{\Delta}$ respectively. It is calculated to be [87]

$$F_{4}(t_{1}, t_{2}, t_{3}, t_{4}) = \langle 0|B(0)|0\rangle \langle 0|\tilde{B}(0)|0\rangle \frac{\Gamma(2r_{0})}{2^{\Delta + \tilde{\Delta} + 2r_{0}}} \\ \times \frac{1}{(t_{13})^{\Delta - r_{0}}(t_{24})^{\tilde{\Delta} - r_{0}}(t_{12})^{\tilde{\Delta} + r_{0}}(t_{34})^{\Delta + r_{0}}(t_{14})^{2r_{0} - \Delta - \tilde{\Delta}}} x^{r_{0}} {}_{2}F_{1}(\Delta, \tilde{\Delta}; 2r_{0}; x) \\ = p(t_{1}, t_{2}, t_{3}, t_{4})x^{r_{0}} {}_{2}F_{1}(\Delta, \tilde{\Delta}; 2r_{0}; x)$$
(2.4.39)

where the parameter *a* set to be one and we have introduced the expressions $t_{ij} := t_i - t_j$ and $x := \frac{t_{12}t_{34}}{t_{13}t_{24}}$. $_2F_1(\Delta, \tilde{\Delta}; 2r_0; x)$ is the hypergeometric function that possesses the Mellin-Barnes representation

$${}_{2}F_{1}(a,b;c;x) = \frac{\Gamma(c)}{\Gamma(a)\Gamma(b)} \frac{1}{2\pi i} \int_{-i\infty}^{+i\infty} ds \frac{\Gamma(a+s)\Gamma(b+s)\Gamma(-s)}{\Gamma(c+s)} (-x)^{s}.$$
(2.4.40)

Note that the single Mellin integral appears in the formula of the 4-point function (2.4.39). This reflects the fact that four points lead to a single invariant in onedimension in contrast to the two invariants in higher dimensions.

It is known that 4-point functions F_4 can be expressed by the superposition of the conformal blocks, or the conformal partial waves *G* in higher dimensional conformal field theories [165]. The conformal block *G* can be determined by requiring that it is the eigenfunction of the quadratic Casimir of the conformal group. As seen from the formula (2.4.39), we can easily read off a single conformal block as [87] ⁸

$$G = x_2^{r_0} F_1(\Delta, \tilde{\Delta}; 2r_0; x), \qquad (2.4.41)$$

which satisfies the differential equation

$$\mathcal{C}_2\left[p(t_1, t_2, t_3, t_4)G\right] = r_0(r_0 - 1)p(t_1, t_2, t_3, t_4)G.$$
(2.4.42)

2.5 Gauged conformal mechanics

It has been pointed out [137] that the DFF-model (2.1.2) can be obtained by the gauged quantum mechanics. Let us consider a simple complex free particle Lagrangian

$$L = \frac{1}{2}\dot{z}\dot{\bar{z}} \tag{2.5.1}$$

⁸ For higher dimensional field theories, the conformal block can be obtained through the operator product expansion.

where *z* is a complex one-dimensional field. The system (2.5.1) is invariant under the following U(1) transformations

$$z' = e^{-i\lambda}z, \qquad \qquad \overline{z}' = e^{i\lambda}\overline{z}$$
 (2.5.2)

where λ is a real parameter. Let us gauge this symmetry by promoting $\lambda \rightarrow \lambda(t)$. Then the gauge invariant Lagrangian is given by

$$L = \frac{1}{2} D_0 z D_0 \overline{z} + c A_0$$

= $\frac{1}{2} (\dot{z} + i A_0 z) (\dot{\overline{z}} - i A_0 \overline{z}) + c A_0$ (2.5.3)

where $A_0(t)$ is the one-dimensional U(1) gauge field. The term cA_0 is a Fayet-Iliopoulos (FI) term with *c* being a constant. This term is gauge invariant itself up to total derivative.

The action (2.5.3) is invariant under the one-dimensional conformal transformations

$$\delta t = f(t) = a + bt + ct^2,$$
 (2.5.4)

$$\delta x = \frac{1}{2}\dot{f}x,\tag{2.5.5}$$

$$\delta A_0 = -\dot{f} A_0. \tag{2.5.6}$$

Here the transformation of the gauge field $A_0(t)$ is the same as that of the time derivative ∂_0 .

Note that the Lagrangian (2.5.3) is quadratic in the U(1) gauge field A_0 and contains no time derivative of A_0 . This immediately implies that the gauge field A_0 is identified with the auxiliary gauge field. Hence we attempt to integrate out the auxiliary gauge field. However, we should be careful of the exclusion of the auxiliary field because it is a gauge field. We need to integrate out the auxiliary gauge field in two steps; firstly we fix a gauge to eliminate residual degrees of freedom and then solve the equation of motion of the auxiliary gauge field or impose the resulting Gauss law constraint to ensure the consistency of the gauge fixing. Let us choose the gauge such that

$$z(t) = \overline{z}(t) = x(t).$$
 (2.5.7)

Then the Lagrangian (2.5.3) becomes

$$L = \frac{1}{2}\dot{x}^2 + \frac{1}{2}A_0^2x^2 + cA_0.$$
 (2.5.8)

Using the algebraic equation of motion for the auxiliary gauge field A_0

$$A_0 = -\frac{c}{x^2},$$
 (2.5.9)

we can integrate out gauge field and obtain the reduce Lagrangian

$$L = \frac{1}{2} \left(\dot{x}^2 - \frac{c^2}{x^2} \right).$$
 (2.5.10)

This is nothing but (2.5.2), the DFF-model Lagrangian. Thus the conformal invariance is preserved under the gauging procedure. This procedure, i.e. the integration of the auxiliary gauge field can be interpreted as the reduction process for the mechanical systems with symmetry. Let us summarize the basic concepts of the classical theory of Hamiltonian dynamical systems.

A manifold \mathcal{M} is said to be endowed with a Poisson structure if there is an operation assigning to every pair of functions $F, G \in \mathcal{F}(\mathcal{M})$ a new function $\{F, G\} \in \mathcal{F}(\mathcal{M})$ which is linear in *F* and *G* and has the following properties

1. skew symmetry

$$\{F,G\} = -\{G,F\}$$
(2.5.11)

2. Jacobi identity

$$\{F, \{G, H\}\} + \{G, \{H, F\}\} + \{H, \{F, G\}\} = 0$$
(2.5.12)

3. Leibniz rule

$$\{F, GH\} = \{F, G\} H + \{F, H\} G.$$
(2.5.13)

As the above three identities (2.5.11)-(2.5.13) are the axioms of the Lie algebra, the space $\mathcal{F}(\mathcal{M})$ is nothing but an infinite dimensional Lie algebra. Then a dynamical system on \mathcal{M} , the so-called Hamiltonian dynamical system can be introduced as

$$\dot{x}^{i} = \left\{ H(x), x^{i} \right\} = X_{H}^{i}$$
 (2.5.14)

where x^i are local coordinates on \mathcal{M} , H(x) is the Hamiltonian of the dynamical system and the vector field X_H^i is referred to as a Hamiltonian vector field. For such system we have

$$\dot{F} = \{H, x\}$$
 (2.5.15)

and the functions which satisfy $\{H, F\} = 0$ are conserved quantities, i.e. the integrals of motion.

The important class of phase spaces is known as a symplectic manifold (\mathcal{M}, ω) , which possesses closed nondegenerate differential two-form ω , i.e. symplectic forms and their Poisson structures are given by

$$\{F(x), G(x)\} = \omega^{ij} \partial_i F \partial_j G$$

= $\omega(X_F, X_G).$ (2.5.16)

Suppose that a Lie group *G* acts on \mathcal{M} . Then one can represent the corresponding Lie algebra \mathfrak{g} of *G* in the Lie algebra of vector fields on \mathcal{M} . In other words there is a vector field X_{ξ} on \mathcal{M} to each $\xi \in \mathfrak{g}$. If one can associate a function H_{ξ} on \mathcal{M} to each X_{ξ} obeying the conditions

$$X^{i}_{\xi} = \left\{ H_{\xi}, x^{i} \right\}, \qquad (2.5.17)$$

$$H_{\xi+\eta} = H_{\xi} + H_{\eta},$$
 (2.5.18)

$$H_{[\xi,\eta]} = \{H_{\xi}, H_{\eta}\}, \qquad (2.5.19)$$

the action of *G* is called Hamiltonian and H_{ξ} the Hamiltonian function. Namely an action of *G* on \mathcal{M} is called Hamiltonian if the map $\xi \mapsto H_{\xi}$ is a homomorphism of the Lie algebra \mathfrak{g} into the Lie algebra $\mathcal{F}(\mathcal{M})$. It is known that any symplectic action of Lie group *G* is Hamiltonian if $H^2(\mathfrak{g}, \mathbb{R}) = 0$.

Since the Hamiltonian function H_{ξ} of *G* depends on $\xi \in \mathfrak{g}$ linearly we may write it as

$$H_{\xi}(x) = \langle \mu(x), \xi \rangle \tag{2.5.20}$$

where the notation $\langle f, \xi \rangle$ denotes the value of f at $\xi \in \mathfrak{g}$ and $\mu(x)$ belongs to \mathfrak{g}^* , the dual of the Lie algebra \mathfrak{g} . Therefore there is a map

$$\mu: \mathcal{M} \mapsto \mathfrak{g}^*$$
 (2.5.21)

for any Hamiltonian action of G on \mathcal{M} . This is called the moment map .

If we have the Hamiltonian action of a Lie group *G* on M which leaves the Hamiltonian H(x) invariant, the quadruple

$$\{\mathcal{M}, \{,\}, H, G\}$$
 (2.5.22)

is called a Hamiltonian system with *G*-symmetry. There is an important property of Hamiltonian system with *G*-symmetry [166]

If the Hamiltonian H(x) is invariant under a Hamiltonian action of a Lie group *G* on \mathcal{M} , then the moment map $\mu(x)$ is an integral of motion.

This is a generalization of the well-known Noether's theorem. Since the symmetries give rise to the integrals of motion, one can reduce the dynamical system to one with fewer degrees of freedom [167, 166, 168, 169]. Suppose we have a Hamiltonian action of *G* on a symplectic manifold \mathcal{M} and the corresponding moment map $\mu : \mathcal{M} \mapsto \mathfrak{g}^*$. We consider the inverse image of a point $c \in \mathfrak{g}^*$ for μ and represent this set by $\tilde{\mathcal{M}}_c$

$$\tilde{\mathcal{M}}_c = \mu^{-1}(c).$$
 (2.5.23)

We require that *c* is a regular value of μ . This implies that either the differential of μ at every point of \tilde{M}_c maps the tangent space to \mathcal{M} onto \mathfrak{g}^* or \tilde{M}_c is empty ⁹. In this case $\tilde{\mathcal{M}}_c$ is a smooth submanifold of \mathcal{M} . The isotropy subgroup of *c* consists of the elements \mathfrak{g} of *G* for which

$$\mathrm{Ad}_{\mathfrak{g}}^* c = c. \tag{2.5.24}$$

Put in another way, the isotropy subgroup is the subgroup relative to the coadjoint action which leaves \tilde{M}_c invariant. Let us denote this isotropy subgroup by

$$G_c = \{g : \mathrm{Ad}_{\mathfrak{g}}^* c = c\}.$$
 (2.5.25)

Now that the space $\tilde{\mathcal{M}}_c$ decomposes into orbits of the action of *G*, we can define the reduced phase space by

$$\mathcal{M}_c = \tilde{\mathcal{M}}_c / G_c. \tag{2.5.26}$$

It has been shown in [166, 168] that if the isotropy subgroup G_c is compact and acts on $\tilde{\mathcal{M}}_c$ without fixed points, the reduced phase space (2.5.26) is shown to symplectic manifold and that the reduced field, the vector field on the reduced phase space \mathcal{M}_c remains Hamiltonian vector field on it and the corresponding Hamiltonian function pulled back to $\tilde{\mathcal{M}}_c$ coincides with the original Hamiltonian function restricted to $\tilde{\mathcal{M}}_c$.

An Abelian version of the Lagrangian reduction with the integrals of motion was firstly proposed by Routh [171]. Recall that there are two formulations for the classical dynamical system; the Lagrangian formalism and the Hamiltonian formalism. The Lagrangian is a functional of coordinates and their time derivatives and it leads to the equations of motion as a set of second order differential

⁹In [170] it has been discussed that almost all *c* are regular values.

equations while the Hamiltonian is a functional of coordinates and their canonical momenta and leads to the equations of motion as a set of first order differential equations at the cost of the twice number of the equations.

Suppose we have a system whose Lagrangian is independent of some subset of coordinates. We will refer them as cyclic coordinates and denote by y^i and the remaining non-cyclic ones by x^i . The Lagrangian can be written as

$$L(x^{i}, y^{i}, \dot{x}^{i}, \dot{y}^{i}; t) = L(x^{i}, \dot{x}^{i}, \dot{y}^{i}; t).$$
(2.5.27)

Note that the canonical momenta of the cyclic coordinates y^i

$$p_{y^i} = \frac{\partial L}{\partial \dot{y}^i} \tag{2.5.28}$$

are conserved quantities. In this case the differential equations associated with these momenta are trivial and therefore the Hamiltonian formulation is more advantageous.

The Routhian *R* is regarded as the new Lagrangian , which is the mixture of the Lagrangian with the Hamiltonian. More precisely it is defined by setting $p_{y^i} = h_i = \text{constant}$ and performing a partial Legendre transformation on the cyclic coordinates y_i

$$R(x^{i}, \dot{x}^{i}, h_{i}; t) := L - \sum_{i} h_{i} \dot{y}^{i}.$$
(2.5.29)

Let us consider the Euler-Lagrange expressions for the Routhian

$$\frac{d}{dt} \left(\frac{\partial R}{\partial \dot{x}^{i}} \right) - \frac{\partial R}{\partial x^{i}}$$

$$= \frac{d}{dt} \left(\frac{\partial L}{\partial \dot{x}^{i}} + \frac{\partial L}{\partial \dot{y}^{i}} \frac{\partial \dot{y}^{i}}{\partial \dot{x}^{i}} \right) - \frac{d}{dt} \left(h_{i} \frac{\partial \dot{y}^{i}}{\partial \dot{x}^{i}} \right)$$

$$- \left(\frac{\partial L}{\partial x^{i}} + \frac{\partial L}{\partial \dot{y}^{i}} \frac{\dot{y}^{i}}{\partial x^{i}} \right) - h^{i} \frac{\partial \dot{y}^{i}}{\partial x^{i}}.$$
(2.5.30)

The first and fourth terms cancel by the original Euler-Lagrange equations and the remaining terms vanish by the definition of the canonical momenta $h_i = \frac{\partial L}{\partial u^i}$.

This shows that the Euler-Lagrange equations for $L(x, \dot{x}, \dot{y})$ together with the conserved quantities $h_i = p_{y^i}$ are equivalent to the Euler-Lagrange equations for the Routhian $R(x, \dot{x})$. The Euler-Lagrange equations for the Routhian are called the reduced Euler-Lagrange equations because the phase space \mathcal{M} with variables $\{x^i, y^i\}$ is now reduced to the small phase space $\tilde{\mathcal{M}}$ with variables $\{x^i\}^{10}$. Note

¹⁰In other words the naive substitution of the conserved quantities into the original Lagrangian spoils the role of the Lagrangian.

that the Hamilton equations for the cyclic coordinates yield the trivial statement; the constant property of h_i (i.e. $\dot{h}_i = 0$) and the definition of h_i (i.e. $h_i = \frac{\partial L}{\partial u^i}$).

Now let us go back to the gauged mechanical Lagrangian (2.5.3) and apply the Routh reduction ¹¹. We will parametrize the complex variable *z* as $z = qe^{i\varphi}$ where $q \ge 0$ and $0 \le \varphi < 2\pi$ are real variables. We then can write the Lagrangian (2.5.3) as

$$L = \frac{1}{2}\dot{q}^2 + \frac{1}{2}(q\dot{\varphi})^2 + q\dot{\varphi}A_0 + \frac{1}{2}q^2A_0^2 + cA_0.$$
(2.5.31)

By choosing the temporal gauge $A_0 = 0$, we get

$$L = \frac{1}{2}\dot{q}^2 + \frac{1}{2}(q\dot{\varphi})^2 \tag{2.5.32}$$

and the Gauss law constraint

$$\phi = q^2 \dot{\phi} + c = 0. \tag{2.5.33}$$

Note that the conserved quantity $h := \frac{\partial L}{\partial \dot{\phi}} = q^2 \dot{\phi}$ appears in the Gauss law. The Gauss law constraint is the result of fixing the gauge action on the phase space. Thus it is interpreted as the moment map condition. Since the variable φ is cyclic coordinate, we can now apply the Routh reduction (2.5.29) and derive the reduced action. We find the new Lagrangian as the Routhian

$$R = \frac{1}{2} \left(\dot{q}^2 - \frac{c^2}{q^2} \right).$$
 (2.5.34)

Again this is exactly the DFF-model Lagrangian (2.5.2) (or (2.5.10)) as expected. Therefore upon the reduction procedure of the gauged mechanical model we get the conformal mechanics (DFF-model).

2.6 Black hole

An interesting connections between black holes and conformal mechanical modelds have been proposed in [65] ¹². Let us consider the d = 4 Einstein-Maxwell theory which has the action

$$S = \frac{1}{16\pi} \int d^4x \sqrt{-g} \left(R - F^2 \right).$$
 (2.6.1)

¹¹ The application of the Routh reduction in the gauged mechanical systems was discussed in [51].

¹²Also see [66] for the conjectural relation between black holes and the Calogero model.

The theory admits a single extreme Reissner-Nordström black hole solution with the metric in isotropic coordinate

$$ds^{2} = -\left(1 + \frac{|Q|l_{p}}{\rho}\right)^{-2} dt^{2} + \left(1 + \frac{|Q|l_{p}}{\rho}\right)^{2} \left(d\rho^{2} + \rho^{2} d\Omega^{2}\right)$$
(2.6.2)

and the gauge field

$$A = \left(1 + \frac{|Q|l_p}{\rho}\right)^{-1} dt \tag{2.6.3}$$

where *Q* is the black hole charge, l_p is the Planck length with the black hole mass $M = \frac{|Q|}{l_p}$, and $d\Omega^2 = d\theta^2 + \sin^2\theta d\varphi^2$ is the *SO*(3) invariant metric on *S*². In the near-horizon limit the metric (2.6.2) becomes the Bertotti-Robinson (BR) metric

$$ds^{2} = -\left(\frac{\rho}{|Q|l_{p}}\right)^{2} dt^{2} + \left(\frac{|Q|l_{p}}{\rho}\right)^{2} d\rho^{2} + \left(|Q|l_{p}\right)^{2} d\Omega^{2}, \qquad (2.6.4)$$

which is $SO(1,2) \times SO(3)$ invariant conformally flat metric on $AdS_2 \times S^2$. Defining the horospherical coordinate as $(t, \phi = \frac{\rho}{Ql_p})$ for AdS_2 part, we can express the BR metric (2.6.4) as

$$ds^{2} = -\phi^{2}dt^{2} + \frac{\left(|Q|l_{p}\right)^{2}}{\phi^{2}}d\phi^{2} + \left(|Q|l_{p}\right)^{2}d\Omega^{2}$$
(2.6.5)

where the quantity $|Q|l_p$ is interpreted as the S^2 radius and also as the radius of the curvature of the AdS₂ space. To go further, let us introduce a new radial coordinate *r* by

$$\phi = \left(\frac{2M}{r}\right)^2. \tag{2.6.6}$$

Putting together the black hole solutions (2.6.2) and (2.4.3) now become

$$ds^{2} = -\left(\frac{2M}{r}\right)^{4} dt^{2} + \left(\frac{2M}{r}\right)^{2} dr^{2} + M^{2} d\Omega^{2}, \qquad (2.6.7)$$

$$A = \left(\frac{2M}{r}\right)^2 dt \tag{2.6.8}$$

where we have chosen the unit so that $l_p = 1$ and $M = |Q_p|$.

Now we consider the test particle with mass m and charge q. The world-line action of the particle is

$$S = -m \int ds + q \int A. \tag{2.6.9}$$

Putting the black hole solutions (2.6.7) and (2.6.8) into (2.6.9), we find the action

$$S = \int dt \left(\frac{2M}{r}\right)^2 \left[q - m\sqrt{1 - \left(\frac{2M}{r}\right)^{-2}\dot{r}^2 - M^2\left(\frac{2M}{r}\right)^{-4}\left(\dot{\theta}^2 + \sin^2\theta\dot{\phi}^2\right)}\right].$$
(2.6.10)

The action is invariant under the conformal transformations [74]

$$\delta t = f(t) + c\left(\frac{1}{M^2}\right)r^4 = a + bt + ct^2 + c\left(\frac{1}{M^2}\right)r^4,$$
 (2.6.11)

$$\delta r = \frac{1}{2}\dot{f}r = \frac{1}{2}(b+2ct)r,$$
(2.6.12)

$$\delta\theta = \delta\varphi = 0. \tag{2.6.13}$$

The corresponding conformal generators, the Hamiltonian H, the dilatation operator D and the conformal boost operator K are given by

$$H = \left(\frac{2M}{r}\right)^2 \left[\sqrt{m^2 + \frac{r^2 p_r^2 + 4L^2}{4M^2}} - q\right]$$
$$= \frac{p_r^2}{2f} + \frac{m\gamma}{2r^2 f'}$$
(2.6.14)

$$D = -\frac{1}{4} (rp_r + p_r r), \qquad (2.6.15)$$

$$K = \frac{1}{2}fr^2$$
 (2.6.16)

where we have introduced

$$L^{2} = p_{\theta}^{2} + \frac{p_{\varphi}^{2}}{\sin^{2}\theta},$$
(2.6.17)

$$f = \frac{1}{2} \left[\sqrt{m^2 + \frac{1}{4M^2} \left(r^2 p_r^2 + 4L^2 \right)} + q \right], \qquad (2.6.18)$$

$$\gamma = 4M^2 \frac{m^2 - q^2}{m} + \frac{4L^2}{m}.$$
(2.6.19)

It can be shown that three generators H, D and K form the one-dimensional conformal $\mathfrak{sl}(2,\mathbb{R})$ algebra under the Poisson brackets.

It has been pointed out [65] that this conformal mechanical model (2.6.10) give rise to the DFF-model (2.1.2) in the specific limit ¹³. Considering the limit

$$M \to \infty$$
, $(m-q) \to 0$, $M^2(m-q) =$ fixed (2.6.20)

¹³However, the physical meaning of this particular limit is not clear and we will see that the mechanical model (2.6.10) and the DFF-model (2.1.2) can be realized as two different non-linear realizations of the one-dimensional conformal group $SL(2, \mathbb{R})$.

and noting that $f \rightarrow m$ in this limit, we obtain the DFF Hamiltonian

$$H = \frac{p_r^2}{2m} + \frac{\gamma}{2r^2}$$
(2.6.21)

with the coupling constant

$$\gamma = 8M^2(m-q) + \frac{4l(l+1)}{m}.$$
(2.6.22)

Here $l(l + 1), l \in \mathbb{Z}$ is the quantum number of the operator L^2 . Note that this quantization corresponds to the freezing of the S^2 angles, θ , φ , i.e. $\theta = \text{const.}, \varphi = \text{const.}$ Therefore the DFF-model (2.6.21) describes the radial motion of the AdS₂ × S^2 particle, i.e. the particle near the horizon of the extreme Reissner-Nordström black hole in the limit (2.6.20).

Let us discuss the procedure proposed by DFF to cure the problem of the absence of the ground state for the Hamiltonian *H* from the perspective of the particle motion near the black hole horizon. Firstly we see that the metric (2.6.5) is singular at $\phi = 0$, however, this is just a coordinate singularity and $\phi = 0$ is a non-singular degenerate Killing horizon of the time-like Killing vector field $\frac{\partial}{\partial t}$. To see this we recall the definition of the AdS₂ space as a Lobachevski-like embedded surface in a three dimensional Minkowski space (see Figure 2.8)

$$-(x^{0})^{2} + (x^{1})^{2} - (x^{2})^{2} = -R^{2}.$$
 (2.6.23)

Using the hypersurface coordinate (ϕ , t) defined by

$$x^0 = t\phi, \tag{2.6.24}$$

$$x^{+} = x^{2} + x^{1} = \frac{R^{2} - t^{2}\phi^{2}}{R\phi},$$
 (2.6.25)

$$x^{-} = x^{2} - x^{1} = R\phi, \qquad (2.6.26)$$

we obtain the AdS₂ factor of the BR metric (2.6.5) with $|Q|l_p = R$. The horospherical coordinates (t, ϕ) can only cover the half of the AdS₂ region. At $\phi = 0$ the metric (2.6.5) is singular and $\phi > 0$ or $\phi < 0$ should be chosen. Correspondingly the coordinate x^- is restricted to $x^- > 0$ or $x^- < 0$. Since the coordinate x^0, x^+ and x^- are smooth on the hypersurface, at the horizon $\phi = 0$ the time coordinate t is ill-defined. To avoid such situation, let us define new coordinates

$$t_1 = \frac{1}{2}(x^+ + x^-), \quad t_2 = x^0, \quad t = t_1 + it_2,$$
 (2.6.27)

$$r = \frac{1}{2}(x^{+} - x^{-}). \tag{2.6.28}$$



Figure 2.8: AdS₂ space viewed as a hyperboloid of one sheet in a three dimensional Minkowski space.

Then the equation (2.6.23) becomes

$$-|\mathfrak{t}|^2 + r^2 = -R^2 \tag{2.6.29}$$

and thus we can write

$$\mathfrak{t} = e^{\frac{i\tau}{R}} \sqrt{R^2 + r^2}$$
(2.6.30)

with $\tau \in \mathbb{R}$ being a new coordinate. In terms of these coordinates the BR metric (2.6.5) can be written as

$$ds^{2} = -\left(\frac{R^{2} + r^{2}}{R^{2}}\right)d\tau^{2} + \left(\frac{R^{2}}{R^{2} + r^{2}}\right)dr^{2} + R^{2}d\Omega^{2}.$$
 (2.6.31)

In fact this shows that the horizon is not a true singularity as we mentioned.

Now we want to get further insights of conformal mechanics from black hole viewpoint. As seen from the expression (2.6.24), the classical analog of an eigenstate vector of the Hamiltonian *H* in conformal mechanics is an orbit of a time-like Killing vector $k = \frac{\partial}{\partial t}$ in the AdS₂ region outside the horizon ($\phi \neq 0$) and the energy eigenvalue *E* is the value of k^2 . This implies that the ground state $|E = 0\rangle$ of *H* with E = 0 in conformal mechanics corresponds to the orbit of *k* with $k^2 = 0$ which is a null geodesic generator of the event horizon. Therefore the absence of the ground state $|E = 0\rangle$ can be interpreted as the fact that the orbit of $k^2 = 0$ cannot be covered by the static coordinate *t* as we discussed.

In classical general relativity it is a general procedure to demonstrate that the horizon is not a true singularity by changing the coordinate system. Note that the AdS₂ isometry *SO*(1, 2) is linearly realized on the coordinates (x^0, x^1, x^2) as rotations $\delta x^{\mu} = \Lambda^{\mu}{}_{\nu}x^{\nu}$ whose generators $J^{\mu\nu} = ix^{[\mu}\partial^{\nu]}$ form the $\mathfrak{so}(1, 2)$ algebra

$$[J^{\mu\nu}, J^{\rho\sigma}] = i \left(\eta^{\mu[\rho} J^{\sigma]\nu} - \eta^{\nu[\rho} J^{\sigma]\mu} \right)$$
(2.6.32)

with $\eta^{\mu\nu} = \text{diag}(-1, +1, -1)$. Then we can find new generators in our new coordinates (t_1, t_2, r) and the corresponding operators in the DFF-model as follows:

1. rotation: $t_1 \leftrightarrow t_2$

This rotation is expressed as the U(1) rotation of the complex coordinate t

$$\mathfrak{t} = e^{\frac{i\tau'}{R}}\sqrt{R^2 + r^2} = e^{i\alpha}\mathfrak{t}$$
(2.6.33)

with $\alpha \in \mathbb{R}$ being the infinitesimal parameter. and thus yields the time τ translation

$$\tau' = \tau + R\alpha. \tag{2.6.34}$$

Since it generates compact rotation in the non-compact SO(1,2) symmetry group, the corresponding generator $J^{t_1t_2}$ is identified with the L_0 in the conformal mechanics.

2. rotation: $(t_1, t_2) \leftrightarrow r$

In this case the rotations are expressed as two boost operations

$$\delta \mathfrak{t} = \beta, \qquad \qquad \delta r = \frac{1}{2} \left(\beta \mathfrak{t}^* + \beta^* \mathfrak{t}\right) \qquad (2.6.35)$$

where $\beta \in \mathbb{C}$ is the infinitesimal parameter. The complexified generator $J^{t_1r} \pm iJ^{t_2r}$ can be regarded as L_{\pm} in the DFF conformal mechanics.

Therefore from the black hole perspective the DFF prescription can be thought of as the different choice of time coordinates in which the world-lines of static particles can pass through the black hole horizon.

2.7 Non-linear realization

The non-linear realization [155, 156, 157, 172] is a useful method to construct the non-linear invariant Lagrangian. The basic idea is the following:

1. Start from the Lie (super)group *G* that reflects the symmetry in the theory.

- 2. Find the invariants under *G* from the Cartan forms ω belonging to the (super)coset *G*/*H* where *H* is the stability subgroup of *G*.
- 3. Construct the invariant Lagrangian under *G* in terms of the Goldstone fields associated with the (super)coset Cartan forms ω .

By making use of the non-linear realization, it has been showed [74] that the DFFmodel (2.1.2) and the black hole conformal mechanics (2.6.10) are essentially equivalent modulo redefinition of the time coordinates and the variables at classical level. Moreover the non-linear realization approach provides us with a powerful method to construct the irreducible supermultiplets for superconformal mechanical models. Much of the irreducible constraints and transformation laws can be automatically obtained from the non-linear realization technique.

Let *G* be a Lie (super)group and *H* be its subgroup. We call Y_i the generator of *H*. and X_i the generator of the remaining generators. We assume that the commutator $[X_i, Y_i]$ is a linear combination of X_i alone

$$[X_i, Y_j] = f_{ij}^k X_k (2.7.1)$$

where f_{ij}^k are the structure constants. (2.7.1) implies that the remaining generators X_i form the representation of the subgroup H, which we will call the stability subgroup. Then a group element g of G can be represented uniquely by [155, 156, 157, 172]

$$g = e^{x \cdot X} h$$

= $\tilde{g}h$ (2.7.2)

where *h* is an element of *H*, $x \cdot X := \sum_{i} x^{i} X_{i}$ and x^{i} are the coordinates parametrizing the coset space *G*/*H*. The actions of the group *G* can be realized by left multiplications on the coset *G*/*H*. This fact is the key statement of the non-linear realization method.

Now we want to apply the basic statement (2.7.2) to find the non-linear realization of *G* symmetry group and to construct the *G*-invariant Lagrangian. In order to achieve this, we classify the parameters x^i in two classes

$$x^{i} = \begin{cases} (\text{super}) \text{space coordinates} & \text{if } X_{i} \text{ is (super) translation} \\ \text{Goldstone (super) fields} & \text{otherwise.} \end{cases}$$
(2.7.3)

In other words, the (super)space and time coordinates are the parameters of the (super)translation generators while the remaining coset parameters are treated as

the (super)fields. We should note that the number of the Goldstone (super)fields is not always same as the number of the coset generators. In fact some of the Goldstone (super)fields may be expressed by other Goldstone (super)fields. This phenomenon is known as the inverse Higgs effect [173].

For one-dimensional conformal algebra $\mathfrak{sl}(2,\mathbb{R})$ given by (2.1.23)-(2.1.25), the stability subgroup *H* is trivial and thus the coset is parametrized by the coordinates for all generators

$$\tilde{g} = e^{itH}e^{iz(t)K}e^{iu(t)D}.$$
(2.7.4)

Since we are now considering one-dimensional field theory, i.e. quantum mechanics, we introduce time coordinate t for the Hamiltonian H. The remaining two coordinates z(t) for K and u(t) for D are Goldstone fields.

Then we can find the realization of the conformal group in our coset (2.7.4). The translation *H* is realized by acting on *g* by $g_0 = e^{iaH}$ from the left

$$g_0 \cdot \tilde{g} = e^{iaH} \cdot e^{itH} e^{iz(t)K} e^{iu(t)D}.$$
(2.7.5)

We thus obtain the translations

$$\delta t = a, \tag{2.7.6}$$

$$\delta u = 0, \qquad \qquad \delta z = 0. \qquad (2.7.7)$$

The dilatation *D* is realized by acting on *g* by $g_0 = e^{ibD}$ from the left

$$g_{0} \cdot \tilde{g} = e^{ibD} \cdot e^{itH} e^{iz(t)K} e^{iu(t)D}$$

$$= (e^{ibD} e^{itH} e^{-ibD})(e^{ibD} e^{izK} e^{-ibD})e^{ibD} e^{iuD}$$

$$= e^{i(t+bt)H} e^{i(z+bz)K} e^{i(u+b)D}.$$
(2.7.8)

One finds the dilatations

$$\delta t = bt, \tag{2.7.9}$$

$$\delta u = b, \qquad \qquad \delta z = -bz. \qquad (2.7.10)$$

The conformal boost *K* is realized by acting on *g* by $g_0 = e^{icK}$ from the left

$$g_0 \cdot \tilde{g} = e^{icK} \cdot e^{itH} e^{izK} e^{iuD}$$

= $e^{itH} (e^{-itH} e^{icK} e^{itH}) e^{izK} e^{iuD}$
= $e^{i(t+ct^2)H} e^{i(z+c-2ctz)K} e^{i(u+2ct)D}$. (2.7.11)

We get the conformal boost transformations

$$\delta t = ct^2, \tag{2.7.12}$$

$$\delta u = 2ct, \qquad \qquad \delta z = c - 2ctz. \qquad (2.7.13)$$

Let us discuss the construction of the *G*-invariant expressions. To this end we introduce the Maurer-Cartan form Ω for the coset *G*/*H* defined by

$$\Omega = \tilde{g}^{-1} d\tilde{g}$$

= $e^{-x \cdot X} d(e^{x \cdot X})$
= $i \omega^i X_i + i \tilde{\omega}^i Y_i$. (2.7.14)

Then one can show [155, 156, 157, 172] that the forms ω^i on the coset transform homogeneously and therefore any expression constructed with ω^i is invariant under *G*. On the other hand it turns out [155, 156, 157, 172] that the forms $\tilde{\omega}^i$ on the stability subgroup *H* transform like connections and can be used to construct covariant derivatives.

The Maurer-Cartan forms for the coset (2.7.4) is

$$\Omega = \tilde{g}^{-1}d\tilde{g} = i\left(\omega_H H + \omega_K K + \omega_D D\right)$$
(2.7.15)

where

$$\omega_H = e^{-u}, \qquad (2.7.16)$$

$$\omega_D = du - 2zdt, \tag{2.7.17}$$

$$\omega_K = e^u \left(dz + z^2 dt \right). \tag{2.7.18}$$

Alternatively the Maurer-Cartan forms associated with the generators T_i , i = 0, 1, 2 defined (2.1.35) are given by

$$\omega_0 = \frac{1}{m} \omega_K + m \omega_K, \qquad (2.7.19)$$

$$\omega_1 = \frac{1}{m} \omega_K - m \omega_H, \qquad (2.7.20)$$

$$\omega_2 = \omega_D \tag{2.7.21}$$

where *m* is a constant parameter. Since the form ω_1 , ω_2 are the coset forms, they transform homogeneously. we can impose the following $SL(2,\mathbb{R})$ invariant conditions ¹⁴

$$\omega_1 = 0,$$
 (2.7.22)

$$\omega_2 = 0.$$
 (2.7.23)

¹⁴ Although the choice of $\omega_0 = 0$ also yields the $SL(2, \mathbb{R})$ invariant constraint, it does not lead to the good dynamical systems.

The first condition (2.7.22) turns out to be the equation of motion for the system and the second condition (2.7.23) leads to the relation

$$z = \frac{1}{2}\dot{u},$$
 (2.7.24)

which implies that the Goldstone field z(t) can be represented by the other Goldstone field u(t). This is the inverse Higgs effect [173]. In terms of the remaining Maurer-Cartan forms ω_0 , one can construct the $SL(2,\mathbb{R})$ action [174]

$$S = -c \int \omega_0$$

= $-c \int dt \left[\frac{c}{m} e^u \left(\dot{z} + z^2 \right) + mc e^{-u} \right]$
= $\int dt \left[x^2 - \frac{c^2}{x^2} \right]$ (2.7.25)

where we have used the relation (2.7.24) and introduced

$$x := \mu^{-\frac{1}{2}} e^{\frac{\mu}{2}}, \qquad (2.7.26)$$

$$\mu = \frac{m}{c} \tag{2.7.27}$$

The action (2.7.25) is just the DFF-model (2.1.4).

Let us introduce

$$\hat{K} = mK - \frac{1}{m}H,$$
 $\hat{D} = mD$ (2.7.28)

where *m* is a constant parameter. Then the one-dimensional conformal $\mathfrak{sl}(2, \mathbb{R})$ algebra can be written as

$$[H, \hat{D}] = imH,$$
 (2.7.29)

$$[\hat{K}, \hat{D}] = -2iH - im\hat{K}, \qquad (2.7.30)$$

$$[H, \hat{K}] = 2i\hat{K}.$$
 (2.7.31)

Defining the corresponding coset by

$$\tilde{g} = e^{i\tau H} e^{i\phi(\tau)\hat{D}} e^{i\Omega(\tau)\hat{K}}$$
(2.7.32)

and acting the corresponding elements on the coset (2.7.32) from the left, one can find the $SL(2,\mathbb{R})$ transformations for the new coordianates

$$\delta \tau = a + b + c\tau + \frac{1}{m^2} c e^{2m\phi},$$
 (2.7.33)

$$\delta\phi = \frac{1}{m} \left(b + 2c\tau \right), \qquad (2.7.34)$$

$$\delta\Omega = \frac{1}{m}ce^{m\phi} \tag{2.7.35}$$

where *a*, *b*, *c* are infinitesimal constant parameters. Note that the transformation of the new time coordinate τ contains the additional term $\frac{1}{m}ce^{2m\phi}$, which is similar to (2.6.11). We can read the Maurer-Cartan forms for the coset (2.7.32) [74]

$$\hat{\omega}_H = \frac{1+\Lambda^2}{1-\Lambda^2} e^{-m\phi} d\tau - 2\frac{\Lambda}{1-\Lambda^2} d\phi, \qquad (2.7.36)$$

$$\hat{\omega}_D = \frac{1 + \Lambda^2}{1 - \Lambda^2} d\phi - 2 \frac{\Lambda}{1 - \Lambda^2} e^{-m\phi} d\tau, \qquad (2.7.37)$$

$$\hat{\omega}_{K} = m \frac{\Lambda}{1 - \Lambda^{2}} \left(\Lambda e^{-m\phi} d\tau - d\phi \right) + \frac{d\Lambda}{1 - \Lambda^{2}}$$
(2.7.38)

where

$$\Lambda = \tanh \Omega. \tag{2.7.39}$$

Let us impose the $SL(2, \mathbb{R})$ invariant conditions as

$$\hat{\omega}_D = 0, \qquad (2.7.40)$$

which results in the inverse Higgs effect [173]

$$\partial_{\tau}\phi = 2e^{-m\phi}\frac{\Lambda}{1+\Lambda^2}.$$
(2.7.41)

So the Goldstone field Λ or Ω can be expressed by ϕ

$$\Lambda = \partial_\tau \phi e^{m\phi} \frac{1}{1 + \sqrt{1 - e^{2m\phi}(\partial_\tau \phi)^2}}.$$
(2.7.42)

Using the non-vanishing Maurer-Cartan forms, one can construct the $SL(2, \mathbb{R})$ invariant action [74]

$$S = \int \left[(q - \tilde{\mu})\hat{\omega}_H - \frac{2}{m}q\hat{\omega}_K \right]$$

= $-\int d\tau e^{-m\phi} \left[\tilde{\mu}\sqrt{1 - e^{2m\phi}(\partial_\tau \phi)^2} - q \right].$ (2.7.43)

We see that the action (2.7.43) is the conformal mechanical model (2.6.10) which describes the radial motion of the $AdS_2 \times S^2$ particle [65].

Therefore we see that the two mechanical model (2.6.10) and the DFF-model (2.1.2) can be realized as two different non-linear realizations of the one-dimensional conformal group $SL(2, \mathbb{R})$. From this point of view, we can conclude that the two conformal mechanical models are equivalent up to the redefinition of the time coordinate and the physical variable.

2.8 Multi-particle conformal mechanics

Let us study the conformal mechanical models with many degrees of freedom for different particles. Generically *n*-particle quantum mechanics can be viewed as a sigma-model with an *n*-dimensional target space \mathcal{M} . So we will see the conditions [69] for the target space \mathcal{M} for the existence of conformal operators D and K.

Consider the Hamiltonian

$$H = \frac{1}{2} p_a^{\dagger} g^{ab} p_b + V(x).$$
 (2.8.1)

Here $g_{ab}(x)$ is the metric of the target space \mathcal{M} where the indices $a, b = 1, \dots, n$ label the particles.

$$p_a = g_{ab} \dot{x}^b \tag{2.8.2}$$

are the canonical momenta obeying

$$[x^{a}, p_{b}] = i\delta_{ab}, \qquad [x^{a}, x^{b}] = 0, \qquad [p_{a}, p_{b}] = 0.$$
(2.8.3)

The Hermitian conjugate of p_a are

$$p_a^{\dagger} = \frac{1}{\sqrt{g}} p_a \sqrt{g}$$
$$= p_a - i\Gamma_{ba}^b$$
(2.8.4)

where Γ_{ab}^{c} is the Christoffel symbol constructed from g_{ab} .

Let us assume that the theory has a dilatational invariance of the form

$$\delta t = bt, \qquad \qquad \delta x^a = \frac{1}{2} D^a(x)b, \qquad (2.8.5)$$

which is a generalization of (2.1.14) and (2.1.15) with *b* being an infinitesimal parameter for the dilatation. Then the dilatation generator *D* is given by

$$D = \frac{1}{4} \left(D^a p_a + p_a^{\dagger} D^{a \dagger} \right).$$
 (2.8.6)

Under the canonical relations (2.8.3) we find the commutation relation of the Hamiltonian (2.8.1) and the dilatation generator (2.8.6) as [69]

$$[H,D] = \frac{i}{4} p_a^{\dagger} \left(\mathcal{L}_D g^{ab} \right) p_b + \frac{i}{2} \mathcal{L}_D V + \frac{i}{8} \nabla^2 \nabla_a D^a$$
(2.8.7)

where \mathcal{L}_d is the Lie derivative

$$\mathcal{L}_{D}g_{ab} = D^{c}g_{ab,c} + D^{c}{}_{,a}g_{cb} + D^{c}{}_{,b}g_{ac}.$$
 (2.8.8)

From the $\mathfrak{sl}(2,\mathbb{R})$ algebra (2.1.23) and the expressions (2.8.1), (2.8.7), the existence of the dilatation generator *D* requires that

$$\mathcal{L}_D g_{ab} = 2g_{ab}, \tag{2.8.9}$$

$$\mathcal{L}_D V(x) = -2V(x),$$
 (2.8.10)

$$\nabla^2 \nabla_a D^a = 0. \tag{2.8.11}$$

A vector field *D* is called homothetic vector field or similarity vector field on \mathcal{M} ¹⁵. A homothetic vector field generates a similarity transformation group. It is shown that along any integral curve of a homothetic vector field the space-like, time-like or null character of the tangent vector does not change and that there is necessarilly a singularity in each orbit of the similarity transformation group [175, 176, 177].

Furthermore the remaining commutation relations (2.1.24) and (2.1.25) lead to [69]

$$\mathcal{L}_D K = 2K, \tag{2.8.13}$$

$$D_a dx^a = dK \tag{2.8.14}$$

respectively. As the solutions to the equations (2.8.13) and (2.8.14), one can express the conformal boost generator K as the norm of D^a

$$K = \frac{1}{2}g_{ab}D^a D^b.$$
 (2.8.15)

The equation (2.8.14) means that the one-form $D = D_a dx^a$ is exact, however, it is shown [69] that closed homothety vector field D is always exact. So it is enough to impose the closeness condition for the homothetic vector field D

$$d(D_a dx^a) = 0. (2.8.16)$$

Therefore we can conclude that in order to obtain conformal quantum mechanical sigma-models,

• the target space \mathcal{M} must admit a homothety vector field D whose associated one-form $D_a dx^a$ is closed ¹⁶; (2.8.9) and (2.8.16)

¹⁵Note that

$$X = \begin{cases} \text{conformal Killing field} & \text{if } \mathcal{L}_X g_{ab} = \rho(x) g_{ab} \\ \text{homothetic vector field} & \text{if } \mathcal{L}_X g_{ab} = c g_{ab} \\ \text{Killing vector field} & \text{if } \mathcal{L}_X g_{ab} = 0 \end{cases}$$
(2.8.12)

where $\rho(x)$ is a function on \mathcal{M} and *c* is a constant on \mathcal{M} .

¹⁶The vector field D with the required properties for conformal mechanical sigma-model is referred to as a closed homothety vector field in [69]

- the potential V(x) must satisfy (2.8.10)
- *D^a* must obey the vanishing condition (2.8.11).

2.9 Calogero model

One of the most celebrated multi-particle conformal mechanical models is the Calogero model, which is the system of multi-particles scattering on the line with inverse-square potentials [134, 135]. The Hamiltonian is given by

$$H = \sum_{i=1}^{n} \frac{1}{2} p_i^2 + \sum_{i < j} \frac{\gamma}{(x_i - x_j)^2}$$
(2.9.1)

where the indices $i = 1, \dots, n$ label the particles and γ is a coupling constant.

In what follows we will discuss that the Calogero model and its generalization can be obtained from gauged matrix models. This formulation not only indicates the intimite relationship between the conformal quantum mechanical models and the gauged quantum mechanical models but also provides us with non-trivial (super)conformal mechanical models.

Let us start with the gauged matrix model action

$$S = \int dt \left[\operatorname{Tr}(DXDX) + \frac{i}{2} \left(\overline{Z}DZ - D\overline{Z}Z \right) + c \operatorname{Tr} A \right].$$
 (2.9.2)

Here $X_a^b(t)$, $\overline{X}_a^b = X_b^a$, $a = 1, \dots, n$ are the bosonic Hermitian $(n \times n)$ matrices, $Z_a(t)$, $\overline{Z}^a = (\overline{Z}_a)$ are bosonic complex matrices and $A_a^b(t)$, $(\overline{A}_a^b) = A_b^a$ are the U(n) gauge fields with n^2 component fields. c is a real constant parameter. In the action (2.9.2) the covariant derivatives are defined as

$$DX := \dot{X} + i[A, X], \qquad DZ := \dot{Z} + iAZ, \qquad D\overline{Z} := \overline{Z} - i\overline{Z}A.$$
 (2.9.3)

Note that in the third term, the Fayet-Iliopoulos term the non-abelian traceless part of the gauge field A drops out and only the U(1) part has contributions.

The action (2.9.2) is invariant under the one-dimensional $SL(2,\mathbb{R})$ conformal transformations

$$\delta t = f(t), \qquad \qquad \delta \partial_0 = -\dot{f} \partial_0, \qquad (2.9.4)$$

$$\delta X = \frac{1}{2}\dot{f}X, \qquad \delta Z = 0, \qquad \delta A = -\dot{f}A \qquad (2.9.5)$$

where $f(t) = a + bt + ct^2$ with *a*, *b*, *c* being infinitesimal real parameters. The action (2.9.3) is invariant under the U(n) gauge transformations

$$X \to g X g^{-1}, \tag{2.9.6}$$

$$Z \to gZ$$
, $\overline{Z} \to \overline{Z}g^{-1}$, (2.9.7)

$$A \to gAg^{-1} + i\dot{g}g^{-1} \tag{2.9.8}$$

where $g \in U(n)$. Let us impose a partial gauge fixing condition

$$X_a^b = x_a \delta_a^b \tag{2.9.9}$$

where x_a are real component fields since X are Hermitian matrices. Then the action (2.9.3) becomes

$$S = \int dt \sum_{a,b} \left[\dot{x}_{a} \dot{x}_{a} + \frac{i}{2} \left(\overline{Z}^{a} \dot{Z}_{a} - \dot{\overline{Z}}^{a} Z_{a} \right) + (x_{a} - x_{b})^{2} A_{a}^{b} A_{b}^{a} - \overline{Z}^{a} A_{a}^{b} Z_{b} + c A_{a}^{a} \right].$$
 (2.9.10)

By noting that the action (2.9.10) is invariant under the $U(1)^n$ gauge transformations

$$x_a \to x_a, \tag{2.9.11}$$

$$Z_a \to e^{i\lambda_a} Z_a, \qquad \qquad \overline{Z}^a \to e^{-i\lambda_a} \overline{Z}^a, \qquad (2.9.12)$$

$$A_a^a \to A_a^a - \dot{\lambda}_a \tag{2.9.13}$$

where $\lambda_a(t)$ are local parameters, we further impose the gauge fixing condition as

$$Z_a = \overline{Z}^a. \tag{2.9.14}$$

Then the action (2.9.10) reduces to

$$S = \int dt \sum_{a,b} \left[\dot{x}_a^2 + (x_a - x_b)^2 A_a^b A_b^a - Z_a Z_b A_a^b + c A_a^a \right].$$
(2.9.15)

At this stage we attempt to integrate out the gauge field *A*. From the action (2.9.15) we obtain the equations of motion for A_a^a and for $A_{a,a}^b \neq b$ as

$$(Z_a)^2 = c (2.9.16)$$

$$A_b^a = \frac{Z_a Z_b}{2(x_a - x_b)^2}.$$
 (2.9.17)

Substituting the equations (2.9.16) and (2.9.17) into the action (2.9.15) and rescaling x_a appropriately, we obtain the Calogero model action

$$S = \frac{1}{2} \int dt \left[\sum_{a} \dot{x}_{a}^{2} - \sum_{a \neq b} \frac{c^{2}}{(x_{a} - x_{b})^{2}} \right].$$
 (2.9.18)

Chapter 3

Superconformal Mechanics

In this chapter we will proceed to the superextension of the conformal quantum mechanics; the superconformal quantum mechanics. Firstly in section 3.1 we will recall the basic facts about Lie superalgebra and Lie supergroup and will clarify the one-dimensional superconformal group. Then in section 3.2 we will stress that supersymmety in one-dimension possesses many peculiar properties. In section 3.3, 3.4, 3.5 and 3.6 we will review the persistent efforts to construct $\mathcal{N} = 1$, 2, 4 and 8 superconformal quantum mechanics by using the superspace and superfield formalism and also review the interesting topics which are relevant to those superconformal mechanical models.

3.1 Superalgebra and supergroup

In *d*-dimensional superconformal field theories the ordinary supersymmetry and the conformal symmetry lead to a second supersymmetry. The corresponding generator S_{α} with α, β, \cdots being spinor indices can be found by taking the commutator of the conformal boost operator K_{μ} with space-time indices $\mu, \nu, \cdots = 0, 1, \cdots, d-1$ and the original supersymmetry Q_{α}

$$[K_{\mu}, Q_{\alpha}] = (\Gamma_{\mu})^{\beta}_{\ \alpha} S_{\beta} \tag{3.1.1}$$

where Γ_{μ} is a *d*-dimensional gamma matrix. Additionally the ani-commutator of supersymmetries Q_{α} and S_{α} generates the bosonic symmetry, the so-called Rsymmetry. In general these generators form the superconformal algebras which are isomorphic to the simple Lie superalgebras. Hence it is expected that one can specify the corresponding Lie superalgebras, i.e. the superconformal algebras which characterize the superconformal field theories.

3.1.1 Lie superalgebra

A superalgebra is a \mathbb{Z}_2 -graded algebra $\mathfrak{g} = \mathfrak{g}_{\overline{0}} \oplus \mathfrak{g}_{\overline{1}}$. This means that if $a \in \mathfrak{g}_{\alpha}$, $b \in \mathfrak{g}_{\beta}$, $\alpha, \beta \in \mathbb{Z}_2 = \{\overline{0}, \overline{1}\}$, then $ab \in \mathfrak{g}_{\alpha+\beta}$. We say that a is of degree α and write deg $a = \alpha$. $\mathfrak{g}_{\overline{0}}$ is a Lie algebra, which is called the even or bosonic part of \mathfrak{g} while $\mathfrak{g}_{\overline{1}}$ is called the odd or fermionic part of \mathfrak{g} , which is not an algebra.

A Lie superalgebra is the superalgebra endowed with the product operation [,] possessing the following axioms:

1. graded anticommutativity

$$[a,b] = -(-1)^{\alpha\beta}[b,a]$$
(3.1.2)

2. generalized Jacobi identity

$$[a, [b, c]] = [[a, b], c] + (-1)^{\alpha \beta} [b, [a, c]]$$
(3.1.3)

where $a \in \mathfrak{g}_{\alpha}$, $b \in \mathfrak{g}_{\beta}$. The product [a, b] is referred to as the Lie superbracket or supercommutator for two elements $a, b \in \mathfrak{g}$.

Let $V = V_{\overline{0}} \oplus V_{\overline{1}}$ be \mathbb{Z}_2 -graded vector space where dim $V_{\overline{0}} = m$ and dim $V_{\overline{1}} = n$. Then the algebra EndV is endowed with a \mathbb{Z}_2 -graded superalgebra structure. Hence the Lie superbracket [,] satisfying (3.1.2) and (3.1.3) turns EndV into a Lie superalgebra l(m, n). The Lie superalgebra l(V) plays the same role as the general linear Lie algebra in the theory of Lie algebra. Let $e_1, \dots, e_m; e_{m+1}, \dots, e_{m+n}$ be a basis of V, formed by the bases of $V_{\overline{0}}$ and $V_{\overline{1}}$. In this basis the matrices of an element a from the Lie superalgebra l(m, n) can be written in the form

$$a = \begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix}$$
(3.1.4)

where α and δ are $\mathfrak{gl}(m)$ and $\mathfrak{gl}(n)$ matrices and β and γ are $m \times n$ and $n \times m$ rectangular matrices. On the Lie superalgebra $\mathfrak{gl}(m, n)$ the supertrace is defined by

$$\operatorname{str}(a) = \operatorname{tr} \alpha - \operatorname{tr} \delta. \tag{3.1.5}$$

In terms of the supertrace (3.1.5), we can define the bilinear form B_R associated with the representation R of \mathfrak{g} by

$$B_R(a,b) = \operatorname{str}(R(a), R(b)), \quad \forall a, b \in \mathfrak{g}$$
(3.1.6)

where R(a) is the matrix of the elements $a \in \mathfrak{g}$ in the representation R. As a special case the Killing form K can be defined as the bilinear form on \mathfrak{g} associated with the adjoint representation

$$K(a,b) = \operatorname{str}(\operatorname{ad}(a),\operatorname{ad}(b)), \quad \forall a,b \in \mathfrak{g}.$$
(3.1.7)

The Lie superalgebra \mathfrak{g} is called simple if it contains no non-trivial ideal. The Lie superalgebra \mathfrak{g} is called semi-simple if it contains no non-trivial solvable ideal. If a Lie superalgebra $\mathfrak{g} = \mathfrak{g}_{\overline{0}} \oplus \mathfrak{g}_{\overline{1}}$ is simple, the representation of $\mathfrak{g}_{\overline{0}}$ on $\mathfrak{g}_{\overline{1}}$ is faithful and $\{\mathfrak{g}_{\overline{1}}, \mathfrak{g}_{\overline{1}}\} = \mathfrak{g}_{\overline{0}}$. If the representation of $\mathfrak{g}_{\overline{0}}$ on $\mathfrak{g}_{\overline{1}}$ is irreducible, then \mathfrak{g} is simple. Unlike the Lie algebras, semi-simple Lie superalgebra cannot be written as the direct sum of simple Lie superalgebras. However, there is a construction which allows us to build finite-dimensional semi-simple Lie superalgebras in terms of simple ones [178].

It is known that simple Lie superalgebras are classified into two families; the classical Lie superalgebras and (non-classical) Cartan type superalgebras. The simple Lie superalgebra is said to be classical the representation of the Lie algebra $\mathfrak{g}_{\overline{0}}$ on $\mathfrak{g}_{\overline{1}}$ is completely reducible.

For the classical Lie superalgebras there are further classifications. Firstly the representation of $\mathfrak{g}_{\overline{0}}$ on $\mathfrak{g}_{\overline{1}}$ can be either (i) irreducible or (ii) the direct sum of two irreducible representations of $\mathfrak{g}_{\overline{0}}$. The superalgebra of the case (i) is called the type I and that of the case (ii) is called type II. In addition, the Lie superalgebra \mathfrak{g} is called basic if there is a non-degenerate invariant bilinear form, the Killing form K on \mathfrak{g} while strange if it is not basic. The basic Lie superalgebras is divided into (a) four infinite series: A(m,n), B(m,n), C(n) and D(m,n), that is $\mathfrak{sl}(m+1|n+1)$, $\mathfrak{osp}(2m+1|2n)$, $\mathfrak{osp}(2|2n)$ and $\mathfrak{osp}(2m|2n)$; (b) three exceptional series: 40-dimensional F(4), 31-dimensional G(3) and 17-dimensional $D(2,1;\alpha)$ which is a one-parameter family of superalgebras. The strange algebras split into two infinite families P(n) and Q(n).

For the Cartan type superalgebras there are four infinite families W(n), S(n), $H(n) \tilde{S}(n)$, where the first three series are analogous to the corresponding series of simple infinite-dimensional Lie algebra of Cartan type and $\tilde{S}(n)$ is a deformation of S(n).

Summarizing the above, the classification of simple Lie superalgebra is illustrated in Figure 3.1.

3.1.2 Lie supergroup

To begin with, let us introduce a supermatrix. A supermatrix *M* is defined as the matrix whose entries valued in a Grassmann algebra $\Gamma = \Gamma_{\overline{0}} \oplus \Gamma_{\overline{1}}$ of the form

$$M = \left(\begin{array}{cc} A & B \\ C & D \end{array}\right) \tag{3.1.8}$$



Figure 3.1: The classification of simple Lie superalgebra.

where *A*,*B*,*C* and *D* are $m \times p$, $m \times q$, $n \times p$ and $n \times q$ matrices respectively. The supermatrix *M* is said to be even and of degree 0 if $A, D \in \Gamma_{\overline{0}}$ and $B, C \in \Gamma_{\overline{1}}$ whereas it is called odd and of degree 1 if $A, D \in \Gamma_{\overline{1}}$ and $B, C \in \Gamma_{\overline{0}}$.

The general linear supergroup GL(m|n) consists of even invertible supermatrices *M* and its product is defined by the multiplication rule of the supermatrices:

$$(MN)_{ij} = \sum_{k=1}^{p+q} M_{ik} N_{kj}$$
(3.1.9)

where *M* and *N* are two $(m + n) \times (p + q)$ and $(p + q) \times (r + s)$ supermatrices and $(MN)_{ij}$ denotes the (i, j) entry of the $(m + n) \times (r + s)$ supermatrix *MN*.

The operations for the supermatrices are defined as follows:

1. transpose M^t and supertranspose M^{st}

$$M^{t} = \begin{pmatrix} A^{t} & C^{t} \\ B^{t} & D^{t} \end{pmatrix},$$

$$M^{st} = \begin{pmatrix} A^{t} & (-1)^{\deg M} C^{t} \\ -(-1)^{\deg M} B^{t} & D^{t} \end{pmatrix} = \begin{cases} \begin{pmatrix} A^{t} & C^{t} \\ -B^{t} & D^{t} \\ A^{t} & -C^{t} \\ B^{t} & D^{t} \end{pmatrix} & \text{if } M \text{ is even} \\ \begin{pmatrix} A^{t} & -C^{t} \\ B^{t} & D^{t} \end{pmatrix} & \text{if } M \text{ is odd} \end{cases}$$

$$(3.1.11)$$

2. supertrace str(M)

$$\operatorname{str}(M) = \operatorname{tr}(A) - (-1)^{\operatorname{deg} M} \operatorname{tr}(D) = \begin{cases} \operatorname{tr}(A) - D & \text{if } M \text{ is even} \\ \operatorname{tr}(A) + D & \text{if } M \text{ is odd} \end{cases}$$
(3.1.12)

3. superdeterminant sdet(M)

$$sdet(M) = \frac{det(A - BD^{-1}C)}{det(D)} = \frac{det(A)}{det(D - CA^{-1}B)}$$
 (3.1.13)

4. adjoint M^{\dagger} and superadjoint M^{\ddagger}

$$M^{\dagger} = (M^{t})^{*}, \qquad (3.1.14)$$

$$M^{\ddagger} = (M^{st})^*. \tag{3.1.15}$$

The relation between the Lie superalgebra \mathfrak{g} and the corresponding Lie supergroup *G* is analogous to the theory of the Lie algebra. Consider the complex Grassmann algebra $\Gamma(n)$ of order *n* with *n* generators $1, \theta_1, \dots, \theta_n$ obeying the anticommutation relations $\{\theta_i, \theta_j\} = 0$. If in the element $\eta = \sum_{0 \le m} \sum_{i_1 < \dots i_m} \eta_{i_1 \dots i_m} \theta_{i_1} \dots \theta_{i_m}$ each complex coefficient $\eta_{i_1 \dots i_m}$ is an even (odd) value of *m*, the corresponding element is called even (odd). In general $\Gamma(n)$ can be decomposed into even and odd parts as a vector space; $\Gamma(n) = \Gamma(n)_{\overline{0}} \oplus \Gamma(n)_{\overline{1}}$. The Grassmann envelope $G(\Gamma)$ of the Lie superalgebra \mathfrak{g} is constructed as a formal linear combinations $\sum_i \eta_i a_i$ where a_i is a basis of \mathfrak{g} and $\eta_i \in \Gamma$ such that the elements a_i and η_i are both even or odd. The Lie supergroup *G* associated with the superalgebra \mathfrak{g} is realized as the exponential mapping of the Grassmann envelope $G(\Gamma)$ of \mathfrak{g} ; the even generators of the superalgebra \mathfrak{g} corresponds to commuting parameters, i.e. even elements of the Grassmann algebra \mathfrak{g} to anticommuting parameters, i.e. odd elements of the Grassmann algebra \mathfrak{g} to anticommuting parameters, i.e. odd elements of the Grassmann algebra \mathfrak{g} to anticommuting parameters, i.e. odd elements of the Grassmann algebra \mathfrak{g} to anticommuting parameters, i.e.

3.1.3 Superconformal algebra

The requirements for the corresponding superconformal algebra have been proposed in [12]:

- 1. The *d*-dimensional conformal algebra $\mathfrak{so}(d, 2)$ should appear as a bosonic factored subgroup.
- 2. The fermionic generators should be spinor representations of the conformal algebra $\mathfrak{so}(d, 2)$.

First of all we can see that these conditions can be satisfied for the simple classical Lie superalgebras. The detail list of the classical Lie superalgebras is given in Table 3.1 [180, 181, 182, 183, 184].

The unitary superalgebra A(m-1, n-1) or $\mathfrak{sl}(m, n)$ with m > n > 0 possesses an even part $\mathfrak{sl}(m) \oplus \mathfrak{sl}(n) \oplus \mathfrak{u}(1)$ and an odd part $(m, \overline{n}) \oplus (\overline{m}, n)$ as a representation of the even part. The unitary superalgbra A(n-1, n-1) with n > 1 has an even part $\mathfrak{sl}(n) \oplus \mathfrak{sl}(n)$ and an odd part $(n, \overline{n}) \oplus (\overline{n}, n)$.

The orthosymplectic superalgebras consist of three infinite series B(m, n), C(n + 1) and D(m, n). The superalgebra B(m, n) or $\mathfrak{osp}(2m + 1|2n)$ with $m \ge 0$, $n \ge 1$ possesses an even part $\mathfrak{so}(2m + 1) \oplus \mathfrak{sp}(2n)$ and an odd part (2m + 1, 2n). The superalgebra C(n + 1) or $\mathfrak{osp}(2|2n)$ with $n \ge 1$ contains an even part $\mathfrak{so}(2) \oplus \mathfrak{sp}(2n)$ and an odd part $\mathfrak{so}(2) \oplus \mathfrak{sp}(2n)$ and an odd part $\mathfrak{so}(2n \oplus 2n)$ as twice the fundamental representation 2n of $\mathfrak{sp}(2n)$. The superalgebra D(m, n) or $\mathfrak{osp}(2m|2n)$ with $m \ge 2$, $n \ge 1$ has an even part $\mathfrak{so}(2m) \oplus \mathfrak{sp}(2n)$ and an odd part (2m, 2n).

The superalgebras $D(2,1;\alpha)$ with $\alpha \neq 0, -1, \infty$ is a one-parameter family of superalgebras of rank 3 and dimension 17. It is a deformation of the superalgebra D(2,1) that corresponds to the case of $\alpha = 1$. It has an even part $\mathfrak{sl}(2) \oplus \mathfrak{sl}(2) \oplus \mathfrak{sl}(2)$ and an odd part (2,2,2) as the spinor representations of $\mathfrak{sl}(2) \oplus \mathfrak{sl}(2) \oplus \mathfrak{sl}(2)$. The three $\mathfrak{sl}(2)$ factors appear as the anticommutator of the fermionic generators with the relative weights $1, \alpha$ and $1 - \alpha$.

The superalgebra F(4) is 40-dimensional algebra of rank 4 and possesses an even part $\mathfrak{sl}(2) \oplus \mathfrak{o}(7)$ and an odd part $(\mathbf{2}, \mathbf{8})$ as the spinor representations of $\mathfrak{sl}(2) \oplus \mathfrak{o}(7)$.

The superalgebra G(3) is 31-dimensional algebra of rank 3 and has an even part $\mathfrak{sl}(2) \oplus G_2$ and an odd part (2,7) as the representations of $\mathfrak{sl}(2) \oplus G_2$.

By scanning through the list in Table 3.1, we can find the superconformal algebras which satisfy the required conditions. For d = 1 superconformal field theory, that is superconformal quantum mechanics, the bosonic conformal algebra is $\mathfrak{so}(1,2) = \mathfrak{sl}(2,\mathbb{R}) = \mathfrak{su}(1,1) = \mathfrak{sp}(2)$ and richer superconformal structures are

g	₿ō	$\mathfrak{g}_{\overline{1}}$	K	type
A(m-1,n-1)	$A_{m-1}\oplus A_{n-1}\oplus \mathfrak{u}(1)\delta_{m,n}$	$(\boldsymbol{m}, \overline{\boldsymbol{n}})$	basic	Ι
		$\oplus(\overline{m},n)$		
$\int \mathfrak{su}(m-p,p n-q,q)$	$\mathfrak{su}(m-p,p)\oplus\mathfrak{su}(n-q,q)\oplus\mathfrak{u}(1)\delta_{m,n}$		basic	I
$\mathfrak{su}^*(2m 2n)$	$\mathfrak{su}^*(2m)\oplus\mathfrak{su}^*(2n)\oplus\mathfrak{so}(1,1)\delta_{m,n}$		basic	I
$\mathfrak{sl}'(n n)$	$\mathfrak{sl}(n,\mathbb{C})$		basic	I
B(m,n)	$B_m\oplus C_n$	(2m+1,2n)	basic	II
	$m \ge 0, n \ge 1$			
C(n+1)	$C_m \oplus \mathfrak{u}(1)$	$2n \oplus 2n$	basic	I
	$n \ge 1$			
D(m,n)	$D_m\oplus C_n$	(2m, 2n)	basic	II
	$m \ge 2, n \ge 1, m \ne n+1$			
$\mathfrak{osp}(m-p,p n)$	$\mathfrak{so}(m-p,p)\oplus\mathfrak{sp}(n)$		basic	II
$\mathfrak{osp}(m^* n-q,q)$	$\mathfrak{so}^*(m)\oplus\mathfrak{usp}(n-q,q)$		basic	II
$D(2,1;\alpha)$	$A_1\oplus A_1\oplus A_1$	(2, 2, 2)	basic	II
	$0 < lpha \leq 1$			
$D^{p}(2, 1; \alpha)$	$\mathfrak{so}(4-p,p)\oplus\mathfrak{sl}(2)$		basic	II
F(4)	$A_1 \oplus B_3$	(2,8)	basic	I
$F^p(4)$	$\mathfrak{so}(7-p,p)\oplus\mathfrak{sl}(2)$		basic	I
	p = 0, 3			
$F^p(4)$	$\mathfrak{so}(7-p,p)\oplus\mathfrak{su}(2)$		basic	I
	p = 1, 2			
G(3)	$A_1\oplus G_2$	(2,7)	basic	I
$G_p(3)$	$\mathfrak{g}_{2,p}\oplus\mathfrak{sl}(2)$		basic	I
	p = -14, 2			
P(m-1)	$\mathfrak{sl}(m)$	$(\boldsymbol{m}\otimes \boldsymbol{m})$	strange	I
	$m \ge 3$			
Q(m-1)	$\mathfrak{su}(m)$	adjoint	strange	II
	$m \ge 3$			
Q(m-1)	$\mathfrak{sl}(m)$		strange	II
$Q((m-1)^*)$	$\mathfrak{su}^*(m)$		strange	II
UQ(p,m-1-p)	$\mathfrak{su}(p,m-p)$		strange	II

Table 3.1: The list of the classical Lie superalgebras $\mathfrak{g} = \mathfrak{g}_{\overline{0}} \oplus \mathfrak{g}_{\overline{1}}$ with Killing forms *K*.

allowed due to the small conformal group. Note that $\mathfrak{so}(1,2)$ may be contained as an even part $\mathfrak{g}_{\overline{0}}$ for the series of the Lie superalgebra $\mathfrak{g} = \mathfrak{osp}(m - p, p|n)$ and thus the corresponding R-symmetry algebras are the series of the non-compact $\mathfrak{sp}(n)$. Therefore if we consider the classical Lie superalgebras with compact Rsymmetry algebras, the corresponding supergroups can be represented in terms of supermatrices as

$$\left(\begin{array}{cc} SL(2,\mathbb{R}) & B\\ C & \text{R-symmetry} \end{array}\right), \qquad (3.1.16)$$

$$\begin{pmatrix} SU(1,1) & B \\ C & \text{R-symmetry} \end{pmatrix}, \qquad (3.1.17)$$

$$\begin{pmatrix} Sp(2) & B \\ C & \text{R-symmetry} \end{pmatrix}$$
(3.1.18)

where *B* and *C* are fermionic matrices. Two supermatrices (3.1.17) and (3.1.18) correspond to the infinite series of the Lie superalgebra and provide us chains of the one-dimensional superconformal groups. The remaining supermatrices (3.1.16) may cover the exceptional Lie superalgebras and other special cases. The one-dimensional superconformal groups are tabulated in Table 3.2 [183, 184, 136].

In the cases of $\mathcal{N} < 4$ supersymmetry the superconformal groups are essentially unique series of $OSp(2|\mathcal{N})$ as the isomorphism $SU(1,1|1) \cong OSp(2|2)$ is taken into account.

For $\mathcal{N} = 4$ supersymmetry the structure of the superconformal group becomes large as the exceptional Lie superalgebra $D(2,1;\alpha)$ is a one-parameter family. Note that SU(1,1|2) for $\mathcal{N} = 4$ case is not simple as SU(m,n|m+n) is not even semi-simple. The quotient $PSU(1,1|2) \cong SU(1,1|2)/U(1)$ is simple and we denote it just by SU(1,1|2). As D(2,1;-1) and D(2,1;0) are semi-direct product $SU(1,1|2) \rtimes SU(2)$ and they are not simple, they are excluded in the Table 3.2.

With $\mathcal{N} = 8$ supersymmetry one-dimensional superconformal groups can be realizes as four different supergroups; OSp(8|2), SU(1,1|4), $OSp(4^*|4)$ and F(4).

When the highly extended supersymmetry with $\mathcal{N} > 8$ exists in the quantum mechanics, one can have three distinct series of one-dimensional superconformal groups for even \mathcal{N} ; $OSp(\mathcal{N}|2)$, $SU(1,1|\frac{\mathcal{N}}{2})$ and $OSp(4^*|\frac{\mathcal{N}}{2})$. The supergroup $OSp(4^*|\frac{\mathcal{N}}{2})$ is the exceptional series which does not appear in the theories with fewer supersymmetries. It has an even part $SO^*(4) \times USp(\mathcal{N})$ where the non-compact bosonic subgroup $SO^*(4) \cong SL(2,\mathbb{R}) \times SU(2)$ contain the one-dimensional conformal group $SL(2,\mathbb{R})$.

supersymmetry	supergroup	R-symmetry	
$\mathcal{N} = 1$	<i>OSp</i> (1 2)	1	
$\mathcal{N}=2$	<i>SU</i> (1,1 1)	<i>U</i> (1)	
$\mathcal{N}=3$	OSp(3 2)	<i>SU</i> (2)	
$\mathcal{N}=4$	<i>SU</i> (1,1 2)	<i>SU</i> (2)	
	$D(2,1;\alpha), \alpha \neq -1, 0,$	$SU(2) \times SU(2)$	
$\mathcal{N} = 5$	OSp(5 2)	SO(5)	
$\mathcal{N}=6$	<i>SU</i> (1,1 3)	$SU(3) \times U(1)$	
	OSp(6 2)	SO(6)	
$\mathcal{N}=7$	OSp(7 2)	SO(7)	
	G(3)	G_2	
$\mathcal{N}=8$	OSp(8 2)	SO(8)	
	SU(1, 1 4)	$SU(4) \times U(1)$	
	$OSp(4^* 4)$	$SU(2) \times SO(5)$	
	F(4)	SO(7)	
$\mathcal{N} > 8$	$OSp(\mathcal{N} 2)$	$SO(\mathcal{N})$	
	$SU(1,1 \frac{N}{2})$	$SU(\frac{N}{2}) \times U(1)$	
	$OSp(4^* rac{\mathcal{N}}{2})$	$SU(2) \times Sp(\frac{N}{2})$	

Table 3.2: The simple classical Lie supergroups that contain the one-dimensional conformal group $SL(2, \mathbb{R})$ as a factored bosonic subgroup. For $\mathcal{N} > 8$ superconformal quantum mechanics there are three different superconformal groups.
3.2 One-dimensional supersymmetry

Now we want to discuss the concrete construction of superconformal quantum mechanical models. To this end we should note that in one dimension the supersymmetry is realized containing various peculiarities which do not appear in higher dimensional cases regardless of whether a conformal symmetry exists or not. It is known that the supersymmetry of a sigma-model imposes strong restrictions on its target space. However, the restrictions of one-dimensional supersymmetric sigma-models are generically weaker than higher dimensional sigma-models. In other words, more couplings among the fields are allowed in one dimension. This is because in higher dimensional cases the Lorentz symmetry rules out particular couplings, however, in one dimension there is no Lorentz symmetry group and much more couplings are possible. Moreover we cannot expect the relation between the number of bosonic and fermionic fields as in higher dimensional supersymmetric field theories.

3.2.1 Supermultiplet

One of the most powerful methods to construct supersymmetric quantum mechanics is to appeal the superspace and superfield formalism. In what follows we will consider a particularly reasonable class of supermultiplets [185, 186, 187] and discuss how many components we need to realize the \mathcal{N} -extended superalgebra ¹ which satisfy

$$[\delta_{\epsilon^A}, \delta_{\epsilon^B}] = -2i\epsilon^A \epsilon^B \partial_t \tag{3.2.1}$$

where $A, B, \dots = 1, \dots, N$ denote the R-symmetry indices and ϵ^A are a set of real anti-commuting supersymmetry parameters.

Now consider the scalar multiplets Φ which consist of a set of d physical bosons $x_i(t)$ and a set of d fermions $\psi_i(t)$ where $i = 1, \dots, d$ and $\hat{i} = 1, \dots, d$ denote the multiplicities, i.e. the numbers of the bosons and the fermions respectively and suppose that their supersymmetric transformations are given by

$$\delta_{\epsilon^A} x_i = -i\epsilon^A (L_A)_i^{\hat{j}} \psi_{\hat{j}}, \qquad (3.2.2)$$

$$\delta_{\epsilon^A}\psi_{\hat{i}} = \epsilon^A (R_A)_{\hat{i}}^{\hat{j}} \dot{x}_{\hat{j}}$$
(3.2.3)

where $(L_A)_i^{j}$ and $(R_A)_i^{j}$ are real $d \times d$ matrices. Then the algebra (3.2.1) imposes

¹Also see [188, 189, 190] for the classification of the supermultiplets.

constraints on the matrices L_A and R_A as

$$(L_A R_B + L_B R_A)_i^{\ j} = -2\delta_{AB}\delta_j^j, \tag{3.2.4}$$

$$(R_A L_B + R_B L_A)_{\hat{i}}^{\hat{j}} = -2\delta_{AB}\delta_{\hat{i}}^{\hat{j}}.$$
(3.2.5)

From the algebraic point of view there is no relationship between two matrices L_A and R_A , however, if we require that the kinetic action for the scalar multiplet Φ with the form

$$S = \int dt \left[\frac{1}{2} \dot{x}_i^2 - \frac{i}{2} \psi_{\hat{t}} \dot{\psi}_{\hat{t}} \right]$$
(3.2.6)

is invarinat under the supersymmetric transformations (3.2.1), we obtain the relation

$$(L_A^T)^{\hat{i}j} = -(R_A)^{\hat{i}j}.$$
(3.2.7)

Likewise let us consider the spinor multiplets Ψ which are composed of a set of *d* real fermions λ_i and a set of *d* real bosons y_i possess the supersymmetry transformations

$$\delta_{\epsilon^A} \lambda_{\hat{i}} = \epsilon^A (R_A)_{\hat{i}}^{j} y_{j}, \qquad (3.2.8)$$

$$\delta_{\epsilon^A} y_i = -i\epsilon^A (L_A)_i{}^j \dot{\lambda}_{\hat{j}}.$$
(3.2.9)

Then one finds the same constraints for the two matrices L_A and R_A as (3.2.4) and (3.2.5). In addition if we require that the quadratic part of the action for the spinor multiplet Ψ

$$S = \int dt \left[-\frac{i}{2} \lambda_{\hat{i}} \dot{\lambda}_{\hat{i}} + \frac{1}{2} y_i y_i \right]$$
(3.2.10)

is invariant under the supersymmetry transformations (3.2.8) and (3.2.9), then we find the precisely same relation as (3.2.7).

Hence the existence of the scalar supermultiplets Φ and Ψ is rooted in the algebra ² defined by three conditions (3.2.4), (3.2.5) and (3.2.7). It is known that there is a minimal value of *d*, called d_N for which N linearly independent real $d \times d$ matrices L_A and R_A satisfying the relations (3.2.4), (3.2.5) and (3.2.7) exist. We see that d_N translates into the minimal number of the bosonic or fermionic

²In [186, 187, 191] this algebra of dimension *d* and rank \mathcal{N} is called $\mathcal{GR}(d, \mathcal{N})$ algebra since the one of the two matrices, say L_A satisfies a general real (\mathcal{GR}) Pauli algebra (3.2.4), (3.2.5) with the other matrix R_A determined by the relation (3.2.7).

п	1	2	3	4	5	6	7	8
$\log_2 n$	0	1	$\log_2 3$	2	$\log_2 5$	$\log_2 6$	log ₂ 7	3
r	0	1	2	2	3	3	3	3
$\rho(2^r)$	1	2	4	4	8	8	8	8

Table 3.3: Hurwitz-Radon function $\rho(2^r)$ where *r* is the nearest integer greater than or equal to $\log_2 n$.

\mathcal{N}	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
$d_{\mathcal{N}}$	1	2	4	4	8	8	8	8	16	32	64	64	128	128	128	128

Table 3.4: Then minimal numbers d_N of the component fields in the N-extended supermultiplets.

component fields in the supermultiplets for a given the number of supersymmetry \mathcal{N} . The value of $d_{\mathcal{N}}$ is given by [189, 187]

$$d_{\mathcal{N}} = 16^m \rho(2^r) \tag{3.2.11}$$

where the number of supersymmetry is written as a mod8 decomposition

$$\mathcal{N} = 8m + n. \tag{3.2.12}$$

Here $\rho(2^r)$ is the so-called Hurwitz-Radon function [192, 193] define by ³

$$\rho(2^{r}) = \begin{cases} 2r+1 & n \equiv 0 \mod 4 \\ 2r & n \equiv 1, 2 \mod 4 \\ 2r+2 & n \equiv 3 \mod 4. \end{cases}$$
(3.2.14)

with *r* being taken as the nearest integer greater than or equal to $\log_2 n$ (see Table 3.3). The results are summarized in Table 3.4 From Table 3.4 one can see that when $\mathcal{N} = 1, 2, 4, 8$ the minimal numbers $d_{\mathcal{N}}$ of the component fields coincide with

$$\left(a_1^2 + \dots + a_{\rho}^2\right)\left(b_1^2 + \dots + b_{2^r}^2\right) = c_1^2 + \dots + c_{2^r}^2$$
(3.2.13)

hold where a_1, \dots, a_ρ and b_1, \dots, b_{2^r} are the independent indeterminates and c_i is a bilinear form in a_1, \dots, a_ρ and b_1, \dots, b_{2^r} . The Hurwitz-Radon function also appears in topology [194] and linear algebra [195]. See also [196, 197, 198].

³The Hurwitz-Radon function $\rho(2^r)$ yields the largest integer ρ for which the square identities can

the numbers \mathcal{N} of supersymmetries. As we will see in the following, the superspace and superfield formalism works well for these four cases. Note that when $\mathcal{N} > 8$ the minimal numbers $d_{\mathcal{N}}$ of the supermultiplets are greater than the numbers \mathcal{N} of supersymmetries and the corresponding supermultiplets become much more complicated and the superspace and superfield formalism is unsuccessful at present.

3.2.2 Automorphic duality

One of the most significant features in one-dimensional supersymmetric field theories, i.e. quantum mechanical models is the fact that the the equal number of bosonic and fermionic physical degrees of freedom, which is valid in higher dimensional field theories, does not take place. This is because in one dimension there is the duality which allows us to convert any physical field to auxiliary field and vice versa [186, 187, 191]. Consequently even if we consider the $\mathcal{N} = 1, 2, 4, 8$ supersymmetric cases, where $d_{\mathcal{N}} = \mathcal{N}$ is realized, a number of supermultiplets can be constructed in one-dimension.

To see this let us take the most basic d = 1 $\mathcal{N} = 1$ superalgebra

$$[\delta_{\epsilon_1}, \delta_{\epsilon_2}] = -2i\epsilon_1\epsilon_2\partial_t. \tag{3.2.15}$$

We introduce $\mathcal{N} = 1$ superspace $\mathbb{R}^{(1|1)}$ parametrized by

$$\mathbb{R}^{(1|1)} = (t,\theta) \tag{3.2.16}$$

where *t* is time and θ is a real Grassmann coordinate. The covariant superderivative *D* is defined by ⁴

$$D = i\frac{\partial}{\partial\theta} - \theta\frac{\partial}{\partial t}, \qquad \{D, D\} = -2i\partial_t \qquad (3.2.17)$$

and the supercharge Q is realized as

$$Q = i\frac{\partial}{\partial\theta} + \theta\frac{\partial}{\partial t}, \qquad \{Q, Q\} = 2i\partial_t \qquad (3.2.18)$$

in the superspace.

In this case there are two irreducible representations of (3.2.15); the scalar multiplet Φ and the spinor multiplet Ψ . The scalar multiplet contains a real bosonic field *x* as the lowest component and a real fermion ψ as the highest component

⁴This convention yields $\{Q, Q\} = 2H$ and leads to simple forms of the supersymmetric Lagrangian and its supersymmetric transformation.

while the spinor multiplet Ψ includes a real fermion λ as the lowest component and a real boson *y* as the highest component. Namely the multiplets can be described by

$$\Phi = x + i\theta\psi, \tag{3.2.19}$$

$$\Psi = \lambda + \theta y. \tag{3.2.20}$$

The supersymmetry transformation laws for the scalar multiplet Φ are $\delta \Phi = -i[\epsilon Q, \Phi]$, which yield

$$\delta_{\epsilon} x = i\epsilon\psi, \qquad (3.2.21)$$

$$\delta_{\epsilon}\psi = \epsilon \dot{x} \tag{3.2.22}$$

and those for the spinor multiplet Ψ are $\delta \Psi = -i[\epsilon Q, \Psi]$, which give rise to

$$\delta_{\epsilon}\lambda = \epsilon y, \tag{3.2.23}$$

$$\delta_{\epsilon} y = i\epsilon \dot{\lambda}. \tag{3.2.24}$$

One can write the supersymmetric action for the scalar multiplet Φ as

$$S = -\frac{1}{2} \int dt d\theta \ D\Phi \dot{\Phi}$$
(3.2.25)

and also write that for the spinor multiplet Ψ as

$$S = -\frac{i}{2} \int dt d\theta \Psi D\Psi. \qquad (3.2.26)$$

In component fields the above supersymmetric action (3.2.25) and (3.2.26) can be expressed by

$$S = \frac{1}{2} \int dt \left[\dot{x}^2 + i \dot{\psi} \psi \right] \tag{3.2.27}$$

and

$$S = \frac{1}{2} \int dt \left[i\lambda\lambda + y^2 \right]$$
(3.2.28)

respectively.

As described in [187], there is a useful operation which maps between the two irreducible $\mathcal{N}=1$ multiplets

$$-D\Phi \leftrightarrow \Psi.$$
 (3.2.29)

In component fields this map is realized by performing the following replacements

$$(\dot{x}, \psi) \leftrightarrow (y, \lambda).$$
 (3.2.30)

We see that the supersymmetry transformations (3.2.21) for the scalar multiplet and the transformations (3.2.23) for the spinor multiplet are exchanged under the replacement (3.2.29) and that the action (3.2.27) for the scalar multiplet and the action (3.2.28) for the spinor multiplet transform into the other under the operation (3.2.29). Therefore a map (3.2.29) or (3.2.30) is the operation which replace a scalar multiplet Φ with a spinor multiplet Ψ and vice-verse. This is called the automorphic duality (AD) map because the operation corresponds to the automorphism on the space of the representations of the superalgebra. Intriguingly the AD map (3.2.30) make it possible to convert the physical field *x* into the auxiliary field *y* and vice versa. It has been pointed out [191] that this remarkable property in quantum mechanics can be interpreted as the Hodge duality in one-dimension. In general the Hodge duality maps a differential *p*-form Ω_p in *d*-dimension into a differential (d - p - 2)-form Ω_{d-p-2} in *d*-dimension by the Hodge star operation as

$$*: d\Omega_p \to d\Omega_{d-p-2}. \tag{3.2.31}$$

If we consider a scalar field, a zero-form in one dimension, then the Hodge duality (3.2.31) gives rise to a dual (-1)-form. Formally the exterior derivative of a 0-form or a scalar x is a (-1)-form. Therefore if we denote the component field of the (-1)-form by y, we then get the relation

$$\dot{x} = y. \tag{3.2.32}$$

This is just the AD map given in (3.2.30).

According to the existence of the AD map in quantum mechanics, we will use the notation $(n, \mathcal{N}, \mathcal{N} - n)$ for $\mathcal{N} = 1, 2, 4, 8$ supermultiplets. Here the first entry denoted by n is the number of physical bosons in the supermultiplet, the second number \mathcal{N} represents the number of fermions which is equal to the number of supersymmetry and the last one $\mathcal{N} - n$ is the number of bosonic auxiliary fields. Using this notation, the $\mathcal{N} = 1$ scalar multiplet Φ is (1, 1, 0) and the spinor multiplet Ψ is (0, 1, 1).

3.3 $\mathcal{N} = 1$ Superconformal mechanics

3.3.1 One particle free action

Consider the $\mathcal{N} = 1$ *n* particle quantum mechanical system which is described by the *n*-dimensional scalar superfield (**1**, **1**, **0**). In general the $\mathcal{N} = 1$ superfield can

be thought of as a map from the superspace $\mathbb{R}^{(1|1)}$ to the target space \mathcal{M} . In terms of component fields we can write the multiplet as

$$\Phi^{i}(t,\theta) = x^{i}(t) + i\theta\psi^{i}(t)$$
(3.3.1)

where $i, j, \dots = 1, \dots, n$. Also consider the (0, 1, 1) spinor superfield Ψ^a which is a section of the bundle on \mathcal{M} with rank k given by

$$\Psi^{a}(t,\theta) = \lambda^{a}(t) + \theta y^{a}(t)$$
(3.3.2)

where $a, b, \dots = 1, \dots, k$. We attach the mass dimension as the following:

$$[t] = -1, \qquad [\theta] = -\frac{1}{2}, \\ [\Phi] = 0, \qquad [\Psi] = \frac{1}{2}, \\ [D] = \frac{1}{2}, \qquad [\partial_t] = 1. \qquad (3.3.3)$$

Then the most general $\mathcal{N} = 1$ action with dimensionless couplings up to cubic terms is given by ⁵

$$S = \int dt d\theta \left[-\frac{1}{2} g_{ij} D \Phi^i \dot{\Phi}^j + \frac{i}{3!} c_{ijk} D \Phi^i D \Phi^j D \Phi^k - \frac{i}{2} h_{ab} \Psi^a \nabla \Psi^b + \frac{1}{3!} l_{abc} \Psi^a \Psi^b \Psi^c + f_{ia} \dot{\Phi}^i \Psi^a + \frac{i}{2} m_{iab} \Psi^a \Psi^b D \Phi^i + \frac{i}{2} n_{ija} D \Phi^i D \Phi^j \Psi^a \right]$$
(3.3.4)

where g_{ij} is a metric on \mathcal{M} and h_{ab} is a fibre metric on the bundle. The covariant derivative for the fermions are defined by

$$\nabla \Psi^a = D\Psi^a + D\Phi^i (A_i)^a{}_b \Psi^b \tag{3.3.5}$$

with $(A_i)^a{}_b$ being the connection on the bundle.

Note that for the one particle case where the corresponding target space $\mathcal{M} = \mathbb{R}$ has no non-trivial bundle over it, the $\mathcal{N} = 1$ superspace action is described by just a free action (3.2.27). This corresponds to the statement that it is not possible to construct one-particle OSp(1|2) superconformal quantum mechanics with inverse-square type potential [65, 67, 137].

⁵See also [199, 70] for the $\mathcal{N} = 1$ superfield action.

3.3.2 Multi-particle model

Let us focus on the sigma-model action constructed only from the (1, 1, 0) scalar supermultiplet Φ^i [199, 200, 69, 136] ⁶

$$S = \int dt d\theta \left[-\frac{1}{2} g_{ij} D \Phi^i \dot{\Phi}^j + \frac{i}{3!} c_{ijk} D \Phi^i D \Phi^j D \Phi^k \right]$$

=
$$\int dt \left[\frac{1}{2} g_{ij} \dot{x}^i \dot{x}^j + \frac{i}{2} \psi^i \left(g_{ij} \frac{D \psi^j}{dt} - \dot{x}^k c_{ijk} \psi^j \right) - \frac{1}{6} \partial_l c_{ijk} \psi^l \psi^i \psi^j \psi^k \right]$$
(3.3.6)

where the covariant derivative is defined as

$$\frac{D\psi^i}{dt} := \dot{\psi}^i + \dot{x}^j \Gamma^i_{jk} \psi^k \tag{3.3.7}$$

with Γ_{ik}^i being the Christoffel symbol on \mathcal{M} .

Instead of the space-time indices *i* for the fermions ψ^i we shall introduce the tangent space indices $\alpha = 1, \dots, n$ by redefining the fermions ψ^{α} as

$$\psi^i = e^i{}_{\alpha}\psi^{\alpha}. \tag{3.3.8}$$

Note that ψ^{α} commute with x^i and p^i while ψ^i does not commute with x^i and p^i

$$[p_i, \lambda^j] = -i \left(\omega_{ik}^j - \Gamma_{ik}^j \right) \psi^k.$$
(3.3.9)

Then the action (3.3.6) can be written as

$$S = \int dt \left[\frac{1}{2} g_{ij} \dot{x}^{i} \dot{x}^{j} + \frac{i}{2} \left(\delta_{\alpha\beta} \psi^{\alpha} \dot{\psi}^{\beta} + \dot{x}^{i} \omega_{i\alpha\beta} \psi^{\alpha} \psi^{\beta} \right) - \frac{i}{2} \dot{x}^{i} c_{i\beta\gamma} \psi^{\alpha} \psi^{\beta} - \frac{1}{6} e^{l}_{\ \delta} \partial_{l} c_{ijk} e^{i}_{\ \alpha} e^{j}_{\ \beta} e^{k}_{\ \gamma} \psi^{\delta} \psi^{\alpha} \psi^{\beta} \psi^{\gamma} \right]$$
(3.3.10)

where ω is the spin connection and $c_{i\alpha\beta} := c_{ijk}e^{j}{}_{\alpha}e^{k}{}_{\beta}$. From the fermionic kinetic terms in the action (3.3.10) we see that the covariant derivatives of the fermions contains the connection with torsion *c*. Although this is similar to the two-dimensional (1,0) supersymmetric sigma models [201], the torsion *c* here is not necessarily closed as opposed to two-dimensional case. This indicates that there exist new supermultiplets in one dimension which have no higher-dimensional ancestors. The canonical momenta p_i is expressed as

$$p_{i} = g_{ij}\dot{x}^{j} + \frac{i}{2} \left(\omega_{ijk} - c_{ijk}\right)\psi^{j}\psi^{k}$$
(3.3.11)

⁶The (**1**, **1**, **0**) supermultiplet is also called $\mathcal{N} = 1B$ superfield.

where $\omega_{ijk} := \omega_{i\gamma}^{\ \beta} e_{j\beta} e_k^{\ \gamma}$. The action (3.3.6) is invariant under the supersymmetry transformations

$$\delta x^i = -i\epsilon\psi^i,\tag{3.3.12}$$

$$\delta \psi^i = \epsilon \dot{x}^i. \tag{3.3.13}$$

By means of the Noether's method we find the supercharge

$$Q = \psi^i \Pi_i - \frac{i}{3} c_{ijk} \psi^i \psi^j \psi^k$$
(3.3.14)

where we have defined

$$\Pi_i = g_{ij} \dot{x}^j. \tag{3.3.15}$$

Note that the supercharge Q is Hermitian though Π_i is not Hermitian. Using the canonical relation for the fermions

$$\left\{\psi^{\alpha},\psi^{\beta}\right\} = \delta^{\alpha\beta} \tag{3.3.16}$$

and the relations (2.8.3) for bosons, one finds

$$\{Q,Q\} = 2H. \tag{3.3.17}$$

where the Hamiltonian is

$$H = \frac{1}{2} p_a^{\dagger} g^{ab} p_b, \qquad (3.3.18)$$

which agrees with the bosonic sigma-model Hamiltonian (2.8.1) with the bosonic potential V(x) vanishing.

At this stage we consider the condition so that the theory (3.3.10) is the OSp(1|2) superconformal quantum mechanics. The corresponding $\mathfrak{osp}(1|2)$ superalgebra is characterized by the following (anti)commutation relations:

$$[H,D] = iH,$$
 $[K,D] = -iD,$ $[H,K] = 2iD,$ (3.3.19)

$$[Q, H] = 0,$$
 $[Q, D] = -\frac{i}{2}Q,$ $[Q, K] = -iS,$ (3.3.20)

$$[S,H] = iQ,$$
 $[S,D] = \frac{i}{2}S,$ $[S,K] = 0,$ (3.3.21)

$$\{Q,Q\} = 2H,$$
 $\{Q,S\} = -2D,$ $\{S,S\} = 2K.$ (3.3.22)

From the commutation relation (3.3.20) and the expressions (3.3.14) and (2.8.15) we can read the superconformal charge *S*

$$S = \psi^i D_i \tag{3.3.23}$$

where D_i has been introduced in (2.8.5) as the generalized dilatation. From the anti-commutator (3.3.22) we obtain the modified dilatation generator

$$D = \frac{1}{4} \left(D^i \Pi_i + \Pi^\dagger D_i^\dagger \right) \tag{3.3.24}$$

with p_i being replaced with Π_i . Using the new dilatation generator (3.3.24), $[S, D] = \frac{i}{2}S$ is satisfied, however, [Q, D] yields

$$[Q,D] = -\frac{i}{2}Q - \frac{i}{2}c_{ijk}D^{i}\psi^{j}p^{k} + \mathcal{O}(\psi^{3}).$$
(3.3.25)

Thus the OSp(1|2) superconformal symmetry imposes the condition so that the second quadratic term in ψ must vanish

$$D^i c_{ijk} = 0,$$
 (3.3.26)

which means that *c* is orthogonal to *D*. With the constraint (3.3.26), the commutator (3.3.25) becomes

$$[Q,D] = -\frac{i}{2}Q - \frac{1}{12}\psi^{i}\psi^{j}\psi^{k}\left(\mathcal{L}_{D} - 2\right)c_{ijk},$$
(3.3.27)

which implies that

$$\mathcal{L}_D c_{ijk} = 2c_{ijk}.\tag{3.3.28}$$

Then one can check that the remaining (anti)commutation relations (3.3.19)-(3.3.22) are satisfied and there are no further constraints for the OSp(1|2) symmetry imposed on the target space \mathcal{M} .

Therefore the conditions so that the $\mathcal{N} = 1$ sigma-model action (3.3.10) realizes the OSp(1|2) superconformal quantum mechanics are the conformal condition (2.8.9), (2.8.16) and the additional two constraints on the torsion *c*

$$D^{i}c_{ijk} = 0, (3.3.29)$$

$$\mathcal{L}_D c_{ijk} = 2c_{ijk}.\tag{3.3.30}$$

3.3.3 Gauged superconformal mechanics

As a generalization of the gauged mechanics (2.5.3) for the DFF-model and the gauged matrix model (2.9.2) for the Calogero model, we will discuss the superextension of the $\mathcal{N} = 1$ supersymmetric gauged mechanical model. As we will see this gauging procedure allows for the explicit construction of the non-trivial

 $\mathcal{N} = 1$ superconformal quantum mechanics [152, 137]. Consider the matrix superfield gauged mechanics action

$$S = -i \int dt d\theta \left[\operatorname{Tr} \left(\nabla_t \mathcal{X} \mathcal{D} \mathcal{X} \right) + \frac{i}{2} \left(\overline{\mathcal{Z}} \mathcal{D} \mathcal{Z} - \mathcal{D} \overline{\mathcal{Z}} \mathcal{Z} \right) + c \operatorname{Tr} \mathcal{A} \right].$$
(3.3.31)

Here we have introduced

- the *N* = 1 Grassmann-even Hermitian *n* × *n* matrix superfield *X*^b_a(*t*, *θ*) which satisfies (*X*)[†] = *X* and transforms as the adjoint representation of *U*(*n*)
- the $\mathcal{N} = 1$ Grassmann-even complex superfield $\mathcal{Z}_a(t, \theta)$ which satisfies $\overline{\mathcal{Z}} = \mathcal{Z}^{\dagger}$ and transform as the fundamental representation of U(n)
- the $\mathcal{N} = 1$ Grassmann-odd anti-Hermitian $n \times n$ matrix superfield $\mathcal{A}_a^b(t, \theta)$ which satisfies $(\mathcal{A})^{\dagger} = -\mathcal{A}$ and transforms as the adjoint representation of U(n).

The covariant derivatives are defined by

$$\nabla_t \mathcal{X} = D\mathcal{X} + i[\mathcal{A}_t, \mathcal{X}], \qquad (3.3.32)$$

$$\mathcal{DX} = \mathcal{DX} + i[\mathcal{A}, \mathcal{X}], \qquad (3.3.33)$$

$$\mathcal{DZ} = \mathcal{DZ} + i\mathcal{AZ} \tag{3.3.34}$$

where 7

$$D = \frac{\partial}{\partial \theta} + i\theta \frac{\partial}{\partial t}, \qquad \{D, D\} = 2i\partial_t, \qquad (3.3.35)$$

$$\mathcal{A}_t = -iD\mathcal{A} - \mathcal{A}\mathcal{A}. \tag{3.3.36}$$

The superconformal boost transformations are found to be

$$\delta t = -i\eta\theta t, \qquad \qquad \delta\theta = \eta t, \qquad (3.3.37)$$

$$\delta(dtd\theta) = -i\eta\theta(dtd\theta), \qquad \qquad \delta D = i\eta\theta D, \qquad (3.3.38)$$

$$\delta \mathcal{X} = -i\eta \theta \mathcal{X}, \qquad \qquad \delta \mathcal{A} = i\eta \theta \mathcal{A}, \qquad (3.3.39)$$

$$\delta \mathcal{Z} = 0. \tag{3.3.40}$$

The action (3.3.31) is invariant under the U(n) gauge transformations

$$\mathcal{X} \to e^{i\Lambda} \mathcal{X} e^{-i\Lambda},$$
 (3.3.41)

$$\mathcal{Z} \to e^{i\Lambda} \mathcal{Z},$$
 (3.3.42)

$$\mathcal{A} \to e^{i\Lambda} \mathcal{A} e^{-i\Lambda} - i e^{i\Lambda} \left(D e^{-i\Lambda} \right),$$
 (3.3.43)

$$\mathcal{A}_t \to e^{i\Lambda} \mathcal{A}_t e^{-i\Lambda} - i e^{i\Lambda} \left(\partial_t e^{-i\Lambda} \right) \tag{3.3.44}$$

⁷Note that the notation here is different from (3.2.17).

where $\Lambda_a^b(t, \theta)$ is the Hermitian $n \times n$ matrix gauge parameter. The $\mathcal{N} = 1$ superfields \mathcal{X} , \mathcal{Z} and \mathcal{A} can be expanded in the component fields as

$$\mathcal{X}_a^b = x_a^b + i\theta\psi_a^b, \tag{3.3.45}$$

$$\mathcal{Z}_a = z_a + \theta \xi_a, \tag{3.3.46}$$

$$\mathcal{A}_a^b = i(\zeta_a^b + \theta A_a^b). \tag{3.3.47}$$

From the gauge transformation (3.3.43) we can fix the gauge so that

$$\mathcal{A}_a^b = i\theta(A_0)_a^b(t). \tag{3.3.48}$$

Inserting (3.3.48) into the action (3.3.31), performing the integration over θ and integrating out the auxiliary fields $\xi, \overline{\xi}$, we find the $\mathcal{N} = 1$ gauged superconformal matrix model action

$$S = \int dt \left[\operatorname{Tr} \left(DxDx \right) - i\operatorname{Tr} \left(\psi D\psi \right) + \frac{i}{2} \left(\overline{z}Dz - D\overline{z}z \right) + c\operatorname{Tr} A_0 \right]$$
(3.3.49)

where the covariant derivative is defined by

$$Dx = \dot{x} + i[A_0, x],$$
 $D\psi = \dot{\psi} + [A_0, \psi].$ (3.3.50)

Note that the action (3.3.49) is the supersymmetric generalization of (2.9.2) that describes the Calogero model.

Instead of the gauge choice (3.3.48), we can fix the gauge as

$$\mathcal{X}_a^b = \mathcal{X}_a \delta_a^b, \tag{3.3.51}$$

$$\mathcal{Z}_a = \overline{\mathcal{Z}}^u \tag{3.3.52}$$

as we have discussed in (2.9.9) and (2.9.14) for the bosonic gauged matrix model. In this gauge the theory contains n^2 real $\mathcal{N} = 1$ superfields \mathcal{A}_a^b , $a \neq b$ and \mathcal{X}_a while the superfields \mathcal{Z}_a and \mathcal{A}_a^a are auxiliary. The superfield action (3.3.31) reads [137]

$$S = -i \int dt d\theta \left[\sum_{a} \dot{\mathcal{X}}_{a} D \mathcal{X}_{a} + \frac{i}{2} \sum_{a} \left(\overline{\mathcal{Z}}^{a} D \mathcal{Z}_{a} - D \overline{\mathcal{Z}}^{a} \mathcal{Z}_{a} \right) - i \sum_{a,b} \left(\mathcal{X}_{a} - \mathcal{X}_{b} \right)^{2} D \mathcal{A}_{a}^{b} \mathcal{A}_{b}^{a} - \sum_{a,b} \left(\mathcal{X}_{a} - \mathcal{X}_{b} \right)^{2} \left(\mathcal{A} \mathcal{A} \right)_{a}^{b} \mathcal{A}_{b}^{a} + \sum_{a,b} \overline{\mathcal{Z}}^{a} \mathcal{A}_{a}^{b} \mathcal{Z}_{b} + c \sum_{a} \mathcal{A}_{[a]}^{a} \right].$$
(3.3.53)

For n = 1, one particle case, the action (3.3.53) becomes free action

$$S = -i \int dt d\theta \dot{\mathcal{X}} D\mathcal{X} \tag{3.3.54}$$

and the theory has no bosonic potential in the component action.

In the case of n = 2, that is two particles case, the action (3.3.53) is written as

$$S = -i \int dt d\theta \left[\frac{1}{2} \dot{\mathcal{X}}_{+} D \mathcal{X}_{+} - \frac{1}{2} \mathcal{A}_{-} D \mathcal{A}_{-} + \frac{1}{2} \dot{\mathcal{X}}_{-} D \mathcal{X}_{-} - \frac{1}{2} \mathcal{A}_{+} D \mathcal{A}_{+} - c \epsilon_{1} \epsilon_{2} \frac{\mathcal{A}_{+}}{\mathcal{X}_{-}} \right]$$
(3.3.55)

where

$$\mathcal{X}_{-} := \mathcal{X}_{1} - \mathcal{X}_{2}, \qquad \qquad \mathcal{X}_{+} := \mathcal{X}_{1} + \mathcal{X}_{2}, \qquad (3.3.56)$$

$$\mathcal{A}_{+} := \mathcal{X}(\mathcal{A}_{1}^{2} + \mathcal{A}_{2}^{1}), \qquad \qquad \mathcal{A}_{-} := i\mathcal{X}(\mathcal{A}_{1}^{2} - \mathcal{A}_{2}^{1}) \qquad (3.3.57)$$

and $\epsilon_1 = \pm 1$, $\epsilon_2 = \pm 1$ are the constants appearing in the constraint $\mathcal{Z}_1 \mathcal{Z}_2 = -\frac{c\epsilon_1\epsilon_2}{2}$. Note that the superfield action (3.3.55) is a sum of two free $\mathcal{N} = 1$ supermultiplets $(\mathcal{X}_+, \mathcal{A}_-)$ and two interacting $\mathcal{N} = 1$ supermultiplets $(\mathcal{X}_-, \mathcal{A}_+)$. It has been argued that the superfield action (3.3.55) is the $\mathcal{N} = 1$ superfield form of the off-shell $\mathcal{N} = 2$ superconformal mechanics based on the supermultiplet (1, 2, 1) [152, 137].

For n = 3 it has been shown [152, 137] that the $\mathcal{N} = 1$ superfield action (3.3.55) cannot be connected to the known $\mathcal{N} = 2$ or $\mathcal{N} = 3$ superconformal mechanical modelds and that in the bosonic limit it yields the three particle Calogero model for the component fields $x_a = \mathcal{X}_a$.

3.4 $\mathcal{N} = 2$ Superconformal mechanics

3.4.1 One particle model

The $\mathcal{N} = 2$ superspace $\mathbb{R}^{(1|2)}$ contains time coordinate *t* and two Grassmann coordinate $\theta, \overline{\theta}$

$$\mathbb{R}^{(1|2)} = (t, \theta, \overline{\theta}). \tag{3.4.1}$$

The covariant superderivatives D and \overline{D} are

$$D = i\frac{\partial}{\partial\theta} - \overline{\theta}\frac{\partial}{\partial t}, \quad \overline{D} = i\frac{\partial}{\partial\overline{\theta}} - \theta\frac{\partial}{\partial t}, \quad \left\{D, \overline{D}\right\} = -2i\partial_t \quad (3.4.2)$$

while the two supercharges Q and \overline{Q} are given by

$$Q = i\frac{\partial}{\partial\theta} + \overline{\theta}\frac{\partial}{\partial t}, \quad \overline{Q} = i\frac{\partial}{\partial\overline{\theta}} + \theta\frac{\partial}{\partial t}, \quad \{Q, \overline{Q}\} = 2i\partial_t \quad (3.4.3)$$

in the superspace.

Now consider the $\mathcal{N} = 2$ superfield (1, 2, 1) in the superspace (3.4.1)

$$\Phi(t,\theta,\overline{\theta}) = x(t) + i\theta\psi(t) + i\overline{\theta}\overline{\psi}(t) + \theta\overline{\theta}y(t).$$
(3.4.4)

The (1, 2, 1) supermultiplet is also called $\mathcal{N} = 2A$ multiplet. This supermultiplet is related to the two-dimensional (1, 1) supersymmetry. Making use of the (1, 2, 1) supermultiplet (3.4.4), we can write $\mathcal{N} = 2$ supersymmetric action in the form

$$S = \frac{1}{2} \int dt d\theta d\overline{\theta} \, \left[\overline{D} \Phi D \Phi - W(\Phi) \right] \tag{3.4.5}$$

where $W(\Phi)$ is a superpotential that is some function of the superfield Φ . In component the superfield action (3.4.5) can be written as

$$S = \frac{1}{2} \int dt \left[\dot{x}^2 + i \bar{\psi} \psi - i \overline{\psi} \dot{\psi} + y^2 - W'(x)y - W''(x)\psi \overline{\psi} \right].$$
(3.4.6)

To obtain the conformal invariant action, let us consider the superpotential in the form

$$W(\Phi) = f \ln \Phi^2. \tag{3.4.7}$$

Then the action (3.4.6) becomes

$$S = \frac{1}{2} \int dt \left[\dot{x}^2 + i \overline{\psi} \psi - i \overline{\psi} \dot{\psi} + y^2 - \frac{2fy}{x} - \frac{2f\psi}{x^2} \right].$$
(3.4.8)

By solving the algebraic equation of motion of y, one can integrate out the auxiliary field y. Then we find the one-particle $\mathcal{N} = 2 OSp(2|2)$ superconformal mechanical model [153, 154]

$$S = \frac{1}{2} \int dt \left[\dot{x}^2 + i \overline{\psi} \psi - i \overline{\psi} \psi - \frac{f(f - 2\psi \overline{\psi})}{x^2} \right].$$
(3.4.9)

In the superspace the generators of the superconformal group can be realized

by the following expressions ⁸

$$H = i\frac{\partial}{\partial t},\tag{3.4.10}$$

$$D = i \left(t \frac{\partial}{\partial t} + \frac{1}{2} \theta \frac{\partial}{\partial \theta} + \frac{1}{2} \overline{\theta} \frac{\partial}{\partial \overline{\theta}} + \Delta \right), \qquad (3.4.11)$$

$$K = i\left(t^2\frac{\partial}{\partial t} + t\theta\frac{\partial}{\partial \theta} + t\overline{\theta}\frac{\partial}{\partial\overline{\theta}} + 2t\Delta\right),$$
(3.4.12)

$$Q = i\frac{\partial}{\partial\theta} + \overline{\theta}\frac{\partial}{\partial t}, \qquad \qquad \overline{Q} = i\frac{\partial}{\partial\overline{\theta}} + \theta\frac{\partial}{\partial t}, \qquad (3.4.13)$$

$$S = tQ - \theta\overline{\theta}\frac{\partial}{\partial\theta} + 2\Delta\overline{\theta}, \qquad \overline{S} = t\overline{Q} - \overline{\theta}\theta\frac{\partial}{\partial\overline{\theta}} + 2\Delta\theta, \qquad (3.4.14)$$

$$B = -i\theta \frac{\partial}{\partial \theta} + i\overline{\theta} \frac{\partial}{\partial \overline{\theta}}.$$
(3.4.15)

One can show that these generators form the $\mathfrak{su}(1,1|1)$ superalgebra

$$[H,D] = iH, \quad [K,D] = -iK, \quad [H,K] = 2iD,$$
 (3.4.16)

$$[B, H] = 0, \ [B, D] = 0, \ [B, K] = 0,$$
 (3.4.17)

$$[H, Q] = 0, \quad [D, Q] = -\frac{i}{2}Q, \quad [K, Q] = -iS, [H, \overline{Q}] = 0, \quad [D, \overline{Q}] = -\frac{i}{2}\overline{Q}, \quad [K, \overline{Q}] = i\overline{S},$$
 (3.4.18)

$$[H,S] = iQ, \quad [D,S] = \frac{i}{2}S, \quad [K,S] = 0, [H,\overline{S}] = i\overline{Q} \quad [D,\overline{S}] = \frac{i}{2}\overline{S}, \quad [K,\overline{S}] = 0,$$

$$(3.4.19)$$

$$\{Q,\overline{Q}\}=2H, \{S,\overline{S}\}=2K, \{Q,\overline{S}\}=2D-B,$$
 (3.4.20)

$$[B,Q] = iQ, [B,S] = iS, [B,\overline{Q}] = -i\overline{Q}, [B,\overline{S}] = -i\overline{S}. (3.4.21)$$

The supersymmetry transformations for the (1, 2, 1) multiplet which follow from $\delta \Phi = -i[\epsilon \overline{Q} + \overline{\epsilon} Q, \Phi]$ are expressed in the component fields as

$$\delta x = i \varepsilon \overline{\psi} + i \overline{\varepsilon} \psi, \qquad (3.4.22)$$

$$\delta \psi = \epsilon \dot{x} - i\epsilon \frac{f}{x}, \qquad (3.4.23)$$

$$\delta\overline{\psi} = \overline{\epsilon}\dot{x} + i\overline{\epsilon}\frac{f}{x}.$$
(3.4.24)

⁸Note that in [153] the Hamiltonian is expresses by $\{Q, \overline{Q}\} = -2H$ while in our notation the additional sign does not appear.

Applying the Noether's method, we find the explicit expressions for the supercharges Q, \overline{Q} , the three $SL(2, \mathbb{R})$ conformal generators H, D, K and we also introduce the superconformal charges S, \overline{S} and the SO(2) R-symmetry generator B as follows:

$$Q = \psi \left(-ip + \frac{f}{x} \right), \qquad \overline{Q} = \overline{\psi} \left(ip + \frac{f}{x} \right), \qquad (3.4.25)$$

$$S = x\psi, \qquad \overline{S} = x\overline{\psi}, \qquad (3.4.26)$$
$$H = \frac{1}{2} \left[p^2 + \frac{f(f+2B)}{x^2} \right] \qquad (3.4.27)$$

$$D = -\frac{1}{4}(xp + px), \tag{3.4.28}$$

$$K = \frac{1}{2}x^2, \tag{3.4.29}$$

$$B = \frac{1}{2} [\psi, \overline{\psi}]. \tag{3.4.30}$$

Note that the potential in the Hamiltonian H is shifted as a quantum effect. Under the canonical relations

$$[x, p] = i,$$
 $\{\psi, \overline{\psi}\} = 1,$ (3.4.31)

the set of operators (3.4.25)-(3.4.30) form the $\mathfrak{osp}(2|2)$ superalgebra (3.4.16)-(3.4.20).

Let us study the spectrum of the one-particle OSp(2|2) superconformal quantum mechanics (3.4.9). In general supersymmetric quantum mechanics has the Hamiltonian H which can be written as the sum of squares of the Hermitian supercharges $Q_A, A = 1, \dots, N$. This implies that the energy of any state is positive or zero [202, 52]. If $H|\Omega\rangle = 0$, then we have $0 = \langle \Omega|H|\Omega\rangle = \sum_A \langle \Omega|Q_A^2|\Omega\rangle =$ $\sum_A |Q_A|\Omega\rangle|^2$, which is only possible if $Q_A|\Omega\rangle$ for any A. Conversely if a state $|\Omega\rangle$ is annihilated by Q_A , then $H|\Omega\rangle = Q_A^2|\Omega\rangle = 0$, i.e. its energy is zero. Therefore the supersymmetry generated by Q_A is broken if the system has no normalizable ground state of H. Now consider the equations defining the ground state of H

$$Q|\Omega\rangle = \overline{Q}|\Omega\rangle = 0.$$
 (3.4.32)

Using the explicit expressions (3.4.25) and (3.4.26), the equation (3.4.32) is written as

$$\left(2iBp - \frac{f}{x^2}\right)|\Omega\rangle = 0 \tag{3.4.33}$$

which can be interpreted as the first order differential equation of x. Then the generic solution of (3.4.33) leads to the x-depgroundence of the ground state of H

$$|\Omega\rangle = x^{-2fB}|\text{phys}\rangle \tag{3.4.34}$$

where $|phys\rangle$ is any *x* independent state. Noting that the *SO*(2) R-symmetry operator *B* has eigenvalue $+\frac{1}{2}$ and $-\frac{1}{2}$, we see that the ground state of *H* may have the two different *x* dependence

$$|\Omega\rangle = \begin{cases} x^{-f}|\text{phys}\rangle & \text{for } B = \frac{1}{2} \\ x^{f}|\text{phys}\rangle & \text{for } B = -\frac{1}{2}. \end{cases}$$
(3.4.35)

As the wavefunction will blow up for either large or small *x* region, there is no normalizable state of *H* and therefore the supersymmetry generated by Q, \overline{Q} is spontaneously broken. Note that the wavefunction with E > 0 energy can be exactly solved by using the result of DFF-model. Comparing the quantum Hamiltonian (3.4.27) with the DFF-model Hamiltonian (2.1.18), we find the relation

$$g^{2} = f(f+2B) = \begin{cases} f(f+1) & \text{for } B = \frac{1}{2} \\ f(f-1) & \text{for } B = -\frac{1}{2}. \end{cases}$$
(3.4.36)

The appearance of two sectors, i.e. the doublet structure of the eigenstates of *H* corresponds to the fact that *H* commutes with two operators Q and \overline{Q} . From the expression (2.2.1) we find the eigenfunctions

$$\psi_{E,B}(x) = \begin{cases} C\sqrt{x}J_{\sqrt{f+\frac{1}{2}}}\left(\sqrt{2E}x\right) & \text{for } B = \frac{1}{2} \\ C\sqrt{x}J_{\sqrt{f-\frac{1}{2}}}\left(\sqrt{2E}x\right) & \text{for } B = -\frac{1}{2}. \end{cases}$$
(3.4.37)

These wavefunctions are shown in Figure 3.2. From Figure 3.2 we see that there are several peaks of the wavefunctions with the nearest one from the origin being the maximum value. For large coupling constant f the relative positions of the particle gradually become far from the origin. At high energy E the number of peaks increases and the probability of the position of the particle is averaged.

Then we can follow the previous discussion for the DFF-model to solve the problem of the absence of the ground state. Instead of the original Hamiltonian we now regard the compact operator $L_0 = \frac{1}{2}(H + K)$ as the new Hamiltonian. Looking at the formulae (2.2.26), (2.2.28) and the relation (3.4.36), one finds

$$r_n = \begin{cases} \frac{1}{2} \left(\frac{3}{2} + f\right) & \text{for } B = \frac{1}{2} \\ \frac{1}{2} \left(\frac{1}{2} + f\right) & \text{for } B = -\frac{1}{2}. \end{cases}$$
(3.4.38)



Figure 3.2: The eigenfunctions $\psi_{E,B}(x)$ of the original Hamiltonian H with $E \neq 0$. There are two sectors labeled by $B = \frac{1}{2}$ and $B = -\frac{1}{2}$.



Figure 3.3: The level structure of the spectrum of the new Hamiltonian L_0 . The spectrum is equally spaced. For a fixed *B* the equal space is 1 while the space with $\Delta B \neq 1$ is $\frac{1}{2}$.

The level structure of the spectrum of L_0 has two series corresponding to the two different eigenvalues $B = -\frac{1}{2}, \frac{1}{2}$. So it can be represented on the plane of the eigenvalue of *B* and L_0 (see Figure 3.3). In order to understand the appearance of the half integer shift in an algebraic way, let us define the fermionic operators [154]

$$M = Q - S = \psi \left(-ip + \frac{f}{x} - x \right), \qquad (3.4.39)$$

$$\overline{M} = \overline{Q} - \overline{S} = \overline{\psi} \left(ip + \frac{f}{x} - x \right), \qquad (3.4.40)$$

$$N = \overline{Q} + \overline{S} = \overline{\psi} \left(ip + \frac{f}{x} + x \right), \qquad (3.4.41)$$

$$\overline{N} = Q + S = \psi \left(-ip + \frac{f}{x} + x \right).$$
(3.4.42)

Then we find the following anti-commutation relations

$$\{M, \overline{M}\} =: 4T_1 = 4L_0 + 2B - 2f,$$
 (3.4.43)

$$\{N,\overline{N}\} =: 4T_2 = 4L_0 - 2B + 2f, \qquad (3.4.44)$$

$$\{M,N\} = 4L_{+} = 2(H - K + 2iD), \qquad (3.4.45)$$

$$\left\{\overline{M},\overline{N}\right\} = 4L_{-} = 2\left(H - K - 2iD\right), \qquad (3.4.46)$$

$$\left\{M,\overline{N}\right\} = \left\{\overline{M},N\right\} = 0. \tag{3.4.47}$$

and the commutation relations

$$[L_0, M] = -\frac{1}{2}M,$$
 $[L_0, \overline{M}] = \frac{1}{2}\overline{M},$ (3.4.48)

$$[L_0, N] = -\frac{1}{2}N,$$
 $[L_0, \overline{N}] = \frac{1}{2}\overline{N},$ (3.4.49)

$$[T_1, N] = -N, [T_1, N] = N, (3.4.50)$$
$$[T_2, N] = -N, [T_2, \overline{N}] = \overline{N}, (3.4.51)$$

$$[T_1, M] = [T_1, \overline{M}] = 0, \qquad (3.4.52)$$

$$[T_2, N] = [T_2, \overline{N}] = 0, \tag{3.4.53}$$

$$[T_1, L_-] = -L_-, \qquad [T_1, L_+] = L_+, \qquad (3.4.54)$$

$$[T_2, L_-] = -L_-,$$
 $[T_2, L_+] = L_+.$ (3.4.55)

Let us consider the ground states eliminated by the supercharges. Since there are now three sets of the supercharges; (Q, \overline{Q}) , (M, \overline{M}) and (N, \overline{N}) , we find six

possible candidates for the *x* dependence of the ground states $|\Omega\rangle$:

1

$$|\Omega\rangle = \begin{cases} x^{-f}|\text{phys}\rangle & \text{for } (H, Q, \overline{Q}, B = \frac{1}{2}) \\ x^{f}|\text{phys}\rangle & \text{for } (H, Q, \overline{Q}, B = -\frac{1}{2}) \\ x^{-f}e^{\frac{x^{2}}{2}}|\text{phys}\rangle & \text{for } (T_{1}, M, \overline{M}, B = \frac{1}{2}) \\ x^{f}e^{-\frac{x^{2}}{2}}|\text{phys}\rangle & \text{for } (T_{1}, M, \overline{M}, B = -\frac{1}{2}) \\ x^{-f}e^{-\frac{x^{2}}{2}}|\text{phys}\rangle & \text{for } (T_{2}, N, \overline{N}, B = \frac{1}{2}) \\ x^{f}e^{\frac{x^{2}}{2}}|\text{phys}\rangle & \text{for } (T_{2}, N, \overline{N}, B = -\frac{1}{2}) \end{cases}$$
(3.4.56)

where $|\text{phys}\rangle$ is a *x* independent state. We see that only the set of generators $(T_1, M, \overline{M}, B = -\frac{1}{2})$ can yield the normalizable eigenfunction of the ground state. In order to obtain the normalizable ground state, $|\text{phys}\rangle$ need to be the eigenstate with $B = -\frac{1}{2}$. Let us define a state $|0\rangle$ annihilated by the operator $\overline{\psi}$

$$\psi|0\rangle = 0. \tag{3.4.57}$$

Then $B|0\rangle = -\frac{1}{2}|0\rangle$ and we thus can choose the state $|0\rangle$ as $|\text{phys}\rangle$. Given the state $|0\rangle$, one can build up a tower of states by multiplying the operator ψ . Since the square of the Grassmann variable is zero $\psi^2 = 0$, the fermionic generators form the two-dimensional space spanned by

$$|0\rangle, \quad \psi|0\rangle \tag{3.4.58}$$

and $\overline{\psi}$ and ψ are identified with the lowering operator and raising operator for fermionic excitation respectively. Therefore we obtain the normalizable ground state

 $|\Omega\rangle = x^f e^{-\frac{x^2}{2}}|0\rangle \tag{3.4.59}$

which satisfies

$$M|\Omega\rangle = \overline{M}|\Omega\rangle = 0,$$
 (3.4.60)

$$N|\Omega\rangle = 0, \tag{3.4.61}$$

$$\overline{\psi}|\Omega\rangle = 0.$$
 (3.4.62)

Having found the eigenfunction of L_0 , we see from (2.2.32) and (3.4.59) that the ground state $|\Omega\rangle$ is the eigenstate of L_0 with the eigenvalue

$$r_0 = \frac{1}{2} \left(f + \frac{1}{2} \right) \tag{3.4.63}$$



Figure 3.4: The bosonic excitations and the fermionic excitations in the L_0 spectrum. For a fixed *B*, i.e. for the bosonic excitation generated by L_+ and L_- , the space is one unit. For a fermionic excitation generated by \overline{N} , N, \overline{M} and M the space is half a unit.

and obtain the two series (3.4.38) labeled by *B*. We observe from the commutation relations (3.4.49) that the fermionic generator *M*, *N* decreases L_0 by $\frac{1}{2}$ while $\overline{M}, \overline{N}$ increase L_0 by $\frac{1}{2}$. As seen from the relations (3.4.60), the fermionic excitation for the ground state $|\Omega\rangle$ can be generated by only \overline{N} . In addition, there are bosonic excitations. As in the DFF-model, L_+ increases L_0 by one and L_- decreases L_0 by one. While the fermionic excitations shift the eigenvalue of *B*, the bosonic excitations does not. The excitations in the L_0 spectrum are drawn in Figure 3.4.

From the relations (3.4.43), (3.4.44), (3.4.52) and (3.4.53) one can see that the two sets of new supercharges (M, \overline{M}) and (N, \overline{N}) yield the bosonic operators T_1 and T_2 respectively. Since the bosonic operators T_1 and T_2 are compact, one may also use T_1 or T_2 as the new Hamiltonian. However, unlike the compact operator L_0 , T_1 and T_2 enjoy the double structures of their spectrums according to the commutation relations (3.4.52) and (3.4.53).

Now consider the spectrum of T_1 . By noting the relations (3.4.43) and (3.4.60), we see that the ground state $|\Omega\rangle$ has zero eigenvalue of T_1 . According to the commutation relations (3.4.50) and (3.4.54), one finds that for the T_1 spectrum the bosonic and fermionic excitations have the same spacing equal to one, which are generated by L_+ , L_- and \overline{N} , N respectively. Note that \overline{M} , M commute with T_1 and do not play the role of the raising and lowering operators. The T_1 spectrum is



Figure 3.5: The level structure of T_1 spectrum and its bosonic and fermionic excitations. Each of the bosonic and fermionic excitations has the equal space of one unit. The ground state $|\Omega\rangle$ has zero eigenvalue.

given by the two series

$$T_1 = \begin{cases} 0, 1, 2, \cdots & \text{for } B = -\frac{1}{2} \\ 1, 2, \cdots & \text{for } B = \frac{1}{2}, \end{cases}$$
(3.4.64)

which is illustrated in Figure 3.5. For all non-zero T_1 states, there are degenerate structures. In other words the bosonic and fermionic states are always paired at the excited level of T_1 . This is due to the relations (3.4.52), which ensure the preservation of the supersymmetry generated by M and \overline{M} . Therefore one can interpret the pairing structure of T_1 spectrum at excited states as the consequence of the preserved supersymmetry generated by M and \overline{M} .

Similarly the spectrum T_2 holds the doublet structure because T_2 commute with N and \overline{N} and the corresponding supersymmetry is preserved as seen from (3.4.53). In this case the bosonic excitation is generated by L_+ , L_- whereas the fermionic one is generated by \overline{M} , M. Also one can see from (3.4.51) and (3.4.55) that both bosonic and fermionic excitations are produced with equal spacing of one unit. In this case, however, there is no normalizable zero T_2 state. The normalizable ground state $|\Omega\rangle$ has the eigenstate of T_2 with the eigenvalue (4f + 2). The T_2 spectrum is



Figure 3.6: The level structure of T_2 spectrum and its bosonic and fermionic excitations. Each of the bosonic and fermionic excitations has the equal space of one unit. The ground state $|\Omega\rangle$ has eigenvalue (4f + 2).

given by

$$T_2 = \begin{cases} 4f + 2 + n & \text{for } B = -\frac{1}{2} \\ 4f + 3 + n & \text{for } B = \frac{1}{2} \end{cases}$$
(3.4.65)

where $n = 0, 1, 2, \cdots$. The T_2 spectrum and its excitation are shown in Figure 3.6.

3.4.2 Multi-particle model

Now we want to discuss the $\mathcal{N} = 2$ superconformal sigma-model. Let us start with n (1,2,1) supermultiplets $\Phi_{a,a} = 1, \dots, n^{-9}$. The generic action without superpotential terms takes the form [200]

$$S = \frac{1}{2} \int dt d^2 \theta \left[(g+b)_{ij} \overline{D} \Phi^i D \Phi^j + l_{ij} D \Phi^i D \Phi^j + m_{ij} \overline{D} \Phi^i \overline{D} \Phi^j \right]$$
(3.4.66)

where g_{ij} is the metric and b_{ij} , l_{ij} and m_{ij} are the two-forms on the target space \mathcal{M} . Note that the terms of l_{ij} and m_{ij} correspond to the non-Lorentz invariant terms in two-dimensions. Notice that the target space \mathcal{M} of the (**1**, **2**, **1**) supermultiplet, or $\mathcal{N} = 2A$ multiplet is a real manifold.

⁹The (1, 2, 1) supermultiplet is also called $\mathcal{N} = 2A$ multiplet while (2, 2, 0) chiral supermultiplet is also called $\mathcal{N} = 2B$ multiplet [200].

We have already defined the two covariant derivatives D and \overline{D} for $\mathcal{N} = 2$ supersymmetry in (3.4.2), however, more generally in terms of the two $\mathcal{N} = 1$ covariant derivatives D_1 , D_2 the $\mathcal{N} = 2$ two covariant derivatives can be chosen as

$$D_2 \Phi^i = I^i{}_j D_1 \Phi^j \tag{3.4.67}$$

where *I* is an endomorphism of the tangent bundle of \mathcal{M} . Then the anti-commutation relations $\{D_i, D_j\} = 2i\partial_t$ impose the conditions

$$I^2 = -1, (3.4.68)$$

$$N(I) = 0$$
 (3.4.69)

where N(I) is the Nijenhuis tensor of the endomorphism *I*. The condition (3.4.68) implies that *I* is the almost complex structure and the condition (3.4.69) further shows that the *I* is an (integrable) complex structure. Thus $\mathcal{N} = 2$ supersymmetry requires a complex structure *I* on the target space [199].

To go further let us follow the strategy in [201] and express the second supersymmetry transformation in terms of the $\mathcal{N} = 1$ superspace formalism as [69]

$$\delta x^i = -i\epsilon I^i{}_j\psi^j, \qquad (3.4.70)$$

$$\delta \psi^{i} = -\epsilon \left[I^{i}{}_{j} \dot{x}^{j} - i \psi^{k} \left(\partial_{k} I^{i}{}_{j} \right) \psi^{j} \right].$$
(3.4.71)

Following the Noether's procedure, we obtain the second supercharge

$$Q_{2} = \psi^{i} I_{i}^{j} \Pi_{j} - \frac{i}{2} \psi^{i} I^{j}_{i} c_{jkl} \psi^{j} \psi^{k} - \frac{i}{6} \psi^{i} \psi^{j} \psi^{k} I^{l}_{i} I^{m}_{j} I^{n}_{k} c_{lmn} - \frac{i}{2} \psi^{i} c_{ijk} I^{jk}$$
(3.4.72)

where $\Pi_i := g_{ij} \dot{x}^j$. Then it turns out that the $\mathcal{N} = 1$ action (3.3.6) is invariant under the $\mathcal{N} = 2$ supersymmetry transformations if we have [200, 203]

$$g_{ij} = I^k{}_i I^l{}_j g_{kl}, (3.4.73)$$

$$\nabla_{(i}^{(+)}I^{k}{}_{j)} = 0, (3.4.74)$$

$$\partial_{[i} \left(I^{m}{}_{j}c_{|m|kl]} \right) - 2I^{m}{}_{[i}\partial_{[m}c_{jkl]]} = 0$$
(3.4.75)

where $\nabla_i^{(+)}$ is the connection with torsion c on \mathcal{M} ; $\Gamma_{jk}^i + c_{jk}^i$. The first constraint (3.4.73) requires that the metric g on \mathcal{M} is Hermitian with respect to the complex structure I. The second condition (3.4.74) is a generalized Yano tensor condition with torsion ¹⁰. This corresponds to the vanishing of $\{Q_1, Q_2\}$ where Q_1 is the $\mathcal{N} = 1$ supercharge given by (3.3.14). The third condition (3.4.75) yields the restriction

¹⁰For vanishing torsion the equation (3.4.74) coincides with the Yano tensor condition as in [204]

on torsion and complex structure, however it has no geometrical interpretation so far.

It is known that the N = 2 supermultiplets in one dimension are related to the N = 1 supersymmetry in two dimensions

$$1d \mathcal{N} = 2A \Leftrightarrow 2d \mathcal{N} = (1,1),$$

$$1d \mathcal{N} = 2B \Leftrightarrow 2d \mathcal{N} = (2,0)$$

$$(3.4.76)$$

by the dimensional reduction. Note that two-dimensional (2,0) supersymmetry sigma-models requires the first condition (3.4.73), however, the last two conditions (3.4.74) and (3.4.75) do not appear in two-dimensional (2,0) sigma-models. Instead of (3.4.74), there appears the covariant constant condition of *I* with respect to the connection $\nabla^{(+)}$

$$\nabla_i^{(+)} I^j{}_k = 0. \tag{3.4.77}$$

Now we consider the $\mathcal{N} = 2$ superconformal condition. Promoted from the $\mathfrak{osp}(1|2)$ algebra (3.3.19)-(3.3.22), the $\mathfrak{su}(1,1|1)$ algebra (3.4.18)-(3.4.21) contain the U(1) R-symmetry generator *B*. From the commutation of the supercharges Q_1 in (3.3.14) and Q_2 in (3.4.72) with the conformal boost generator *K* we can read the superconformal charges

$$S_1 = \psi^i D_i,$$
 $S_2 = \psi^i I^j{}_i D_j.$ (3.4.78)

Then the R-symmetry generator *B* can be found from the commutator of *Q* and S_2 as

$$B = D_2^i \Pi_i - i I_{ij} \psi^i \psi^j - i D_2^i c_{ijk} \psi^j \psi^k.$$
(3.4.79)

The constraint can be found from the commutation relation $[D, Q_2] = \frac{i}{2}Q_2$, which leads to

$$\mathcal{L}_D I^j{}_i = 0. \tag{3.4.80}$$

This implies that *D* preserves the complex structure *I*, that is *D* acts holomorphically. Combining the constraint (3.4.80) with the other required conditions (3.3.29) and (3.4.74), we also find

$$\mathcal{L}_{\tilde{D}}I^{j}{}_{i}=0, \qquad \qquad \mathcal{L}_{\tilde{D}}g_{ij}=0, \qquad (3.4.81)$$

which means that $\tilde{D}^i := D^j I^i{}_j$ generates a holomorphic isometry.

Therefore the SU(1,1|1) superconformal quantum sigma-model with vanishing bosonic potential can be realized if the conformal invariant conditions (2.8.9), (2.8.16), the $\mathcal{N} = 2$ supersymmetry invariant conditions (3.4.73)-(3.4.75) and the SU(1,1|1) superconformal invariant conditions (3.4.80), (3.4.81) are satisfied. The last additional constraints on the target space \mathcal{M} require that D acts holomorphically and $\tilde{D}^i : D^j I^i{}_i$ generates a holomorphic isometry.

3.4.3 Freedman-Mende model

Let us consider *n* (1, 2, 1) supermultiplets $\Phi_{a,a} = 1, \dots, n$ and a simple superfield action given by

$$S = \frac{1}{2} \int dt d^2 \theta \left[\sum_{a=1}^{n} \overline{D} \Phi_a D \Phi_a - W(\Phi) \right]$$
(3.4.82)

where $W(\Phi)$ is the superpotential. In terms of the component fields the action (3.4.82) is expressed as

$$S = \frac{1}{2} \int dt \left[\sum_{a=1}^{n} \left(\dot{x}_{a}^{2} + i \overline{\psi}_{a} \psi_{a} - i \overline{\psi}_{a} \dot{\psi}_{a} \right) - \frac{1}{4} \sum_{a=1}^{N} \partial_{a} W \partial_{a} W - \sum_{a,b} \left(\partial_{a} \partial_{b} W \right) \psi_{a} \overline{\psi}_{b} \right]$$
(3.4.83)

where $\partial_a := \frac{\partial}{\partial x_a}$. Taking into account the superconformal boost transformation on the (**1**, **2**, **1**) multiplet

$$\delta \Phi_a = -i \left(\eta \overline{\theta} + \overline{\eta} \theta \right) \Phi_a \tag{3.4.84}$$

and the invariance of the measure δ ($dtd\theta$) = 0 we find that the action (3.4.83) is invariant under the superconformal boost transformation only if we have

$$\Phi_a \partial_a W(\Phi) = c \tag{3.4.85}$$

with *c* being a constant. It has been shown [205] that *c* characterizes the central charge in $\mathfrak{su}(1,1|1)$ superconformal algebra and that the superpotential $W(\Phi)$ is a harmonic function of Φ_a if quantum Hamiltonian contain boson-fermion interaction but no boson-boson interaction.

It is interesting to note that the superpotential [206]

$$W(\Phi) = f \sum_{a \neq b} \ln \left(\Phi_a - \Phi_b \right)$$
(3.4.86)

where f is a constant gives rise to the Freedman-Mende model [207]

$$S = \frac{1}{2} \int dt \left[\sum_{a=1}^{n} \left(\dot{x}_{a}^{2} + i \dot{\overline{\psi}}_{a} \psi_{a} - i \overline{\psi}_{a} \dot{\psi}_{a} \right) - \sum_{a \neq b} \frac{f^{2} + 4f \psi_{a} \overline{\psi}_{b}}{4(x_{a} - x_{b})^{2}} \right].$$
(3.4.87)

This is the $\mathcal{N} = 2$ superconformal generalization of the Calogero model. For the Freedman-Mende model the central charge *Z* in the $\mathfrak{su}(1,1|1)$ superconformal algebra can be identified with

$$Z = n(n-1)f.$$
 (3.4.88)

The Freedman-Mende model is the supersymmetric rational A_{n+1} Calogero model in the sense that the original Calogero model is obtained by projecting the supersymmetric Hamiltonian onto the zero fermion sector.

If we have the superpotential

$$W(\Phi) = f \ln\left(\sum_{a} \Phi_{a} \Phi_{a}\right)$$
(3.4.89)

with f being a constant, then we find [208]

$$S = \frac{1}{2} \int dt \sum_{a=1}^{N} \left[\dot{x}_a^2 + i \overline{\psi}_a \psi_a - i \overline{\psi}_a \dot{\psi}_a - \frac{f(f - 2\psi_a \overline{\psi}_a)}{x_a^2} \right].$$
(3.4.90)

Unlike the Freedman-Mende model (3.4.88), the interaction terms are not pairwise but still possess the inverse square behavior. This is the $\mathcal{N} = 2$ superconformal mechanics describing the motion of the *n*-particle center of mass and the corresponding central charge *Z* in the superconformal algebra $\mathfrak{su}(1,1|1)$ is [208]

$$Z = 2f.$$
 (3.4.91)

3.4.4 Gauged superconformal mechanics

We start with the $\mathcal{N} = 2$ matrix superfield gauged mechanical action [152, 209, 137]

$$S = \int dt d^{2}\theta \left[\operatorname{Tr} \left(\overline{\mathcal{D}} \mathcal{X} \mathcal{D} \mathcal{X} \right) + \frac{1}{2} \overline{\mathcal{Z}} e^{2V} \mathcal{Z} - c \operatorname{Tr} V \right].$$
(3.4.92)

Here we have

• the $\mathcal{N} = 2$ Grassmann-even Hermitian $n \times n$ matrix superfield $\mathcal{X}_a^b(t, \theta, \overline{\theta})$ which satisfies $(\mathcal{X})^{\dagger} = \mathcal{X}$ and transforms as the adjoint representation of U(n); the $(\mathbf{1}, \mathbf{2}, \mathbf{1})$ supermultiplet

- the $\mathcal{N} = 2$ Grassmann-even chiral superfield $\mathcal{Z}_a(t_L, \theta)$, $\overline{\mathcal{Z}}^a(t_R, \overline{\theta})$, $t_{L,R} = t \pm i\theta\overline{\theta}$ which transform as the fundamental representations of U(n); the (**2**, **2**, **0**) supermultiplets
- the *N* = 2 Grassmann-even complex *n* × *n* matrix bridge superfield *b_a^b(t, θ, θ̄)* which satisfies *b̄* := *b*[†].

Note that gauge superfields are described by the complex $n \times n$ matrix bridge superfields *b* or by the prepotential *V* defined by

$$e^{2V} = e^{-i\bar{b}}e^{ib}.$$
 (3.4.93)

The covariant derivatives are defined by

$$\mathcal{DX} = D\mathcal{X} + i[\mathcal{A}, \mathcal{X}], \qquad \qquad \overline{\mathcal{D}}\mathcal{X} = \overline{\mathcal{D}}\mathcal{X} + i[\overline{\mathcal{A}}, \mathcal{X}] \qquad (3.4.94)$$

where 11

$$D = \frac{\partial}{\partial \theta} + i\overline{\theta}\frac{\partial}{\partial t}, \qquad \overline{D} = -\frac{\partial}{\partial \overline{\theta}} + i\theta\frac{\partial}{\partial t}, \qquad \left\{D, \overline{D}\right\} = -2i\partial_t \qquad (3.4.95)$$

where the connections \mathcal{A} are deduced from the bridge superfields

$$\mathcal{A} = -ie^{i\overline{b}} \left(De^{-i\overline{b}} \right), \qquad \qquad \overline{\mathcal{A}} = -ie^{ib} \left(\overline{D}e^{-ib} \right). \qquad (3.4.96)$$

The superconformal boost transformations are [210]

$$\delta t = -i \left(\eta \overline{\theta} + \overline{\eta} \theta \right) t, \qquad \qquad \delta \theta = -\eta \left(t + i \theta \overline{\theta} \right), \qquad (3.4.97)$$

$$\delta\overline{\theta} = -\eta(t - i\theta\overline{\theta}), \qquad \delta(dtd^2\theta) = 0, \qquad (3.4.98)$$

$$\delta \mathcal{X} = -i \left(\eta \theta + \overline{\eta} \theta \right) \mathcal{X}, \qquad \qquad \delta \mathcal{Z} = 0, \qquad (3.4.99)$$

$$\delta b = 0, \qquad \qquad \delta V = 0. \qquad (3.4.100)$$

The action (3.4.92) is invariant under the U(n) transformations [152, 209, 137]

$$e^{ib} \rightarrow e^{i\Lambda} e^{ib} e^{-i\lambda}, \qquad e^{i\overline{b}} \rightarrow e^{i\Lambda} e^{i\overline{b}} e^{-i\overline{\lambda}}, \qquad e^{2V} \rightarrow e^{i\overline{\lambda}} e^{2V} e^{-i\lambda}, \qquad (3.4.101)$$

$$\mathcal{X} \to e^{i\Lambda} \mathcal{X} e^{-i\Lambda}, \qquad \mathcal{Z} \to e^{i\lambda} \mathcal{Z}, \qquad \qquad \overline{\mathcal{Z}} \to \overline{\mathcal{Z}} e^{-i\overline{\lambda}}.$$
 (3.4.102)

Here Λ is the Hermitian $n \times n$ matrix gauge parameter and λ are the complex $n \times n$ gauge parameters.

Alternatively if we use the Λ gauge invariant superfields V, Z and the new Hermitian $n \times n$ matrix superfield

$$\mathcal{Y} = e^{-ib} \mathcal{X} e^{ib}, \tag{3.4.103}$$

¹¹The notation here is different from (3.4.2).

the action (3.4.92) can be written as

$$S = \int dt d^{2}\theta \left[\operatorname{Tr} \left(\overline{\mathcal{D}} \mathcal{Y} e^{2V} \mathcal{D} \mathcal{Y} e^{2V} \right) + \frac{1}{2} \overline{\mathcal{Z}} e^{2V} \mathcal{Z} - c \operatorname{Tr} V \right]$$
(3.4.104)

where the covariant derivatives are

$$\mathcal{D}\mathcal{Y} = \mathcal{D}\mathcal{Y} + e^{-2V}(\mathcal{D}e^{2V})\mathcal{Y}, \qquad \overline{\mathcal{D}}\mathcal{Y} = \overline{\mathcal{D}}\mathcal{Y} - \mathcal{Y}e^{2V}\left(\overline{\mathcal{D}}e^{-2V}\right).$$
(3.4.105)

The $\mathcal{N} = 2$ superfields V, \mathcal{Y} , \mathcal{Z} and $\overline{\mathcal{Z}}$ can be expressed in terms of the component fields as

$$V = v + \theta \xi - \overline{\theta \xi} + \theta \overline{\theta} A, \qquad (3.4.106)$$

$$\mathcal{Y} = x + \theta \psi - \overline{\theta} \overline{\psi} + \theta \overline{\theta} y, \qquad (3.4.107)$$

$$\mathcal{Z} = z + 2i\theta\zeta + i\theta\overline{\theta}\dot{z},\tag{3.4.108}$$

$$\overline{\mathcal{Z}} = \overline{z} + 2i\overline{\theta}\overline{\zeta} - i\theta\overline{\theta}\overline{z}.$$
(3.4.109)

According to the gauge transformation (3.4.101), let us choose the gauge so that

$$V(t,\theta,\overline{\theta}) = \theta\overline{\theta}A_0(t). \tag{3.4.110}$$

After integrating out the auxiliary fields $\zeta, \overline{\zeta}$ and performing the Grassmann integrations, we get the $\mathcal{N} = 2$ superconformal gauged mechanical action

$$S = \int dt \left[\operatorname{Tr} \left(DxDx \right) + \frac{i}{2} \left(\overline{z}Dz - D\overline{z}z \right) + i\operatorname{Tr} \left(\overline{\psi}D\psi - D\overline{\psi}\psi \right) - c\operatorname{Tr}A_0 \right]$$
(3.4.111)

where the covariant derivatives are

$$Dx = \dot{x} + i[A_0, x],$$
 $Dz = \dot{z} + iA_0z,$ (3.4.112)

$$D\psi = \dot{\psi} + i[\psi, A_0], \qquad \qquad D\overline{\psi} = \dot{\overline{\psi}} + i[\overline{\psi}, A_0]. \qquad (3.4.113)$$

The action (3.4.111) is the supersymmetric generalization of the Calogero model whose bosonic part agrees with the Calogero model (2.9.2). The action is invariant with respect to the U(n) gauge transformations

$$x \to gXg^{-1}, \qquad z \to gz, \qquad (3.4.114)$$

$$\psi \to g\psi g^{-1}$$
, $A_0 \to gA_0g^{-1} + i\dot{g}g^{-1}$ (3.4.115)

where $g \in U(n)$. By fixing the gauge as in (2.9.9) and (2.9.14), z, \overline{z} and the nondiagonal part of x are eliminated and thus we have

n physical bosons
$$x_a^a$$
,
 $2n^2$ physical fermions $\psi_a^b, \overline{\psi}_a^b$. (3.4.116)

This is different from the Freedman-Mende model (3.4.88) which possesses *n* physical bosons and 2n physical fermions. which can be realized as

$$(n, 2n^2, 2n^2 - n) = n(1, 2, 1) \oplus (n^2 - n)(0, 2, 2).$$
 (3.4.117)

It has been pointed out [211] that the supermultiplet (3.4.117) can be obtained from n (1, 2, 1) supermultiplets by gauging procedure.

3.5 $\mathcal{N} = 4$ Superconformal mechanics

As we have discussed in subsection 3.1.3, the most general superconformal algebra of $\mathcal{N} = 4$ superconformal quantum mechanics is $D(2,1;\alpha)$. As opposed to the $\mathcal{N} = 1$ superconformal algebra $\mathfrak{osp}(1|2)$ and $\mathcal{N} = 2$ superconformal algebra $\mathfrak{su}(1,1|1) \cong \mathfrak{osp}(2|2)$, the Lie superalgebra $D(2,1;\alpha)$ is a one-parameter family of superalgebra characterized by a real parameter α . In order to construct the corresponding family of $\mathcal{N} = 4$ superconformal quantum mechanical modelds parametrized by α , it is desirable to find the inequivalent irreducible off-shell supermultiplets in a systematic way.

To this end there is the methodical way proposed in [212] by means of the nonlinear realizations technique [155, 156, 157]. We shall start from the superconformal algebra $D(2,1;\alpha)$, which contains three conformal charges H, D, K which form $\mathfrak{sl}(2,\mathbb{R})$, four supercharges $Q^i, \overline{Q}_i, i = 1, 2$, four superconformal charges S^i, \overline{S}_i and two commuting sets of $\mathfrak{su}(2)$ R-symmetry generators J, \overline{J}, J_3 and I, \overline{I}, I_3 ¹²

$$[H,D] = iH, \quad [K,D] = -iK, \quad [H,K] = 2iD, \tag{3.5.1}$$

$$[H, Q^{i}] = 0, \qquad [D, Q^{i}] = -\frac{i}{2}Q^{i}, \qquad [K, Q^{i}] = -iS^{i}, \qquad (3.5.2)$$

$$[H, S^{i}] = iQ^{i}, \qquad [D, S^{i}] = \frac{i}{2}Q^{i}, \qquad [K, S^{i}] = 0, \qquad (3.5.3)$$

¹² Here we use the notation in [183, 213, 212], which is slightly different from our previous $\mathcal{N} = 1$ and $\mathcal{N} = 2$ cases in that the signs of the anti-commutators (3.5.4) and the covariant derivatives (3.5.28) and the supercharges (3.5.29).

$$\{Q^{i}, \overline{Q}_{j}\} = -2\delta^{i}_{j}H, \quad \{S^{i}, \overline{S}_{j}\} = -2\delta^{i}_{j}K, \qquad \{Q^{i}, S^{j}\} = -2(1+\alpha)\epsilon^{ij}\overline{I}, \\ \{Q^{1}, \overline{S}_{2}\} = 2\alpha\overline{J} \qquad \{Q^{1}, \overline{S}_{1}\} = -2D - 2\alpha J_{3} + 2(1+\alpha)I_{3}, \\ \{Q^{2}, \overline{S}_{1}\} = -2\alpha J, \qquad \{Q^{2}, \overline{S}_{2}\} = -2D + 2\alpha J_{3} + 2(1+\alpha)I_{3},$$
(3.5.4)

$$[J_{3}, Q^{1}] = -\frac{i}{2}Q^{1}, \quad [J_{3}, Q^{2}] = \frac{i}{2}Q^{2}, \quad [J, Q^{1}] = -iQ^{2}, \quad [J, \overline{Q}_{2}] = i\overline{Q}_{1},$$

$$[J_{3}, S^{1}] = -\frac{i}{2}S^{1}, \quad [J_{3}, S^{2}] = \frac{i}{2}S^{2}, \quad [J, S^{1}] = -iS^{2}, \quad [J, \overline{S}_{2}] = i\overline{S}_{1},$$

$$[I_{3}, Q^{i}] = -\frac{i}{2}Q^{i}, \quad [I, Q^{i}] = -i\overline{Q}^{i}, \quad [I_{3}, S^{i}] = -\frac{i}{2}S^{i}, \quad [I, S^{i}] = -i\overline{S}^{i}, \quad (3.5.5)$$

$$[J_3, J] = iJ, [J_3, \overline{J}] = -i\overline{J}, [J, \overline{J}] = -2iJ_3, [I_3, I] = iI, [I_3, \overline{I}] = -i\overline{J}, [I, \overline{I}] = -2iJ_3. (3.5.6)$$

The R-symmetry group contains two SU(2) factors generated by J, \overline{J}, J_3 and I, \overline{I}, I_3 . Looking at the commutation relations (3.5.5), J corresponds to the rotations indices i of θ_i while I mixes θ_i with their complex conjugates.

Here we take bosonic conformal generators *H*, *D*, *K* as Hermitian operators

$$(H)^{\dagger} = H,$$
 $(D)^{\dagger} = D,$ $(K)^{\dagger} = K$ (3.5.7)

while the other operators are chosen so that

$$(J)^{\dagger} = \overline{J},$$
 $(J_3)^{\dagger} = -J_3,$ (3.5.8)

$$(I)^{\dagger} = \overline{I},$$
 $(I_3)^{\dagger} = -I_3,$ (3.5.9)

$$(Q^i) = \overline{Q}_i, \qquad (S^i) = \overline{S}_i. \qquad (3.5.10)$$

The parameter α only appears in the anti-commutation relations (3.5.4), from which we see that two $\mathfrak{su}(2)$ R-symmetry algebras appear with relative weights α and $-(1 + \alpha)$. Note that the conformal algebra $\mathfrak{sl}(2, \mathbb{R})$ has relative weight 1. Thus the transformation

$$\alpha \leftrightarrow -(1+\alpha) \tag{3.5.11}$$

exchanges the role of two R-symmetry algebras; $J \leftrightarrow I$. On the other hand, the transformation

$$\alpha \leftrightarrow \frac{1}{\alpha}$$
 (3.5.12)

is not well-defined for our real $D(2,1;\alpha)$ superalgebra because it exchanges the role of the non-compact conformal algebra $\mathfrak{sl}(2,\mathbb{R})$ and the compact R-symmetry algebra $\mathfrak{su}(2)$.

In particular we have the isomorphism

$$D(2,1;\alpha) \cong \begin{cases} \mathfrak{su}(1,1|2) + \mathfrak{su}(2) & \text{for } \alpha = -1,0 \\ \mathfrak{osp}(4^*|2) & \text{for } \alpha = 1,-2 \\ \mathfrak{osp}(4|2) & \text{for } \alpha = -\frac{1}{2} \end{cases}$$
(3.5.13)

At $\alpha = -1$ and $\alpha = 0$ one of the R-symmetry $\mathfrak{su}(2)$ algebra is decoupled and the superalgebra D(2,1;-1) is isomorphic to the semi-direct sum $\mathfrak{su}(1,1|2) + \mathfrak{su}(2)$ ¹³. In this case one can extend the $\mathfrak{su}(1,1|2)$ superalgebra by adding the central charges. To see this let us put the $\mathfrak{su}(2)$ generators J, \overline{J} and J_3 as

$$Z \equiv \alpha J,$$
 $\overline{Z} \equiv \alpha \overline{J},$ $Z_3 \equiv \alpha J_3$ (3.5.14)

where Z, \overline{Z} and Z_3 commute with everything. Then the new generators Z, \overline{Z} and Z_3 only appear in the anti-commutations (3.5.4) and they now become

$$\{Q^{i}, \overline{Q}_{j}\} = -2\delta^{i}_{j}H, \qquad \{S^{i}, \overline{S}_{j}\} = -2\delta^{i}_{j}K, \qquad \{Q^{i}, S^{j}\} = -2\epsilon^{ij}\overline{I}, \\ \{Q^{1}, \overline{S}_{2}\} = 2\overline{Z}, \qquad \{Q^{1}, \overline{S}_{1}\} = -2D - 2Z_{3} + 2I_{3}, \\ \{Q^{2}, \overline{S}_{1}\} = -2Z, \qquad \{Q^{2}, \overline{S}_{2}\} = -2D + 2Z_{3} + 2I_{3}.$$
(3.5.15)

Hence the three generators Z, \overline{Z} and Z_3 are identified with the central charges. Note that we can only have single nonvanishing central charge by taking into account the SU(2) transformation on the three central charges.

As its name suggests, $D(2,1;\alpha)$ is regarded as a deformation of the superalgebra $D(2,1) = \mathfrak{osp}(4|2)$ that corresponds to the case $\alpha = 1$, however, we are now considering the even part of $D(2,1;\alpha)$ as $\mathfrak{sl}(2) \oplus \mathfrak{su}(2) \oplus \mathfrak{su}(2)$ not as $\mathfrak{sl}(2) \oplus \mathfrak{sl}(2) \oplus \mathfrak{sl}(2)$. The first $\mathfrak{sl}(2)$ factor is the conformal algebra and the remaining two factors are replaced with the compact algebras $\mathfrak{su}(2)$. Consequently $\mathfrak{so}^*(4)$, the non-compact version of the original factor $\mathfrak{so}(4)$ shows up for $\alpha = 1$. We see that the case of $\alpha = -\frac{1}{2}$ is self-dual under the transformation (3.5.11). In our case this degenerate case realizes the $\mathfrak{so}(4) \cong \mathfrak{su}(2) \oplus \mathfrak{su}(2)$ factor and all the other cases can be thought of as the deformations of $D(2,1;\frac{1}{2}) = \mathfrak{osp}(4|2)$.

Using the generators of $D(2, 1; \alpha)$, let us consider the supercoset of $D(2, 1; \alpha)$

$$g = e^{itH} e^{iuD} e^{izK} e^{\theta_i Q^i + \overline{\theta}^i \overline{Q}_i} e^{\psi_i S^i + \overline{\psi}^i \overline{S}_i} e^{i\varphi J + i\overline{\varphi}\overline{J}} e^{\phi J_3}$$
(3.5.16)

where the parameters $t, \theta_i, \overline{\theta}^i$ are the coordinates of the $\mathcal{N} = 4$ superspace $\mathbb{R}^{(1|4)}$ and the other parameters $u, z, \psi_i, \overline{\psi}^i, \varphi, \overline{\varphi}$ and ϕ are the $\mathcal{N} = 4$ Goldstone superfields. Note that the R-symmetry group SU(2) generated by (I, \overline{I}, I_3) , which mixes

¹³We use \oplus for the direct sum and + for the semi-direct sum.

the fermionic charges with their conjugates, is taken into the supercoset but considered as the stability subgroup. Note that our chice of the supercoset (3.5.16) is allowed for the case of $\alpha \neq 0$ where the generators (J, \bar{J}, J_3) exist.

From the supercoset one can extract left-covariant Cartan one-form Ω

$$\Omega = g^{-1} dg. \tag{3.5.17}$$

Expanding Ω over the generators, we find the the corresponding one-forms [212]

$$\omega_D = idu - 2\left(\overline{\psi}^i d\theta_i + \psi_i d\overline{\theta}^i\right) - 2izd\tilde{t}, \qquad (3.5.18)$$

$$\omega_V = \frac{e^{-i\phi}}{1 + \Lambda\overline{\Lambda}} \left(id\Lambda + \hat{\omega}_J + \Lambda^2 \hat{\overline{\omega}}_J - \Lambda \hat{\omega}_{J_3} \right), \qquad (3.5.19)$$

$$\overline{\omega}_{V} = \frac{e^{i\phi}}{1 + \Lambda\overline{\Lambda}} \left(id\overline{\Lambda} + \hat{\overline{\omega}}_{J} + \overline{\Lambda}^{2}\hat{\omega}_{J} + \overline{\Lambda}\hat{\omega}_{J_{3}} \right), \qquad (3.5.20)$$

$$\omega_{J_3} = d\phi + \frac{1}{1 + \Lambda\overline{\Lambda}} \left[i \left(d\Lambda\overline{\Lambda} - \Lambda d\overline{\Lambda} \right) + \left(1 - \Lambda\overline{\Lambda} \right) \hat{\omega}_{J_3} - 2 \left(\Lambda \hat{\overline{\omega}}_J - \overline{\Lambda} \hat{\omega}_J \right) \right] \quad (3.5.21)$$

where

$$\hat{\omega}_J = 2\alpha \left[\psi_2 d\bar{\theta}^1 - \overline{\psi}^1 \left(d\theta_2 - \psi_2 d\tilde{t} \right) \right], \qquad (3.5.22)$$

$$\hat{\overline{\omega}}_{J} = 2\alpha \left[\overline{\psi}^{2} d\theta_{1} - \psi_{1} \left(d\overline{\theta}^{2} - \overline{\psi}^{2} d\tilde{t} \right) \right], \qquad (3.5.23)$$

$$\hat{\omega}_{J_3} = 2\alpha \left[\psi_1 d\overline{\theta}^1 - \overline{\psi}^2 d\theta_1 - \psi_2 d\overline{\theta}^2 + \overline{\psi}^2 d\theta_2 + \left(\overline{\psi}^1 \psi_1 - \overline{\psi}^2 \psi_2 \right) d\tilde{t} \right], \qquad (3.5.24)$$

$$d\tilde{t} = dt + i\left(\theta_i d\bar{\theta}^i + \bar{\theta}^i d\theta_i\right), \qquad (3.5.25)$$

$$\Lambda = \frac{\tan\sqrt{\overline{\phi}\overline{\phi}}}{\sqrt{\overline{\phi}\overline{\phi}}}.$$
(3.5.26)

For the $\mathcal{N} = 4$ superspace $\mathbb{R}^{(1|4)}$ parametrized by [210]

$$\mathbb{R}^{(1|4)} = (t, \theta_i, \overline{\theta}^j), \qquad (\theta_i)^{\dagger} = \overline{\theta}^i, \qquad i, j = 1, 2 \qquad (3.5.27)$$

we will introduce the covariant derivatives

$$D^{i} = \frac{\partial}{\partial \theta_{i}} + i\overline{\theta}^{i}\frac{\partial}{\partial t}, \qquad \overline{D}_{j} = \frac{\partial}{\partial\overline{\theta}^{j}} + i\theta_{j}\frac{\partial}{\partial t}, \qquad \left\{D^{i}, \overline{D}_{j}\right\} = 2i\delta_{j}^{i}\partial_{t}. \qquad (3.5.28)$$

The supercharges *Q* and \overline{Q} can be expressed by

$$Q^{i} = \frac{\partial}{\partial \theta_{i}} - i\overline{\theta}^{i}\frac{\partial}{\partial t}, \qquad \overline{Q}_{j} = \frac{\partial}{\partial\overline{\theta}^{j}} - i\theta_{j}\frac{\partial}{\partial t}, \qquad \left\{Q^{i}, \overline{Q}_{j}\right\} = -2i\delta_{j}^{i}\partial_{t} \qquad (3.5.29)$$

in the superspace.

By acting a particular element on the supercoset element (3.5.16) from the left, we can find the corresponding transformations.

1. supersymmetry transformations

Acting the element

$$g_{\epsilon} = e^{\epsilon_i Q^i + \overline{\epsilon}^i \overline{Q}_i} \in D(2, 1; \alpha), \qquad (3.5.30)$$

we obtain the supersymmetry transformations

$$\delta t = i \left(\theta \overline{\epsilon} - \epsilon \overline{\theta} \right), \qquad (3.5.31)$$

$$\delta\theta_i = \epsilon_i, \tag{3.5.32}$$

$$\delta \overline{\theta}^i = \overline{\epsilon}^i. \tag{3.5.33}$$

2. superconformal boost transformations

Acting the element

$$g_{\eta} = e^{\eta_i S^i + \overline{\eta}_i \overline{S}_i}, \qquad (3.5.34)$$

one finds the superconformal boost transformations [213, 212, 214]

.

$$\delta t = -it \left(\eta \overline{\theta} + \overline{\eta} \theta \right) + (1 + 2\alpha) \left(\theta \overline{\theta} \right) \left(\eta \overline{\theta} - \overline{\eta} \theta \right), \qquad (3.5.35)$$

_

$$\delta\theta_{i} = \eta_{i}t - 2i\alpha\theta_{i}(\theta\overline{\eta}) + 2i(1+\alpha)\theta_{i}(\overline{\theta}\eta) - i(1+2\alpha)\eta_{i}(\theta\overline{\theta}), \qquad (3.5.36)$$

$$\delta u = -2i\left(\eta \overline{\theta} + \overline{\eta} \theta\right), \qquad (3.5.37)$$

$$\delta \phi = 2\alpha \left[\overline{\eta}^{1} \theta_{1} - \overline{\eta}^{2} \theta_{2} - \eta_{1} \overline{\theta}^{1} + \eta_{2} \overline{\theta}^{2} + \left(\overline{\eta}^{2} \theta_{1} - \eta_{1} \overline{\theta}^{2} \right) \Lambda + \left(\overline{\eta}^{1} \theta_{2} - \eta_{2} \overline{\theta}^{1} \right) \overline{\Lambda} \right], \qquad (3.5.38)$$
$$\delta \Lambda = 2i\alpha \left[\theta_{2} \overline{\eta}^{1} - \overline{\theta}^{1} \eta_{2} + \left(\overline{\theta}^{2} \eta_{1} - \theta_{1} \overline{\eta}^{2} \right) \Lambda^{2} \right]$$

$$+\left(\overline{\theta}^{1}\eta_{1}-\theta_{1}\overline{\eta}^{1}+\theta_{2}\overline{\eta}^{2}-\overline{\theta}^{2}\eta_{2}\right)\Lambda\right]$$
(3.5.39)

and

$$\delta(dtd^{4}\theta) = 2i\left(\eta\overline{\theta} + \overline{\eta}\theta\right)dtd^{4}\theta, \qquad (3.5.40)$$
$$\delta D^{i} = i\left[(2+\alpha)(\eta\overline{\theta}) + \alpha(\theta\overline{\eta})\right]D^{i}$$

$$-2i(1+\alpha)(\overline{\eta}\overline{\theta})\overline{D}^{i} - 2i\alpha \left[\eta^{(i}\overline{\theta}_{k)} + \theta^{(i}\overline{\epsilon}_{k)}\right]D^{k}, \qquad (3.5.41)$$

$$\delta \overline{D}_{i} = i \left[(2+\alpha)(\overline{\eta}\theta) + \alpha(\overline{\theta}\eta)\overline{D}_{i} \right] - 2i(1+\alpha)(\theta\eta)D_{i} - 2i\alpha \left[\eta_{(i}\overline{\theta}_{k)} + \theta_{(i}\overline{\epsilon}_{k)} \right] \overline{D}^{k}.$$
(3.5.42)

At this stage we are ready to derive the irreducible off-shell supermultiplets which allow us to construct the $D(2,1;\alpha)$ superconformal mechanics. The strategy is to extract the irreducible superfields from the Goldstone superfields $u, z, \psi_i, \overline{\psi}^i, \varphi, \overline{\varphi}$ by imposing the appropriate constraints. Since the number of the fermionic Goldstone superfields is four which coincides with the minimal number of the fermionic fields in $\mathcal{N} = 4$ supermultiplets, we attempt to reduce the number of the bosonic Goldstone superfields. It has been discussed [213, 212] that such irreducibility condition can be achieved by requiring that all spinor derivatives of all bosonic superfields are expressed in terms of the fermionic fields ψ_i and $\overline{\psi}^i$. From the equations (3.5.18)-(3.5.22), we see that this requirement corresponds to the constraints on the corresponding Cartan forms ω_D , ω_I , $\overline{\omega}_V \omega_{I_3}$.

3.5.1 (4, 4, 0) supermultiplet

Let us begin with the most general case where the supercoset (3.5.17) holds all four bosonic Goldstone superfields u, φ , $\overline{\varphi}$ and ϕ . If we require that the all spinor covariant derivatives of these bosonic superfields can be expressed by ψ_i , $\overline{\psi}^i$, then (3.5.18)-(3.5.22) lead to

$$\omega_D = \omega_J |= \overline{\omega}_J |= \omega_{J_3} = 0 \tag{3.5.43}$$

where | represents the restriction to spinor projection. The set of constraints (3.5.43) can be rewritten as

$$D^{(i}q^{j)} = 0,$$
 $\overline{D}^{(i}q^{j)} = 0,$ $D^{(i}\overline{q}^{j)} = 0,$ $\overline{D}^{(i}\overline{q}^{j)} = 0$ (3.5.44)

where

$$q^{1} = \frac{e^{\frac{1}{2}(\alpha u - i\phi)}}{\sqrt{1 + \Lambda\overline{\Lambda}}}\Lambda, \qquad \qquad q^{2} = \frac{e^{\frac{1}{2}(\alpha u - i\phi)}}{\sqrt{1 + \Lambda\overline{\Lambda}}},$$
$$\overline{q}_{1} = \frac{e^{\frac{1}{2}(\alpha u + i\phi)}}{\sqrt{1 + \Lambda\overline{\Lambda}}}, \qquad \qquad \overline{q}_{2} = \frac{e^{\frac{1}{2}(\alpha u + i\phi)}}{\sqrt{1 + \Lambda\overline{\Lambda}}} \qquad (3.5.45)$$

are identified with four $\mathcal{N} = 4$ superfields. This multiplet was discussed in [199, 200, 215, 216, 212, 217, 211, 218, 219] and was considered in $\mathcal{N} = 4$ harmonic superspace [214]. The constraints (3.5.44) lead to the following independent fields:

$$\begin{cases} q^{i} & 4 \text{ physical bosons} \\ D_{i}q^{i}, \overline{D}_{i}q^{i}, D_{i}\overline{q}^{i}, \overline{D}_{i}\overline{q}^{i} & 4 \text{ fermions.} \end{cases}$$
(3.5.46)

The superfield q_i contains 4 bosonic, 4 fermionic fields and no auxiliary fields and is diagnosed as the (4,4,0) supermultiplet. Since q^i and their set of constraints (3.5.44) are similar to the d = 4 $\mathcal{N} = 2$ hypermultiplet, it is also called hypermultiplet. However, the conditions (3.5.44) for the (4,4,0) multiplet defines the off-shell multiplet as opposed to the d = 4 $\mathcal{N} = 2$ hypermultiplet.

Remarkably it has been discussed that all other $\mathcal{N} = 4$ supermultiplets can be obtained from (4, 4, 0) multiplet via reduction process either on the component action [217] or on the superfield action [211, 218, 219]. Accordingly the (4, 4, 0) multiplet can be viewed as a fundamental multiplet.

Since we know the superconformal boost transformations (3.5.35)-(3.5.39) for the original Goldstone superfields, we can read off the superconformal boost transformations for the superfields q^i , \bar{q}_i

$$\delta q^{i} = 2i\alpha \left(\overline{\theta}^{i} \eta_{j} - \theta^{i} \overline{\eta}_{j}\right) q^{j}.$$
(3.5.47)

This leads to the transformations $\delta(q\bar{q}) = -2i\alpha(\eta\bar{\theta} + \bar{\eta}\theta)(q\bar{q})$, which cancel the transformation (3.5.40) of the integration measure. Therefore we can write super-conformally invariant superfield action

$$S = \int dt d^4\theta \ (q\overline{q})^{\frac{1}{\alpha}}.$$
 (3.5.48)

Note that this vanishes when $\alpha = -1$ due to the constraints (3.5.44). For $\alpha = -1$ the superconformal superfield action is given by [214, 212]

$$S = \int dt d^4\theta \; \frac{\ln\left(q\overline{q}\right)}{q\overline{q}}.\tag{3.5.49}$$

It is worthwhile to remark that these two expressions (3.5.48) and (3.5.49) can be written uniformly by adding the overall factor as

$$\frac{1}{1+\alpha} \int dt d^4\theta \ (q\overline{q})^{\frac{1}{\alpha}}.$$
(3.5.50)

One can check that (3.5.50) is regular for any α and coincides with (3.5.49) at $\alpha = -1$.

Although there is a superpotential term for the (4, 4, 0) multiplet [214] which is a Wess-Zumino type term ¹⁴ of first order in time derivative, it does not produce any non-trivial potential for physical bosons. Therefore one cannot construct $\mathcal{N} =$ 4 superconformal mechanics with the non-trivial potential for the physical bosons by using the (4, 4, 0) multiplet only. On the other hand, it has been discussed [211, 218] that the gauged action of the (4, 4, 0) multiplet generates more generic actions.

¹⁴ The superfield Wezz-Zumino type potential term for all $\mathcal{N} = 4$ multiplets can be represented manifestly only in the harmonic superspace [137].
3.5.2 (3, 4, 1) supermultiplet

Let us set $\phi = 0$ in the supercoset (3.5.16). This enforces us to put the corresponding subgroup $U(1) \subset SU(2)$ into the stability subgroup and thus the resulting supercoset involves $SL(2,\mathbb{R}) \times SU(2)/U(1)$. To realize the spinor covariant derivatives of the remaining bosonic superfields u, Λ , $\overline{\Lambda}$, we should impose the conditions

$$\omega_D = \omega_I |= \overline{\omega}_I |= 0. \tag{3.5.51}$$

The set of conditions (3.5.51) can be expressed as

$$D^{(i}V^{jk)} = 0,$$
 $\overline{D}^{(i}V^{jk)} = 0$ (3.5.52)

where

$$V^{11} = -i\sqrt{2}e^{\alpha u}\frac{\Lambda}{1+\Lambda\overline{\Lambda}}, \quad V^{22} = i\sqrt{2}e^{\alpha u}\frac{\overline{\Lambda}}{1+\Lambda\overline{\Lambda}}, \quad V^{12} = \frac{i}{\sqrt{2}}e^{\alpha u}\frac{1-\Lambda\overline{\Lambda}}{1+\Lambda\overline{\Lambda}}.$$
 (3.5.53)

Note that the $\mathcal{N} = 4$ superfields V^{ij} is real and satisfy the relations

$$V^{ij} = V^{ji}, \qquad \overline{V^{ij}} = \epsilon_{ik}\epsilon_{jl}V^{kl}, \qquad V^2 := V^{ij}V_{ij} = e^{2\alpha u}. \qquad (3.5.54)$$

The superfield V^{ij} obeying the constraints (3.5.52) was firstly introduced in [220] and later discussed in [221, 222, 70, 213, 212, 214, 211, 218]. The constraints (3.5.52) give rise to the independent components

$$\begin{cases} V^{11}, V^{12}, V^{12} & 3 \text{ physical bosons} \\ D^{1}V^{12}, D^{2}V^{12}, \overline{D}^{1}V^{12}, \overline{D}^{2}V^{12} & 4 \text{ fermions} \\ D^{i}\overline{D}^{j}V_{ij} & 1 \text{ auxiliary boson} \end{cases}$$
(3.5.55)

Thus we can identify the superfield V^{ij} with the (3, 4, 1) supermultiplet. Since the constraints (3.5.52) are obtained by the dimensional reduction from the constraints of the d = 4 $\mathcal{N} = 2$ tensor multiplet [223], the (3, 4, 1) multiplet is also called tensor multiplet ¹⁵.

From (3.5.37)-(3.5.39), one can read the $D(2,1;\alpha)$ superconformal boost transformations of V^{ij}

$$\delta V^{ij} = -2i\alpha \left[\left(\eta \overline{\theta} + \overline{\eta} \theta \right) V^{ij} + \left(\eta^{(i} \overline{\theta}_k - \overline{\eta}_k \theta^{(i)} \right) V^{j)k} + \left(\eta_k \overline{\theta}^{(i)} - \overline{\eta}^{(i)} \theta_k \right) V^{j)k} \right]$$
(3.5.56)

¹⁵ The superfield V^{ij} can also be obtained by the dimensional reduction from $d = 4 \mathcal{N} = 1$ vector multiplet [221] as the spatial component of d = 4 Abelian gauge vector connection superfield.

The superfield action for the kinetic term is given by [213, 214, 212]

$$S_{\rm kin} = \int dt d\theta \ (V^2)^{\frac{1}{2\alpha}} \tag{3.5.57}$$

where V^2 is defined in (3.5.54). The action (3.5.57) vanishes for $\alpha = -1$. The superfield action for the kinetic term in the case of $\alpha = -1$ is

$$S_{\rm kin} = -\frac{1}{2} \int dt d^4 \theta \ (V^2)^{-\frac{1}{2}} \ln V^2. \tag{3.5.58}$$

It has been pointed out [213] that both of the action (3.5.57) and (3.5.58) can be described in a unified form as

$$S_{\rm kin} = \frac{1}{1+\alpha} \int dt d\theta \ (V^2)^{\frac{1}{2\alpha}}.$$
 (3.5.59)

The superconformally invariant potential term for the (3, 4, 1) multiplet can be written as [213]

$$S_{\text{pot}} = -i\sqrt{2} \int dt d^4\theta \,\left[\int_0^1 dy \partial_y \mathcal{W} \frac{1}{\sqrt{V^2}}\right]$$
(3.5.60)

where \mathcal{W} is the prepotential satisfying

$$V^{ij} = D^{(i}\overline{D}^{j)}\mathcal{W}, \qquad \overline{W} = -\mathcal{W}.$$
 (3.5.61)

Note that the constraints (3.5.52) are solved by an unconstraint prepotential W. Alternative way to obtain the potential term for the (3,4,1) multiplet has been proposed as an integral over the analytic harmonic superspace [214].

Combining the kinetic terms (3.5.59) and the potential terms (3.5.60), we find the bosonic superconformal actions in component fields as [213]

$$S_{\text{bosonic}} = \mu^{-1} \frac{\alpha^2}{1+\alpha} (S_{\text{kin}})_{\text{bosonic}} + \nu (S_{\text{pot}})_{\text{bosonic}}$$

$$= \int dt \left[\mu^{-1} \alpha^2 e^u \dot{u}^2 + 4\mu^{-1} e^u \frac{\dot{\Lambda}\dot{\Lambda}}{(1+\Lambda\overline{\Lambda})^2} - \frac{1}{4}\mu \nu^2 e^{-u} + i\nu \frac{\overline{\Lambda}\dot{\Lambda} - \Lambda\dot{\overline{\Lambda}}}{1+\Lambda\overline{\Lambda}} \right]$$

$$= \frac{1}{2} \int dt \left[4\alpha^2 \mu \dot{r}^2 + \mu r^2 \left(\dot{\vartheta}^2 + \sin^2 \vartheta \dot{\varphi}^2 \right) - \frac{\nu^2}{\mu r^2} + 2\nu \cos \vartheta \dot{\varphi} \right]$$

$$= \int dt \left[\mu g_{ij}(X) \dot{X}^i \dot{X}^j - \frac{1}{4\mu} \frac{\nu^2}{|\mathbf{X}|^2} + 2i\nu \frac{\epsilon^{3ij} X^i \dot{X}^j}{(X^2 + |\mathbf{X}|) |\mathbf{X}|} \right] \qquad (3.5.62)$$

where

$$\Lambda = \tan \frac{\vartheta}{2} e^{i\varphi}, \qquad \qquad e^{\frac{u}{2}} = \frac{1}{\sqrt{2}} \mu r, \qquad (3.5.63)$$

$$g_{ij}(X) = \delta_{ij} + (4\alpha^2 - 1)\frac{X_i X_j}{|\mathbf{X}|^2}.$$
(3.5.64)

Observing the two explicit expressions (3.5.45) and (3.5.54) for the two superfields q^i and V^{ij} in terms of the initial Goldstone superfields, we can express the superfields V^{ij} as

$$V^{11} = -i\sqrt{2}q^{1}\overline{q}^{1}, \quad V^{22} = -i\sqrt{2}q^{2}\overline{q}^{2}, \quad V^{12} = -\frac{i}{\sqrt{2}}\left(q^{1}\overline{q}^{2} + q^{2}\overline{q}^{1}\right). \quad (3.5.65)$$

Also one can check that if the the irreducible constraints (3.5.44) for the (4,4,0) multiplet are satisfied by q^i , \bar{q}^i , then the constraints (3.5.52) for the (3,4,1) multiplet are also solved by (3.5.65) [214]. However, it is important to note that (3.5.65) are not general but rather special solutions to the(3,4,1) multiplet. So the generic (3,4,1) multiplet cannot be covered by (3.5.65).

3.5.3 (2, 4, 2) supermultiplet

Now we will put u = 0, z = 0, $\phi = 0$ in the supercoset (3.5.16). Then the supercoset contain only two bosonic fields φ , $\overline{\varphi}$ or equivalently Λ , $\overline{\Lambda}$, which parametrize the two-sphere $S^2 \sim SU(2)/U(1)$. The condition that the spinor covariant derivatives of φ , $\overline{\varphi}$ can be expressed in terms of ψ , $\overline{\psi}$ is

$$\omega_{\bar{I}}| = \omega_{\bar{I}}| = 0. \tag{3.5.66}$$

For $\alpha \neq -1$ these the conditions (3.5.66) are written as

$$D^1 \Lambda = -\Lambda D^2 \Lambda, \qquad \qquad \overline{D}_2 \Lambda = \Lambda \overline{D}_1 \Lambda. \qquad (3.5.67)$$

Under the constraints (3.5.67) the superfield Λ , $\overline{\Lambda}$ yields the independent component fields

$$\begin{cases} \Lambda, \overline{\Lambda} & 2 \text{ physical bosons} \\ -D^1 \overline{\Lambda}, \overline{D}_1 \Lambda, D^2 \Lambda, -\overline{D}_2 \overline{\Lambda} & 4 \text{ fermions} \\ \overline{D}_1 D^2 \Lambda, \overline{D}_2 D^1 \overline{\Lambda} & 2 \text{ auxiliary bosons,} \end{cases}$$
(3.5.68)

which implies the (2, 4, 2) supermultiplet. This multiplet is called non-linear chiral multiplet because the constraints (3.5.67) can be viewed as the modified chirality conditions so that they are also covariant with respect to $D(2, 1; \alpha)$. Note that, apart from the non-linear realization of $D(2, 1; \alpha)$, the $\mathcal{N} = 4$ chiral multiplet (2, 4, 2) is constructed by a complex superfields $\phi, \overline{\phi}$ obeying the constraints

$$D^i \phi = 0,$$
 $\overline{D}_i \overline{\phi} = 0.$ (3.5.69)

It has been discussed that one cannot construct superconformal superfield actions out of the (2, 4, 2) multiplet alone due to the absence of the dilaton u [212]. In order to obtain superconformal superfield actions, the coupling to some other $\mathcal{N} = 4$ supermultiplets is needed.

In terms of the hypermultiplet q, \overline{q} , the superfield Λ , $\overline{\Lambda}$ can be written as

$$\Lambda = -\frac{q^1}{q^2}, \qquad \qquad \overline{\Lambda} = -\frac{\overline{q}_1}{\overline{q}_2}. \qquad (3.5.70)$$

These are just the special solutions to the constraint equations (3.5.67) for the nonlinear chiral multiplet.

3.5.4 (1,4,3) supermultiplet

Let us retain the dilaton u alone in the supercoset (3.5.16). This corresponds to putting two SU(2) R-symmetry factors into the stability subgroup. The irreducible condition

$$\omega_D| = 0 \tag{3.5.71}$$

just implies that the four spinor derivatives of u is expressed by the four fermionic Goldstone superfield $\psi, \overline{\psi}$. Therefore the equation (3.5.71) does not impose any constraints on the superfield u. The independent component fields are [210]

$$\begin{cases} e^{u} & 1 \text{ physical bosons} \\ D^{i}u, \overline{D}_{i}u & 4 \text{ fermions} \\ [D^{(i}, \overline{D}^{j)}]e^{u}, [D^{i}, \overline{D}_{i}]e^{u} & 3 \text{ auxiliary bosons} \end{cases}$$
(3.5.72)

and this means the (1, 4, 3) supermultiplet. However, as was shown in [210], one should impose additional irreducible constraints on the dilaton u

$$D^{i}D_{i}e^{-\alpha u} = \overline{D}_{i}\overline{D}^{i}e^{-\alpha u} = [D^{i},\overline{D}_{i}]e^{-\alpha u} = 0$$
(3.5.73)

for the minimal (1, 4, 3) multiplet. It has been pointed out [213] that if we build up the *u* superfield out of the (3, 4, 1) superfield V^{ij} satisfying (3.5.52) as

$$e^{-\alpha u} = \frac{1}{\sqrt{V^2}},\tag{3.5.74}$$

then u automatically obeys the minimal constraints (3.5.73). Substituting the relation (3.5.74) into (3.5.60) and (3.5.58), we obtain the superconformal superfield

action 16

$$S = \int dt d^4\theta \ e^u \tag{3.5.75}$$

for $\alpha \neq -1$ and

$$S = \int dt d^4 \theta \ e^u u \tag{3.5.76}$$

for $\alpha = -1$. By putting the overall factor, we can express the superconformal superfield actions for both cases as [213]

$$S = \frac{1}{1+\alpha} \int dt d^4\theta \ e^u. \tag{3.5.77}$$

Combining (3.5.37) and (3.5.40), one can check that the superfield action (3.5.77) is invariant under the superconformal boost transformations. Note that (3.5.77) is not defined at $\alpha = 0$ because of our choice of the supercoset (3.5.16) and it should be treated separately [211, 218].

Inserting the appropriate set of component fields which solve the minimal constraints (3.5.73) into the superfield action (3.5.77), integrating over the Grassmann coordinates θ_i , $\overline{\theta}^i$ and integrating out the auxiliary fields, one finds the one particle $D(2,1;\alpha)$ superconformal mechanical model [224, 137]

$$S = \frac{1}{2} \int dt \left[\dot{x}^2 + i \left(\overline{\psi}_i \dot{\psi}^i - \dot{\overline{\psi}}_i \psi^i \right) + \frac{2}{3} (1 + 2\alpha) \frac{\psi^i \overline{\psi}^j \psi_{(i} \overline{\psi}_{j)}}{x^2} \right].$$
(3.5.78)

Although the action (3.5.78) does not possess bosonic potential at the classical level, upon the quantization the anti-commutation for the fermions may yield a purely bosonic potential term. We see that the potential terms just flip the overall sign under the transformation α (3.5.11).

As we have already seen (3.5.13), when $\alpha = -1, 0$ the $\mathcal{N} = 4$ superconformal algebra $D(2,1;\alpha)$ is isomorphic to the semi-direct sum of $\mathfrak{su}(1,1|2)$ and $\mathfrak{su}(2)$, which implies that one of the SU(2) symmetry is broken and the superalgebra $\mathfrak{su}(1,1|2)$ allows for the central charge. So the irreducible constraints for the bosonic Goldstone superfields can be weakened by adding the central charge [224, 137]. The constraints (3.5.73) can be modified as

$$D^i D_i e^{-\alpha u} = 0, \qquad \overline{D}_i \overline{D}^i e^{-\alpha u} = 0, \qquad [D^i, \overline{D}_i] e^{-\alpha u} = c \qquad (3.5.79)$$

or

$$D^i D_i e^{-\alpha u} = c, \qquad \overline{D}_i \overline{D}^i e^{-\alpha u} = c, \qquad [D^i, \overline{D}_i] e^{-\alpha u} = 0 \qquad (3.5.80)$$

¹⁶In the original work in [210] only the SU(1,1|2) invariant action with $\alpha = -1$ was considered.

where *c* is the central charge of the $\mathfrak{su}(1,1|2)$ superalgebra. The two constraints correspond to the case where the broken SU(2) symmetry is taken as the rotation of θ coordinates and $\overline{\theta}$ coordinates respectively [210]. The solutions to the new constraint equations acquire the additional term proportional to $\theta\overline{\theta}c$. Then one obtains the one particle SU(1,1|2) superconformal mechanical action [210, 224, 137]

$$S = \frac{1}{2} \int dt \left[\dot{x}^2 + i \left(\overline{\psi}_i \dot{\psi}^i - \dot{\overline{\psi}}_i \psi^i \right) - \frac{\left(c + \overline{\psi}_i \psi^i \right)^2}{x^2} \right].$$
(3.5.81)

Note that the additional contribution from the central charge *c* yields the inverse square type bosonic potential at the classical level.

3.5.5 (0, 4, 4) supermultiplet

Although we have seen that the irreducibility conditions for the supermultiplet can be systematically obtained by means of the non-linear realization method, there is a further possible supermultiplet (0, 4, 4)¹⁷. It is described by a fermionic analytic superfield in the harmonic superspace (HSS) [214].

The harmonic superspace (HSS) is the extension of the original superspace by introducing the new commuting harmonic coordinate u_i^{\pm} , i = 1, 2 parametrizing the internal degrees of freedom as the two-sphere $S^2 \sim SU(2)/U(1)$ with SU(2) being the R-symmetry [225].

$$\mathbb{H}\mathbb{R}^{(1+2|4)} = (t_A, \theta^+, \overline{\theta}^+, \theta^-, \overline{\theta}^-, u_i^+, u_k^-)$$
$$= (\zeta, u_i^+, u_k^-, \theta^-, \overline{\theta}^-)$$
(3.5.82)

where

$$t_{A} := t - i(\theta^{+}\overline{\theta}^{-} + \theta^{-}\overline{\theta}^{+}), \qquad \theta^{\pm} = \theta^{i}u_{i}^{\pm}, \qquad \overline{\theta}^{\pm} = \overline{\theta}^{i}u_{i}^{\pm}, \qquad (3.5.83)$$
$$u^{+i}u_{i}^{-} = 1, \qquad u_{i}^{+}u_{j}^{-} - u_{j}^{+}u_{i}^{-} = \epsilon_{ij}. \qquad (3.5.84)$$

The significant property is the existence of an analytic subspace (ASS), which is the quotient of
$$\mathbb{H}^{(1+2|4)}$$
 by $\{\theta^-, \overline{\theta}^-\}$

$$\mathbb{AR}^{(1+2|2)} = (\zeta, u) = (t_A, \theta^+, \overline{\theta}^+, u_i^+, u_k^-).$$
(3.5.85)

¹⁷At least the author does not know the (**0**, **4**, **4**) supermultiplet based on the non-linear realization of the superconformal group $D(2, 1; \alpha)$.

The covariant derivatives in the analytic basis of HSS, $(\zeta, u, \theta^-, \overline{\theta}^-)$ are defined by

$$D^+ = \frac{\partial}{\partial \theta^-}, \qquad \qquad \overline{D}^+ = -\frac{\partial}{\partial \overline{\theta}^-}, \qquad (3.5.86)$$

$$D^{-} = -\frac{\partial}{\partial\theta^{+}} - 2i\overline{\theta}^{-}\frac{\partial}{\partial t_{A}}, \qquad \overline{D}^{-} = \frac{\partial}{\partial\overline{\theta}^{+}} - 2i\theta^{-}\frac{\partial}{\partial t_{A}} \qquad (3.5.87)$$

and the harmonic covariant derivatives in the analytic basis of HSS are

$$D^{\pm\pm} = \partial^{\pm\pm} + 2i\theta^{\pm}\overline{\theta}^{\pm}\frac{\partial}{\partial t_A} + \theta^{\pm}\frac{\partial}{\partial\theta^{\mp}} + \overline{\theta}^{\pm}\frac{\partial}{\partial\overline{\theta}^{\mp}}.$$
 (3.5.88)

The constraints for the (0, 4, 4) superfield $\Psi^{+a}(\zeta, u)$, $a = 1, 2^{-18}$ are given by [214]

$$D^{++}\Psi^{+a} = 0. (3.5.89)$$

The solution of the constraint (3.5.89) is written as

$$\Psi^{+a}(\zeta, u) = \psi^{ia}u_i^+ + \theta^+ \xi^a + \overline{\theta}^+ \overline{\xi}^a + 2i\theta^+ \overline{\theta}^+ \dot{\psi}^{ia}u_i^-$$
(3.5.90)

and the independent component fields are

$$\begin{cases} \psi^{ia} & \text{4 fermions} \\ \bar{\zeta}^{a}, \bar{\zeta}^{a} & \text{4 auxiliary bosons.} \end{cases}$$
(3.5.91)

The (0, 4, 4) superfield Ψ^{+a} has been discussed in [211, 218, 227]. The action takes the form

$$S = \frac{1}{2} \int du d\zeta^{--} \Psi^{+a} \Psi^{+}_{a}$$
$$= \int dt \left[i \psi^{ia} \dot{\psi}_{ia} + \xi^{a} \overline{\xi}_{a} \right]. \qquad (3.5.92)$$

Although the action (3.5.92) contains only the kinetic term of the free fermions and the quadratic term of the bosonic auxiliary fields, if we appropriately couple the (0,4,4) multiplet to the other $\mathcal{N} = 4$ supermultiplets, we may produce bosonic potentials [211, 218, 227].

3.5.6 Multi-particle model

WDVV equation

We have seen that the superspace and superfield formalism based on the nonlinear realization technique is useful to build up $\mathcal{N} = 4$ superconformal mechanical models possessing $D(2, 1; \alpha)$ symmetry. However, it is known that the direct

¹⁸The indices a = 1, 2 denote the doublet of the extra SU(2) called the Pauli-Gürsey group [226].

generalization of the one particle analysis does not work well for the construction of the $D(2,1;\alpha)$ multi-particle superconformal mechanical systems ¹⁹. Hence it is insightful to investigate the construction for the $\mathcal{N} = 4$ multi-particle superconformal mechanics in the component level.

Let us consider *N* particles on \mathbb{R} with canonical variables x^a and their momenta p_p where $a = 1, \dots, N$ label the particles. The $\mathcal{N} = 4$ supersymmetry leads to two complex fermions $\psi_i^a, \overline{\psi}^{ai}$, i = 1, 2. In addition, we also consider a one pair of bosonic isospin variables u^i , i = 1, 2 which parametrize the internal degrees of freedom ²⁰.

Now we impose the ansatz for the supercharges Q_i and \overline{Q}^i of the form [228]

$$Q_{i} = p_{a}\psi_{i}^{a} + U_{a}(x)K_{ij}\psi^{aj} + iF_{abc}(x)\psi^{aj}\psi_{j}^{b}\overline{\psi}_{i}^{c}, \qquad (3.5.93)$$

$$\overline{Q}^{i} = p_{a}\overline{\psi}^{ai} + U_{a}(x)K_{ij}\overline{\psi}^{aj} - iF_{abc}(x)\overline{\psi}^{aj}\overline{\psi}^{b}_{j}\psi^{c}_{i}$$
(3.5.94)

where $U_a(x)$ and $F_{abc}(x)$ are homogeneous functions of degree -1 in x^a and

$$K_{ij} = \frac{i}{2} \left(u_i \overline{u}_j + u_j \overline{u}_i \right).$$
(3.5.95)

Let us consider the Dirac brackets

$$\{x^a, p_b\} = \delta^a_b, \qquad \left\{\psi^a_i, \overline{\psi}^{bj}\right\} = -\frac{i}{2}\delta^k_i\delta^{ab}, \qquad \left\{u^i, \overline{u}_k\right\} = -i\delta^i_k. \tag{3.5.96}$$

Then the $\mathcal{N} = 4$ superalgebra

$$\left\{Q_i, \overline{Q}^j\right\} = 2i\delta_i^j H \tag{3.5.97}$$

implies that [229, 228]

$$\partial_a U_b - \partial_b U_a = 0, \tag{3.5.98}$$

$$\partial_a F_{bcd} - \partial_b F_{acd} = 0 \tag{3.5.99}$$

and

$$F_{cae}F_{ebd} - F_{cbe}F_{ead} = 0, (3.5.100)$$

$$\partial_a U_b - U_a U_b - F_{abc} U_c = 0. \tag{3.5.101}$$

¹⁹In the case of $\alpha = -1, 0$ with SU(1, 1|2) symmetry, the standard $\mathcal{N} = 4$ superspace description can be generalized to the multi-particle case [228].

²⁰It has been discussed [228] that isospin variables is needed in order to obtain the multi-particle $D(2,1;\alpha)$ superconformal mechanics for $\alpha \neq -1, 0$. See [229] for $\alpha = -1, 0$, i.e. SU(1,1|2) superconformal mechanics.

The first set of equations (3.5.98) and (3.5.99) can be solved by

$$U_a(x) = \partial_a U(x), \qquad \qquad F_{abc}(x) = \partial_a \partial_b \partial_c F(x) \qquad (3.5.102)$$

where U(x) and F(x) are the prepotentials, the scalar functions defined up to polynomials of degree 0 and 2 in x^a respectively. Therefore we have two nonlinear differential equations (3.5.100) and (3.5.101) for the prepotential U(x) and F(x). Quite interestingly the equation (3.5.100) is the so-called Witten-Dijkgraaf-Verlinde-Verlinde (WDVV) equation [230, 231]. It has been established that the solution of WDVV equations determines the structure of a Frobenius manifold. The other equation (3.5.101) describes the so-called twisted periods U_a of the Frobenius manifold [232, 233]²¹.

Under the conditions (3.5.100) and (3.5.101) the Hamiltonian can be written as

$$H = \frac{1}{4} p_a p_a + \frac{1}{8} J^{ij} J_{ij} U_a U_a$$

$$- i U_{ab} K_{ij} \psi^{ai} \overline{\psi}^{bj} - \frac{1}{2} F_{abcd} \psi^{ai} \psi^b_i \overline{\psi}^c_j \overline{\psi}^{dj}. \qquad (3.5.104)$$

We should note that the $\mathcal{N} = 4$ superconformal algebra $D(2,1;\alpha)$ has not been taken into account so far. So the WDVV equation and the twisted period equation are just the requirement for the conservation of $\mathcal{N} = 4$ supersymmetry.

To realize $D(2, 1; \alpha)$ superconformal algebra let us introduce the conformal generators D, K, superconformal generators S_i , \overline{S}^i and the R-symmetry generators J_{ij}

$$D = -\frac{1}{4} \{x^p p_a\}, \qquad K = x^a x_a, \qquad (3.5.105)$$

$$S_i = -2x^a \psi_i^a, \qquad \overline{S}^i = -2x^a \overline{\psi}^{ia}, \qquad (3.5.106)$$

$$J_{ij} = K_{ij} + 2i\psi^a_{(i}\overline{\psi}^a_{j)},$$
(3.5.107)

$$I_{11} = i\psi_i^a\psi^{ia}, \qquad \qquad I_{22} = -i\overline{\psi}^{ia}\overline{\psi}_i^a, \qquad \qquad I_{12} = i\psi_i^a\overline{\psi}^{ia}. \qquad (3.5.108)$$

From the dilatation invariance we require the homogeneity

$$\partial_b(x^a U_a) = (x^a \partial_a + 1) U_b = 0,$$
 (3.5.109)

$$\partial_b(x^a F_{acd}) = (x^a \partial_a + 1) F_{bcd} = 0.$$
 (3.5.110)

²¹ Any function \tilde{p} satisfying

$$\frac{\partial \xi_a}{\partial p^b} = \nu G^{cd} \frac{\partial^3 F_*(p)}{\partial p^d \partial p^b} \xi_c, \qquad \qquad \xi_a = \frac{\partial \tilde{p}(p;\nu)}{\partial p^a} \tag{3.5.103}$$

is called twisted period of the Frobenius manifold where p^a are periods.

The remaining $D(2, 1; \alpha)$ superconformal algebra (3.5.1)-(3.5.6) then leads to

$$x^a U_a = 2\alpha, \qquad (3.5.111)$$

$$x^{a}F_{abc} = -(1+2\alpha)\delta_{bc}.$$
 (3.5.112)

For $\alpha \neq -\frac{1}{2}$ the prepotential *F* is non-vanishing and any two values of α are related by a rescaling under the transformation (3.5.11). In this sense the two conditions (3.5.111) and (3.5.112) can be viewed as the normalization conditions. In the case of $\alpha = -\frac{1}{2}$ which realizes the OSp(4|2) superconformal mechanics, the prepotential *F* cannot be normalized. This corresponds to the fact that under the reflection (3.5.11) $\alpha = -\frac{1}{2}$ is self-dual ²². Therefore we can utilize the families of the solutions (*U*, *F*) along with the expression (3.5.104) to construct $\mathcal{N} = 4$ multiparticle superconformal mechanics. Since the number of independent equations are given by [137]

$$\begin{cases} \frac{1}{12}(N-1)(N-2)^2(N-3) & \text{for WDVV equation} \\ \frac{1}{2}(N-1)(N-2) & \text{for twisted periods,} \end{cases}$$
(3.5.113)

when $N \ge 4$, i.e. the system contains more than four particles, the non-trivial WDVV equation (3.5.100) appears and the twisted periods equation (3.5.101) gives rise to the non-trivial conditions.

At this stage we with to look for the solution *F* to the WDVV equation (3.5.100) and the twisted periods U_a defined by (3.5.101). However, up to date it is an open mathematical problem to list up all the solutions to the WDVV equation and only part of the solutions are known [234, 235, 236, 237, 238, 239, 240]. In [234] it was shown that one can construct the solutions to the WDVV equation (3.5.100) by imposing the ansatz

$$F(x) = \sum_{\alpha} f_{\alpha} K(\alpha \cdot x)$$
(3.5.114)

where

$$K(z) = \begin{cases} -\frac{1}{4}z^2 \ln z^2 & \text{rational case} \\ -\frac{1}{4}\text{Li}_3(e^{2iz}) + \frac{1}{6}z^3 & \text{trigonometric case} \\ -\frac{1}{4}\mathcal{L}i_3(e^{2iz}|\tau) & \text{elliptic case} \end{cases}$$
(3.5.115)

with $f_{\alpha} \in \mathbb{R}$ and $\alpha \cdot x = \alpha^a x_a$. Here $\{\alpha\}$ are the covectors constructing a deformed Lie (super)algebra root system ²³. Li₃ is the trilogarithm and $\mathcal{L}i_3$ is an elliptic generalization [242, 243, 244, 245]. Among the above known solutions to the WDVV

²²The induced metric defined in (3.5.117) is degenerate for $\alpha = -\frac{1}{2}$.

 $^{^{23}}$ It is known that the root systems of some Lie superalgebras give rise to the solutions to the WDVV equation [237, 241].

equation, only the rational case satisfies the normalization conditions (3.5.111) and (3.5.112). Thus the $D(2, 1; \alpha)$ superconformal models may arise for

$$F(x) = -\frac{1}{4} \sum_{\alpha} f_{\alpha}(\alpha \cdot x)^2 \ln |\alpha \cdot x|^2$$
(3.5.116)

The ansatz (3.5.116) defines the constant metric

$$g_{ab} = -x^c F_{cab} = \sum_{\alpha} f_{\alpha} \alpha \otimes \alpha.$$
 (3.5.117)

Then it was established [235] that certain deformations of root systems can solve the WDVV equation (3.5.100) and the corresponding collections of covectors { α } is called \lor -systems [246].

On the other hand, it was observed [233] that the ansatz for the twisted periods U_a

$$U(x) = \sum_{\beta} u_{\beta} \ln P_{\beta}(x)$$
 (3.5.118)

can solve the equation (3.5.101) where $P_{\beta}(x)$ are homogeneous polynomials of degree n_{β} in x and u_{β} is chosen so that $\sum_{\beta} n_{\beta} u_{\beta} = 2\alpha$.

Now let us assume that $\alpha \neq -\frac{1}{2}$ and consider the special solutions to the twisted periods as the form

$$U(x) = \sum_{\alpha} u_{\alpha} \ln(\alpha \cdot x)$$
(3.5.119)

where the same covectors α are chosen for U(x) and F(x). Then the normalization conditions (3.5.111) and (3.5.112) reduce to

$$\sum_{\alpha} u_{\alpha} = 2\alpha, \qquad (3.5.120)$$

$$\sum_{\alpha} f_{\alpha} \alpha_a \alpha_b = (1+2\alpha) \delta_{ab}$$
(3.5.121)

and we get the potential term

$$V(x) = \frac{K^{ij}K_{ij}}{8} \sum_{\alpha,\beta} u_{\alpha} u_{\beta} \frac{\alpha \cdot \beta}{(\alpha \cdot x) (\beta \cdot x)}.$$
(3.5.122)

By requiring the invariance under permutations of the particle labels, the WDVV solutions *F* based on deformed root systems of the Lie algebras A_n , BCD_n and EF_n and the Lie superalgebras have been discussed [247]. It is an interesting question to reveal the geometrical understanding for the relevant WDVV solutions and the relation to the construction of the $\mathcal{N} = 4$ superconformal mechanical models.

On the contrary, we cannot apply the same method to the OSp(4|2) superconformal mechanical models for $\alpha = -\frac{1}{2}$ since some formulae become singular. One of the illness is the degenerate induced metric

$$\sum_{\alpha} f_{\alpha} \alpha \otimes \alpha = 0, \qquad (3.5.123)$$

which can be seen from (3.5.117) and (3.5.121). Since this implies the degenerate covectors α , it is natural to consider the degenerate limit of the deformed root systems which solve the WDVV equation. By observing that there exists a degenerate limit in the moduli space of the deformed A_n root systems, the prepotentials for the OSp(4|2) superconformal mechanics have been proposed as [247]

$$F(x) = \frac{1}{4N} \sum_{a < b} (x^a - x^b)^2 \ln(x^a - x^b)^2 - \frac{1}{4N^2} \sum_{a} (Nx^a - X)^2 \ln(Nx^a - X)^2,$$
(3.5.124)

$$U(x) = -\frac{1}{2N} \sum_{a} \ln(Nx^{a} - X)$$
(3.5.125)

where $X = \sum_{a} x^{a}$. Correspondingly we get the potential [247, 137]

$$V(x) = \frac{K^{ij}K_{ij}}{8} \left[\sum_{a} \frac{1}{(Nx^{a} - X)^{2}} - \frac{1}{N} \left(\sum_{a} \frac{1}{Nx^{a} - X} \right)^{2} \right]$$
$$= \frac{K^{ij}K_{ij}}{8N} \sum_{a < b} \left[\frac{1}{Nx^{a} - X} - \frac{1}{Nx^{b} - X} \right].$$
(3.5.126)

We should note that the potential (3.5.126) does not take the form of the Calogero type pairwise interaction albeit it is the inverse-square type interaction.

Sigma-model

We shall study the $\mathcal{N} = 4$ superconformal sigma-model which is more general multi-particle $\mathcal{N} = 4$ superconformal quantum mechanical system ²⁴.

In order to find the condition on the target space geometry, we assume that the second, third and fourth supersymmetry transformations are expressed as [200]

$$\delta \Phi^{i} = \epsilon^{r} (I_{r})^{i}{}_{j} D \Phi^{j} \tag{3.5.127}$$

where Φ^i is the (1,2,1) superfields and ϵ^r , r = 1,2,3 are the supersymmetry parameters and I_r are the endomorphisms of the tangent bundle of the target space.

²⁴Note that in the $\mathcal{N} = 4$ superconformal multi-particle mechanical models relevant to the WDVV equation, the metric is trivial due to the ansatz (3.5.93), (3.5.94).

The corresponding $\mathcal{N} = 4$ supermultiplet is referred to as $\mathcal{N} = 4B$ multiplet. This is related to the two-dimensional $\mathcal{N} = (4,0)$ supersymmetry. Then the $\mathcal{N} = 4$ superalgebra imposes the conditions [199, 200]

$$I_r I_s + I_s I_r = -2\delta_{rs}, (3.5.128)$$

$$N(I_r, I_s) = 0 \tag{3.5.129}$$

where a N(F, G) is Nijenhuis concomitant [248, 249]

$$N(F,G)(X,Y) = [FX,GY] - F[X,GY] - F[GX,Y] + FG[X,Y] + G \leftrightarrow F \quad (3.5.130)$$

where X, Y are vector fields on \mathcal{M} . Thus the target space \mathcal{M} possesses three complex sructures I_r which have vanishing mixed Nijenhuis tensors and obey the Clifford algebra (3.5.128). Furthermore the three complex structures turn out to satisfy the algebra of imaginary unit quaternions

$$I_r I_s = -\delta_{rs} + \epsilon_{rst} I_t \tag{3.5.131}$$

or the $\mathfrak{su}(2)$ R-symmetry algebra

$$[I_r, I_s] = 2\epsilon^{rst} I^t \tag{3.5.132}$$

since one can construct a third complex structures from other two by multiplication.

Also the supersymmetry invariance of the action requires that

$$g_{ij} = (I_r)^k{}_i (I_r)^l{}_j g_{kl}, (3.5.133)$$

$$\nabla_{(i}^{(+)}(I_r)_{\ j}^k = 0, \tag{3.5.134}$$

$$\partial_{[i} \left(I^{m}{}_{j}c_{|m|kl]} \right) - 2(I_{r})^{m}{}_{[i}\partial_{[m}c_{jkl]]} = 0.$$
(3.5.135)

The first condition(3.5.133) implies that the metric g on \mathcal{M} is Hermitian with respect to the three complex structures. The second condition (3.5.134) is a generalized Yano tensor condition with torsion and the third condition (3.5.135) is imposed on torsion and complex structures.

It has been pointed out [200] that the above constraints on the target space \mathcal{M} are similar to the defining conditions for a weak hyperkähler manifold with torsion (HKT) [250]. A weak HKT manifold is a Riemannian manifold { \mathcal{M} , g, c} with a metric g, a torsion three-form c and three complex structures I_r , r = 1, 2, 3 which obey the following conditions ²⁵

²⁵If *c* is closed in addition to (3.5.131), (3.5.133), (3.5.136), it is called a strong HKT[250, 200].

- 1. the three complex structures I_r satisfy the algebra of imaginary of unit quaternions (3.5.131)
- 2. the metric is Hermitian with respect to the three complex structures; (3.5.133)
- 3. the complex structures are covariant constant

$$\nabla_k^{(+)} (I_r)^i_{\ j} = 0 \tag{3.5.136}$$

with respect to the covariant derivative $\nabla^{(+)}$ with the torsion.

We see that the conditions for the weak HKT geometry are only different from the constraints on the target space \mathcal{M} in that the covariant constant properties for the complex structures (3.5.136) are replaced with (3.5.134) and (3.5.135). It turns out that the equation (3.5.136) always solves the constraints (3.5.134) and (3.5.135). Therefore a weak HKT geometry satisfies the constraints (3.5.130)-(3.5.135) on the $\mathcal{N} = 4B$ supersymmetric sigma-models.

Although it is known that the $\mathcal{N} = 4$ supermultiplets in one-dimension hold the connections to the $\mathcal{N} = 2$ supersymmetry in two-dimensions as

$$d \mathcal{N} = 4A \Leftrightarrow 2d \mathcal{N} = (2, 2),$$

$$d \mathcal{N} = 4B \Leftrightarrow 2d \mathcal{N} = (4, 0),$$
 (3.5.137)

we have seen that the target space \mathcal{M} of the $\mathcal{N} = 4B$ sigma-model is not the HKT geometry in two-dimensions, but rather a weak HKT geometry. This shows that there are one-dimensional supermultiplets which cannot be obtained from higher-dimensional supermultiplets.

Furthermore the $D(2, 1; \alpha)$ superconformal algebra (3.5.1)-(3.5.6) imposes the additional conditions [69] ²⁶

$$\mathcal{L}_{D^{r}}(I_{r})^{i}{}_{j} = -\frac{2}{1+\alpha} \epsilon^{rst} (I^{t})^{j}{}_{i}, \qquad \qquad \mathcal{L}_{D^{r}} g_{ij} = 0 \qquad (3.5.138)$$

where $D^r := D^i (I^r)_i{}^j \partial_j$. These conditions (3.5.138) can be viewed as the generalizations of the $\mathcal{N} = 2$ superconformal constraints (3.4.81).

3.5.7 Gauged superconformal mechanics

Consider the $\mathcal{N} = 4$ matrix superfield gauged mechanical action in the harmonic superspace [152]

$$S = S_{\mathcal{X}} + S_{WZ} + S_{FI}$$
 (3.5.139)

²⁶Here the value $\alpha = -1, 0$ are excluded.

where

$$S_{\mathcal{X}} = -\frac{1}{4(1+\alpha)} \int \mu_H \operatorname{Tr}\left(\mathcal{X}^{-\frac{1}{\alpha}}\right), \qquad (3.5.140)$$

$$S_{WZ} = \frac{1}{2} \int \mu_A^{(-2)} \mathcal{V}_0 \tilde{\mathcal{Z}}^+ \mathcal{Z}^+, \qquad (3.5.141)$$

$$S_{\rm FI} = \frac{i}{2}c \int \mu_A^{(-2)} {\rm Tr} V^{++}$$
(3.5.142)

where the integration measures are defined

$$\mu_H = dudt d^4 \theta, \qquad \mu_A^{(-2)} = dud\zeta^{(-2)}$$
(3.5.143)

with harmonic superspace parametrized by the coordinates (3.5.82)-(3.5.85). The superfields are

• the $\mathcal{N} = 4$ Grassmann-even Hermitian $n \times n$ matrix superfield $\mathcal{X}_a^b(t, \theta^{\pm}, \overline{\theta}^{\pm}, u^{\pm})$ which obeys

$$\mathcal{D}^{++}\mathcal{X} = 0, \qquad \mathcal{D}^+\mathcal{D}^-\mathcal{X} = 0, \qquad \left(\mathcal{D}^+\overline{\mathcal{D}}^- + \overline{\mathcal{D}}^+\mathcal{D}^-\right)\mathcal{X} = 0, \qquad (3.5.144)$$

which is the (1, 4, 3) supermultiplet

• the $\mathcal{N} = 4$ Grassmann-even analytic superfield $\mathcal{Z}_a^+(\zeta, u)$ which satisfies

$$D^{++}Z^{+} = 0,$$
 $D^{+}Z^{+} = 0,$ $\overline{D}^{+}Z^{+} = 0,$ (3.5.145)

which is the (4, 4, 0) supermultiplet and \tilde{Z}^+ being its Hermitian conjugation preserving analyticity [226, 214]

- the $\mathcal{N} = 4$ Grassmann-even $n \times n$ matrix gauge superfield $V_a^{++b}(\zeta, u)$
- the unconstrained real analytic superfield $\mathcal{V}_0(\zeta, u)$ defined by

$$\int du \mathcal{V}_0(t_A, \theta^+, \overline{\theta}^+, u^{\pm})|_{\theta^{\pm} = \theta^i u_i^{\pm}, \overline{\theta}^{\pm} = \overline{\theta}^i u_i^{\pm}} = \operatorname{Tr}\left(\mathcal{X}\right)$$
(3.5.146)

where the covariant derivative \mathcal{D}^{++} is given by

$$\mathcal{D}^{++}\mathcal{X} = D^{++}\mathcal{X} + i[V^{++}, \mathcal{X}], \qquad (3.5.147)$$

$$\mathcal{D}^{++}\mathcal{Z} = D^{++}\mathcal{Z}^{+} + iV^{++}\mathcal{Z}^{+}.$$
(3.5.148)

The first term $S_{\mathcal{X}}$ of the action (3.5.139) is the superconformal action (3.5.77) for the (**1**, **4**, **3**) superfield \mathcal{X} . The second term S_{WZ} is the Wess-Zumino (WZ) term describing \mathcal{Z}_a^+ [211]. The third term S_{FI} is the Fayet-Iliopoulos (FI) term for the gauge superfield V^{++} . The superconformal boost transformations are [214, 211]

$$\delta t_A = \alpha^{-1} \Lambda t_A, \quad \delta \theta_A = -\eta^+ t_A + 2i(1+\alpha)\eta^- \theta^+ \overline{\theta}^+, \quad \delta u_i^+ = \Lambda^{++} u_i^-, \quad (3.5.149)$$

$$\delta\mu_H = \mu_H \left(2\Lambda - \frac{1+\alpha}{\alpha}\Lambda_0\right), \qquad \qquad \delta\mu_A^{(-2)} = 0, \qquad (3.5.150)$$

$$\delta \mathcal{X} = -\Lambda_0 \mathcal{X}, \qquad \quad \delta \mathcal{Z}^+ = \Lambda \mathcal{Z}^+, \qquad \quad \delta V^{++} = 0 \qquad (3.5.151)$$

where

$$\lambda = 2i\alpha \left(\overline{\eta}^{-} \theta^{+} - \eta^{-} \overline{\theta}^{+} \right), \qquad (3.5.152)$$

$$\Lambda^{++} = D^{++}\Lambda = 2i\alpha \left(\overline{\eta}^+ \theta^+ - \eta^+ \overline{\theta}^+\right), \qquad (3.5.153)$$

$$\Lambda_0 = 2\Lambda - D^{--}\Lambda^{++}.$$
 (3.5.154)

The action (3.5.139) is invariant under the U(n) transformations [152]

$$\mathcal{X} \to e^{i\Lambda} \mathcal{X} e^{-i\Lambda},$$
 (3.5.155)

$$\mathcal{Z}^+ \to e^{i\Lambda} \mathcal{Z}^+,$$
 (3.5.156)

$$\tilde{\mathcal{Z}}^+ \to e^{i\Lambda} \mathcal{Z}^+,$$
 (3.5.157)

$$V^{++} \to e^{i\Lambda}V^{++}e^{-i\Lambda} - ie^{i\Lambda}\left(D^{++}e^{-i\Lambda}\right)$$
(3.5.158)

where $\Lambda_a^b(\zeta, u^{\pm})$ is the Hermitian analytic matrix gauge parameter. From the gauge freedom (3.5.155)-(3.5.158) let us fix the gauge as

$$V^{++} = -2i\theta^{+}\overline{\theta}^{+}A(t_{A}).$$
(3.5.159)

Integrating out the auxiliary fields by means of their algebraic equations of motion and performing the Grassmann integral, we obtain the $D(2,1;\alpha)$ superconformal mechanics [209]

$$S = \int dt \left[\dot{x}^{2} + \frac{i}{2} \left(\overline{z}_{i} z^{i} - c \right) - i \overline{\psi}_{i} \dot{\psi}^{i} - \dot{\overline{\psi}}_{i} \psi^{i} - \frac{i}{\sqrt{2}} \psi^{i} \psi^{i} \psi^{j} - \frac{i}{\sqrt{2}} \psi^{i} \overline{\psi}_{i} \overline{\psi}_{j} + 2\alpha \frac{\psi^{i} \overline{\psi}^{j} z_{(i} \overline{z}_{j)}}{x^{2}} + \frac{2}{3} (1 + 2\alpha) \frac{\psi^{i} \overline{\psi}^{j} \psi_{(i} \overline{\psi}_{j)}}{x^{2}} - A \left(\overline{z}_{i} z^{i} - c \right) \right]. \quad (3.5.160)$$

Using the Noether's method the set of generators are evaluated to be [209]

$$H = \frac{1}{4}p^{2} + \alpha^{2} \frac{(\overline{z}_{i}z^{i})^{2} + 2\overline{z}_{i}z^{i}}{4x^{2}} - 2\alpha \frac{z^{(i\overline{z}^{j})}\psi_{(i}\psi_{k)}}{x^{2}},$$

- $(1 + 2\alpha)\frac{\psi_{i}\psi^{i}\overline{\psi}^{j}\overline{\psi}_{j}}{2x^{2}} + \frac{(1 + 2\alpha)^{2}}{16x^{2}},$ (3.5.161)

$$D = tH - \frac{1}{4} \{x, p\}, \qquad (3.5.162)$$

$$K = t^{2}H - \frac{1}{2}t\{x, p\} + x^{2}, \qquad (3.5.163)$$

$$Q^{i} = p\psi^{i} + 2i\alpha \frac{z^{(i}\overline{z}^{j)}\psi_{j}}{x} + i(1+2\alpha)\frac{\langle\psi_{j}\psi^{j}\overline{\psi}^{i}\rangle}{x}, \qquad (3.5.164)$$

$$\overline{Q}_i = p\overline{\psi}_i - 2i\alpha \frac{z_{(i}\overline{z}_j)\overline{\psi}^j}{+}i(1+2\alpha)\frac{\langle \overline{\psi}^j\overline{\psi}_j\psi_i\rangle}{x}, \qquad (3.5.165)$$

$$S^i = tQ^i - 2x\psi^i \tag{3.5.166}$$

$$\overline{S}_i = t\overline{Q}_i - 2x\overline{\psi}_i \tag{3.5.167}$$

$$J^{ij} = i \left(z^{(i}\overline{z}^{k)} + 2\psi^{(i}\overline{\psi}^{k)} \right), \qquad (3.5.168)$$

$$I^{11} = -i\psi_i\psi^k, (3.5.169)$$

$$I^{22} = i\overline{\psi}^i\overline{\psi}^i, \qquad (3.5.170)$$

$$I^{12} = -\frac{i}{2} [\psi_i, \overline{\psi}^i]$$
(3.5.171)

.

where $\langle \cdot \rangle$ denotes the Weyl ordering. One can show that under the canonical relations

$$[x, p] = i, \qquad [z^i, \overline{z}_j] = \delta^i_j, \qquad \left\{\psi^i, \overline{\psi}_j\right\} = -\frac{1}{2}\delta^i_j \qquad (3.5.172)$$

the generators form the $D(2, 1; \alpha)$ superalgebra [209].

3.6 $\mathcal{N} = 8$ Superconformal mechanics

Up to now much less has been known about higher extended N > 4 supersymmetric quantum mechanics. A study on N > 4 supersymmetric quantum mechanics was initiated in [220] within the on-shell Hamiltonian approach. As we have discussed in subsection 3.2.1, the N = 8 supersymmetry is the maximum case in which only the same number of supersymmetry is required for the component

fields in the minimal supermultiplet ²⁷. In other words, the $\mathcal{N} = 8$ supersymmetry is the highest supersymmetric case in which the superspace and superfield formalism is applicable. In fact off-shell actions of the $\mathcal{N} = 8$ superconformal mechanical models are only known for a few cases. From the Table 3.2 we see that there are four different possible superconformal group for $\mathcal{N} = 8$ superconformal mechanics ²⁸:

- 1. *SU*(1,1|4)
- 2. *OSp*(8|2)
- 3. $OSp(4^*|4)$
- 4. *F*(4).

As we will see, the $OSp(4^*|4)$ superconformal mechanics has been constructed from the (**3**, **8**, **5**) and the (**5**, **8**, **3**) supermultiplets [252, 253] and *F*(4) superconformal mechanics has been proposed from the (**1**, **8**, **7**) supermultiplet [254].

3.6.1 On-shell $SU(1,1|\frac{N}{2})$ action

It has been discussed [210, 255, 224, 137] that the on-shell one particle component action of the $SU(1,1|\frac{N}{2})$, N > 4 superconformal mechanical models generically take the form

$$S = \int dt \left[\dot{x}^2 + i \left(\overline{\psi}_i \dot{\psi}^i - \dot{\psi}_i \psi^i \right) - \frac{\left(c + \overline{\psi}_i \psi^i \right)^2}{x^2} \right]$$
(3.6.1)

where the fermionic fields ψ^i are the spinor representation of the R-symmetry group $SU(\frac{N}{2})$. It has been pointed out [224] that the generators of the superconformal group $SU(1,1|\frac{N}{2})$ can be found from those of the SU(1,1|2) jus by replacing the SU(2) spinor ψ^i with the $SU(\frac{N}{2})$ spinors and c is a constant parameter. Correspondingly the supercharges Q^i, \overline{Q}_i and the Hamiltonian H can be expressed

²⁷Note that this statement has not been strictly proven without the assumptions for the particular forms of supersymmetric transformations (3.2.2), (3.2.3) and the relevant algebras.

²⁸ The relevant *D*-module representations for the d = 1 $\mathcal{N} = 2$, 4 and 8 superconformal algebras have been discussed in [251].

as

$$Q^{i} = \psi^{i} \left(p - 2i \frac{c + \overline{\psi}_{i} \psi^{i}}{x} \right), \qquad (3.6.2)$$

$$\overline{Q}_i = \overline{\psi}_i \left(p + 2i \frac{c + \psi_i \overline{\psi}^i}{x} \right), \qquad (3.6.3)$$

$$H = \frac{p^2}{4} + \frac{\left[c + \overline{\psi}_i \psi^i\right]^2}{x^2}.$$
 (3.6.4)

However, it has not been completely understood how to realize the on-shell action (3.6.1) from the off-shell superspace and superfield formalism.

3.6.2 Superspace and supermultiplet

The $\mathcal{N} = 8$ superspace $\mathbb{R}^{(1|8)}$ is parametrized by [252, 253]

$$\mathbb{R}^{(1|8)} = (t, \theta_{ia}, \vartheta_{\alpha A}), \qquad \overline{(\theta_{ia})} = \theta^{ia}, \qquad \overline{(\vartheta_{\alpha A})} = \vartheta^{\alpha A} \qquad (3.6.5)$$

with *i*, *a*, α , A = 1, 2. In terms of (3.6.5) four commuting SU(2) factors of the R-symmetry will be manifest. The covariant derivatives are defined by

$$D^{ia} = \frac{\partial}{\partial \theta_{ia}} + i\theta^{ia} \frac{\partial}{\partial t}, \qquad \nabla^{\alpha A} = \frac{\partial}{\partial \theta_{\alpha A}} \qquad (3.6.6)$$

and they satisfy

$$\left\{D^{ia}, D^{jb}\right\} = 2i\epsilon^{ij}\epsilon^{ab}\frac{\partial}{\partial t}, \qquad \left\{\nabla^{\alpha A}, \nabla^{\beta B}\right\} = 2i\epsilon^{\alpha\beta}\epsilon^{AB}\frac{\partial}{\partial t}. \qquad (3.6.7)$$

Although the $\mathcal{N} = 8$ superfields are useful to find the irreducibility constraints and the transformation properties, it is hard to reproduce the supersymmetric action in terms of the component fields because of the large dimension of the integration measure. The efficient strategy is to split the $\mathcal{N} = 8$ supermultiplets into the $\mathcal{N} = 4$ supermultiplets and to deal with the $\mathcal{N} = 4$ superspace and superfield formalism. Such decompositions of the $\mathcal{N} = 8$ supermultiplets in terms of the $\mathcal{N} = 4$ supermultiplets can be written as the direct sum [252, 256]

$$(n, 8, 8 - n) = (n_1, 4, 4 - n_1) \oplus (n_2, 4, 4 - n_2)$$
(3.6.8)

with $n = n_1 + n_2$. Here *n* represents the number of physical bosonic fields in the $\mathcal{N} = 8$ supermultiplets while n_1 and n_2 denote the numbers of physical bosons in the two $\mathcal{N} = 4$ supermultiplets respectively.

(0, 8, 8) supermultiplet

The (**0**, **8**, **8**) supermultiplet is described by two real fermionic superfields Ψ^{aA} , $\Xi^{i\alpha}$ satisfying the constraints

$$\nabla^{(\alpha A}\Xi_i^{\beta)} = 0, \qquad D^{(ia}\Xi_{\alpha}^j = 0, \qquad (3.6.9)$$

$$\nabla^{\alpha(A} \Psi_{a}^{B)} = 0, \qquad D^{i(a} \Psi_{A}^{b)} = 0, \qquad (3.6.10)$$

$$\nabla^{\alpha A} \Psi^a_A = D^{ia} \Xi^{\alpha}_i, \qquad \nabla^{\alpha A} \Xi^i_{\alpha} = -D^{ia} \Psi^A_a. \qquad (3.6.11)$$

(3.6.11) implies that the covariant derivative with respect to $\vartheta_{\alpha A}$ can be represented by the covariant derivatives with respect to θ_{ia} .

The (0, 8, 8) supermultiplet possesses a unique splitting

$$(0, 8, 8) = (0, 4, 4) \oplus (0, 4, 4).$$
 (3.6.12)

In order to describe the (0, 0, 8) supermultiplet in terms of the $\mathcal{N} = 4$ superfields, we pick up the appropriate $\mathcal{N} = 4$ superspace as

$$\mathbb{R}^{(1|4)} = (t, \theta_{ia}) \subset \mathbb{R}^{(1|8)} = (t, \theta_{ia}, \vartheta_{\alpha A}).$$
(3.6.13)

Expanding the superfields in ϑ^{iA} , the constraints (3.6.11) leave the independent $\mathcal{N} = 4$ superfields

$$\psi^{aA} = \Psi^{aA}|_{\vartheta=0}, \qquad \qquad \xi^{i\alpha} = \Xi^{i\alpha}|_{\vartheta=0}. \qquad (3.6.14)$$

Then the constraints (3.6.9) and (3.6.10) imply that

$$D^{a(i}\xi^{j)\alpha} = 0,$$
 $D^{i(a}\psi^{b)A} = 0.$ (3.6.15)

The conditions (3.6.15) correspond to the constraints (3.5.89) for (**0**, **4**, **4**) supermultiplets on the superfields ξ^{ia} , ψ^{aA} .

The $\mathcal{N} = 8$ supersymmetric action can be written as

$$S = \int dt d^4\theta \left[\theta^{ia} \theta^b_i \psi^A_a \psi_{bA} + \theta^{ia} \theta^j_a \xi^\alpha_i \xi_{j\alpha} \right].$$
(3.6.16)

Although the action (3.6.16) is not manifestly invariant due to the existence of the Grassmann coordinates, one can show that it is invariant.

(1, 8, 7) supermultiplet

The (1, 8, 7) supermultiplet is described by a single scalar superfield \mathcal{U} obeying the conditions

$$\nabla^{(\alpha i} \nabla^{\beta) j} \mathcal{U} = 0, \qquad D^{i(a} D^{jb)} \mathcal{U} = 0, \qquad (3.6.17)$$

$$D^{ia}D^{j}_{a}\mathcal{U} = -\nabla^{\alpha j}\nabla^{i}_{\alpha}\mathcal{U}$$
(3.6.18)

The condition (3.6.18) reduce the manifest R-symmetry into three SU(2) factors due to the identification of the indices A and *i* of the covariant derivatives $\nabla^{\alpha A}$ and D^{ia} .

The (1, 8, 7) has a unique decomposition into the $\mathcal{N} = 4$ multiplets as

$$(\mathbf{1}, \mathbf{8}, \mathbf{7}) = (\mathbf{1}, \mathbf{4}, \mathbf{3}) \oplus (\mathbf{0}, \mathbf{4}, \mathbf{4}).$$
 (3.6.19)

By choosing the $\mathcal{N} = 4$ superspace $\mathbb{R}^{(1|4)}$ as in (3.6.13) and expanding the superfields in ϑ^{iA} , we find the projected $\mathcal{N} = 4$ superfields

$$u = \mathcal{U}|_{\vartheta=0}, \qquad \qquad \psi^{i\alpha} = \nabla^{i\alpha}\mathcal{U}|_{\vartheta=0}$$
(3.6.20)

obeying

$$D^{(ia}\psi^{j)\alpha} = 0,$$
 $D^{i(a}D^{jb)}u = 0,$ (3.6.21)

which are viewed as the constraint equations (3.5.89) and (3.5.72). Thus we can identify $\psi^{i\alpha}$ and *u* with the (**0**, **4**, **4**) and (**1**, **4**, **3**) superfields respectively.

The general $\mathcal{N} = 8$ supersymmetric component action of the (1, 8, 7) supermultiplet can be found in [190]. The harmonic superspace action can be found in [257, 258, 226].

Taking into account the decomposition (3.6.19) of the (**1**, **8**, **7**) supermultiplet, $\mathcal{N} = 8$ superconformal mechanical model has been constructed by combining the two supermultiplets (**1**, **4**, **3**) and (**0**, **4**, **4**) for $D(2, 1; \alpha = -\frac{1}{3})$ [254]. Since the possible $\mathcal{N} = 8$ superconformal group into which one can embed $D(2, 1; \alpha = -\frac{1}{3})$ is only F(4), the resulting $\mathcal{N} = 8$ superconformal mechanical model is identified with F(4) superconformal mechanics.

(2, 8, 6) supermultiplet

The (2, 8, 6) supermultiplet contains two scalar bosonic superfields \mathcal{U}, Φ which satisfy

$$\nabla^{(ai}\nabla^{b)j}\mathcal{U} = 0, \qquad \nabla^{a(i}\nabla^{bj)}\Phi = 0, \qquad (3.6.22)$$

$$\nabla^{ai}\mathcal{U} = D^{ia}\Phi, \qquad \nabla^{ai}\Phi = -D^{ia}\mathcal{U} \qquad (3.6.23)$$

where the indices *i*, *A* being identified and the indices *a*, α being identified and thus only two *SU*(2) factors are manifest. The (**2**, **8**, **6**) multiplets can be regarded as the two (**1**, **8**, **7**) multiplets with the additional conditions because the two constraints (3.6.22) and (3.6.23) lead to

$$D^{(ia}D^{j)b}\Phi = 0,$$
 $D^{i(a}D^{jb)}\mathcal{U} = 0,$ (3.6.24)

$$D^{ia}D^{j}_{a}\mathcal{U} = -\nabla^{aj}\nabla^{i}_{a}\mathcal{U}, \qquad D^{ia}D^{b}_{i}\Phi = -\nabla^{bi}\nabla^{a}_{i}\Phi. \qquad (3.6.25)$$

The (2, 8, 6) multiplet has two different decompositions

$$(\mathbf{2},\mathbf{8},\mathbf{6}) = \begin{cases} (\mathbf{1},\mathbf{4},\mathbf{3}) \oplus (\mathbf{1},\mathbf{4},\mathbf{3}) \\ (\mathbf{2},\mathbf{4},\mathbf{2}) \oplus (\mathbf{0},\mathbf{4},\mathbf{4}). \end{cases}$$
(3.6.26)

1. (**1**, **4**, **3**) ⊕ (**1**, **4**, **3**)

Choosing the $\mathcal{N} = 4$ superspace (3.6.13) and expanding the superfields in ϑ^{iA} , we find from (3.6.22) and (3.6.23) the independent $\mathcal{N} = 4$ superfields

$$u = \mathcal{U}|_{\vartheta=0}, \qquad \qquad \phi = \Phi|_{\vartheta=0} \qquad (3.6.27)$$

satisfying

$$D^{i(a}D^{jb)}u = 0,$$
 $D^{(ia}D^{j)b}\phi = 0,$ (3.6.28)

which are the constraints equations (3.5.73). Therefore the two superfields u, ϕ are regarded as the (1,4,3) superfields.

The action can be written as

$$S = \int dt d^4\theta \ F(u,\phi) \tag{3.6.29}$$

where the function *F* satisfies the Laplace equation

$$\frac{\partial^2 F}{\partial u^2} + \frac{\partial^2 F}{\partial \phi^2} = 0. \tag{3.6.30}$$

2. $(2,4,2) \oplus (0,4,4)$

To realize the decomposition $(2, 8, 6) = (2, 4, 2) \oplus (0, 4, 4)$ we need to modify the choice of the $\mathcal{N} = 4$ superspace and the superfields. Let us introduce the covariant derivatives

$$\mathcal{D}^{ia} = \frac{1}{\sqrt{2}} \left(D^{ia} - i \nabla^{ai} \right), \qquad \overline{\mathcal{D}}^{ia} = \frac{1}{\sqrt{2}} \left(D^{ia} + i \nabla^{ai} \right) \qquad (3.6.31)$$

and the superfields $\mathcal{V}, \overline{\mathcal{V}}$ as

$$\mathcal{V} = \mathcal{U} + i\Phi,$$
 $\overline{\mathcal{V}} = \mathcal{U} - i\Phi.$ (3.6.32)

Then we find a set of constraint equations

$$D^i \mathcal{V} = 0,$$
 $\nabla^i \mathcal{V} = 0,$ (3.6.33)

$$\overline{D}_i \overline{\mathcal{V}} = 0, \qquad \qquad \overline{\nabla}_i \overline{\mathcal{V}} = 0, \qquad (3.6.34)$$

$$D^{i}D_{i}\overline{\mathcal{V}} = \overline{\nabla}_{i}\overline{\nabla}^{i}\mathcal{V}, \qquad D^{i}\overline{\nabla}^{j}\overline{\mathcal{V}} = \overline{D}^{i}\overline{\nabla}^{j}\mathcal{V} = 0 \qquad (3.6.35)$$

where we have defined

$$D^i := \mathcal{D}^{i1}, \qquad \qquad \overline{D}^i := \overline{\mathcal{D}}^{i2}, \qquad (3.6.36)$$

$$abla^i := \mathcal{D}^{i2}, \qquad \qquad \overline{\nabla}^i = -\overline{\mathcal{D}}^{i1}. \qquad (3.6.37)$$

Considering a new set of coordinates for the $\mathcal{N} = 4$ superspace as

$$\mathbb{R}^{(1|4)} = (t, \theta_{i1} + i\vartheta_{i1}, \theta_{i2} - i\vartheta_{i2}) \subset \mathbb{R}^{(1|8)}, \qquad (3.6.38)$$

we find from the constraints (3.6.33)-(3.6.35) the independent $\mathcal{N}=4$ superfields

$$v = \mathcal{V}|, \qquad \overline{v} = \overline{\mathcal{V}}|, \qquad (3.6.39)$$

$$\psi^{i} = \overline{\nabla}^{i} \mathcal{V}|, \qquad \qquad \overline{\psi}^{i} = -\nabla^{i} \overline{V}| \qquad (3.6.40)$$

satisfying

$$D^i v = 0, \qquad \overline{D}^i \overline{v} = 0, \qquad (3.6.41)$$

$$D^{i}\psi^{j} = 0, \qquad \overline{D}^{i}\overline{\psi}^{j} = 0, \qquad D^{i}\overline{\psi}^{j} = -\overline{D}^{i}\psi^{j} \qquad (3.6.42)$$

Thus we can identify the two sets of the superfields, v, \overline{v} and $\psi^i, \overline{\psi}^i$ with the (2, 4, 2) and (0, 4, 4) superfields.

The invariant action is given by

$$S = \int dt d^4 \theta v \overline{v} - \frac{1}{2} \int dt d^2 \overline{\theta} \psi^i \psi_i - \frac{1}{2} \int dt d^2 \theta \overline{\psi}_i \overline{\psi}^i.$$
(3.6.43)

We should note that the form of the action (3.6.43) depend on the choice of the $\mathcal{N} = 4$ superspace. Although the superfield action (3.6.43) looks different from the previous action (3.5.92), it turns out to be the same in the component level.

(3, 8, 5) supermultiplet

The (3, 8, 5) supermultiplet includes the three bosonic superfields $V^{ij} = V^{ji}$ obeying

$$D_a^{(i}\mathcal{V}^{jk)} = 0, \qquad \qquad \nabla_{\alpha}^{(i}\mathcal{V}^{jk)} = 0 \qquad (3.6.44)$$

and three SU(2) factors are manifest. (3.6.44) yield to a further condition

$$\partial_t \left(D_i^a D_{ja} V^{ij} + \nabla_i^\alpha \nabla_{j\alpha} V^{ij} \right) = 0, \qquad (3.6.45)$$

which leads to [252]

$$\nabla_i^{\alpha} \nabla_{j\alpha} V^{ij} = 6m - D_i^a D_{ja} V^{ij} \tag{3.6.46}$$

where *m* is a constant parameter.

The (3, 8, 5) multiplet has two decompositions

$$(\mathbf{3},\mathbf{8},\mathbf{5}) = \begin{cases} (\mathbf{3},\mathbf{4},\mathbf{1}) \oplus (\mathbf{0},\mathbf{4},\mathbf{4}) \\ (\mathbf{1},\mathbf{4},\mathbf{3}) \oplus (\mathbf{2},\mathbf{4},\mathbf{2}). \end{cases}$$
(3.6.47)

1. $(3,4,1) \oplus (0,4,4)$

Let us choose the $\mathcal{N} = 4$ superspace (3.6.13) and expand the superfields in $\vartheta_{i\alpha}$. Then the constraints (3.6.44) leave in \mathcal{V}^{ij} the four bosonic and four fermionic $\mathcal{N} = 4$ superfields

$$v^{ij} = \mathcal{V}^{ij}|, \qquad \qquad \xi^i_{\alpha} = \nabla_{j\alpha} \mathcal{V}^{ij}|, \qquad (3.6.48)$$

$$A = \nabla_i^{\alpha} \nabla_{j\alpha} V^{ij} | \tag{3.6.49}$$

which obey

$$D_a^{(i}v^{jk)} = 0,$$
 $D_a^{(i}\xi_{\alpha}^{j)} = 0,$ (3.6.50)

$$A = 6m - D_i^a D_{aj} v^{ij}.$$
 (3.6.51)

Since (3.6.50) are identified with the constraint equations (3.5.52) and (3.5.89), we see that the superfields v^{ij} and ξ^i_{α} are the (3,4,1) and (0,4,4) superfields respectively. The remaining equation (3.6.51) is the conservation law type condition which gives rise to a constant *m*. As observed in [210], this is the reminiscent of the d = 4 $\mathcal{N} = 1$ tensor multiplet constraints [259].

To write down the invariant action let us project out the $\mathcal{N} = 4$ superfields $v^{(ij)}$ and ξ^i_{α} onto the harmonic superspace as

$$v^{++} = v^{ij}u_i^+u_j^+, \quad v^{+-} = \frac{1}{2}D^{--}v^{++}, \quad v^{--} = D^{--}v^{+-}, \quad (3.6.52)$$

$$\xi^+ = \xi^i u_i^+, \qquad \overline{\xi}^+ = \overline{\xi}^i u_i^+. \qquad (3.6.53)$$

Then the $OSp(4^*|4)$ superconformal action is given by [252]

$$S = \int dt d^{4}\theta \sqrt{v^{2}}$$

- $\frac{1}{\sqrt{2}} \int du d\zeta^{--} \left[\frac{\zeta^{+} \overline{\zeta}^{+}}{(1 + c^{--} \widehat{v}^{++})^{\frac{3}{2}}} + 12m \frac{\widehat{v}^{++}}{\sqrt{1 + c^{--} \widehat{v}^{++}}(1 + \sqrt{1 + c^{--} \widehat{v}^{++}})} \right]$
(3.6.54)

where

$$dud\zeta^{--} = dudt_A d\theta^+ d\overline{\theta}^+, \qquad (3.6.55)$$

$$c^{\pm}c^{\pm} = c^{ik}u_i^{\pm}u_k^{\pm}, \qquad c^{ik} = \text{const.}, \qquad (3.6.56)$$

$$v^{++} = \hat{v}^{++} + c^{++}. \tag{3.6.57}$$

2. $(1,4,3) \oplus (2,4,2)$

To obtain the decomposition $(3, 8, 5) = (1, 4, 3) \oplus (2, 4, 2)$, we shall introduce the new covariant derivatives

$$D^{a} = \frac{1}{\sqrt{2}} \left(D^{1a} + i\nabla^{a1} \right), \qquad \overline{D}_{a} = \sqrt{1}\sqrt{2} \left(D_{a}^{2} - i\nabla_{a}^{2} \right), \qquad (3.6.58)$$

$$\nabla^{a} = \frac{i}{\sqrt{2}} \left(D^{2a} + i \nabla^{a2} \right), \qquad \overline{\nabla}_{a} = \frac{i}{\sqrt{2}} \left(D^{1}_{a} - i \nabla^{1}_{a} \right) \qquad (3.6.59)$$

and the set of coordinates closed under the action of D^a , \overline{D}_a

$$\mathbb{R}^{(1|4)} = (t, \theta_{1a} - i\vartheta_{a1}, \theta^{1a} + i\vartheta^{a1}) \subset \mathbb{R}^{(1|8)}.$$
(3.6.60)

Defining the $\mathcal{N} = 4$ superfields as

$$v = -2i\mathcal{V}^{12}, \qquad \varphi = \mathcal{V}^{11}, \qquad \overline{\varphi} = \mathcal{V}^{22}, \qquad (3.6.61)$$

we find the constraints (3.5.73) for the (1,4,3) supermultiplet and the constraints (3.5.69) for the (2,4,2) chiral supermultiplet

$$D^a D_a v = 0, \qquad \qquad \overline{D}_a \overline{D}^a v = 0, \qquad (3.6.62)$$

$$D^a \varphi = 0, \qquad \qquad \overline{D}_a \overline{\varphi}.$$
 (3.6.63)

from the constraints (3.6.44). Therefore the superfields v can be viewed as the (1,4,3) superfield and φ as the (2,4,2) superfield. From the constraints (3.6.62) and (3.6.63) it follows that

$$\frac{\partial}{\partial t} \left[D^a, \overline{D}_a \right] v = 0. \tag{3.6.64}$$

Combining (3.6.46) and (3.6.64), we obtain the constant m [210]

$$[D^a, \overline{D}_a]v = -2m. \tag{3.6.65}$$

In this case the $\mathcal{N} = 8$ supersymmetric free action takes the form [252]

$$S = -\frac{1}{4} \int dt d^4 \theta \left(v^2 - 2\varphi \overline{\varphi} \right).$$
(3.6.66)

However, the action (3.6.66) is not invariant under the superconformal transformations. Following the strategy of [260, 213], the $OSp(4^*|4)$ superconformal action is given by [252]

$$S = -\frac{1}{4} \int dt d^4 \theta \left[v \ln \left(v + \sqrt{v^2 + \varphi \overline{\varphi}} \right) - \sqrt{v^2 + 4\varphi \overline{\varphi}} \right]$$
(3.6.67)

whose bosonic part is

$$S_{\text{bosonic}} = \int dt \frac{1}{\sqrt{v^2 + 4\phi\overline{\phi}}} \left[\dot{v}^2 + 4\dot{\phi}\overline{\phi} - m^2 - 2im\dot{v} - \frac{4im\phi\overline{\phi}}{v + \sqrt{v^2 + \phi\overline{\phi}}} \right].$$
(3.6.68)

Therefore the (3, 8, 5) supermultiplet can describe the $OSp(4^*|4)$ superconformal mechanics [252]. By means of the non-linear realization method parametrize a coset of the supergroup $OSp(4^*|4)$ such that $SO(5) \subset OSp(4^*|4)$ belongs to the stability subgroup while one out of three Goldstone bosons is the coset parameter associated with the dilatation, the dilaton and the remaining two Goldstone bosons parametrize the R-symmetry coset $SU(2)_R/U(1)_R$. Although the action (3.6.54) and (3.6.68) have different manifest $\mathcal{N} = 4$ superconformal symmetries $OSp(4^*|2)$ and SU(1,1|2) respectively, both of them form $OSp(4^*|4)$ superconformal group together with the hidden symmetries. Hence the two superfield actions (3.6.54) and (3.6.68) exhibit different symmetry aspects of the same $\mathcal{N} = 8$ superconformal mechanics. Note that the two actions (3.6.54) and (3.6.68) produce the same actions (3.6.68) in terms of the component fields as they can be obtained from the single $\mathcal{N} = 8$ superfield formulation.

(4, 8, 4) supermultiplet

The (4, 8, 4) supermultiplet includes a four superfields $Q^{a\alpha}$ which obeys

$$D_i^{(a}\mathcal{Q}^{b)\alpha} = 0, \qquad \qquad \nabla_i^{(\alpha}\mathcal{Q}_a^{\beta)} = 0. \qquad (3.6.69)$$

The constraints (3.6.69) are manifestly covariant with respect to the three SU(2) factors for the indices *i*, *a* and *a*.

There are three different decompositions of the (4, 8, 4) supermultiplet

$$(4,8,4) = \begin{cases} (4,4,0) \oplus (0,4,4) \\ (3,4,1) \oplus (1,4,3) \\ (2,4,2) \oplus (2,4,2). \end{cases}$$
(3.6.70)

1. $(4,4,0) \oplus (0,4,4)$

Making the choice of the $\mathcal{N} = 4$ superspace (3.6.13) and expanding the superfields in $\vartheta_{i\alpha}$, the constraints (3.6.69) yield the independent $\mathcal{N} = 4$ superfields

$$q^{a\alpha} = \mathcal{Q}^{a\alpha}|, \qquad \qquad \psi^{ia} = \nabla^i_{\alpha} \mathcal{Q}^{a\alpha}| \qquad (3.6.71)$$

satisfying the constraint conditions (3.5.44) for the (4, 4, 0) supermultiplet and (3.5.90) for the (0, 4, 4) supermultiplet

$$D^{i(a}q^{b)\alpha} = 0,$$
 $D^{i(a}\psi^{b)i} = 0.$ (3.6.72)

Thus the superfields $q^{i\alpha}$ and ψ^{ia} are the (4, 4, 0) and (0, 4, 4) superfields respectively.

2. (**3**, **4**, **1**) ⊕ (**1**, **4**, **3**)

Let us introduce the $\mathcal{N} = 8$ superfields \mathcal{V}^{ab} , \mathcal{V} as

$$Q^{a\alpha} = \delta^{\alpha}_{b} \mathcal{V}^{ab} - \epsilon^{a\alpha} \mathcal{V}, \qquad \qquad \mathcal{V}^{ab} = \mathcal{V}^{ba} \qquad (3.6.73)$$

and pick up the $\mathcal{N} = 4$ superspace

$$\mathbb{R}^{(1|4)} = (t, \theta_{1a} + i\vartheta_{1a}, \theta_{2a} - i\vartheta_{2a}) \subset \mathbb{R}^{(1|8)}.$$
(3.6.74)

Correspondingly we will consider the covariant derivatives D^a , \overline{D}^a and $\overline{\nabla}^a$, ∇^a as

$$(D^a, \overline{D}^a) = \left(\mathcal{D}^{1a}, \overline{\mathcal{D}}^{2a}\right), \qquad (\overline{\nabla}^a, \nabla^a) = \left(\mathcal{D}^{2a}, \overline{\mathcal{D}}^{1a}\right) \qquad (3.6.75)$$

where \mathcal{D}^{ia} , $\overline{\mathcal{D}}^{ia}$ are defined in (3.6.31). Then the constraints (3.6.69) lead to the independent $\mathcal{N} = 4$ superfields

$$v^{ab} = \mathcal{V}^{ab}, \qquad v = \mathcal{V}$$
 (3.6.76)

which are subjected to

$$D^{(a}v^{bc)} = 0,$$
 $\overline{D}^{(a}v^{bc)} = 0,$ (3.6.77)

$$D^{(a}\overline{D}^{b)}v = 0. (3.6.78)$$

Thus the superfields v^{ab} and v are the (3, 4, 1) and (1, 4, 3) superfields respectively.

The supersymmetric invariant free action is given by

$$S = \int dt d^4\theta \left[v^2 - \frac{3}{8} v^{ab} v_{ab} \right].$$
(3.6.79)

3. $(2,4,2) \oplus (2,4,2)$

We shall define the new set of $\mathcal{N} = 8$ superfields \mathcal{W} , Φ in terms of \mathcal{V} , \mathcal{V}^{ab} introduced in (3.6.73) as

$$\mathcal{W} = \mathcal{V}^{11}, \qquad \qquad \overline{\mathcal{W}} = \mathcal{V}^{22}, \qquad (3.6.8o)$$

$$\Phi = \frac{2}{3} \left(\mathcal{V} + \frac{3}{2} \mathcal{V}^{12} \right), \qquad \overline{\Phi} = \frac{2}{3} \left(\mathcal{V} - \frac{3}{2} \mathcal{V}^{12} \right) \qquad (3.6.81)$$

and the new set of the $\mathcal{N} = 4$ covariant derivatives D^i , ∇^i as

$$D^{i} = \frac{1}{\sqrt{2}} \left(\mathcal{D}^{i1} + \overline{\mathcal{D}}^{i1} \right), \qquad \overline{D}^{i} = \frac{1}{\sqrt{2}} \left(\mathcal{D}^{i2} + \overline{\mathcal{D}}^{i2} \right), \qquad (3.6.82)$$

$$\nabla^{i} = \frac{1}{\sqrt{2}} \left(\mathcal{D}^{i1} - \overline{\mathcal{D}}^{i1} \right), \qquad \overline{\nabla}^{i} = -\frac{1}{\sqrt{2}} \left(\mathcal{D}^{i2} - \overline{\mathcal{D}}^{i2} \right) \qquad (3.6.83)$$

where \mathcal{D}^{ia} , $\overline{\mathcal{D}}^{ia}$ are introduced in (3.6.31). Then the constraints (3.6.69) provides us with the two independent (**2**, **4**, **2**) superfields

$$w = \mathcal{W}|, \qquad \phi = \Phi|.$$
 (3.6.84)

The free supersymmetric action can be written as

$$S = \int dt d^4\theta \left[w \overline{w} - \phi \overline{\phi} \right].$$
 (3.6.85)

The (4, 8, 4) supermultiplet can be constructed by reducing two-dimensional $\mathcal{N} = (4, 4)$ or heterotic $\mathcal{N} = (8, 0)$ sigma model [261].

(5, 8, 3) supermultiplet

The (5, 8, 3) supermultiplet is described by the five bosonic superfields $V_{\alpha a}$, U which satisfy

$$D^{ib}\mathcal{V}_{\alpha a} = -\delta^b_a \nabla^i_{\alpha} \mathcal{U}, \qquad \nabla^{\beta i}\mathcal{V}_{\alpha a} = -\delta^{\beta}_{\alpha} D^i_{\alpha} \mathcal{U}. \qquad (3.6.86)$$

The constraints (3.6.86) are covariant not only with respect to three SU(2) factors for the indices *i*, *a*, α but also with respect to the SO(5) R-symmetry. The SO(5)R-symmetry transformations mix the spinor derivatives

The (5, 8, 3) supermultiplet may have two decompositions

$$(5,8,3) = \begin{cases} (1,4,3) \oplus (4,4,0) \\ (3,4,1) \oplus (2,4,2) \end{cases}$$
(3.6.87)

1. (**1**, **4**, **3**) ⊕ (**4**, **4**, **0**)

Using the $\mathcal{N} = 4$ superspace (3.6.13) and carrying out the expansion of the superfields in $\vartheta_{i\alpha}$, we find the independent $\mathcal{N} = 4$ superfields

$$v_{\alpha a} = \mathcal{V}_{\alpha a}|, \qquad \qquad u = \mathcal{U}| \qquad (3.6.88)$$

which satisfy

$$D^{i(a}v^{b)\alpha} = 0,$$
 $D^{i(a}D^{b)}_{i}u = 0.$ (3.6.89)

Hence we obtain the (4, 4, 0) superfield $v_{a\alpha}$ and the (1, 4, 3) superfield u.

2. $(3,4,1) \oplus (2,4,2)$

In order to present the decomposition $(5, 8, 3) = (2, 4, 2) \oplus (2, 4, 2)$, we introduce the new set of superfields W, \overline{W} and $W^{\alpha\beta}$ as

$$\mathcal{W}^{\alpha\beta} = \frac{1}{2} \left(\mathcal{V}^{\alpha\beta} + \mathcal{V}^{\beta\alpha} \right), \qquad \qquad \mathcal{W} = -\epsilon_{\alpha a} \mathcal{V}^{\alpha a} + i\mathcal{U}$$
(3.6.90)

and the new $\mathcal{N} = 4$ superspace

$$\mathbb{R}^{(1|4)} = (t, \theta_{i\alpha} + \vartheta_{\alpha i}, \theta^{i\alpha} - i\vartheta^{\alpha i}) \subset \mathbb{R}^{(1|8)}.$$
(3.6.91)

Then the constraints (3.6.86) leave us with the independent $\mathcal{N} = 4$ superfields

$$\phi = \mathcal{W}, \qquad e^{\alpha\beta} = \mathcal{W}^{\alpha\beta} \qquad (3.6.92)$$

which obey

$$\mathcal{D}^{\alpha}\overline{\phi} = 0,$$
 $\overline{\mathcal{D}}_{\alpha}\phi = 0,$ (3.6.93)

$$\mathcal{D}^{(\alpha}w^{\beta\gamma)} = 0, \qquad \qquad \overline{\mathcal{D}}^{(\alpha}w^{\beta\gamma)} = 0. \qquad (3.6.94)$$

Here the $\mathcal{N} = 4$ covariant derivatives \mathcal{D}^{α} , $\overline{\mathcal{D}}_{\alpha}$ are defined as

$$\mathcal{D}^{\alpha} = \mathcal{D}^{1\alpha} \qquad \qquad \overline{\mathcal{D}}_{\alpha} = \overline{\mathcal{D}}_{1\alpha} \qquad (3.6.95)$$

in terms of the covariant derivatives introduced in (3.6.31). Therefore the $\mathcal{N} = 4$ superfields ϕ and $w^{\alpha\beta}$ are the (2,4,2) superfield and the (3,4,1) superfield respectively.

The $\mathcal{N} = 4$ supersymmetric free action is given by [252]

$$S = \int dt d^4\theta \left[w^2 - \frac{3}{4}\phi\overline{\phi} \right].$$
 (3.6.96)

The $OSp(4^*|4)$ superconformal action can be written as [252]

$$S = 2 \int dt d^4\theta \frac{\ln\left(\sqrt{w^2} + \sqrt{w^2 + \frac{1}{2}\phi\overline{\phi}}\right)}{\sqrt{w^2}}$$
(3.6.97)

whose bosonic part has the form

$$S_{\text{bosonic}} = \int dt \frac{\dot{w}_{\alpha} \dot{w}_{\alpha\beta} + \frac{1}{2} \dot{\phi} \overline{\phi}}{\left(w^2 + \frac{1}{2} \phi \overline{\phi}\right)^{\frac{3}{2}}}.$$
(3.6.98)

. .

The action (3.6.98) can be regarded as a conformal invariant type of the SO(5) invariant sigma-model action of [262].

The (5, 8, 3) supermultiplet can be obtained by the dimensional reduction of the d = 4 $\mathcal{N} = 2$ Abelian multiplet [262]. The three extra physical scalar fields originate from the spatial component fields of the d = 4 gauge vector potential.

Using the non-linear realization technique, it has been shown [252] that the (5, 8, 3) supermultiplet can parametrize a coset of $OSp(4^*|4)$ such that the four out of five Goldstone bosons parametrize the SO(5)/SO(4) coset while the remaining one Goldstone boson is the dilaton.

(6, 8, 2) supermultiplet

The (6, 8, 2) supermultiplet has two tensor superfields $\mathcal{V}^{ij}, \mathcal{W}^{ab}$ subjected to the conditions

$$D_a^{(i} \mathcal{V}^{jk)} = 0,$$
 $\nabla_a^{(i} \mathcal{V}^{jk)} = 0,$ (3.6.99)

$$D_i^{(a} \mathcal{W}^{bc)} = 0,$$
 $\nabla_i^{(a} \mathcal{W}^{bc)} = 0,$ (3.6.100)

$$D_j^a \mathcal{V}^{ij} = \nabla^{bi} \mathcal{W}_b^a, \qquad \nabla_j^a \mathcal{V}^{ij} = -D_b^i \mathcal{W}^{ab}. \qquad (3.6.101)$$

The conditions (3.6.101) identify the eight fermions in \mathcal{V}^{ij} with those in \mathcal{W}^{ab} and also reduce the number of the auxiliary fields to two.

The (6, 8, 2) supermultiplet can be decomposed as

$$(6,8,2) = \begin{cases} (3,4,1) \oplus (3,4,1) \\ (4,4,0) \oplus (2,4,2). \end{cases}$$
(3.6.102)

1. $(3,4,1) \oplus (3,4,1)$

Using the $\mathcal{N} = 4$ superspace (3.6.13) and expanding the superfields in ϑ , we can project out the $\mathcal{N} = 4$ superfields

$$v^{ij} = \mathcal{V}^{ij}, \qquad \qquad w^{ab} = \mathcal{W}^{ab} \tag{3.6.103}$$

obeying

$$D^{a(i}v^{jk)} = 0,$$
 $D^{i(a}w^{bc)} = 0.$ (3.6.104)

Thus we obtain the two (3, 4, 1) superfields v^{ij} and w^{ab} .

The supersymmetric free action reads

$$S = \int dt d^4\theta \left(v^2 - w^2 \right). \tag{3.6.105}$$

2. (**4**, **4**, **0**) ⊕ (**2**, **4**, **2**)

This decomposition can be realized by combining the (**2**, **4**, **2**) chiral multiplet $\phi, \overline{\phi}$ and the (**4**, **4**, **0**) hypermultiplet q^{ia} .

The invariant free action takes the form

$$S = \int dt d^4\theta \left(q^2 - 4\phi\overline{\phi}\right). \tag{3.6.106}$$

(7, 8, 1) supermultiplet

The (7,8,1) supermultiplet contains two different types of superfields \mathcal{V}^{ij} , $\mathcal{Q}^{a\alpha}$ which obey

$$D^{(ia}\mathcal{V}^{jk)} = 0, \qquad \nabla^{\alpha(i}\mathcal{V}^{jk)} = 0, \qquad (3.6.107)$$

$$D^{i(a}\mathcal{Q}^{\alpha b)} = 0, \qquad \nabla^{(\alpha}_{i}\mathcal{Q}^{\beta)}_{a} = 0, \qquad (3.6.108)$$

$$D_j^a \mathcal{V}^{ij} = i \nabla_{\alpha}^i \mathcal{Q}^{a\alpha}, \qquad \nabla_j^{\alpha} \mathcal{V}^{ij} = -i D_a^i \mathcal{Q}^{a\alpha}. \qquad (3.6.109)$$

The constraints (3.6.107) extract the (**3**, **8**, **5**) and (**4**, **8**, **4**) supermultiplets from the superfields \mathcal{V}^{ij} and $\mathcal{Q}^{a\alpha}$ respectively. The constraints (3.6.108) identify the fermions in the superfields \mathcal{V}^{ij} and $\mathcal{Q}^{a\alpha}$ and reduce the number of the auxiliary fields to one.

The (7, 8, 1) supermultiplet has a unique splitting

$$(7, 8, 1) = (3, 4, 1) \oplus (4, 4, 0).$$
 (3.6.110)

By using the $\mathcal{N} = 4$ superspace (3.6.13) and expanding the superspace in ϑ , we find the independent $\mathcal{N} = 4$ superfields

$$v^{ij} = \mathcal{V}^{ij}|, \qquad \qquad q^{a\alpha} = \mathcal{Q}^{a\alpha}| \qquad (3.6.111)$$

which satisfy the constraints

$$D^{a(i}v^{jk)} = 0,$$
 $D^{i(a}q^{b)\alpha} = 0.$ (3.6.112)

We thus obtain the (3, 4, 1) superfield v^{ij} and the (4, 4, 0) superfield $q^{a\alpha}$.

The invariant free action is given by [253]

$$S = \int dt d^4\theta \left[v^2 - \frac{4}{3}q^2 \right].$$
 (3.6.113)

(8, 8, 0) supermultiplet

The (8, 8, 0) supermultiplet possesses two real bosonic superfields Q^{aA} , $\Phi^{i\alpha}$ which obey

$$D^{(ia}\Phi^{j)\alpha} = 0,$$
 $\nabla^{(\alpha A}\Phi^{\beta)}_{i} = 0,$ (3.6.114)

$$D^{i(a}\mathcal{Q}^{b)A} = 0, \qquad \nabla^{\alpha(A}\mathcal{Q}^{aB)} = 0, \qquad (3.6.115)$$

$$\nabla^{\alpha A} \Phi^{i}_{\alpha} = D^{ia} \mathcal{Q}^{A}_{a}, \qquad \nabla^{\alpha A} \mathcal{Q}^{a}_{A} = -D^{ia} \Phi^{\alpha}_{i}. \qquad (3.6.116)$$

Similar to the (**0**, **8**, **8**) supermultiplet, the two conditions (3.6.115) and (3.6.116) means that the covariant derivatives with respect to $\vartheta_{\alpha A}$ can be written in terms of the covariant derivatives with respect to θ_{ia} .

The (0, 8, 8) supermultiplet has a unique decomposition

$$(\mathbf{8},\mathbf{8},\mathbf{0}) = (\mathbf{4},\mathbf{4},\mathbf{0}) \oplus (\mathbf{4},\mathbf{4},\mathbf{0}).$$
 (3.6.117)

Choosing the $\mathcal{N} = 4$ superspace (3.6.13) and expanding the superfields in ϑ , one find the independent $\mathcal{N} = 4$ superfields

$$q^{aA} = \mathcal{Q}^{aA}|, \qquad \qquad \phi^{i\alpha} = \Phi^{i\alpha}| \qquad (3.6.118)$$

satisfying the constraints for (4, 4, 0) supermultiplet

$$D^{a(i}\phi^{j)\alpha} = 0,$$
 $D^{i(a}q^{b)A} = 0.$ (3.6.119)

This implies that the (8, 8, 0) supermultiplet can be decomposed as the sum of the two (4, 4, 0) supermultiplets as in (3.6.117).

The invariant free action can be written as [253]

$$S = \int dt d^4\theta \left[q^2 - \phi^2 \right]. \tag{3.6.120}$$

3.6.3 Multi-particle model

Let us consider the $\mathcal{N} = 8$ supersymmetric sigma-model ²⁹.

Suppose we have the extended supersymmetry transformations as the form

$$\delta \Phi^i = \epsilon^A (I_A)^i{}_j D \Phi^j \tag{3.6.121}$$

where Φ^i is the (1,2,1) superfields and ϵ^A , $A = 1, \dots, 7$ are the supersymmetry parameters and I_A are the endomorphism of the tangent bundle of the target space. This $\mathcal{N} = 8$ supermultiplet is called $\mathcal{N} = 8B$ multiplet. This is related to the twodimensional $\mathcal{N} = (4,0)$ supersymmetry. The closure of the $\mathcal{N} = 8$ superalgebra requires that [200]

$$I_A I_B + I_B I_A = -2\delta_{AB}, (3.6.122)$$

$$N(I_A, I_B) = 0 (3.6.123)$$

where a N(F,G) is Nijenhuis concomitant defined in (3.5.130). Thus the target space \mathcal{M} has seven complex structures I_r which have vanishing mixed Nijenhuis tensors and the underlying algebraic structure is associated with that of octonions.

The invariance of the action under the $\mathcal{N} = 8B$ supersymmetry leads to

$$g_{ij} = (I_A)^k{}_i (I_A)^l{}_j g_{kl}, (3.6.124)$$

$$\nabla_{(i}^{(+)}(I_A)^k_{\ j} = 0, \tag{3.6.125}$$

$$\partial_{[i} \left(I^{m}{}_{j} c_{|m|kl]} \right) - 2 (I_{A})^{m}{}_{[i} \partial_{[m} c_{jkl]]} = 0.$$
(3.6.126)

The first condition(3.5.133) implies that the metric g on \mathcal{M} is Hermitian with respect to the seven complex structures. The second condition (3.5.134) is a generalized Yano tensor condition with torsion and the third condition (3.5.135) is imposed on torsion and complex structures.

The Riemannian manifold $\{\mathcal{M}, g, c\}$ with a metric g, a torsion three-form c and three complex structures I_A , $A = 1, \dots, 7$ which obey the conditions (3.6.122)-(3.6.126) is called Octonionic Kähler with torsion manifold (OKT) [200].

²⁹The $\mathcal{N} = 8$ superconformal sigma-model has not been well understood. We will only discuss the $\mathcal{N} = 8$ supersymmetric sigma-model in this thesis.

Part II

M2-branes

Chapter 4

BLG-model

The dominant theme of this chapter and the next chapter is the world-volume theories of the multiple planar M2-branes ¹. We will begin in this chapter with the BLG-model [21, 22, 23, 24, 25], which is one of the candidate descriptions of the low-energy dynamics of the multiple planar M2-branes. In section 4.1 we will set our notations and conventions and review the basic properties. In section 4.2 we will focus on the study of the A_4 BLG-model that is the non-trivial finite dimensional Lie 3-algebra with positive definite metric, which may describe two membranes.

4.1 Construction

The BLG-model is a three-dimensional $\mathcal{N} = 8$ supersymmetric Chern-Simons matter theory found by Bagger, Lambert [21, 22, 23] and Gustavsson [24, 25]. It is characterized by a Lie 3-algebra \mathcal{A} , which is a generalization of a Lie algebra. The action has a manifest $\mathcal{N} = 8$ supersymmetry and the $SO(8)_R$ R-symmetry. It has been shown [264] that the SO(4) BLG theory has an OSp(4|8) superconformal symmetry at the classical level.

The field content is

- 8 real scalar fields $X^I = X_a^I T^a$
- 16 (8 on-shell) real fermionic fields $\Psi_{\dot{A}} = \Psi_{\dot{A}a}T^a$
- gauge fields $A_{\mu} = A_{\mu ab} T^{ab}$.

¹See [263] for the excellent review on the world-volume theories of the multiple planar M2branes.

Here T^a , $a = 1, \dots, \dim A$ is a basis of the Lie 3-algebra A and T^{ab} is the fundamental object in A which will be introduced in (4.1.18). Under the $SO(8)_R$ R-symmetry the bosonic scalar fields X^I , $I = 1, \dots, 8$ are the vector representations $\mathbf{8}_v$ while the fermionic fields $\Psi_{\dot{A}}$, $\dot{A} = 1, \dots, 8$ are the conjugate spinor representations $\mathbf{8}_c$ respectively.

They also carry the $(\dim A)$ -dimensional representations of the Lie 3-algebra. The gauge fields $A_{\mu ab}$ are 3-algebra A valued world-volume vector fields. They are antisymmetric under two indices a, b of the Lie 3-algebra; $A_{\mu ab} = -A_{\mu ba}^2$. The mass dimensions of the field content and the supersymmetry parameter ϵ are given by

$$[X_a^I] = \frac{1}{2}, \quad [\Psi_a] = 1, \quad [A_\mu] = 1, \quad [\epsilon] = -\frac{1}{2}$$
 (4.1.1)

 Ψ_{Aa} is defined as an SO(1, 10) Majorana fermion and its conjugate is given by

$$\overline{\Psi} := \Psi^T \mathcal{C}, \tag{4.1.2}$$

where C is a SO(1, 10) charge conjugation matrix satisfying

$$\mathcal{C}^{T} = -\mathcal{C}, \qquad \qquad \mathcal{C}\Gamma^{M}\mathcal{C}^{-1} = -(\Gamma^{M})^{T}. \qquad (4.1.3)$$

Gamma matrix Γ^M is the representation of eleven-dimensional Clifford algebra

$$\{\Gamma^M, \Gamma^N\} = 2g^{MN} \tag{4.1.4}$$

$$\Gamma^{10} := \Gamma^{0\dots9}, \tag{4.1.5}$$

where $g^{MN} = \eta^{MN} = \text{diag}(-1, +1, +1, \cdots, +1)$. Γ^M can be decomposed as

$$\begin{cases} \Gamma^{\mu} = \gamma^{\mu} \otimes \tilde{\Gamma}^{9} & \mu = 0, 1, 2\\ \Gamma^{I} = \mathbb{I}_{2} \otimes \tilde{\Gamma}^{I-2} & I = 3, \cdots, 10 \end{cases}$$
(4.1.6)

where

$$\gamma^{0} = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} = i\sigma_{2}, \quad \gamma^{1} = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} = \sigma_{1}, \quad \gamma^{2} = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} = \sigma_{3} \quad (4.1.7)$$

and $\tilde{\Gamma}^{I}$ is an SO(8) 16 × 16 gamma matrix whose chirality matrix is $\tilde{\Gamma}^{9} := \tilde{\Gamma}^{1...8}$. The fermionic field Ψ is a real $\frac{1}{2} \cdot 2^{\left[\frac{11}{2}\right]} = 32$ -component Majorana spinor of elevendimensional space-time, obeying the chirality condition³

$$\Gamma^{012}\Psi = -\Psi.$$
 (4.1.8)

²For the \mathcal{A}_4 algebra we have a one-to-one correspondence between the fundamental object \mathcal{T} and the element T^{ab} of the associated Lie algebra $\mathfrak{so}(4)$. Hence $A_{\mu ab}$ is Lie $\mathfrak{so}(4)$ -valued. Moreover matter fields $X_a^I, \Psi_{\dot{A}a}$ are interpreted as the fundamental representations **4** of $\mathfrak{so}(4)$.

³ 32 supercharges in M-theory is broken to 16 due to the existence of M2-branes and Ψ is identified with the Goldstino corresponding to the broken supersymmetry. Therefore the chirality condition on Ψ is opposite to that of supersymmetry parameters ϵ .
Although at this stage Ψ contains 16 independent real components, the number is reduced to 8 when we treat it on-shell. From (4.1.6) it follows that

$$\Gamma^{012} = \Gamma^{34\dots10} = \mathbb{I}_2 \otimes \tilde{\Gamma}^9 \tag{4.1.9}$$

and

$$\Gamma^{34\dots10}\Psi = -\Psi.$$
 (4.1.10)

Thus Ψ is the conjugate spinor representation $\mathbf{8}_c$ of the $SO(8)_R$ R-symmetry group.

4.1.1 Lie 3-algebra

The construction of the BLG model is based on the Lie 3-algebra A. The Lie 3-algebra is an *N*-dimensional vector space endowed with the totally antisymmetric multi-linear triple product [A, B, C] satisfying the fundamental identity

$$[A, B[C, D, E]] = [[A, B, C], D, E] + [C, [A, B, D], E] + [C, D, [A, B, E]],$$
(4.1.11)

which is a generalization of the Jacobi identity in Lie algebra and requires that the gauge symmetry $\delta_{AB}X = [A, B, X]$ acts as the derivation⁴

$$\delta_{AB}([C, D, E]) = [\delta_{AB}C, D, E] + [C, \delta_{AB}D, E] + [C, D, \delta_{AB}E].$$
(4.1.12)

The supersymmetry algebra of the BLG model is closed on-shell when the fundamental identity (4.1.11) is satisfied [22]. Let us introduce the basis $\{T^a\}_{1 \le a \le N}$ of 3-algebra. Then the 3-algebra is specified by the metric h^{ab} and the structure constant f^{abc}_{d}

$$h^{ab} = (T^a, T^b), (4.1.13)$$

$$[T^a, T^b, T^c] = f^{abc}_{\ \ d} T^d.$$
(4.1.14)

In terms of the structure constant, the fundamental identity (4.1.11) can be expressed as

$$f^{abc}_{\ g} f^{deg}_{\ f} = f^{dea}_{\ g} f^{bcg}_{\ f} + f^{deb}_{\ g} f^{cag}_{\ f} + f^{dec}_{\ g} f^{abg}_{\ f}$$
(4.1.15)

$$=3f^{de[a}{}_{g}f^{bc]g}{}_{f}, (4.1.16)$$

which turns out to be equivalent to the relation [265]

$$f^{[abc}_{\ g} f^{d]eg}_{\ f} = 0. (4.1.17)$$

⁴Jacobi identity [A, [B, C]] = [[A, B], C] + [B, [A, C]] ensures that the transformation $\delta_A X = [A, X]$ behaves as derivation $\delta_A [B, C] = [\delta_A B, C] + [B, \delta_A C]$.

Here we will define the fundamental object $T = T^{ab}$ as

$$\mathcal{T} \cdot X := [T^a, T^b, X], \quad \forall X \in \mathcal{A}.$$
(4.1.18)

The fundamental object induces derivation and gives the adjoint map

$$\operatorname{ad}_{T^{a}T^{b}}: X \mapsto [T^{a}, T^{b}, X], \quad \forall X \in \mathcal{A}.$$
 (4.1.19)

If we require that the action of the derivation on the scalar product is invariant

$$\mathcal{T} \cdot (T^c, T^d) = (\mathcal{T} \cdot T^c, T^d) + (T^c, \mathcal{T} \cdot T^d) = 0, \qquad (4.1.20)$$

then we obtain the relation

$$(T^{a}, [T^{b}, T^{c}, T^{d}]) = -([T^{a}, T^{b}, T^{c}], T^{d}).$$
(4.1.21)

A Lie 3-algebra is called "metric" if it satisfies the relation (4.1.21). This metric property is assumed for all of the BLG theories. In terms of the structure constant, (4.1.21) is rewritten as

$$f^{abcd} = f^{[abcd]}. (4.1.22)$$

This antisymmetry of f^{abcd} indicates that the symmetry algebra is contained in $\mathfrak{so}(N)$. To be more precise, we rewrite the fundamental identity (4.1.11) as

$$ad_{AB}(ad_{CD}X) - ad_{CD}(ad_{AB}X) = ad_{([A,B,C],D)+(C,[A,B,D])}X$$
 (4.1.23)

or equivalently

$$\operatorname{ad}_{\mathcal{T}}(\operatorname{ad}_{\mathcal{S}} X) - \operatorname{ad}_{\mathcal{S}}(\operatorname{ad}_{\mathcal{T}} X) = \operatorname{ad}_{\mathcal{T}\mathcal{S}} X, \quad \forall \mathcal{T}, \mathcal{S} \in \wedge^{2} \mathcal{A}, \quad X \in \mathcal{A}.$$
 (4.1.24)

Introducing the coordinates of the $(\dim \mathcal{A} \times \dim \mathcal{A})$ matrices $[T^{a_1}, T^{a_2},] =: \mathcal{T}^{a_1 a_2} =: ad_{a_1 a_2} \in End\mathcal{A}$ as

$$\operatorname{ad}_{\mathcal{T}^{a_1 a_2} k}^{l} = (\mathcal{T}^{a_1 a_2})_k^{l} := f^{a_1 a_2 l}_{k}$$
 (4.1.25)

$$\mathcal{T}^{a_1 a_2} \cdot T^k = [T^{a_1}, T^{a_2}, T^k] = f^{a_1 a_2 k}{}_l T^l, \qquad (4.1.26)$$

then the equations (4.1.23) and (4.1.24) may be written in the form

$$[(\mathcal{T}^{a_1 a_2}), (\mathcal{T}^{b_1 b_2})]_k^s = -f^{a_1 a_2 [b_1 l} f^{b_2]}_{lk} {}^s, \tag{4.1.27}$$

which means that

$$[(\mathcal{T}^{a_1 a_2}), (\mathcal{T}^{b_1 b_2})]_k^s = \frac{1}{2} C^{a_1 a_2 b_1 b_2}{}_{c_1 c_2} (\mathcal{T}^{c_1 c_2})_k^s$$
(4.1.28)

where

$$C^{a_1a_2b_1b_2}{}_{c_1c_2} = f^{a_1a_2[b_1}{}_{[c_1}\delta^{b_2]}_{c_2]}.$$
(4.1.29)

Although (4.1.28) is the same form of the commutator in the Lie algebra, this does not mean $C^{a_1a_2b_1b_2}{}_{c_1c_2}$ are the structure constants of the Lie algebra g because

- 1. $C^{a_1a_2b_1b_2}{}_{c_1c_2}$ may not be antisymmetric under $(a_1, a_2) \leftrightarrow (b_1, b_2)$
- 2. $T^{c_1c_2}$ may not be the basis of the Lie algebra \mathfrak{g} .

However, it has been shown [266] that when the Lie 3-algebra is simple, $C^{a_1a_2b_1b_2}{}_{c_1c_2}$ are antisymmetric in the upper indices

$$f^{a_1a_2[b_1}{}_{[c_1}\delta^{b_2]}_{c_2]} = -f^{b_1b_2[a_1}{}_{[c_1}\delta^{a_2]}_{c_2]}$$
(4.1.30)

and define the structure constants of Lie algebra \mathfrak{g} . Moreover one can find the cases where $\mathcal{T}^{c_1c_2}$ can be viewd as the basis of \mathfrak{g} .

4.1.2 Lagrangian

The BLG-model Lagrangian is

$$\mathcal{L} = -\frac{1}{2} D^{\mu} X^{Ia} D_{\mu} X^{I}_{a} + \frac{i}{2} \overline{\Psi}^{a}_{\dot{A}} \Gamma^{\mu}_{\dot{A}\dot{B}} D_{\mu} \Psi_{\dot{B}a} + \frac{i}{4} \overline{\Psi}_{\dot{A}b} \Gamma^{IJ}_{\dot{A}\dot{B}} X^{I}_{c} X^{J}_{d} \Psi_{\dot{B}a} f^{abcd} - V(X) + \mathcal{L}_{TCS}$$
(4.1.31)

where

$$V(X) = \frac{1}{12} f^{abcd} f^{efg}{}_{d} X^{I}_{a} X^{J}_{b} X^{K}_{c} X^{I}_{e} X^{J}_{f} X^{K}_{g}, \qquad (4.1.32)$$

$$\mathcal{L}_{TCS} = \frac{1}{2} \epsilon^{\mu\nu\lambda} \left(f^{abcd} A_{\mu ab} \partial_{\nu} A_{\lambda cd} + \frac{2}{3} f^{cda}_{\ g} f^{efgb} A_{\mu ab} A_{\nu cd} A_{\lambda ef} \right).$$
(4.1.33)

The covariant derivative is defined as

$$D_{\mu}X_{a} := \partial_{\mu}X_{a} - A_{\mu cd}[T^{c}, T^{d}, X]_{a}$$
$$= \partial_{\mu}X_{a} - \tilde{A}^{b}_{\mu a}X_{b}$$
(4.1.34)

where $\tilde{A}^{a}_{\mu b} := f^{cda}{}_{b}A_{\mu cd}$. Alternatively we can express the Lagrangian in terms of the trace and the triple product of Lie 3-algebra:

$$\mathcal{L} = -\frac{1}{2} (D_{\mu} X^{I}, D^{\mu} X^{I}) + \frac{i}{2} (\overline{\Psi}, \Gamma^{\mu} D_{\mu} \Psi) + \frac{i}{4} \left(\overline{\Psi} \Gamma^{IJ} [X^{I}, X^{J}, \Psi] \right) - \frac{1}{12} \left([X^{I}, X^{J}, X^{K}], [X^{I}, X^{J}, X^{K}] \right) + \frac{1}{2} \epsilon^{\mu\nu\lambda} \left[\operatorname{Tr} \left(A_{\mu a b} \partial_{\nu} \tilde{A}^{a b}_{\lambda} \right) + \frac{2}{3} \operatorname{Tr} \left(A_{\mu a b} \tilde{A}^{a}_{\nu g} \tilde{A}^{b}_{\lambda g} \right) \right]$$
(4.1.35)

Although the kinetic term of the gauge fields is similar to the conventional Chern-Simons term, it is twisted by the structure constant of the 3-algebra. Notice that the gauge fields are non-propagating since it has at most first order derivative terms. This is consistent with the degrees of freedom required from supersymmetry.

From (4.1.31), we obtain the equations of motion

$$D^{\mu}D_{\mu}X_{a}^{I} - \frac{i}{2}\overline{\Psi}_{c}\Gamma^{IJ}X_{d}^{J}\Psi_{b}f^{cdb}{}_{a} + \frac{1}{2}f^{bcd}{}_{a}f^{efg}{}_{d}X_{b}^{J}X_{c}^{K}X_{e}^{I}X_{f}^{J}X_{g}^{K} = 0, \qquad (4.1.36)$$

$$\Gamma^{\mu}D_{\mu}\Psi_{a} + \frac{1}{2}\Gamma^{IJ}X_{c}^{I}X_{d}^{J}\Psi_{b}f^{cdb}_{\ a} = 0, \qquad (4.1.37)$$

$$\tilde{F}^{b}_{\mu\nu a} + \epsilon_{\mu\nu\lambda} (X^{J}_{c}D^{\lambda}X^{J}_{d} + \frac{i}{2}\overline{\Psi}_{c}\Gamma^{\lambda}\Psi_{d})f^{cdb}_{\ a} = 0.$$
(4.1.38)

Here the field strength of the gauge field is defined as

$$\tilde{F}^{b}_{\mu\nu a}X_{b} := [D_{\mu}, D_{\nu}]X_{a}.$$
(4.1.39)

Combining the definition (4.1.34) of the covariant derivative, we can express it as

$$\tilde{F}^b_{\mu\nu a} = \partial_\nu \tilde{A}^b_{\mu a} - \partial_\mu \tilde{A}^b_{\nu a} - \tilde{A}^b_{\mu c} \tilde{A}^c_{\nu a} + \tilde{A}^b_{\nu c} \tilde{A}^c_{\mu a}.$$
(4.1.40)

The field strength satisfies Bianchi identity

$$\epsilon^{\mu\nu\lambda}D_{\mu}\tilde{F}^{a}_{\nu\lambda b}=0. \tag{4.1.41}$$

The stress-energy tensor can be computed as

$$T_{\mu\nu} = D_{\mu} X_{a}^{I} D_{\nu} X^{Ia} - \eta_{\mu\nu} \left(\frac{1}{2} D_{\lambda} X^{Ia} D^{\lambda} X_{a}^{I} + V(X)\right)$$
(4.1.42)

where we set fermionic fields to zero. Thus bosonic part of the Hamiltonian density is

$$\mathcal{H} = T_{00} = \frac{1}{2} D_0 X^{Ia} X_0 X_a^I + \frac{1}{2} D_\alpha X^{Ia} D^\alpha X_a^I + V(X)$$
(4.1.43)

and the momentum density is

$$p_{\alpha} = T_{0\alpha} = D_0 X^{Ia} D_{\alpha} X^I_a. \tag{4.1.44}$$

4.1.3 Gauge transformation

The gauge transformations of the BLG-model are given by

$$\delta_{\Lambda} X_a^I = \Lambda_{cd} [T^c, T^d, X^I]_a$$

= $\Lambda_{cd} f^{cdb}_a X_b^I = \tilde{\Lambda}_a^b X_b^I,$ (4.1.45)

$$\delta_{\Lambda} \Psi_{a} = \Lambda_{cd} [T^{c}, T^{d}, \Psi]_{a}$$
$$= \Lambda_{cd} f^{cdb}{}_{a} \Psi_{b} = \tilde{\Lambda}^{b}_{a} \Psi_{b}, \qquad (4.1.46)$$

$$\delta_{\Lambda}\tilde{A}^{b}_{\mu a} = \partial_{\mu}\tilde{\Lambda}^{b}_{\mu a} - \tilde{\Lambda}^{b}_{c}\tilde{A}^{c}_{\mu a} + \tilde{A}^{b}_{\mu c}\tilde{\Lambda}^{c}_{a}$$

$$=D_{\mu}\Lambda_{a}^{\nu}, \qquad (4.1.47)$$

$$\delta \tilde{F}^b_{\mu\nu a} = -\tilde{\Lambda}^b_c \tilde{F}^c_{\mu\nu a} + \tilde{F}^b_{\mu\nu c} \tilde{\Lambda}^c_a, \qquad (4.1.48)$$

where $\tilde{\Lambda}_{a}^{b} := f^{cdb}{}_{a}\Lambda_{cd}$ is a gauge parameter. Lagrangian (4.1.31) is invariant up to a total derivative terms under the above gauge transformations.

4.1.4 Supersymmetry transformation

The $\mathcal{N} = 8$ supersymmetry transformations of the BLG-model are

$$\delta X_a^I = i\overline{\epsilon}_A \Gamma^I_{A\dot{B}} \Psi_{\dot{B}a}, \tag{4.1.49}$$

$$\delta \Psi_{\dot{A}a} = D_{\mu} X_a^I \Gamma^{\mu} \Gamma^I_{\dot{A}B} \epsilon_B - \frac{1}{6} X_b^I X_c^J X_d^K f^{bcd}_{a} \Gamma^{IJK}_{\dot{A}B} \epsilon_B, \qquad (4.1.50)$$

$$\delta \tilde{A}^{b}_{\mu a} = i \overline{\epsilon}_{A} \Gamma_{\mu} \Gamma^{I}_{A\dot{B}} X^{I}_{c} \Psi_{\dot{B}d} f^{cdb}_{\ a}.$$
(4.1.51)

Here ϵ_A , $A = 1, \dots, 8$ is the unbroken supersymmetry parameter obeying the chirality condition

$$\Gamma^{012}\epsilon = \Gamma^{34\dots10}\epsilon = \epsilon. \tag{4.1.52}$$

This implies that ϵ_A is a two component three-dimensional Majorana spinor and transforms as the spinor representation $\mathbf{8}_s$ of the $SO(8)_R$ R-symmetry. Lagrangian (4.1.31) is invariant under the supersymmetric transformations up to a total derivative.

Using the equations of motion (4.1.36), (4.1.37) and (4.1.38), we find the following relations from (4.1.49), (4.1.50) and (4.1.51):

$$[\delta_1, \delta_2] X_a^I = v^\lambda D_\lambda X_a^I + \tilde{\Lambda}_a^b X_b^I, \qquad (4.1.53)$$

$$[\delta_1, \delta_2] \Psi_a = v^{\lambda} D_{\lambda} \Psi_a + \tilde{\Lambda}^b_a \Psi_b, \qquad (4.1.54)$$

$$[\delta_1, \delta_2]\tilde{A}^b_{\mu a} = v^\lambda \tilde{F}^b_{\mu\lambda a} + D_\mu \tilde{\Lambda}^b_a \tag{4.1.55}$$

where $v^{\lambda} = -2i\overline{\epsilon}_2\Gamma^{\lambda}\epsilon_1$ and $\tilde{\Lambda}^b_a = -2i\overline{\epsilon}_2\Gamma^{JK}\epsilon_1X^J_cX^K_df^{cdb}_a$ are identified with a translation parameter and a gauge parameter respectively. Thus the supersymmetry transformations close into a translation (the first term) and a gauge transformation (the second term) on-shell and the theory is invariant under 16 supersymmetries and $SO(8)_R$ R-symmetry at the classical level.

Allowing the supersymmetry parameter ϵ to has x dependence and taking supersymmetry variations of the action, we obtain

$$\delta S = -i \int d^3 x D_\mu \overline{\epsilon} \left(D_\nu X^I_a \Gamma^\nu \Gamma^I \Gamma^\mu \Psi^a + \frac{1}{6} X^I_a X^J_b X^K_c f^{abcd} \Gamma^{IJK} \Gamma^\mu \Psi_d \right).$$
(4.1.56)

This gives

$$J^{\mu} = -D_{\nu}X^{I}_{a}\Gamma^{\nu}\Gamma^{I}\Gamma^{\mu}\Psi^{a} - \frac{1}{6}X^{I}_{a}X^{J}_{b}X^{K}_{c}f^{abcd}\Gamma^{IJK}\Gamma^{\mu}\Psi_{d}.$$
(4.1.57)

Then the supercharge is

$$Q = \int dx^2 x J^0 = -\int d^2 x \left(D_{\nu} X^I_a \Gamma^{\nu} \Gamma^I \Gamma^0 \Psi^a + \frac{1}{6} X^I_a X^J_b X^K_c f^{abcd} \Gamma^{IJK} \Gamma^0 \Psi_d \right).$$
(4.1.58)

From (4.1.1) one can check that Q has the correct mass dimension $[Q] = \frac{1}{2}$ and J^0 has $[J^0] = \frac{5}{2}$. The supercharge Q is the SUSY generator in the sense that⁵

$$\delta_{\epsilon} \Phi = i[\bar{\epsilon}Q, \Phi] = \begin{cases} i\bar{\epsilon}_{\dot{P}}[Q^{\dot{P}}, \Phi_{B}] & \text{(bosonic field)}\\ i\bar{\epsilon}_{\dot{P}}\{Q^{\dot{P}}, \Phi_{F}^{\dot{Q}}\} & \text{(fermionic field)} \end{cases}$$
(4.1.59)

where \dot{P}, \dot{Q}, \cdots are 11-dimensional spinor indices. As an example, we can generate the SUSY transformation for the scalar fields X^{I}

$$\delta X^{I} = i\overline{\epsilon}[Q, X^{I}]$$

$$= i\overline{\epsilon} \left[-\int d^{2}x \partial_{\nu} X^{J}(x) \Gamma^{\nu} \Gamma^{J} \Gamma^{0} \Psi(x), X^{I}(x') \right]$$

$$= -i\overline{\epsilon} \Gamma^{0} \Gamma^{J} \Gamma^{0} \int d^{2}x \Psi(x) \left[\partial_{0} X^{J}(x), X^{I}(x') \right]$$

$$= i\overline{\epsilon} \Gamma^{J} \int d^{2}x \Psi(x) \delta^{IJ} \delta(x - x') = i\overline{\epsilon} \Gamma^{I} \Psi. \qquad (4.1.60)$$

4.1.5 M2-brane algebra

Now we want to discuss the algebraic structure of the M2-brane by studying the BLG-model. Noting that ⁶

$$i\overline{\epsilon}_{\dot{p}}\{Q^{\dot{p}}, Q^{\dot{Q}}\} = \int d^2 x \overline{\epsilon}_{\dot{p}} \left\{ Q^{\dot{p}}, J^{0\dot{Q}}(x) \right\}$$
$$= \int d^2 x (\delta_{\epsilon} J^{0\dot{Q}}(x)), \qquad (4.1.61)$$

we obtain [267, 268]

$$\left\{ Q^{\dot{P}}, Q^{\dot{Q}} \right\} = -2P_{\mu} (\Gamma^{\mu} \Gamma^{0})^{\dot{P}\dot{Q}} + Z_{IJ} \Gamma^{IJ} \Gamma^{0} + Z_{\alpha IJKL} (\Gamma^{IJKL} \Gamma^{\alpha} \Gamma^{0})^{\dot{P}\dot{Q}} + Z_{IJKL} (\Gamma^{IJKL})^{\dot{P}\dot{Q}}$$
(4.1.62)

where α is the two-dimensional spatial indice of the M2-brane world-volume and P^{μ} is the energy momnetum vector $P^{\mu} := \int d^2x T^{0\mu}$. The central charges are given

⁵ The symbol [A, B] means $AB - (-1)^{AB}BA$ in a \mathbb{Z}_2 -graded algebra.

⁶The central charges are proportional to the world-volume of M2-branes and can be infinite for infinitely extended M2-branes. Focusing on the charge density, we can avoid the infinities.

by

$$Z_{IJ} = -\int d^2 x \operatorname{Tr} \left(D_{\alpha} X^I D_{\beta} X^J \epsilon^{\alpha\beta} - D_0 X^K [X^I, X^J, X^K] \right), \qquad (4.1.63)$$

$$Z_{\alpha IJKL} = \frac{1}{3} \int d^2 x \operatorname{Tr} \left(D_{\beta} X^{[I}[X^J, X^K, X^{L]}] \epsilon^{\alpha \beta} \right), \qquad (4.1.64)$$

$$Z_{IJKL} = \frac{1}{4} \int d^2 x \operatorname{Tr} \left([X^M, X^{[I}, X^J], [X^M, X^K, X^L]] \right).$$
(4.1.65)

Introducing the expression

$$\tilde{\Gamma}^{I} = \begin{pmatrix} 0 & \overline{\Gamma}^{I}_{\dot{A}A} \\ \overline{\Gamma}^{I}_{B\dot{B}} & 0 \end{pmatrix}$$
(4.1.66)

where $(\overline{\Gamma}_{A\dot{A}}^{I})^{T} = \overline{\Gamma}_{\dot{A}A}^{I}$ are 8 × 8 real gamma matrices satisfying

$$\overline{\Gamma}_{\dot{A}\dot{A}}^{I}\overline{\Gamma}_{\dot{A}\dot{B}}^{J} + \overline{\Gamma}_{\dot{A}\dot{A}}^{J}\overline{\Gamma}_{\dot{A}\dot{B}}^{I} = 2\delta^{IJ}\delta_{AB},$$

$$\overline{\Gamma}_{\dot{A}A}^{I}\overline{\Gamma}_{\dot{A}\dot{B}}^{J} + \overline{\Gamma}_{\dot{A}\dot{A}}^{J}\overline{\Gamma}_{\dot{A}\dot{B}}^{I} = 2\delta^{IJ}\delta_{\dot{A}\dot{B}}$$
(4.1.67)

we can rewrite (4.1.63) and (4.1.64) as surface integrals [267]

$$Z^{[AB]} = -\int d^2 x \partial_{\alpha} \operatorname{Tr} \left(X^I, D_{\beta} X^J \right) \epsilon^{\alpha \beta} (\overline{\Gamma}^{IJ})^{AB}, \qquad (4.1.68)$$

$$Z_{\mu}^{(AB)} = -\frac{1}{12} \int d^2 x \partial_{\alpha} \operatorname{Tr} \left(X^I, [X^J, X^K, X^L] \right) \epsilon^{0\alpha}{}_{\mu} (\overline{\Gamma}^{IJKL})^{AB}$$
(4.1.69)

where the symmetric central charge is traceless $\delta_{AB}Z_{\mu}^{(AB)} = 0$ and $A, B, \dots = 1, \dots, 8$ are the SO(8) indices. (4.1.62) and (4.1.63)-(4.1.65) are the field realization of the M2-brane algebra and the central charges [269]. These are useful tools to investigate five constitutes in M-theory, that is M-wave, M2-brane, M5-brane, M-KK monopole, M9-brane.

1. $Z^{[AB]}$

 $Z^{[AB]}$ is a world-volume o-form transforming **28** of SO(8). o-form corresponds to a o-brane (point) on the M2-brane. **28** defines a 2-form or 6-form in the transverse space to the M2-brane. In the case of 2-form, o-brane is the result of the intersection with two another M2-brane over a point and defines the 2-plane along which the second M2-brane is aligned [270].

	0	1	2	3	4	5	6	7	8	9	10	
M2	0	0	0	\times	\times	\times	\times	\times	\times	×	×	(4.1.70)
M2	0	×	×	0	0	×	×	×	×	×	×	

When choosing 6-form, o-brane acquires the interpretation as the intersection of M2-brane with M-KK monopole over a point [271].

	0	1	2	3	4	5	6	7	8	9	10		
M2	0	0	0	\times	\times	\times	×	\times	×	×	\times	(4.1.7	71)
MKK	0	×	×	0	0	0	0	0	0	X	×		

2. $Z_{\mu}^{(AB)}$

 $Z_{\mu}^{(AB)}$ is a world-volume 1-form and **35**⁺ of SO(8). 1-form corresponds to a 1-brane (string) on the M2-brane. **35**⁺ defines a 4-form in the 8-dimensional transverse space. 1-brane is determined by 4-plane along which four of the spatial spaces of the M5-brane are aligned. Thus 1-brane has the interpretation as the intersection of M2-brane with M5-brane.

	0	1	2	3	4	5	6	7	8	9	10	
M2	0	0	0	×	\times	\times	\times	\times	\times	\times	×	(4.1.72)
M5	0	0	Х	0	0	0	0	×	×	X	×	

3. Z_{IJKL}

Due to the total antisymmetry and the fundamental identity, $Z_{IJKL} = Z_{[IJKL]}$ vanishes when we consider trace elements.

However, it is discussed [268] that if we take into account constant background configurations of X^I that take values in non-trace elements⁷, such configurations may give rise to BPS charges although non-abelian fields are infinite dimensional and have an infinite norm⁸.

4. *P*_μ

 P_{μ} is a 1-form on a world-volume and a singlet **1** of SO(8). 1-form corresponds to a 1-brane (string) on the M2-brane. **1** defines a o-form or 8-form in the transverse space. In the case of o-form, 1-brane can be viewed as the

⁷Configurations with non-trace elements are discussed in the matrix theory conjecture for Mtheory in the light-cone quantization [272].

⁸By the novel Higgs mechanism, we can reduce Z_{IJKL} to the form $\text{Tr}[X^I, X^J][X^I, X^J]$ which is similar to D4-brane charge in the Do-brane action in the matrix model. It is natural to think that Z_{IJKL} is identified with D6-brane charge because the BLG theory action reduces to the D2-brane action rather than Do-brane action. Furthermore D6-brane is uplifted to M-KK monopole, so Z_{IJKL} is expected to produce the energy bound of the configuration of M2-brane and M-KK monopole.

intersection of M2-brane with an M-wave over a 1-dimensional string.

In the case of 8-form, 1-brane is the intersection of the M2-brane with M9brane over a string.

4.2 \mathcal{A}_4 BLG-theory

If we assume that

- 1. the metric h^{ab} of the 3-algebra A is positive definite so that the kinetic term and the potential term are all positive,
- 2. the dimension *N* of 3-algebra A is finite,

then the 3-algebra A is uniquely determined by [273, 274]

$$f^{abcd} = \frac{2\pi}{k} \epsilon^{abcd} = f \epsilon^{abcd}, \qquad (4.2.1)$$

$$h^{ab} = \delta^{ab} \tag{4.2.2}$$

with $a, b = 1, \dots, 4$. Here e^{abcd} is an antisymmetric tensor and k is the integer determined by the quantization of the Chern-Simons level for a non-simply connected gauge group SO(4) [275]. The correct normalization can be checked by using the expression (4.2.14) and noting that the coefficient of the Chern-Simons term is $\frac{k}{4\pi}$.

The 3-algebra characterized by (4.2.1) and (4.2.2) is called the A_4 algebra. For the A_4 algebra we do not distinguish superscripts and subscripts since gauge indices a, b, \cdots are raised and lowered with Kronecker delta. However, A and \tilde{A} should be distinguished because of the existence of f. The corresponding BLG theory has no continuous coupling constant but admit a discrete coupling k. The uniqueness up to the Chern-Simons level k makes it difficult to describe an arbitrary number of coincident M2-branes because the rank of the gauge algebra is expected to be related to the number of M2-branes in analogy with D-branes. In terms of the antisymmetric tensor ϵ_{abcd} let us introduce the dual generators

$$M_{a_1a_2} := \frac{1}{2} \epsilon_{a_1a_2b_1b_2} \mathcal{T}^{b_1b_2}$$
(4.2.3)

for the fundamental object \mathcal{T} . Then from the relation

$$\epsilon_{j_1\cdots j_n}^{i_1\cdots i_n} = \sum_{k=1}^n (-1)^{k+1} \delta_{j_k}^{i_1} \epsilon_{j_1\cdots j_k\cdots j_n}^{i_2\cdots i_n} = \sum_{k=1}^n (-1)^{k+n} \epsilon_{j_1\cdots j_k\cdots j_n}^{i_1\cdots i_{n-1}} \delta_{j_k}^{i_n}, \qquad (4.2.4)$$

we obtain the commutation relations

$$[M_{a_1a_2}, M_{b_1b_2}] = -\delta_{a_1b_2}M_{a_2b_1} - \delta_{a_2b_1}M_{a_1b_2} + \delta_{a_1b_1}M_{a_2b_2} + \delta_{a_2b_2}M_{a_1b_1}$$
(4.2.5)

The algebraic relation (4.2.5) is recognized as commutators of semisimple $\mathfrak{so}(4)$ algebra. Thus from the ordinary Lie algebra point of view, the \mathcal{A}_4 BLG theory is based on the $\mathfrak{so}(4)$ gauge algebra. It has been discussed [276] that for the \mathcal{A}_4 BLG-model there are two possible inequivalent gauge groups *G*;

$$G = \begin{cases} SO(4) &\cong (SU(2) \times SU(2)) / \mathbb{Z}_2 \\ Spin(4) &\cong SU(2) \times SU(2). \end{cases}$$
(4.2.6)

4.2.1 Quiver gauge structure

Now we want to discuss the connection between the BLG-model based on the Lie 3-algebras and the ordinary gauge theories based on the Lie algebras. This has been accomplished by the remarkable observation [277] that the A_4 BLG-model can be rewritten as an ordinary gauge theory with quiver type gauge group and matters in the bifundamental representation ⁹.

Since in the A_4 theory the Higgs fields X^I and Ψ are the fundamental representation **4** of the $\mathfrak{so}(4)$ we can denote them by the four-vectors

$$X^{I} = \begin{pmatrix} x_{1}^{I} \\ x_{2}^{I} \\ x_{3}^{I} \\ x_{4}^{I} \end{pmatrix}, \qquad \Psi = \begin{pmatrix} \Psi_{1} \\ \Psi_{2} \\ \Psi_{3} \\ \Psi_{4} \end{pmatrix}. \qquad (4.2.7)$$

In terms of the Pauli matrices σ_i^{10} , one may express these in the bi-fundamental

⁹This alternative expression of the A_4 BLG-model triggered the discovery of the ABJM-model.

¹⁰Pauli matrices σ_i are given in (4.1.7) and normalized such that $\text{Tr}(\sigma_i \sigma_j) = 2\delta_{ij}$.

representation (2, 2) of the $\mathfrak{su}(2) \oplus \mathfrak{su}(2)$ gauge algebra as

$$X^{I} = \frac{1}{2} (x_{4}^{I} \mathbb{I}_{2} + ix_{i}^{I} \sigma^{i}) = \frac{1}{2} \begin{pmatrix} x_{4}^{I} + ix_{3}^{I} & x_{2}^{I} + ix_{1}^{I} \\ -x_{2}^{I} + ix_{1}^{I} & x_{4}^{I} - ix_{3}^{I} \end{pmatrix},$$

$$\Psi = \frac{1}{2} (\Psi_{4} \mathbb{I}_{2} + i\Psi_{i} \sigma^{i}) = \frac{1}{2} \begin{pmatrix} \Psi_{4} + i\Psi_{3} & \Psi_{2} + i\Psi_{1} \\ -\Psi_{2} + i\Psi_{1} & \Psi_{4} - i\Psi_{3} \end{pmatrix}.$$
 (4.2.8)

They obey the reality conditions

$$X^{I}_{\alpha\dot{\beta}} = \epsilon_{\alpha\beta} \epsilon_{\dot{\beta}\dot{\alpha}} (X^{I\dagger})^{\dot{\alpha}\beta},$$

$$\Psi_{\alpha\dot{\beta}} = \epsilon_{\alpha\beta} \epsilon_{\dot{\beta}\dot{\alpha}} (\Psi^{\dagger})^{\dot{\alpha}\beta},$$
(4.2.9)

where α , $\beta = 1, 2$ and $\dot{\alpha}$, $\dot{\beta} = 1, 2$ denote bi-fundamental representation (**2**, **2**) of the $\mathfrak{su}(2) \times \mathfrak{su}(2)$ gauge algebra.

In order to find the adjoint gauge field for each $\mathfrak{su}(2)$ gauge symmetry factor, we decompose gauge fields $A_{\mu ab}$ into the sum of the selfdual and anti-selfdual parts

$$A_{\mu ab} := -\frac{1}{2f} (A^+_{\mu ab} + A^-_{\mu ab})$$
(4.2.10)

where

$$*A^{+}_{\mu ab} = \frac{1}{2} \epsilon_{ab}{}^{cd} A^{+}_{\mu cd} = A^{+}_{\mu ab},$$

$$*A^{-}_{\mu ab} = \frac{1}{2} \epsilon_{ab}{}^{cd} A^{-}_{\mu cd} = -A^{-}_{\mu ab}$$
(4.2.11)

and * is the Hodge star acting on the gauge indices and satisfying $*^2 = 1$. Noting that $\tilde{A}^{ab}_{\mu} = f^{cdab} A_{\mu cd}$, we also have

$$\tilde{A}^{cd}_{\mu} = -(A^{+cd}_{\mu} - A^{-cd}_{\mu}).$$
(4.2.12)

Then we define

$$A_{\mu} := A_{\mu 4i}^{+} \sigma_{i},$$

$$\hat{A}_{\mu} := A_{\mu 4i}^{-} \sigma_{i}.$$
(4.2.13)

Using the expressions (4.2.8) and (4.2.13), we rewrite the original BLG-theory La-

grangian (4.1.31) as

$$\mathcal{L} = -\operatorname{Tr}\left(D^{\mu}X^{I\dagger}D_{\mu}X^{I}\right) + i\operatorname{Tr}\left(\overline{\Psi}^{\dagger}\Gamma^{\mu}D_{\mu}\Psi\right) - \frac{2}{3}if\operatorname{Tr}\left(\overline{\Psi}^{\dagger}\Gamma^{IJ}X^{I}X^{J\dagger}\Psi + \overline{\Psi}^{\dagger}\Gamma^{IJ}X^{J}\Psi^{\dagger}X^{I} + \overline{\Psi}^{\dagger}\Gamma^{IJ}\Psi X^{I\dagger}X^{J}\right) - \frac{8}{3}f^{2}\operatorname{Tr}\left(X^{[I}X^{J\dagger}X^{k]}X^{K\dagger}X^{J}X^{I\dagger}\right) + \frac{1}{2f}\epsilon^{\mu\nu\lambda}\operatorname{Tr}\left(A_{\mu}\partial_{\nu}A_{\lambda} + \frac{2}{3}iA_{\mu}A_{\nu}A_{\lambda}\right) - \frac{1}{2f}\epsilon^{\mu\nu\lambda}\operatorname{Tr}\left(\hat{A}_{\mu}\partial_{\nu}\tilde{A}_{\lambda} + \frac{2}{3}i\hat{A}_{\mu}\hat{A}_{\nu}\hat{A}_{\lambda}\right).$$

$$(4.2.14)$$

Here the covariant derivative is defined by

$$D_{\mu}X^{I} = \partial_{\mu}X^{I} + iA_{\mu}X^{I} - iX^{I}\hat{A}_{\mu}.$$
(4.2.15)

Notice that now the twisted Chern-Simons terms in the original BLG-model is decomposed into two ordinary Chern-Simons terms for A and \hat{A} with opposite signs. This observation was crucial for the discovery of the ABJM-model as it opens up the highly extended supersymmetric Chern-Simons matter theories with quiver type gauge group. We see that the theory becomes weakly coupled in the large k limit since after rescaling $A \rightarrow \sqrt{f}A$, all interaction terms are proportional to positive power of $f = \frac{2\pi}{k}$.

The Lagrangian (4.2.14) is invariant under a new set of supersymmetry transformations

$$\delta X^{I} = i\overline{\varepsilon}\Gamma^{I}\Psi, \qquad (4.2.16)$$

$$\delta \Psi = D_{\mu} X^{I} \Gamma^{\mu} \Gamma^{I} \epsilon + \frac{4\pi}{3} X^{I} X^{J\dagger} X^{K} \Gamma^{IJK} \epsilon, \qquad (4.2.17)$$

$$\delta A_{\mu} = f \overline{\epsilon} \Gamma^{I} (X^{I} \Psi^{\dagger} - \Psi X^{I\dagger}), \qquad (4.2.18)$$

$$\delta \hat{A}_{\mu} = f \overline{\epsilon} \Gamma_{\mu} \Gamma^{I} (\Psi^{\dagger} X^{I} - X^{I \dagger} \Psi).$$
(4.2.19)

4.2.2 Superconformal symmetry

It has been proven [264] that the \mathcal{A}_4 BLG theory has OSp(8|4) superconformal symmetry that contains the $SO(8)_R$ R-symmetry group and the three-dimensional $Sp(4) \cong Spin(2,3)$ conformal symmetry group as bosonic factor groups at the classical level. To see the superconformal symmetry explicitly, we replace supersymmetry parameter ϵ_A by $\Gamma^{\mu} x_{\mu} \eta_A$ where η_A is a superconformal symmetry parameter and add a term $-\Gamma^I X_a^I \eta$ to $\delta \Psi_a$ in the supersymmetry transformations of the BLG-model. Then the superconformal symmetry is given by

$$\delta X_a^I = i\overline{\eta}\Gamma^\mu x_\mu \Gamma^I \Psi_a, \tag{4.2.20}$$

$$\delta \Psi_a = D_\mu X_a^I \Gamma^\mu \Gamma^I \Gamma^\nu x_\nu \eta - \frac{1}{6} X_b^I X_c^J X_d^K f^{bcd}_{a} \Gamma^{IJK} \Gamma^\nu x_\nu \eta - \Gamma^I X_a^I \eta, \qquad (4.2.21)$$

$$\delta \tilde{A}^{b}_{\mu a} = i\overline{\eta} x_{\nu} \Gamma^{\nu} \Gamma_{\mu} \Gamma^{I} X^{I}_{c} \Psi_{d} f^{cdb}_{\ a}$$
(4.2.22)

and one can check that the action (4.1.31) is invariant under the superconformal transformations (4.2.20)-(4.2.22) up to total derivative terms.

Parity invariance 4.2.3

Although Chern-Simons theories are parity violating, we can make the A_4 BLG Lagrangian (4.2.14) parity invariant by defining parity transformation as a spatial reflection together with interchange of two SU(2) gauge groups [264, 22, 277]. This implies that we assign an odd parity to f^{abcd} . In particular, under the reflection $x^2 \rightarrow -x^2$ we require that

$$X_a^I \to X_{a'}^I$$
, $\tilde{A}_{2b}^a \to -\tilde{A}_{2b'}^a$, (4.2.23)

$$f^{abcd} \to -f^{abcd},$$
 (4.2.24)

$$\begin{split} \tilde{A}^{a}_{0b} &\to \tilde{A}^{a}_{0b}, \qquad \qquad f^{abcd} \to -f^{abcd}, \qquad (4.2.24) \\ \tilde{A}^{a}_{1b} &\to \tilde{A}^{a}_{1b}, \qquad \qquad \Psi_{a} \to \Gamma_{2} \Psi_{a}. \qquad (4.2.25) \end{split}$$

Then (4.2.14) turns out ot be parity conserving.

Moduli space 4.2.4

The vacuum moduli space of the theory is the configuration space that minimise the potential modulo gauge transformations. For the A_4 BLG-model it was initially investigated in [277, 278, 275]. Since A_4 BLG theory has the Euclidean inner product, the potential is positive definite and the potential is minimal when

$$[X^I, X^J, X^K] = 0. (4.2.26)$$

From the fact that the bosonic scalar fields X_a^I are eight vectors in an \mathbb{R}^4 rotated by the gauge symmetry SO(4), the triple product $X_a^I X_b^J X_c^K$ produces a new vector perpendicular to the three vectors X_a^I , X_b^J and X_c^K whose length is the signed volume of the parallelepiped spanned by the three vectors in \mathbb{R}^4 (see Figure 4.1).

The bosonic potential is proportional to the square of this volume summed over each possible triple of vectors. Therefore the bosonic potential vanishes if and only if all the three vectors lie in the same plane. This space is labeled by ordered sets



Figure 4.1: The parallelepiped spanned by the three vectors X_a^I , X_b^J and X_c^K . A new vector produced by the triple product has the length as the signed volume of the parallelepiped. The triple product is zero if and only if all the vectors lie in the same plane.

of eight vectors in the same plane. One can assume that all vectors lie in the x_1 - x_2 plane without losing generality where x_a are the coordinates of T^a . Then eight x_1 coordinates r_1^I and the eight x_2 coordinates r_2^I form two octuplets which are rotated into each other by the residual O(2) symmetry. Thus, up to gauge transformation, the vacuum moduli space is parametrized by

$$X_{a}^{I} = r_{1}^{I}T^{1} + r_{2}^{I}T^{2} = \begin{pmatrix} r_{1}^{I} \\ r_{2}^{I} \\ 0 \\ 0 \end{pmatrix}, \quad r_{1}^{I}, r_{2}^{I} \in \mathbb{R}^{8}.$$
(4.2.27)

In the bi-fundamental notation (4.2.8), (4.2.27) is expressed as

$$X^{I} = \frac{1}{\sqrt{2}} \begin{pmatrix} z^{I} & 0\\ 0 & \overline{z}^{I} \end{pmatrix} = \frac{1}{\sqrt{2}} \begin{pmatrix} r_{1} + ir_{2}^{I} & 0\\ 0 & r_{1}^{I} - ir_{2}^{I} \end{pmatrix}.$$
 (4.2.28)

Then one can see that the residual gauge symmetries $g \in SO(4)$ that preserve the form X^{I} is the block diagonal form

$$g = \left(\begin{array}{cc} g_1 & 0\\ 0 & g_2 \end{array}\right) \tag{4.2.29}$$

where $g_1, g_2 \in O(2)$ act on (x_1, x_2) and (x_3, x_4) respectively, with det $g_1 = \det g_2$. Since g_2 acts trivially on (4.2.27), we can ignore it and simply look at $g_1 \in O(2)$. Let us discuss the residual gauge symmetry in the diagonal configurations (4.2.28). The residual O(2) gauge symmetry in which g_1 is contained consists of two types of symmetries:

1. simultaneous rotation on z^{I} (continuous symmetry)

$$U(1)_{12}: z^I \to e^{i\theta} z^I, \quad \theta \in [0, 2\pi)$$
 (4.2.30)

2. simultaneous complex conjugation (discrete symmetry)

$$z^I o \overline{z}^I$$
 (4.2.31)

However, the continuous symmetry U_{12} is generically broken down for the diagonal configuration (4.2.27). Therefore the remaining component of gauge field become massive by the Higgs mechanism. To see this we shall write down the effective action. Let us firstly define the gauge field B_{μ} associated with the broken $U(1)_{12}$ that rotate z^{I} and the preserved gauge field C_{μ} associated with the preserved U(1) by

$$B_{\mu} := \frac{4\pi}{k} A_{\mu}^{34}, \qquad (4.2.32)$$

$$C_{\mu} := \frac{4\pi}{k} A_{\mu}^{12}. \tag{4.2.33}$$

Note that because of ϵ^{abcd} in the covariant derivative (4.1.34) the broken $U(1)_{12}$ gauge field is associated with A^{34}_{μ} not A^{12}_{μ} . Substituting the configurations (4.2.27), (4.2.32) and (4.2.33) into the BLG Lagrangian (4.1.31), one can write the kinetic terms on the moduli space and the twisted Chern-Simons terms as [278, 275]

$$\mathcal{L}_{kin} + \mathcal{L}_{TCS} = -\frac{1}{2} |\mathcal{D}_{\mu} z^{I}|^{2} + \frac{k}{2\pi} \epsilon^{\mu\nu\lambda} B_{\mu} \partial_{\nu} C_{\lambda}$$
(4.2.34)

where $\mathcal{D}_{\mu}z^{I} = \partial_{\mu}z^{I} + iB_{\mu}z^{I}$.

Moreover we can replace the unbroken gauge field *C* with its dual photon σ that plays a role of a Lagrange multiplier to impose the Bianchi identity $\epsilon^{\mu\nu\lambda}\partial_{\mu}G_{\nu\lambda}$ on the field strength $G_{\mu\nu} := \partial_{\mu}C_{\nu} - \partial_{\nu}C_{\mu}$ by introducing the additional term ¹¹

$$\mathcal{L}_{dual} = \frac{1}{4\pi} \sigma \epsilon^{\mu\nu\lambda} \partial_{\mu} G_{\nu\lambda}. \tag{4.2.35}$$

¹¹In the original work of [278, 275] the normalization is chosen as $\mathcal{L}_{dual} = \frac{1}{8\pi} \sigma \epsilon^{\mu\nu\lambda} \partial_{\mu} G_{\nu\lambda}$ so that $\sigma \in [0, 2\pi)$. However, this does depend on the two choice of the gauge group; $SU(2) \times SU(2)$ and $(SU(2) \times SU(2))/\mathbb{Z}_2$ as pointed in [276].

Combining (4.2.34) and (4.2.35), we can write the low-energy effective action as [278, 275]

$$\mathcal{L}_{kin} + \mathcal{L}_{TCS} + \mathcal{L}_{dual} = -\frac{1}{2} |\mathcal{D}_{\mu} z^{I}|^{2} + \frac{1}{4\pi} \epsilon^{\mu\nu\lambda} (kB_{\mu} - \partial_{\mu}\sigma) G_{\nu\lambda}.$$
(4.2.36)

The action (4.2.36) is invariant under the $U(1)_{12}$ gauge symmetry transformations

$$z^{I} \rightarrow e^{i\theta} z^{I}, \qquad \sigma \rightarrow \sigma + k\theta, \qquad B_{\mu} \rightarrow B_{\mu} + \partial_{\mu}\theta.$$
 (4.2.37)

Using the equation of motion for $G_{\mu\nu}$

$$B_{\mu} = \frac{\partial_{\mu}\sigma}{k}, \qquad (4.2.38)$$

the action (4.2.36) further reduces to

$$\mathcal{L} = -\frac{1}{2} |\partial_{\mu} z^{I} - \frac{i}{k} z^{I} \partial_{\mu} \sigma|^{2}.$$
(4.2.39)

By defining the fields

$$w^I := e^{-\frac{i\sigma}{k}} z^I, \tag{4.2.40}$$

we can absorb the Lagrange multiplier σ and the action (4.2.39) finally becomes

$$\mathcal{L} = -\frac{1}{2} \partial_{\mu} w^{I} \partial^{\mu} \overline{w}^{I}. \tag{4.2.41}$$

As a next step we need to determine the periodicity of σ which yields the gauge symmetry of the moduli parameter z^{I} as seen from the redefinition (4.2.40). The periodicity of σ occurs from the Dirac quantization of the flux of the field strength. Let us consider the case where some field ϕ couples to a U(1) gauge field A_{μ} as $D_{\mu}\phi = \partial_{\mu}\phi + iA_{\mu}\phi$. If we go around a closed path γ , then ϕ is parallel transported into $\phi_{\gamma} = e^{i\oint_{\gamma}A}\phi = e^{i\int_{\Sigma}F}\phi$ where D is a two-dimensional surface whose boundary is γ and F is the field strength of A. Since the choice of the surface Σ is not unique, we require that $q := \int_{\Sigma} F = 2\pi\mathbb{Z}$. This is the Dirac quantization for the charge q. Now we are interested in the Dirac quantization of the field strength G = dC of the preserved gauge field C since it yields the periodicity for σ as we see from the action (4.2.36). However, in our case the charge of the field strength G = dC turns out to be different as the Dirac value. The result is given by [275, 276]

$$\int_{\Sigma} G \in \begin{cases} 4\pi \mathbb{Z} & \text{for } Spin(4) = SU(2) \times SU(2) \\ 2\pi \mathbb{Z} & \text{for } SO(4) = (SU(2) \times SU(2))/\mathbb{Z}_2. \end{cases}$$
(4.2.42)

This is because at the generic point of the moduli space the U(1) gauge field C sits inside the diagonal $SO(3) \in (SU(2) \times SU(2))/\mathbb{Z}_2$ or $SU(2) \times SU(2)$ and the Higgs fields does not transform as the adjoint representations of the U(1) but that of the SO(3). This situation is similar to the 't Hooft-Polyakov monopoles [279] where all the fields transform in the adjoint representation of $SU(2) \cong SO(3)$. For the $SU(2) \times SU(2)$ group G is thus essentially the sum of two independent field strengths and we need the additional factor 2 as $\int_{\Sigma} G \in 4\pi\mathbb{Z}$. For the $(SU(2) \times SU(2))/\mathbb{Z}_2$ gauge group the phase is equal to one only up to a \mathbb{Z}_2 action and we require that $\int_{\Sigma} G \in 2\pi\mathbb{Z}$. Noting that $dG = \frac{1}{2}\epsilon^{\mu\nu\lambda}\partial_{\mu}G_{\nu\lambda}$ and lifting the relation (4.2.42) to the integral of dG over the three-manifold, we get

$$\frac{1}{4\pi} \int \epsilon^{\mu\nu\lambda} \partial_{\mu} G_{\nu\lambda} \in \begin{cases} 2\mathbb{Z} & \text{for } Spin(4) = SU(2) \times SU(2) \\ \mathbb{Z} & \text{for } SO(4) = (SU(2) \times SU(2))/\mathbb{Z}_2. \end{cases}$$
(4.2.43)

Since σ appears in the action (4.2.36) as the coupling to $\frac{1}{4\pi}\epsilon^{\mu\nu\lambda}\partial_{\mu}G_{\nu\lambda}$, which takes the discrete value in (4.2.42), σ must be periodic as

$$\sigma \sim \begin{cases} \sigma + \pi & \text{for } Spin(4) = SU(2) \times SU(2) \\ \sigma + 2\pi & \text{for } SO(4) = (SU(2) \times SU(2)) / \mathbb{Z}_2. \end{cases}$$
(4.2.44)

Combining the periodicity (4.2.44) and the expression (4.2.40), we can read the gauge identification of z^{I} from the continuous transformation (4.2.30) as

$$z^{I} \cong \begin{cases} e^{\frac{\pi i}{k}} z^{I} & \text{for } Spin(4) = SU(2) \times SU(2) \\ e^{\frac{2\pi i}{k}} z^{I} & \text{for } SO(4) = (SU(2) \times SU(2)) / \mathbb{Z}_{2}. \end{cases}$$
(4.2.45)

At this stage we have two types of the gauge equivalences; one is (4.2.45) from the continuous symmetry (4.2.45) yielding \mathbb{Z}_{2k} or \mathbb{Z}_k and the other is from the discrete one (4.2.31) corresponding to \mathbb{Z}_2 . Since both of them do not commute, we finally obtain the moduli space \mathcal{M}_k of the \mathcal{A}_4 BLG-model with the Chern-Simons level k as [276]

$$\mathcal{M}_{k} = \begin{cases} \frac{\mathbb{R}^{8} \times \mathbb{R}^{8}}{D_{4k}} & \text{for } Spin(4) = SU(2) \times SU(2) \\ \frac{\mathbb{R}^{8} \times \mathbb{R}^{8}}{D_{2k}} & \text{for } SO(4) = (SU(2) \times SU(2))/\mathbb{Z}_{2}. \end{cases}$$
(4.2.46)

For generic k we do not know whether these moduli spaces can have a geometrical interpretation of the M2-branes. However, for k = 1, 2, 4 there is a conjectural space-time interpretation of the M2-branes.

Chapter 5

ABJM-model

In this chapter we will review the ABJM-model [26] which may describe an arbitrary number of M2-branes. We will introduce the notations and conventions in section 5.1. We will turn to the analysis of the moduli space in section 5.2. Then we will discuss the conjectural duality between the BLG-model and the ABJM-model in section 5.3.

5.1 Construction

The ABJM-model is a three-dimensional $\mathcal{N} = 6$ superconformal $U(N)_k \times \hat{U}(N)_{-k}$ Chern-Simons-matter theory proposed as a generalization of the BLG-model in that it may describe the dynamics of an arbitrary number of coincident M2-branes [26]. The theory has manifestly only $\mathcal{N} = 6$ supersymmetry and the corresponding $SU(4)_R$ R-symmetry at the classical level. It has been discussed that [26, 280, 281] at k = 1 and k = 2 these symmetries are enhanced to $\mathcal{N} = 8$ supersymmetry and $SO(8)_R$ R-symmetry as a quantum effect. The theory contains

- 4 complex scalar fields Y^A
- 4 Weyl spinors ψ_A
- 2 types of gauge fields A_{μ} , \hat{A}_{μ} .

Here the upper and lower indices $A, B, \dots = 1, 2, 3, 4$ denote 4 and $\overline{4}$ of the $SU(4)_R$ respectively. The matter fields are $N \times N$ matrices so that Y^A and ψ_A transform as (N, \overline{N}) bi-fundamental representations of $U(N)_k \times \hat{U}(N)_{-k}$ gauge group, while Y_A^{\dagger} and $\psi^{\dagger A}$ do as (\overline{N}, N) . A_{μ} is a Chern-Simons U(N) gauge field of level +k and \hat{A}_{μ} is that of level -k. Also in the theory there is a $U(1)_B$ flavor symmetry and the corresponding baryonic charges are assigned +1 for bi-fundamental fields,

	U(N)	$\hat{U}(N)$	$SU(4)_R$	$U(1)_B$
Y^A	N	\overline{N}	4	+1
Y_A^{\dagger}	\overline{N}	N	$\overline{4}$	-1
ψ_A	N	\overline{N}	$\overline{4}$	+1
$\psi^{\dagger A}$	\overline{N}	N	4	-1
A_{μ}	N^2	1	1	0
\hat{A}_{μ}	1	N^2	1	0

Table 5.1: The symmetries and their representations for fields in the ABJM-model. The bold letters for U(N), $\hat{U}(N)$ and $SU(4)_R$ symmetries denote the representations for the symmetry groups and the quantities for $U(1)_B$ symmetry are the corresponding charges.

-1 for anti-bi-fundamental fields and 0 for gauge fields. The symmetries in the ABJM-model are summarized in Table 5.1.

5.1.1 Lagrangian

The Lagrangian of the ABJM-model is given by [282]

$$\mathcal{L}_{ABJM} = -\operatorname{Tr}(D_{\mu}Y_{A}^{\dagger}D^{\mu}Y^{A}) - i\operatorname{Tr}(\psi^{\dagger A}\gamma^{\mu}D_{\mu}\psi_{A}) - V_{\text{ferm}} - V_{\text{bos}} + \frac{k}{4\pi}\epsilon^{\mu\nu\lambda}\operatorname{Tr}\left[A_{\mu}\partial_{\nu}A_{\lambda} + \frac{2i}{3}A_{\mu}A_{\nu}A_{\lambda} - \hat{A}_{\mu}\partial_{\nu}\hat{A}_{\lambda} - \frac{2i}{3}\hat{A}_{\mu}\hat{A}_{\nu}\hat{A}_{\lambda}\right]$$
(5.1.1)

where

$$V_{\text{ferm}} = -\frac{2\pi i}{k} \text{Tr} \left(Y_{A}^{\dagger} Y^{A} \psi^{\dagger B} \psi_{B} - \psi^{\dagger B} Y^{A} Y_{A}^{\dagger} \psi_{B} - 2Y_{A}^{\dagger} Y^{B} \psi^{\dagger A} \psi_{B} + 2Y^{A} Y_{B}^{\dagger} \psi_{A} \psi^{\dagger B} - e^{ABCD} Y_{A}^{\dagger} \psi_{B} Y_{C}^{\dagger} \psi_{D} + e_{ABCD} Y^{A} \psi^{\dagger B} Y^{C} \psi^{\dagger D} \right), \quad (5.1.2)$$

$$V_{\text{bos}} = -\frac{4\pi^{2}}{3k^{2}} \text{Tr} \left(Y^{A} Y_{A}^{\dagger} Y^{B} Y_{B}^{\dagger} Y^{C} Y_{C}^{\dagger} + Y_{A}^{\dagger} Y^{A} Y_{B}^{\dagger} Y^{B} Y_{C}^{\dagger} Y^{C} + 4Y^{A} Y_{B}^{\dagger} Y^{C} Y_{A}^{\dagger} Y^{B} Y_{C}^{\dagger} - 6Y^{A} Y_{B}^{\dagger} Y^{B} Y_{A}^{\dagger} Y^{C} Y_{C}^{\dagger} \right). \quad (5.1.3)$$

Here we use the Dirac matrix $(\gamma^{\mu})_{\alpha}{}^{\beta} = (i\sigma_2, \sigma_1, \sigma_3)$. The spinor indices are raised, $\theta^{\alpha} = \epsilon^{\alpha\beta}\theta_{\beta}$, and lowered, $\theta_{\alpha} = \epsilon_{\alpha\beta}\theta^{\beta}$ with $\epsilon^{12} = -\epsilon_{12} = 1$. Note that this makes the Dirac matrix $\gamma^{\mu}_{\alpha\beta} := (\gamma^{\mu})_{\alpha}{}^{\gamma}\epsilon_{\beta\gamma} = (-\mathbb{I}_2, -\sigma_3, \sigma_1)$ symmetric and guarantees the

Hermiticity of the fermionic kinetic term. The covariant derivatives are defined by

$$D_{\mu}Y^{A} = \partial_{\mu}Y^{A} + iA_{\mu}Y^{A} - iY^{A}\hat{A}_{\mu}, \quad D_{\mu}\psi_{A} = \partial_{\mu}\psi_{A} + iA_{\mu}\psi_{A} - i\psi_{A}\hat{A}_{\mu},$$
$$D_{\mu}Y^{\dagger}_{A} = \partial_{\mu}Y^{\dagger}_{A} - iA_{\mu}Y^{\dagger}_{A} + iY^{\dagger}_{A}\hat{A}_{\mu}, \quad D_{\mu}\psi^{\dagger A} = \partial_{\mu}\psi^{\dagger A} - iA_{\mu}\psi^{\dagger A} + i\psi^{\dagger A}\hat{A}_{\mu}.$$
(5.1.4)

5.1.2 Supersymmetry transformation

The supersymmetry transformation laws are

$$\delta Y^A = i\omega^{AB}\psi_B,\tag{5.1.5}$$

$$\delta Y_A^{\dagger} = i \psi^{\dagger B} \omega_{AB}, \tag{5.1.6}$$

$$\delta\psi_A = -\gamma^{\mu}\omega_{AB}D_{\mu}Y^B + \frac{2\pi}{k} \left[-\omega_{AB}(Y^C Y^+_C Y^B - Y^B Y^+_C Y^C) + 2\omega_{CD}Y^C Y^+_A Y^D \right],$$
(5.1.7)

$$\delta\psi^{\dagger A} = D_{\mu}Y_{B}^{\dagger}\omega^{AB}\gamma^{\mu} + \frac{2\pi}{k} \left[-(Y_{B}Y^{C}Y_{C}^{\dagger} - Y_{C}^{\dagger}Y^{C}Y_{B}^{\dagger})\omega^{AB} + 2Y_{D}^{\dagger}Y^{A}Y_{C}^{\dagger}\omega^{CD} \right], \quad (5.1.8)$$

$$\delta A_{\mu} = \frac{\pi}{k} \left(-\Upsilon^{A} \psi^{\dagger B} \gamma_{\mu} \omega_{AB} + \omega^{AB} \gamma_{\mu} \psi_{A} \Upsilon^{\dagger}_{B} \right),$$
(5.1.9)

$$\delta \hat{A}_{\mu} = \frac{\pi}{k} \left(-\psi^{\dagger A} \Upsilon^{B} \gamma_{\mu} \omega_{AB} + \omega^{AB} \gamma_{\mu} \Upsilon^{\dagger}_{A} \psi_{B} \right).$$
(5.1.10)

The parameter ω_{AB} is defined by

$$\omega_{AB} := \epsilon_i (\Gamma^i)_{AB}, \quad \omega^{AB} := \epsilon_i (\Gamma^{i*})^{AB}$$
(5.1.11)

where the $SL(2, \mathbb{R})$ spinor ϵ^i , $i = 1, \dots, 6$ transforms as the representation **6** under the $SU(4)_R$ and Γ^i is the six-dimensional 4×4 matrix satisfying

$$(\Gamma^i)_{AB} = -(\Gamma^i)_{BA},$$
 (5.1.12)

$$\frac{1}{2}\epsilon^{ABCD}(\Gamma^{i})_{CD} = -(\Gamma^{i\dagger})^{AB} = (\Gamma^{i\ast})^{AB},$$
(5.1.13)

$$\left\{\Gamma^{i},\Gamma^{j}\right\} = 2\delta_{ij}.\tag{5.1.14}$$

Note that the supersymmetry parameter ω_{AB} obeys

$$\omega^{AB} = \omega_{AB}^* = \frac{1}{2} \epsilon^{ABCD} \omega_{CD}. \tag{5.1.15}$$

5.2 Moduli space

In order to determine the vacuum moduli space of the $U(N)_k \times \hat{U}(N)_{-k}$ ABJMmodel, we need to consider the minimum of the scalar potential. Since the potential turns out to be a perfect square, the potential is minimal when the potential vanishes. The vanishing condition of the bosonic potential is given by

$$Y^{C}Y_{C}^{\dagger}Y^{B} = 0, (5.2.1)$$

$$Y^{C}Y_{A}^{\dagger}Y^{D} = 0. (5.2.2)$$

The generic solution is given by diagonal configurations

$$Y^A = \operatorname{diag}(y_1^A, \cdots, y_N^A) \tag{5.2.3}$$

up to gauge equivalences. The configurations (5.2.3) are the full moduli space because for generic diagonal elements one obtains positive definite mass matrix for the off-diagonal elements and all off-diagonal elements turn out to be massive. The solutions (5.2.3) break the gauge group $U(N) \times \hat{U}(N)$ to $U(1)^N \times U(1)^N \times S_N$ where S_N is the Weyl group of U(N) that permutes the diagonal elements of all matrices. At a generic point of the moduli space, only a $U(1)^N$ subgroup that does not act on the eigenvalues remains unbroken and its gauge transformations keep There are gauge transformations that Y^A diagonal. Quotienting by such gauge symmetries, one finds The moduli space of the $U(N)_k \times \hat{U}(N)_{-k}$ ABJM-model is [26]

$$\mathcal{M}_{N,k} = \frac{(\mathbb{C}^4/\mathbb{Z}_k)^N}{S_N} = \operatorname{Sym}^N(\mathbb{C}^4/\mathbb{Z}_k).$$
(5.2.4)

This can be identified with the moduli space of N indistinguishable M2-branes moving in $\mathbb{C}^4/\mathbb{Z}_k$ transverse space. Therefore the ABJM-model is expected to describe the low-energy world-volume theory of N coincident M2-branes probing an orbifold $\mathbb{C}^4/\mathbb{Z}_k$. The four complex scalar fields Υ^A represent the positions of the membranes in \mathbb{C}^4 .

The orbifold \mathbb{Z}_k acts on the four complex coordinates y^A as

$$y^A \to e^{\frac{2\pi i}{k}} y^A. \tag{5.2.5}$$

This preserves SU(4) rotational symmetry, which is realized as the R-symmetry in the ABJM theory. The action of the \mathbb{Z}_k on the fermionic fields is

$$\psi \to e^{\frac{2\pi(s_1+s_2+s_3+s_4)}{k}}\psi$$
 (5.2.6)

where $s_i = \pm \frac{1}{2}$ are the spinor weights. The chirality projection implies that the sum of all s_i must be even, which produces an eight-dimensional representation. The spinors that are left invariant by the orbifold have $\sum_{i=1}^{4} s_i = 0$, mod k. This selects six out of eight spinors, so the M2-brane theory has 12 supercharges. This agrees with ABJM theory. Therefore this is consistent to the conjecture that the ABJM theory is dual to M-theory on $AdS_4 \times S^7 / \mathbb{Z}_k$ with N units of flux [26].

5.3 Duality between BLG and ABJ(M)

In [276] it has been discussed that if N and k are co-prime, then the vacuum moduli space of the $U(N)_k \times \hat{U}(N)_{-k}$ theory is equivalent to that of the $SU(N) \times SU(N)/\mathbb{Z}_N$ theory. Consequently there are conjectural dualities between the ABJ(M) theory and the BLG theory

$$U(2)_1 \times \hat{U}(2)_{-1}$$
 ABJM theory $\Leftrightarrow SO(4)$ BLG theory with $k = 1$, (5.3.1)

$$U(2)_2 \times \hat{U}(2)_{-2}$$
 ABJM theory $\Leftrightarrow Spin(4)$ BLG theory with $k = 2$, (5.3.2)

$$U(3)_2 \times \hat{U}(2)_{-2}$$
 ABJ theory $\Leftrightarrow SO(4)$ BLG theory with $k = 4$. (5.3.3)

These proposed dualities have been tested by the computations of the superconformal indices [283]. Hence we may regard the SO(4) BLG-model with k = 1 as the world-volume theory of two planar M2-branes propagating in a flat space.

Part III

SCQM from M2-branes

Chapter 6

$\mathcal{N} = 16$ Superconformal Mechanics

Let us turn to the most important part of this thesis in which we will see how the two subjects discussed so far are connected with each other. We will initiate our study in this chapter by considering the BLG-model wrapped on a torus and derive the IR quantum mechanics by shrinking the torus. We will see that the IR quantum mechanics is the $\mathcal{N} = 16$ superconformal gauged quantum mechanics and also find the OSp(16|2) superconformal quantum mechanics from the reduced systems.

6.1 $\mathcal{N} = 16$ gauged quantum mechanics

We shall start our analysis of the wrapped M2-branes with the case where the two membranes wrap a torus T^2 and propagate in a transverse space with an SO(8) holonomy group. For a torus there is no non-trivial spin connection and the world-volume theory of M2-branes is given by the BLG action (4.1.31) defined on $M_3 = \mathbb{R} \times T^2$.

A torus is a compact Riemann surface of genus one and it is characterized by two periods which are defined as the integration of a holomorphic differential ω along two canonical homology basis *a*, *b* of a torus (see Figure 6.1). Let us define the periods by

$$\int_{a} \omega = 1, \quad \int_{b} \omega = \tau. \tag{6.1.1}$$

Here τ is the moduli of the torus and it should not be real.

We now want to consider the limit in which T^2 has vanishingly small size and derive the low-energy effective one-dimensional theory on \mathbb{R} . In order to obtain such a theory we need to determine the configurations with the lowest



Figure 6.1: A torus with two canonical homology basis *a* and *b*.

energy. Since we are now considering supersymmetric theories, the low-energy configurations can be determined by solving the BPS equations. As we are interested in bosonic BPS configurations, we require that the background values of the fermionic fields vanish. Then the bosonic fields are automatically invariant under their supersymmetry transformations. Therefore the BPS equations correspond to the vanishing of the supersymmetry transformations (4.1.50) for fermionic fields. Also we discard the terms which include the covariant derivatives with respect to time because we are now interested in the low energy dynamics as a fluctuation around gauge invariant static configurations. Then one finds the BPS equations

$$D_z X_a^I = 0, \quad D_{\overline{z}} X_a^I = 0,$$
 (6.1.2)

$$[X^I, X^J, X^K] = 0. (6.1.3)$$

To go further we consider the SO(4) BLG-model that may describe two M2branes. In this case the Higgs fields transform as fundamental representations of the SO(4) gauge group and we assume that these Higgs fields have non-zero values. Then the generic solution to (6.1.3) is given by $X_a^I = (X_1^I, X_2^I, 0, 0)^T$. For these solutions, the remaining BPS equations (6.1.2) reduce to

$$\partial_z X_1^I + \tilde{A}_{z2}^1 X_2^I = 0, \qquad \qquad \partial_z X_2^I - \tilde{A}_{z2}^1 X_1^I = 0, \qquad (6.1.4)$$

$$\tilde{A}_{z3}^{1}X_{1}^{I} + \tilde{A}_{z3}^{2}X_{2}^{I} = 0, \qquad \qquad \tilde{A}_{z4}^{1}X_{1}^{I} + \tilde{A}_{z4}^{2}X_{2}^{I} = 0, \qquad (6.1.5)$$

and their complex conjugates. First of all, the equations (6.1.4) tell us that the sum of the squares $(X_1^I)^2 + (X_2^I)^2$ for $I = 1, \dots, 8$ is independent of the locus of the Riemann surface. Thus we can write

$$X_1^{I+2} + iX_2^{I+2} = r^I e^{i(\theta^I + \varphi(z,\bar{z}))}$$
(6.1.6)

where $r^{I}, \theta^{I} \in \mathbb{R}$ are constant on the torus and represent the configuration of the two membranes in the *I*-th direction while $\varphi(z, \overline{z})$ may depend on z and \overline{z} . Furthermore the equations (6.1.4) enable us to write $\tilde{A}_{z2}^{1} = \partial_{z}\varphi$. The second set of equations (6.1.5) forces us to turn off four of six gauge fields; $\tilde{A}_{z3}^{1} = \tilde{A}_{z3}^{2} =$ $\tilde{A}_{z4}^1 = \tilde{A}_{z4}^2 = 0$. These components of the gauge field become massive by the Higgs mechanism. Note that the above set of solutions automatically satisfies the integrability condition for (6.1.2) because the gauge field \tilde{A}_{z2}^1 is flat.

One can find further restrictions by noting that the flat gauge fields \tilde{A}_{z2}^1 on a torus have specific expressions. Cutting a torus along the canonical basis *a* and *b*, the sections of a flat bundle are described by their transition functions, i.e. constant phases around *a* and *b*. Thus they can be completely classified by their twists $e^{2\pi i \zeta}$, $e^{-2\pi i \zeta}$ on the homology along cycles *a*, *b* where ζ and ζ are real parameters. This space is the torus \mathbb{C}/L_{τ} where L_{τ} is the lattice generated by $\mathbb{Z} + \tau \mathbb{Z}$. It is referred to as the Jacobi variety of T^2 denoted by $Jac(T^2)$. The twists on the homology can be described as a point on the Jacobi variety. Hence the flat gauge field can be expressed in the form [284]

$$\tilde{A}_{z2}^{1} = -2\pi \frac{\Theta}{\tau - \overline{\tau}} \omega, \qquad \tilde{A}_{\overline{z}2}^{1} = 2\pi \frac{\overline{\Theta}}{\tau - \overline{\tau}} \overline{\omega}$$
(6.1.7)

where $\Theta := \zeta + \overline{\tau}\xi$ is the complex parameter representing the twists on the homology along two cycles. Subsequently we can write

$$\varphi(z,\overline{z}) = 2\pi \frac{\overline{\Theta}}{\tau - \overline{\tau}} \overline{z} - 2\pi \frac{\Theta}{\tau - \overline{\tau}} z.$$
(6.1.8)

Since the angular variable $\varphi(z, \overline{z})$ in the $X_1^I X_2^I$ -plane characterizes the ratio of two bosonic degrees of freedom for the two membranes, it must take same values modulo $2\pi\mathbb{Z}$ under the shifts $z \to z + 1$ and $z \to z + \tau$ around two cycles. Therefore both the coordinates ξ and ζ are required to be integer values. From the expression (6.1.7), the discretization of these coordinates implies that \tilde{A}_{z2}^1 and \tilde{A}_{z2}^1 are quantized. We therefore conclude that the generic BPS solutions are given by

$$X^{I+2} = \begin{pmatrix} X_A^I \\ X_B^I \\ 0 \\ 0 \end{pmatrix} = \begin{pmatrix} \cos(\theta^I + \varphi(z, \bar{z})) \\ \sin(\theta^I + \varphi(z, \bar{z})) \\ 0 \\ 0 \end{pmatrix} r^I,$$
$$\tilde{A}_z = \begin{pmatrix} 0 & -2\pi \frac{\Theta}{\tau - \bar{\tau}} \omega_z & 0 & 0 \\ 2\pi \frac{\Theta}{\tau - \bar{\tau}} \omega_z & 0 & 0 \\ 0 & 0 & 0 & \tilde{A}_{z4}^3(z, \bar{z}) \\ 0 & 0 & -\tilde{A}_{z4}^3(z, \bar{z}) & 0 \end{pmatrix}.$$
(6.1.9)

Note that the Abelian gauge fields \tilde{A}_{z4}^3 and \tilde{A}_{z4}^3 which are associated with the preserved U(1) symmetry do not have any constraints from the BPS conditions. Taking into account the bosonic configurations (6.1.9) and the supersymmetry transformations (4.1.49), we can write down the fermionic partners

$$\Psi_{\pm} = \begin{pmatrix} \Psi_{\pm A} \\ \Psi_{\pm B} \\ 0 \\ 0 \end{pmatrix}, \quad \overline{\Psi}^{\pm} = \begin{pmatrix} \overline{\Psi}_{A}^{\pm} \\ \overline{\Psi}_{B}^{\pm} \\ 0 \\ 0 \end{pmatrix}$$
(6.1.10)

where $\overline{\Psi}$ is the conjugate spinor defined by $\overline{\Psi} := \Psi^T \tilde{C}$ in terms of the SO(8) charge conjugation matrix \tilde{C} . Ψ^a_+ and $\overline{\Psi}^{+a}$ are the $SO(2)_E$ spinors with the positive chiralities while Ψ^a_- and $\overline{\Psi}^{-a}$ carry the negative ones. Both of them transform as $\mathbf{8}_c$ of the $SO(8)_R$. The subscripts A, B are just the label of the gauge indices 1 and 2¹.

Under retaining the above static BPS configurations (6.1.9) and (6.1.10), we now wish to consider the evolution of time and compactify the system on T^2 . Substitution of the configurations (6.1.9) and (6.1.10) into the action (4.1.31) yields

$$S = \int_{\mathbb{R}} dt \int_{T^2} d^2 z \left[\frac{1}{2} D_0 X^{Ia} D_0 X^I_a - \frac{i}{2} \overline{\Psi}^{\alpha a} D_0 \Psi_{\alpha a} - \frac{k}{2\pi} \tilde{A}^{1}_{02} \tilde{F}^3_{z\bar{z}4} - \frac{k}{4\pi} \left(\tilde{A}^1_{z2} \dot{A}^3_{\bar{z}4} - \tilde{A}^1_{\bar{z}2} \dot{A}^3_{z4} \right) \right]$$
(6.1.11)

where the Greek letters $\alpha = +, -$ denote the $SO(2)_E$ spinor indices. The terms in the first line of the action (6.1.11) come from the kinetic terms of the BLG action while those in the second correspond to the twisted topological Chern-Simons terms.

Firstly since the gauge fields \tilde{A}_{z2}^1 and \tilde{A}_{z2}^1 are quantized and their time derivatives do not appear in the action, these fields are just auxiliary fields. Exploiting the equations of motion they can be excluded and we find the constraints $\tilde{A}_{z4}^3 = \tilde{A}_{z4}^3 = 0$. Hence the corresponding field strength $\tilde{F}_{z\overline{z}4}^3$ has no time dependence. In order to dimensionally reduce the theory on the torus, we rescale the fields as

$$X^{I'} = R^2 X^I, \quad \Psi'_{\alpha a} = R^2 \Psi_{\alpha a}, \quad \overline{\Psi}^{\alpha a'} = R^2 \overline{\Psi}^{\alpha a}$$
(6.1.12)

where *R* is the circumference of the torus. Note that they get the canonical dimensions in the reduced theory; the bosonic variable $X^{I'}$ has mass dimension -1/2 and the fermionic variable Ψ' acquires mass dimension 0.

¹In order to avoid the confusion coming from the various possible explicit numerical subscripts, we here relabel the gauge indices a = 1, 2 as a = A, B.

Let us carry out the integration on the torus with respect to the coordinates z, \overline{z} by applying the Kaluza-Klein ansatz for the Abelian gauge field \tilde{A}_0^{12} and omit the unimportant primes on the fields. Then we find the effective action

$$S = \int_{\mathbb{R}} dt \left[\frac{1}{2} D_0 X^{Ia} D_0 X^I_a - \frac{i}{2} \overline{\Psi}^{\alpha a} D_0 \Psi_{\alpha a} - k C_1(E) \tilde{A}^1_{02} \right].$$
(6.1.13)

Here

$$C_1(E) = \int_{T^2} c_1(E) := \frac{1}{2\pi} \int_{T^2} d^2 z \tilde{F}_{z\overline{z}4}^3$$
(6.1.14)

is the Chern number resulting from the integration of the first Chern class $c_1(E)$ of the U(1) principal bundle $E \rightarrow T^2$ over the torus, which is associated with the preserved U(1) gauge field \tilde{A}^3_{z4} . Hence the last term in the action (6.1.13) is recognized as a Fayet-Iliopoulos (FI) term as in (2.5.3).

The action (6.1.13) is invariant under the one-dimensional conformal transformations

$$\delta t = f(t) = a + bt + ct^2, \qquad \qquad \delta \partial_0 = -\dot{f} \partial_0, \qquad (6.1.15)$$

$$\delta X_a^I = \frac{1}{2} \dot{f} X_a^I, \qquad \qquad \delta \tilde{A}_{02}^1 = -\dot{f} \tilde{A}_{02}^1, \qquad (6.1.16)$$

$$\delta \Psi_{\alpha a} = 0, \qquad \qquad \delta \overline{\Psi}^{\alpha a} = 0 \qquad (6.1.17)$$

where f(t) is a quadratic function of time with real infinitesimal parameters *a*, *b* and *c*.

The action (6.1.13) is also invariant under the $\mathcal{N} = 16$ supersymmetry transformations

$$\delta X_a^I = i\overline{\epsilon}^+ \widetilde{\Gamma}^I \Psi_{-a} - i\overline{\epsilon}_- \widetilde{\Gamma}^I \Psi_{+a}, \qquad \delta \widetilde{A}_{02}^1 = 0, \qquad (6.1.18)$$

$$\delta \Psi_{+a} = -D_0 X_a^I \tilde{\Gamma}^I \epsilon_{-}, \qquad \qquad \delta \Psi_{-a} = D_0 X_a^I \tilde{\Gamma}^I \epsilon_{+}. \qquad (6.1.19)$$

Therefore the low-energy effective theory (6.1.13) is the $\mathcal{N} = 16$ superconformal gauged quantum mechanics with the FI term.

6.2 Reduction

As we have already argued, gauged conformal mechanics and the Calogero model reduce to conformal mechanical models with inverse-square type potentials after integrating out the auxiliary gauge fields. In fact our gauged mechanical action (6.1.13) is quadratic in the U(1) gauge field \tilde{A}_{02}^1 and does not contain the time

derivative of the Abelian gauge field. So \tilde{A}_{02}^1 can be identified with an auxiliary field and has no contribution to the Hamiltonian. Namely, the Hamiltonian is invariant under the action of the corresponding U(1) gauge group on the phase space \mathcal{M} . This implies that the corresponding moment map $\mu : \mathcal{M} \to \mathfrak{u}(1)^*$ is the integral of motion [166] and we can reduce the original phase space \mathcal{M} to a smaller phase space $\mathcal{M}_c = \mu^{-1}(c)$ with fewer degrees of freedom by fixing the inverse of the moment map at a point $c \in \mathfrak{u}(1)^{*2}$.

In order to obtain our reduced system, we shall eliminate the auxiliary field \tilde{A}_{02}^1 in two steps; first we choose a specific gauge and then impose the Gauss law constraint to ensure the consistency of the gauge fixing. Let us choose the temporal gauge $\tilde{A}_0 = 0$. Together with the solutions

$$\tilde{A}_{02}^{1} = \frac{kC_{1}(E) + \sum_{I} (r^{I})^{2} \dot{\theta}^{I} + i \overline{\Psi}_{A}^{\alpha} \Psi_{\alpha B}}{\sum_{I} (r^{I})^{2}},$$
(6.2.1)

$$\tilde{A}_{03}^1 = \tilde{A}_{04}^1 = \tilde{A}_{03}^2 = \tilde{A}_{04}^2 = 0$$
(6.2.2)

to the equations of motion for \tilde{A}_0 , we can read off the Gauss law constraint

$$\phi_0 := kC_1(E) + \sum_I (r^I)^2 \dot{\theta}^I + i \overline{\Psi}^{\alpha}_A \Psi_{\alpha B} = 0.$$
(6.2.3)

This equation is the moment map condition. To see the physical meaning of this constraint, we observe that $(r^I)^2 \dot{\theta}^I$ represents the "angular momentum", the SO(2)-charge corresponding to the rotation in the $X_1^I X_2^I$ -plane while the fermionic bilinear term $i \overline{\Psi}_A^{\alpha} \Psi_{\alpha B}$ produces the charge of the SO(2) rotational group of the two types of fermionic variables Ψ_A and Ψ_B . Accordingly the equation (6.2.3) says that the total SO(2) charge which rotates the internal degrees of freedom for the two M2-branes is fixed by the Chern-Simons level k and the Chern number $C_1(E)$.

With the constraint function ϕ_0 , one can write a new Lagrangian by adding $\lambda \phi_0$ where λ is the Lagrange multiplier. The resulting action is

$$S = \int_{\mathbb{R}} dt \left[\frac{1}{2} \sum_{I} (\dot{r}^{I})^{2} + \frac{1}{2} \sum_{I} (r^{I} \dot{\theta}^{I})^{2} - \frac{i}{2} \overline{\Psi}^{\alpha a} \dot{\Psi}_{\alpha a} + \lambda \left(kC_{1}(E) + \sum_{I} (r^{I})^{2} \dot{\theta}^{I} + i \overline{\Psi}^{\alpha}_{A} \Psi_{\alpha B} \right) \right].$$

$$(6.2.4)$$

The absence of the variables $\theta^{I'}$ s in the action (6.2.4) immediately implies that they are cyclic coordinates and their canonical momenta $p_{\theta^{I}} = (r^{I})^{2} \dot{\theta}^{I}$ are just the integrals of motion.

² The components of the moment map form a system being in involution since the gauge group is Abelian. So we do not need to divide by the non-trivial coadjoint isotropy subgroup to obtain the reduced phase space.

At this state we can eliminate cyclic coordinates from the Lagrangian by introducing the Routhian. As we have discussed in section 2.5, the Routhian is a hybrid between the Lagrangian and the Hamiltonian, defined by performing a Legendre transformation on the cyclic coordinates

$$R(r^{I}, \dot{r}^{I}, h^{I}, \Psi) := L(r^{I}, \dot{r}^{I}, \dot{\theta}^{I}, \Psi) - \sum_{I} \dot{\theta}^{I} p_{\theta^{I}}.$$
(6.2.5)

Due to the partial Legendre transformation, the variables r^{I} and Ψ still follow the Euler-Lagrange equations while the cyclic coordinates θ^{I} and their momenta $h^{I} := p_{\theta^{I}}$ obey the Hamilton equations. However, the latter set of equations results in trivial statements; the constant property of h^{I} (i.e. $\dot{h}^{I} = 0$) and the definition of h^{I} (i.e. $\dot{\theta}^{I} = \frac{h^{I}}{(r^{I})^{2}}$). So classically the Routhian is not really $R(r^{I}, \dot{r}^{I}, h^{I}, \Psi)$ but rather $R(r^{I}, \dot{r}^{I}, \Psi)$ along with the integrals of motion h^{I} 's. Hence we can rewrite (6.2.4) as

$$S = \int_{\mathbb{R}} dt \left[\frac{1}{2} \sum_{I} (\dot{r}^{I})^{2} - \frac{1}{2} \sum_{I} \frac{(h^{I})^{2}}{(r^{I})^{2}} - \frac{i}{2} \overline{\Psi}^{\alpha a} \dot{\Psi}_{\alpha a} + \lambda \left(kC_{1}(E) + \sum_{I} h^{I} + i \overline{\Psi}^{\alpha}_{A} \Psi_{\alpha B} \right) \right].$$

$$(6.2.6)$$

Integrating out λ , we finally obtain the reduced effective action

$$S = \frac{1}{2} \int_{\mathbb{R}} dt \left[\dot{q}^2 + \sum_{I \neq K} (\dot{r}^I)^2 - i \Psi^{\alpha a} \dot{\Psi}_{\alpha a} - \frac{\left[k C_1(E) + \sum_{I \neq K} h^I + i \Psi^{\alpha}_A \Psi_{\alpha B} \right]^2}{q^2} - \sum_{I \neq K} \frac{(h^I)^2}{(r^I)^2} \right].$$
(6.2.7)

Here we have taken the SO(8) charge conjugation matrix \tilde{C} as an identity matrix for simplicity ³. We have defined $q := r^K$ where *K* denotes the specific direction in which h^K is automatically determined by other conserved quantities h^I 's. Note that the terms appearing in the numerator of the potential are the integrals of motion, namely they commute with the Hamiltonian.

Let us study the classical properties of the theory (6.2.7). The action (6.2.7)

³For the symmetric charge conjugation matrix one can reduce it to an identity matrix by an appropriate unitary transformation.

leads to the classical equations of motion

$$\ddot{q} = \frac{[kC_1(E) + \sum_{I \neq K} h^I + i\Psi^{\alpha}_A \Psi_{\alpha B}]^2}{q^3},$$
(6.2.8)

$$\ddot{r}^{I} = \frac{(h^{I})^{2}}{(r^{I})^{3}},\tag{6.2.9}$$

$$\dot{\Psi}_{\alpha A} = -\frac{[kC_1(E) + \sum_{I \neq K} h^I + i\Psi^{\alpha}_A \Psi_{\alpha B}]}{q^2} \Psi_{\alpha B}, \qquad (6.2.10)$$

$$\dot{\Psi}_{\alpha B} = \frac{[kC_1(E) + \sum_{I \neq K} h^I + i\Psi^{\alpha}_A \Psi_{\alpha B}]}{q^2} \Psi_{\alpha A}.$$
(6.2.11)

Making use of the equations of motion (6.2.10) and (6.2.11), one can check that the differentiation of the Gauss law constraint (6.2.3) with respect to time *t* vanishes. In other words, ϕ_0 is the constant of motion.

The canonical momenta are

$$p := \frac{\partial L}{\partial \dot{q}} = \dot{q}, \qquad \qquad p_I := \frac{\partial L}{\partial \dot{r}^I} = \dot{r}^I, \qquad (6.2.12)$$

$$\pi^{\alpha a} := \frac{\vec{\partial} L}{\partial \dot{\Psi}_{\alpha a}} = \frac{i}{2} \Psi^{\alpha a}.$$
(6.2.13)

The fermionic momenta $\pi^{\alpha a}$ do not depend on the velocities but on the fermionic degrees of freedom themselves. Hence one can read second-class constraints

$$\phi_1^{\alpha a} := \pi^{\alpha a} - \frac{i}{2} \Psi^{\alpha a} = 0. \tag{6.2.14}$$

Under the constraints, we get the Dirac brackets

$$[q, p]_{DB} = 1,$$
 $[r^{I}, p_{J}]_{DB} = \delta^{I}{}_{J},$ (6.2.15)

$$\left[\Psi_{\alpha a\dot{A}},\pi^{\beta b\dot{B}}\right]_{DB} = \frac{1}{2}\delta_{\alpha\beta}\delta_{ab}\delta_{\dot{A}\dot{B}},\qquad \left[\Psi_{\alpha a\dot{A}},\Psi^{\beta b\dot{B}}\right]_{DB} = -i\delta_{\alpha\beta}\delta_{ab}\delta_{\dot{A}\dot{B}}.\tag{6.2.16}$$

The action (6.2.7) is invariant under the following one-dimensional conformal transformations

$$\delta t = f(t) = a + bt + ct^2, \qquad \qquad \delta \partial_0 = -\dot{f} \partial_0, \qquad (6.2.17)$$

$$\delta q = \frac{1}{2}\dot{f}q, \qquad \qquad \delta r^{I} = \frac{1}{2}\dot{f}r^{I}, \qquad (6.2.18)$$

$$\delta \Psi_{\alpha a} = 0. \tag{6.2.19}$$

Here the constant parameters *a*, *b* and *c* are infinitesimal parameters of translation, dilatation and conformal boost respectively. The corresponding Noether charges,

the Hamiltonian H, the dilatation operator D and the conformal boost operator K are found to be

$$H = \frac{1}{2} \left[p^2 + \frac{\left(kC_1(E) + \sum_{I \neq K} h^I + i\Psi_A^{\alpha} \Psi_{\alpha B} \right)^2}{q^2} + \sum_{I \neq K} \left(p_I^2 + \frac{(h^I)^2}{(r^I)^2} \right) \right], \quad (6.2.20)$$

$$D = tH - \frac{1}{4} \left[(qp + pq) + \sum_{I \neq K} \left(r^{I} p_{I} + p_{I} r^{I} \right) \right], \qquad (6.2.21)$$

$$K = t^{2}H - \frac{1}{2}t\left[(qp + pq) + \sum_{I \neq K} \left(r^{I}p_{I} + p_{I}r^{I}\right)\right] + \frac{1}{2}\left[q^{2} + \sum_{I \neq K} (r^{I})^{2}\right].$$
 (6.2.22)

The action (6.2.7) is invariant under the following fermionic transformations

$$\delta q = \frac{i}{\sqrt{2}} \left(\epsilon^{-} \Psi_{-A} - \epsilon^{+} \Psi_{+A} \right) + \frac{i}{\sqrt{2}} \left(\epsilon^{-} \Psi_{-B} - \epsilon^{+} \Psi_{+B} \right), \qquad (6.2.23)$$

$$\delta r^{I} = i \cos \theta^{I} \left(\epsilon^{+} \tilde{\Gamma}^{I} \Psi_{-A} - \epsilon^{-} \tilde{\Gamma}^{I} \Psi_{+A} \right) + i \sin \theta^{I} \left(\epsilon^{+} \tilde{\Gamma}^{I} \Psi_{-B} - \epsilon^{-} \tilde{\Gamma}^{I} \Psi_{+B} \right),$$
(6.2.24)

$$\delta \Psi_{+A\dot{A}} = -\frac{1}{\sqrt{2}} \left(\dot{q} - \frac{h^{K}}{q} \right) \epsilon_{+\dot{A}} - \frac{i}{\sqrt{2}} \frac{l}{q} \Psi_{+B\dot{A}} - \sum_{I \neq K} \left(\dot{r}^{I} \cos \theta^{I} - \sin \theta^{I} \frac{h^{I}}{r^{I}} \right) \tilde{\Gamma}^{I} \epsilon_{-\dot{A}},$$
(6.2.25)

$$\delta \Psi_{-A\dot{A}} = \frac{1}{\sqrt{2}} \left(\dot{q} - \frac{h^{K}}{q} \right) \epsilon_{-\dot{A}} - \frac{i}{\sqrt{2}} \frac{l}{q} \Psi_{-B\dot{A}} + \sum_{I \neq K} \left(\dot{r}^{I} \cos \theta^{I} - \sin \theta^{I} \frac{h^{I}}{r^{I}} \right) \tilde{\Gamma}^{I} \epsilon_{+\dot{A}}, \tag{6.2.26}$$

$$\delta \Psi_{+B\dot{A}} = -\frac{1}{\sqrt{2}} \left(\dot{q} + \frac{h^{K}}{q} \right) \epsilon_{+\dot{A}} + \frac{i}{\sqrt{2}} \frac{l}{q} \Psi_{+A\dot{A}} - \sum_{I \neq K} \left(\dot{r}^{I} \sin \theta^{I} + \cos \theta^{I} \frac{h^{I}}{r^{I}} \right) \tilde{\Gamma}^{I} \epsilon_{-\dot{A}},$$
(6.2.27)

$$\delta \Psi_{-B\dot{A}} = \frac{1}{\sqrt{2}} \left(\dot{q} + \frac{h^{K}}{q} \right) \epsilon_{-\dot{A}} + \frac{i}{\sqrt{2}} \frac{l}{q} \Psi_{-A\dot{A}} + \sum_{I \neq K} \left(\dot{r}^{I} \sin \theta^{I} + \cos \theta^{I} \frac{h^{I}}{r^{I}} \right) \tilde{\Gamma}^{I} \epsilon_{+\dot{A}}$$
(6.2.28)

where we have defined

$$\theta^{I}(t) := h^{I} \int^{t} \frac{dt'}{(r^{I}(t'))^{2}},$$
(6.2.29)

$$l := \left(\Psi^{+A}\epsilon_{+} - \Psi^{-A}\epsilon_{-}\right) - \left(\Psi^{+B}\epsilon_{+} - \Psi^{-B}\epsilon_{-}\right).$$
(6.2.30)

We should note that the supersymmetry is generically non-local in the sense that the transformations contain the integrals of the function of the non-local variables $r^{I'}$ s with respect to time. The non-locality is the consequence of the Routh reduction. Hence the infinite number of the associated conserved charges may exist and

things may become much more exotic ⁴. However, as seen from the (6.2.20), the motion in the *K*-th direction endowed with the local supersymmetry and others with non-local ones are essentially decoupled because their Hamiltonians commute with each other. Thus we can treat them separately. This indicates that the theory possesses the local conserved supercurrents and the non-local supercurrents which are in involution.

6.3 OSp(16|2) superconformal mechanics

We shall focus on the study of the motion in the *K*-th direction which is associated with the local charges and investigate the algebraic structure of the symmetry group in the quantum mechanics. From now on we will consider the case where the all independent conserved charges $h^{I'}$ s are zeros. This is realized when the internal degrees of freedom for two M2-branes are unbiased. Note that for the purpose of the exploration of the algebraic structure for the *K*-th motion, this specific charge assignment does not affect the following discussion since non-vanishing $h^{I'}$ s can only give rise to a constant shift of the coupling constant in the potential. From (6.2.7) one can read the effective action for the dynamics in the *K*-th direction

$$S = \frac{1}{2} \int_{\mathbb{R}} dt \left[\dot{q}^2 - i \Psi^{\alpha a} \dot{\Psi}_{\alpha a} - \frac{\left(k C_1(E) + i \Psi^{\alpha}_A \Psi_{\alpha B} \right)^2}{q^2} \right].$$
(6.3.1)

We see that our reduced action (6.3.1) contains the inverse-square potential which is similar to the known N > 4 superconformal mechanical potentials discussed in (3.6.1) (also see [210, 255, 224, 137]).

We shall study the existing symmetry in the effective action (6.3.1). The action (6.3.1) may be rewritten as SU(1,1|16) superconformal quantum mechanics in the form of (3.6.1). However, we should note that the same form of the Lagrangian does not necessarily lead to the same symmetry in the theory if we have additional constraints or symmetries. In fact in our setup the bilinear terms for fermions are treated as conserved quantities due to the Gauss constraint (6.2.3). This implies that the gauge indices $a, b, \dots = A, B$ should be distinguished from other indices α, β, \dots and \dot{A}, \dot{B}, \dots and prevents us from forming 32 supercharges. Put it another way, our theory describes the radial motion of the wrapped membranes and

⁴The action (6.2.7) is invariant under the fermionic transformations (6.2.23)-(6.2.28), however, the Gauss constraint (6.2.14) may not be invariant under those transformations. Although in that case the original system may be modified, we here just want to study the reduced system without the auxiliary gauge field.

thus we have at most 16 supercharges on the branes by the projection. So we will only focus on the remaining $\mathcal{N} = 16$ supersymmetry due to the constraint. The simplest way to read the consistent supersymmetry for the wrapped branes is just look at the supersymmetry transformations for the original BLG-model. From (4.1.49)-(4.1.51) we see that the action (6.3.1) is invariant under the following $\mathcal{N} = 16$ supersymmetry transformation laws

$$\delta q = \frac{i}{\sqrt{2}} \left(\epsilon^{-} \Psi_{-A} - \epsilon^{+} \Psi_{+A} \right) + \frac{i}{\sqrt{2}} \left(\epsilon^{-} \Psi_{-B} - \epsilon^{+} \Psi_{+B} \right), \qquad (6.3.2)$$

$$\delta \Psi_{+A\dot{A}} = -\frac{1}{\sqrt{2}} \left(\dot{q} + \frac{g}{q} \right) \epsilon_{+\dot{A}} - \frac{i}{\sqrt{2}} \frac{l}{q} \Psi_{+B\dot{A}}, \tag{6.3.3}$$

$$\delta \Psi_{-A\dot{A}} = \frac{1}{\sqrt{2}} \left(\dot{q} + \frac{g}{q} \right) \epsilon_{-\dot{A}} - \frac{i}{\sqrt{2}} \frac{l}{q} \Psi_{-B\dot{A}'}$$
(6.3.4)

$$\delta \Psi_{+B\dot{A}} = -\frac{1}{\sqrt{2}} \left(\dot{q} - \frac{g}{q} \right) \epsilon_{+\dot{A}} + \frac{i}{\sqrt{2}} \frac{l}{q} \Psi_{+A\dot{A}'}$$
(6.3.5)

$$\delta \Psi_{-B\dot{A}} = \frac{1}{\sqrt{2}} \left(\dot{q} - \frac{g}{q} \right) \epsilon_{-\dot{A}} + \frac{i}{\sqrt{2}} \frac{l}{g} \Psi_{-A\dot{A}}$$
(6.3.6)

where we have defined

$$g := kC_1(E) + i\Psi_A^{\alpha}\Psi_{\alpha B}. \tag{6.3.7}$$

Unlike the transformations (6.2.23)-(6.2.28), the supersymmetry transformations (6.3.2)-(6.3.6) are local and we therefore can apply the conventional Noether's procedure. By means of the Noether's method, the corresponding supercharges are calculated to be

$$Q_{+\dot{A}} = \frac{1}{\sqrt{2}} \left(p + \frac{g}{q} \right) \Psi_{+A\dot{A}} + \frac{1}{\sqrt{2}} \left(p - \frac{g}{q} \right) \Psi_{+B\dot{A}},$$
 (6.3.8)

$$Q_{-\dot{A}} = -\frac{1}{\sqrt{2}} \left(p + \frac{g}{q} \right) \Psi_{-A\dot{A}} - \frac{1}{\sqrt{2}} \left(p - \frac{g}{q} \right) \Psi_{-B\dot{A}}.$$
 (6.3.9)

Since the action (6.3.1) is invariant under the conformal transformations $\delta t = f(t)$, $\delta q = \frac{1}{2}\dot{f}q$ and $\delta \Psi_{\alpha a} = 0$, three generators, the Hamiltonian *H*, the dilatation generator *D* and the conformal boost generator *K* are explicitly expressed as

$$H = \frac{1}{2}p^2 + \frac{\left[kC_1(E) + i\Psi_A^{\alpha}\Psi_{\alpha B}\right]^2}{2q^2},$$
(6.3.10)

$$D = -\frac{1}{4} \{q, p\}, \tag{6.3.11}$$

$$K = \frac{1}{2}q^2 \tag{6.3.12}$$

where $\{,\}$ represents an anti-commutator.

In order to quantize the theory, we impose the (anti)commutation relations for the canonical variables obtained from the Dirac brackets (6.2.15) and (6.2.16)

$$[q, p] = i, \qquad \left\{ \Psi_{\alpha a \dot{A}}, \Psi_{\beta b \dot{B}} \right\} = \delta_{\alpha \beta} \delta_{a b} \delta_{\dot{A} \dot{B}}. \qquad (6.3.13)$$

The combination of the conformal symmetry and the supersymmetry leads to the superconformal symmetry. Let us define the superconformal boost generators

$$S_{+\dot{A}} = \frac{1}{\sqrt{2}}q \left(\Psi_{+A\dot{A}} + \Psi_{+B\dot{A}}\right), \qquad (6.3.14)$$

$$S_{-\dot{A}} = -\frac{1}{\sqrt{2}}q \left(\Psi_{-A\dot{A}} + \Psi_{-B\dot{A}}\right).$$
(6.3.15)

Because of the extended supersymmetry the theory has the internal R-symmetry which rotates the fermionic charges. Let us define the R-symmetry generators by

$$(J_{\alpha\beta})_{\dot{A}\dot{B}} = i\Psi_{\alpha a\dot{A}}\Psi^a_{\beta\dot{B}}.$$
(6.3.16)

Notice that the R-symmetry generators satisfy the relations

$$(J_{++})_{\dot{A}\dot{B}} = -(J_{++})_{\dot{B}\dot{A}}, \tag{6.3.17}$$

$$(J_{--})_{\dot{A}\dot{B}} = -(J_{--})_{\dot{B}\dot{A}}, \tag{6.3.18}$$

$$(J_{+-})_{\dot{A}\dot{B}} = -(J_{-+})_{\dot{B}\dot{A}}$$
(6.3.19)

and therefore the matrices J_{++} , J_{--} and J_{-+} contain 28, 28 and 64 independent entries respectively while J_{-+} yields no independent ones because of the relations (6.3.19). Therefore the R-symmetry matrix totally carries 28 + 28 + 64 = 120 elements.

Using the canonical (anti)commutation relations (6.3.13) and considering the Weyl ordering for the fermionic bilinear terms, one can find the complete set of (anti)commutators among the generators

$$[H,D] = iH, \quad [K,D] = -iK, \quad [H,K] = 2iD,$$
 (6.3.20)

$$[(J_{\alpha\beta})_{\dot{A}\dot{B}}, H] = 0, \quad [(J_{\alpha\beta})_{\dot{A}\dot{B}}, D] = 0, \quad [(J_{\alpha\beta})_{\dot{A}\dot{B}}, K] = 0, \tag{6.3.21}$$

$$[(J_{\alpha\beta})_{\dot{A}\dot{B}}, (J_{\gamma\delta})_{\dot{C}\dot{D}}] = i(J_{\gamma\beta})_{\dot{C}\dot{B}}\delta_{\alpha\delta}\delta_{\dot{A}\dot{D}} - i(J_{\alpha\delta})_{\dot{A}\dot{D}}\delta_{\beta\gamma}\delta_{\dot{B}\dot{C}}, + i(J_{\delta\beta})_{\dot{D}\dot{B}}\delta_{\alpha\gamma}\delta_{\dot{A}\dot{C}} - i(J_{\alpha\gamma})_{\dot{A}\dot{C}}\delta_{\beta\delta}\delta_{\dot{B}\dot{D}},$$
(6.3.22)

$$[H, Q_{\alpha \dot{A}}] = 0, \quad [D, Q_{\alpha \dot{A}}] = -\frac{i}{2}Q_{\alpha \dot{A}}, \quad [K, Q_{\alpha \dot{A}}] = iS_{\alpha \dot{A}}, \quad (6.3.23)$$

$$[H, S_{\alpha \dot{A}}] = -iQ_{\alpha \dot{A}}, \quad [D, S_{\alpha \dot{A}}] = \frac{i}{2}S_{\alpha \dot{A}}, \quad [K, S_{\alpha \dot{A}}] = 0, \tag{6.3.24}$$

$$\{Q_{\alpha\dot{A}}, Q_{\beta\dot{B}}\} = 2H\delta_{\alpha\beta}\delta_{\dot{A}\dot{B}},$$

$$\{S_{\alpha\dot{A}}, S_{\beta\dot{B}}\} = 2K\delta_{\alpha\beta}\delta_{\dot{A}\dot{B}},$$

$$\{Q_{\alpha\dot{A}}, S_{\beta\dot{B}}\} = -2D\delta_{\alpha\beta}\delta_{\dot{A}\dot{B}} + (J_{\alpha\beta})_{\dot{A}\dot{B}},$$

(6.3.25)

$$[(J_{\alpha\beta})_{\dot{A}\dot{B}}, Q_{\gamma\dot{C}}] = i \left(Q_{\alpha\dot{A}} \delta_{\beta\gamma} \delta_{\dot{B}\dot{C}} - Q_{\beta\dot{B}} \delta_{\alpha\gamma} \delta_{\dot{A}\dot{C}} \right), [(J_{\alpha\beta})_{\dot{A}\dot{B}}, S_{\gamma\dot{C}}] = i \left(S_{\alpha\dot{A}} \delta_{\beta\gamma} \delta_{\dot{B}\dot{C}} - S_{\beta\dot{B}} \delta_{\alpha\gamma} \delta_{\dot{A}\dot{C}} \right).$$
(6.3.26)

The Hamiltonian H, the dilatation generator D and the conformal boost generator K satisfy the one-dimensional conformal algebra (6.3.20).

As the superpartners of the conformal generators there are sixteen supercharges $Q_{\alpha\dot{A}}$ and as many superconformal generators $S_{\alpha\dot{A}}$. As seen from (6.3.21) and (6.3.26), the R-symmetry generators $(J_{\alpha\beta})_{\dot{A}\dot{B}}$ commute with the bosonic generators H, D and K while they yield the rotations of the fermionic generators $Q_{\alpha\dot{A}}$ and $S_{\alpha\dot{A}}$. The commutation relation (6.3.22) implies that $(J_{\alpha\beta})_{\dot{A}\dot{B}}$ obey the $\mathfrak{so}(16)$ algebra. Therefore we can conclude that the theory (6.3.1) is the OSp(16|2) invariant $\mathcal{N} = 16$ superconformal mechanics. In fact this fits in the list of the possible simple supergroup for superconformal quantum mechanics, which we have already given in Table 3.2.

It is true that the quantum mechanics (6.3.1) possesses the $\mathcal{N} = 16$ superconformal symmetry, however, it is not clear that the theory (6.3.1) actually captures the dynamics of the wrapped membranes around a torus since it is not totally same as the superconformal gauged quantum mechanics (6.1.13) due to the reduction process.

Note that the original SO(8) R-symmetry is now enhanced to SO(16) in our quantum mechanics. It is not so strange as a similar phenomenon has been already observed in d = 11 supergravity. In d = 11 supergravity the original tangent space symmetry SO(1,10) can break down into the subgroup $SO(1,2) \times SO(8)$ through a partial choice of gauge for the elfbein. However, it has been pointed out in [285, 286, 287] that one can find the enhanced $SO(1,2) \times SO(16)$ tangent space symmetry by introducing new gauge degrees of freedom. It would be an intriguing open question to investigate whether such enhanced R-symmetry of our quantum mechanics reflects that of d = 11 supergravity.
Chapter 7

$\mathcal{N} = 12$ Superconformal Mechanics

Similar to the previous chapter, we will consider the ABJM-model wrapped on a torus and derive the IR quantum mechanics by shrinking the torus in this chapter. We will derive the IR $\mathcal{N} = 12$ superconformal gauged quantum mechanic-sãĂĂ and extract the corresponding SU(1,1|6) superconformal quantum mechanics from the reduced systems.

7.1 $\mathcal{N} = 12$ gauged quantum mechanics

We now want to consider an arbitrary number of M2-branes wrapped around a torus, which may be described by the $U(N)_k \times \hat{U}(N)_{-k}$ ABJM-model on $\mathbb{R} \times T^2$. The theory may describe the dynamics of *N* coincident M2-branes with the world-volume $M_3 = \mathbb{R} \times T^2$ moving in a transverse space with an SU(4) holonomy. The crucial point is now that the volume of the torus yields a typical energy scale in the theory and we can take a further limit where the energy is lower than the inverse of the size of the torus. Such low-effective theory describes the fluctuations around static BPS configurations obeying the BPS equations. From the supersymmetry transformations (5.1.7), (5.1.8) for fermions we find the following set of the BPS equations:

$$D_z Y^A = 0, \quad D_{\overline{z}} Y^A = 0,$$
 (7.1.1)

$$Y^{C}Y_{C}^{\dagger}Y^{B} - Y^{B}Y_{C}^{\dagger}Y^{C} = 0, (7.1.2)$$

$$Y^{C}Y_{A}^{\dagger}Y^{D} = 0. (7.1.3)$$

To satisfy the algebraic equations (7.1.2) and (7.1.3), the bosonic Higgs fields Y^A and Y^{\dagger}_A should take the diagonal form

$$Y^{A} = \operatorname{diag}(y_{1}^{A}, \cdots, y_{N}^{A}), \qquad Y_{A}^{\dagger} = \operatorname{diag}(\overline{y}_{A1}, \cdots, \overline{y}_{AN}) \qquad (7.1.4)$$

where y_a^A is a complex scalar field. For the above diagonal configurations, all the off-diagonal elements are massive and the gauge group $U(N) \times \hat{U}(N)$ is spontaneously broken to $U(1)^N$ [26]. Let us define

$$\mathcal{A}^{+}_{\mu a} := A_{\mu a a} + \hat{A}_{\mu a a}, \qquad \qquad \mathcal{A}^{-}_{\mu a} := A_{\mu a a} - \hat{A}_{\mu a a} \qquad (7.1.5)$$

where the indices $a = 1, \dots, N$ characterize the gauge degrees of freedom, i.e. the internal degrees of freedom of the multiple M2-branes. Note that all the couplings involve the gauge fields $\mathcal{A}_{\mu a}^{-}$ while the other gauge fields $\mathcal{A}_{\mu a}^{+}$ are associated with the preserved U(1) gauge group. In terms of the expressions (7.1.4) and (7.1.5), we can rewrite the equations (7.1.1) as

$$\partial_z y_a^A + i \mathcal{A}_{za}^- y_a^A = 0, \qquad \qquad \partial_z \overline{y}_{Aa} - i \mathcal{A}_{za}^- \overline{y}_{Aa} = 0, \qquad (7.1.6)$$

$$\partial_{\overline{z}}y_a^A + i\mathcal{A}_{\overline{z}a}^- y_a^A = 0, \qquad \qquad \partial_{\overline{z}}\overline{y}_{Aa} - i\mathcal{A}_{\overline{z}a}^- \overline{y}_{Aa} = 0, \qquad (7.1.7)$$

$$A_{zab} = \hat{A}_{zab} = A_{\bar{z}ab} = \hat{A}_{\bar{z}ab} = 0$$
 for $a \neq b$. (7.1.8)

The first and second lines correspond to the equations for diagonal elements and last one is for the off-diagonal elements. The general solutions to the equations (7.1.6) and (7.1.7) are given by

$$y_a^A = r_a^A e^{i\left(\varphi_a(z,\bar{z}) + \theta_a^A\right)},\tag{7.1.9}$$

$$\mathcal{A}_{za}^{-} = -\partial_z \varphi_a(z, \overline{z}) \tag{7.1.10}$$

where r_a^A , $\theta_a^A \in \mathbb{R}$ have no dependence on z and \overline{z} while $\varphi_a(z,\overline{z}) \in \mathbb{R}$ is a function of z and \overline{z} . The expression (7.1.10) ensures the flatness of the U(1) gauge field \mathcal{A}_z^- . Hence φ_a , \mathcal{A}_{za}^- and $\mathcal{A}_{\overline{z}a}^-$ take the form [284]

$$\varphi_a(z,\overline{z}) = -2\pi \frac{\Theta_a}{\tau - \overline{\tau}} z + 2\pi \frac{\overline{\Theta}_a}{\tau - \overline{\tau}} \overline{z}, \qquad (7.1.11)$$

$$\mathcal{A}_{za}^{-} = 2\pi \frac{\Theta_a}{\tau - \overline{\tau}} \omega, \qquad \mathcal{A}_{\overline{z}a}^{-} = -2\pi \frac{\Theta_a}{\tau - \overline{\tau}} \overline{\omega}.$$
 (7.1.12)

Here τ is the moduli of the torus defined in (6.1.1) and $\Theta_a := \zeta_a + \overline{\tau}\xi_a$, $a = 1, \dots, N$ are the coordinates of the product space of the *N* Jacobi varieties characterizing the *N U*(1) flat bundles. For the bosonic Higgs fields to describe the positions of the membranes, we should impose the single-valuedness of y_a^A as

$$y_{a}^{A}(z+1,\bar{z}+1) = y_{a}^{A}(z,\bar{z}),$$

$$y_{a}^{A}(z+\tau,\bar{z}+\bar{\tau}) = y_{a}^{A}(z,\bar{z}).$$
(7.1.13)

These conditions require that ξ_a and ζ_a are integers, which result in the quantization of the variables φ_a , \mathcal{A}_{za}^- and $\mathcal{A}_{\overline{z}a}^-$. Then the resulting static BPS configurations

are

$$Y^{A} = \operatorname{diag}(y_{1}^{A}, \cdots, y_{N}^{A}) = \operatorname{diag}\left(r_{1}^{A}e^{i(\varphi_{1}(z,\overline{z})+\theta_{1}^{A})}, \cdots, r_{N}^{A}e^{i(\varphi_{N}(z,\overline{z})+\theta_{N}^{A})}\right),$$

$$Y_{A}^{\dagger} = \operatorname{diag}(\overline{y}_{A1}, \cdots, \overline{y}_{AN}) = \operatorname{diag}\left(r_{1}^{A}e^{-i(\varphi_{1}(z,\overline{z})+\theta_{1}^{A})}, \cdots, r_{N}^{A}e^{-i(\varphi_{N}(z,\overline{z})+\theta_{N}^{A})}\right),$$

$$A_{z} = \operatorname{diag}\left(A_{z11}, \cdots, A_{zNN}\right),$$

$$\hat{A}_{z} = A_{z} + \partial_{z}\varphi = \operatorname{diag}\left(A_{z11} + \partial_{z}\varphi_{1}, \cdots, A_{zNN} + \partial_{z}\varphi_{N}\right).$$

(7.1.14)

By the supersymmetry the above bosonic configurations are paired with the fermionic fields

$$\psi_{\pm A} = \operatorname{diag}\left(\psi_{\pm A1}, \cdots, \psi_{\pm AN}\right), \qquad \psi_{\pm}^{\dagger A} = \operatorname{diag}\left(\psi_{\pm}^{\dagger A1}, \cdots, \psi_{\pm}^{\dagger AN}\right) \qquad (7.1.15)$$

where the subscripts \pm label the $SO(2)_E$ spinor representation.

Inserting the set of BPS configurations (7.1.14) and (7.1.15) into the ABJM action (5.1.1) one finds

$$S = \int_{\mathbb{R}} dt \int_{T^2} d^2 z \sum_{A} \sum_{a=1}^{N} \left[D_0 \overline{y}^a_A D_0 y^A_a - i \psi^{\dagger A a}_+ D_0 \psi_{+Aa} - i \psi^{\dagger A a}_- D_0 \psi_{-Aa} + \frac{k}{4\pi} \left(\mathcal{A}^-_{0a} \mathcal{F}^+_{z\overline{z}a} + \frac{1}{2} \mathcal{A}^-_{\overline{z}a} \dot{\mathcal{A}}^+_{za} - \frac{1}{2} \mathcal{A}^-_{za} \dot{\mathcal{A}}^+_{\overline{z}a} \right) \right].$$
(7.1.16)

Recall that \mathcal{A}_z^- and $\mathcal{A}_{\overline{z}}^-$ are quantized and their time derivative terms do not show up in the action. Thus we can treat them as auxiliary fields and integrate out them. Consequently we get constraints $\dot{\mathcal{A}}_{za}^+ = \dot{\mathcal{A}}_{\overline{z}a}^+ = 0$, which imply that the gauge fields \mathcal{A}_{za}^+ and $\mathcal{A}_{\overline{z}a}^+$ on the Riemann surface have no time dependence.

Taking these constraints into account and proceeding the integration over the torus, we obtain the low-energy effective action

$$S = \int_{\mathbb{R}} dt \left[D_0 \overline{y}^a_A D_0 y^A_a - i \psi^{\dagger \alpha A a} D_0 \psi_{\alpha A a} + k C_1(E_a) \mathcal{A}^-_{0a} \right].$$
(7.1.17)

Here the repeated indices are summed over and α , β , $\cdots = +, -$ denote the $SO(2)_E$ spinor indices. The covariant derivatives are defined by

$$D_0 y_a^A = \dot{y}_a^A + i \mathcal{A}_{0a}^- y_a^A, \qquad D_0 \overline{y}_{Aa} = \dot{\overline{y}}_{Aa} - i \mathcal{A}_{0a}^- \overline{y}_{Aa'}, D_0 \psi_{\alpha Aa} = \dot{\psi}_{\alpha Aa} + i \mathcal{A}_{0a}^- \psi_{\alpha Aa}, \qquad D_0 \psi_{\alpha a}^{\dagger A} = \dot{\psi}_{\alpha a}^{\dagger A} - i \mathcal{A}_{0a}^- \psi_{\alpha a}^{\dagger A}.$$
(7.1.18)

and

$$C_1(E_a) := \frac{1}{2\pi} \int_{T^2} F_{z\overline{z}aa} = \frac{1}{4\pi} \int_{T^2} \mathcal{F}_{z\overline{z}a}^+$$
(7.1.19)

is the Chern number of the *a*-th U(1) principal bundle $E_a \rightarrow T^2$ over the torus associated with the preserved U(1) gauge fields A_{zaa} .

The action (7.1.17) is invariant under the one-dimensional conformal transformations

$$\delta t = f(t) = a + bt + ct^2, \qquad \qquad \delta \partial_0 = -\dot{f} \partial_0, \qquad (7.1.20)$$

$$\delta y_a^A = \frac{1}{2} \dot{f} y_a^A, \qquad \qquad \delta \overline{y}_{Aa} = \frac{1}{2} \dot{f} \overline{y}_{Aa'} \tag{7.1.21}$$

$$\delta\psi_{\alpha Aa}=0, \qquad \qquad \delta\psi_{\alpha a}^{\dagger A}=0, \qquad (7.1.22)$$

$$\delta \mathcal{A}_{0a}^{-} = -\dot{f} \mathcal{A}_{0a}^{-} \tag{7.1.23}$$

and $\mathcal{N} = 12$ supersymmetry transformations

$$\delta y_a^A = i\omega^{\alpha AB}\psi_{\alpha Ba}, \qquad \qquad \delta \overline{y}_{Aa} = i\psi_a^{\dagger\alpha B}\omega_{\alpha AB}, \qquad (7.1.24)$$

$$\delta\psi_{\alpha Aa} = \omega_{\alpha AB} D_0 y_a^B, \qquad \qquad \delta\psi_{\alpha a}^{\dagger A} = -D_0 \overline{y}_{Ba} \omega_{\alpha}^{AB}, \qquad (7.1.25)$$

$$\delta \mathcal{A}_{0a}^{-} = 0 \tag{7.1.26}$$

where the supersymmetry parameters $\omega_{+AB} := \epsilon_{+i}(\Gamma^i)_{AB}$ and $\omega_{-AB} := \epsilon_{-i}(\Gamma^i)_{AB}$ transform as $\mathbf{6}_+$ and $\mathbf{6}_-$ under $SU(4) \times SO(2)_E$ respectively. Therefore the lowenergy effective theory is described by the $\mathcal{N} = 12$ superconformal gauged quantum mechanics (7.1.17).

7.2 Reduction

The low-energy effective action (7.1.17) is quadratic in \mathcal{A}_{0a}^- and contains no time derivatives of \mathcal{A}_{0a}^- . So they are auxiliary fields and we want to integrate them out. Let us fix the gauge as $\mathcal{A}_{0a}^- = 0$. Then the algebraic equations of motion of \mathcal{A}_{0a}^- yield the Gauss law constraints, the moment map conditions

$$\phi_{0a} := kC_1(E_a) + 2\sum_A (r_a^A)^2 \dot{\theta}_a^A + \sum_A \psi^{\dagger \alpha Aa} \psi_{\alpha Aa} = 0$$
(7.2.1)

for $a = 1, \dots, N$. Note that although the set of equations (7.2.1) has the same form as that of (6.2.3), the physical meaning of these constraints are different because the angular variable $\theta_a^{A'}$ s are defined not in the abstract space of the internal degrees of freedom as in (6.2.3), but in the actual configuration space of the *a*-th M2-brane in the *A*-th complex plane.

Defining the conserved charges $h_a^A := 2(r_a^A)^2 \dot{\theta}_a^A$, using the above constraints (7.2.1) and following the reduction procedure as in the derivation of (6.2.7), we can

integrate out the auxiliary gauge fields \mathcal{A}_{0a}^{-} and find the reduced effective action with the inverse-square type interaction

$$S = \int_{\mathbb{R}} dt \sum_{a=1}^{N} \left[\dot{x}_{a}^{2} - \frac{i}{2} \sum_{A \neq B} \left(\psi^{\dagger \alpha A a} \dot{\psi}_{\alpha A a} - \dot{\psi}^{\dagger A a} \psi_{\alpha A a} \right) \right. \\ \left. + \sum_{A \neq B} (\dot{r}_{a}^{A})^{2} - \frac{i}{2} \left(\lambda^{\dagger \alpha a} \dot{\lambda}_{\alpha a} - \dot{\lambda}^{\dagger \alpha a} \lambda_{\alpha a} \right) \right. \\ \left. - \frac{\left[kC_{1}(E_{a}) + \sum_{A \neq B} h_{a}^{A} + \sum_{A \neq B} \psi^{\dagger \alpha A a} \psi_{\alpha A a} + \lambda^{\dagger \alpha a} \lambda_{\alpha a} \right]^{2}}{4x_{a}^{2}} - \sum_{A \neq B} \frac{\left(h_{a}^{A} \right)^{2}}{4(r_{a}^{A})^{2}} \right].$$

$$(7.2.2)$$

Here $x_a := r_a^B$ describes the motion of the *a*-th M2-brane in the *B*-th complex plane in which the corresponding "angular momentum" h_a^B is determined by the assignment of the other preserved charges. We have also introduced the fermionic variable $\lambda_{\alpha a} := \psi_{\alpha Ba}$ with A = B, which turns out to be the superpartner of r_a^C , C = 1, 2, 3, as we will see the supersymmetry transformations (7.2.27) and (7.2.28).

The action (7.2.2) leads to the following equations of motion

$$\ddot{x}_{a} = \frac{\left[kC_{1}(E_{a}) + \sum_{A \neq B} h_{a}^{A} + \sum_{A \neq B} \psi^{\dagger \alpha A a} \psi_{\alpha A a} + \lambda^{\dagger \alpha a} \lambda_{\alpha a}\right]^{2}}{4x_{a}^{3}}, \qquad (7.2.3)$$

$$\ddot{r}_a^A = \frac{(h_a^A)^2}{4(r_a^A)^3},\tag{7.2.4}$$

$$\dot{\psi}_{\alpha Aa} = i \frac{kC_1(E_a) + \sum_{A \neq B} h_a^A + \sum_{A \neq B} \psi^{\dagger \alpha Aa} \psi_{\alpha Aa} + \lambda^{\dagger \alpha a} \lambda_{\alpha a}}{2x_a} \psi_{\alpha Aa}, \qquad (7.2.5)$$

$$\dot{\psi}^{\dagger \alpha Aa} = -i \frac{kC_1(E_a) + \sum_{A \neq B} h_a^A + \sum_{A \neq B} \psi^{\dagger \alpha Aa} \psi_{\alpha Aa} + \lambda^{\dagger \alpha a} \lambda_{\alpha a}}{2x_a} \psi^{\dagger \alpha Aa}, \qquad (7.2.6)$$

$$\dot{\lambda}_{\alpha a} = i \frac{kC_1(E_a) + \sum_{A \neq B} h_a^A + \sum_{A \neq B} \psi^{\dagger \alpha A a} \psi_{\alpha A a} + \lambda^{\dagger \alpha a} \lambda_{\alpha a}}{2x_a} \lambda_{\alpha a}, \qquad (7.2.7)$$

$$\dot{\lambda}^{\dagger \alpha a} = -i \frac{kC_1(E_a) + \sum_{A \neq B} h_a^A + \sum_{A \neq B} \psi^{\dagger \alpha A a} \psi_{\alpha A a} + \lambda^{\dagger \alpha a} \lambda_{\alpha a}}{2x_a} \lambda^{\dagger \alpha a}.$$
 (7.2.8)

Using the fermionic equations of motion (7.2.5)-(7.2.8), we can check that the Gauss law constraint (7.2.1) has no time dependence, i.e. $\dot{\phi}_{0a} = 0$.

The canonical momenta are given by

$$p^{a} := \frac{\partial L}{\partial \dot{x}_{a}} = 2\dot{x}^{a}, \qquad P^{a}_{A} := \frac{\partial L}{\partial \dot{r}^{A}_{a}} = 2\dot{r}^{a}_{A}, \qquad (7.2.9)$$

$$\pi^{\alpha Aa} := \frac{\vec{\partial}L}{\partial \dot{\psi}_{\alpha Aa}} = \frac{i}{2} \psi^{\dagger \alpha Aa}, \qquad \qquad \tilde{\pi}_{\alpha Aa} := \frac{\vec{\partial}L}{\partial \dot{\psi}^{\dagger \alpha Aa}} = \frac{i}{2} \psi_{\alpha Aa}, \qquad (7.2.10)$$

$$\Pi^{\alpha a} := \frac{\vec{\partial}L}{\partial \dot{\lambda}_{\alpha a}} = \frac{i}{2} \lambda^{\dagger \alpha a}, \qquad \qquad \tilde{\Pi}_{\alpha a} := \frac{\vec{\partial}L}{\partial \dot{\lambda}^{\dagger \alpha a}} = \frac{i}{2} \lambda_{\alpha a}. \qquad (7.2.11)$$

The fermionic canonical momenta provide the second class constraints

$$\phi_1^{\alpha Aa} := \pi^{\alpha Aa} - \frac{i}{2} \psi^{\dagger Aa} = 0, \qquad \phi_{2\alpha Aa} := \tilde{\pi}_{\alpha Aa} - \frac{i}{2} \psi_{\alpha Aa} = 0, \qquad (7.2.12)$$

$$\phi_3^{\alpha a} := \Pi^{\alpha a} - \frac{i}{2} \lambda^{\dagger \alpha a} = 0, \qquad \phi_{4\alpha a} := \tilde{\Pi}_{\alpha a} - \frac{i}{2} \lambda_{\alpha a} = 0.$$
 (7.2.13)

Taking account into the constraints (7.2.12) and (7.2.13), we find the Dirac brackets

$$[x_a, p^b]_{DB} = \delta_{ab}, \qquad [r_a^A, P_B^b]_{DB} = \delta_{AB}\delta_{ab}, \qquad (7.2.14)$$

$$\left[\psi_{\alpha Aa},\psi^{\dagger\beta Bb}\right]_{DB} = i\delta_{\alpha\beta}\delta_{AB}\delta_{ab}, \qquad \left[\lambda_{\alpha a},\lambda^{\dagger\beta b}\right]_{DB} = i\delta_{\alpha\beta}\delta_{ab}. \tag{7.2.15}$$

The action (7.2.2) possesses the one-dimensional conformal invariance

$$\delta t = f(t) = a + bt + ct^2, \qquad \qquad \delta \partial_0 = -\dot{f}\partial_0, \qquad (7.2.16)$$

$$\delta x_a = \frac{1}{2}\dot{f}x_a, \qquad \qquad \delta r_a^A = \frac{1}{2}\dot{f}r_a^A, \qquad (7.2.17)$$

$$\delta\psi_{\alpha Aa} = 0, \qquad \qquad \delta\psi_a^{\dagger\alpha A} = 0, \qquad (7.2.18)$$

$$\delta\lambda_{\alpha a} = 0, \qquad \qquad \delta\lambda_a^{\dagger \alpha} = 0. \tag{7.2.19}$$

Using the Noether's procedure we find the $SL(2, \mathbb{R})$ generators

$$H = \sum_{a=1}^{N} \left[\frac{p_a^2}{4} + \frac{\left(kC_1(E_a) + \sum_{A \neq B} h_a^A + \sum_A \psi^{\dagger \alpha A a} \psi_{\alpha A a} + \lambda^{\dagger \alpha a} \lambda_{\alpha a}\right)^2}{4x_a^2} + \sum_{A \neq B} \left(\frac{(P_a^A)^2}{4} + \frac{(h_a^A)^2}{4(r_a^A)^2} \right) \right],$$
(7.2.20)

$$D = -\frac{1}{4} \sum_{a=1}^{N} \left[\{x_a, p_a\} + \sum_{A \neq B} \{r_a^A, P_a^A\} \right],$$
(7.2.21)

$$K = \sum_{a=1}^{N} \left[x_a^2 + \sum_{A \neq B} (r_a^A)^2 \right].$$
 (7.2.22)

Note that we have absorbed the time dependent part of D and K by using the similarity transformations (2.1.34).

Also the action (7.2.2) is invariant under the following fermionic transforma-

tions

$$\delta x_{a} = \frac{i}{\sqrt{2}} \left(\epsilon^{\alpha C} \psi_{\alpha Ca} + \epsilon^{\dagger}_{\alpha C} \psi^{\dagger \alpha C}_{a} \right), \qquad (7.2.23)$$

$$\delta r_{a}^{C} = \frac{i}{2} \left[\left(\omega^{\alpha CD} \psi_{\alpha Da} \right) e^{-i\theta_{a}^{C}} + \left(\psi^{\dagger \alpha D}_{a} \omega_{\alpha CD} \right) e^{i\theta_{a}^{C}} - \left(\epsilon^{\alpha C} \lambda_{\alpha a} \right) e^{-i\theta_{a}^{C}} - \left(\epsilon^{\dagger}_{\alpha C} \lambda^{\dagger \alpha}_{a} \right) e^{i\theta_{a}^{C}} \right], \qquad (7.2.24)$$

$$\begin{split} \delta\psi_{\alpha Ca} &= \left(\dot{r}_{a}^{D} + i\frac{h_{a}^{D}}{2r_{a}^{D}}\right)e^{i\theta_{a}^{D}}\omega_{\alpha CD} \\ &+ \sqrt{2}\left(\dot{x}_{a} - i\frac{kC_{1}(E_{a}) + \sum_{D\neq B}h_{a}^{D} + \psi^{\dagger\alpha Da}\psi_{\alpha Da} + \lambda^{\dagger\alpha a}\lambda_{\alpha a}}{2x_{a}}\right)\epsilon_{\alpha C}^{\dagger} - \frac{i}{\sqrt{2}}\frac{l_{a}}{x_{a}}\psi_{\alpha Ca}, \end{split}$$
(7.2.25)

$$\begin{split} \delta\psi_{a}^{\dagger\alpha C} &= -\left(\dot{r}_{a}^{D} - i\frac{h_{a}^{D}}{2r_{a}^{D}}\right)e^{-i\theta_{a}^{D}}\omega_{\alpha}^{CD} \\ &+ \sqrt{2}\left(\dot{x}_{a} + i\frac{kC_{1}(E_{a}) + \sum_{D\neq B}h_{a}^{D} + \psi^{\dagger\alpha Da}\psi_{\alpha Da} + \lambda^{\dagger\alpha a}\lambda_{\alpha a}}{2x_{a}}\right)e^{\alpha C} + \frac{i}{\sqrt{2}}\frac{l_{a}}{x_{a}}\psi_{a}^{\dagger\alpha C}, \end{split}$$
(7.2.26)

$$\delta\lambda_{\alpha a} = -\epsilon_{\alpha C}^{\dagger} \left(\dot{r}_{a}^{C} + i\frac{h_{a}^{C}}{2r_{a}^{C}} \right) e^{i\theta_{a}^{C}}, \tag{7.2.27}$$

$$\delta\lambda_a^{\dagger\alpha} = -\left(\dot{r}_a^C - i\frac{h_a^C}{2r_a^C}\right)e^{-i\theta_a^C}\epsilon^{\alpha C}$$
(7.2.28)

with C, D = 1, 2, 3. Here $\epsilon^{\alpha C}$ and their Hermitian conjugate $\epsilon^{\dagger}_{\alpha C}$ are infinitesimal fermionic parameters and we have defined

$$\theta_a^{\rm C}(t) = h_a^{\rm C} \int^t \frac{dt'}{(r_a^{\rm C}(t'))^2},$$
(7.2.29)

$$l_a = \epsilon \psi_a - \epsilon^{\dagger} \psi_a^{\dagger}. \tag{7.2.30}$$

7.3 SU(1,1|6) superconformal mechanics

Since the non-local quantities are included in the fermionic transformations (7.2.23)-(7.2.28), there may exist infinitely many conserved non-local charges. However, we see from (7.2.20) that the Hamiltonian describing the motion in the *B*-th complex plane associated with the variable x_a and the local charges commute with the others associated with the variables $r_a^{C'}$ s and the non-local charges. Therefore they are decoupled with one another and we thus can analyze the dynamics in the *B*-th direction separately. As in the subsection 6.3, it is convenient to assign the conserved charges h_a^A and $\lambda^{\dagger \alpha a} \lambda_{\alpha a}$ to be zeros. Then the low-energy dynamics in the *B*-th complex plane is described by the action

$$S = \int_{\mathbb{R}} dt \sum_{a=1}^{N} \left[\dot{x}_{a}^{2} - i\psi^{\dagger \alpha Aa} \dot{\psi}_{\alpha Aa} - \frac{\left(kC_{1}(E_{a}) + \psi^{\dagger \alpha Aa} \psi_{\alpha Aa} \right)^{2}}{4x_{a}^{2}} \right]$$
(7.3.1)

where A = 1, 2, 3 denote the R-symmetry indices. Note that the action (7.3.1) has the same structure as (3.6.1) [210, 224, 137] for $SU(1, 1|\frac{N}{2})$, N > 4 superconformal quantum mechanics.

The action (7.3.1) has the invariance under the $\mathcal{N}=12$ supersymmetry transformation laws

$$\delta x_a = \frac{i}{\sqrt{2}} \left(\epsilon^{\alpha A} \psi^{\alpha}_{Aa} + \epsilon^{\dagger}_{\alpha A} \psi^{\dagger \alpha A}_{a} \right), \qquad (7.3.2)$$

$$\delta\psi_{\alpha Aa} = \sqrt{2} \left(\dot{x}_a - i \frac{g_a}{2x_a} \right) \epsilon^{\dagger}_{\alpha A} - \frac{i}{\sqrt{2}} \frac{l_a}{x_a} \psi_{\alpha Aa}, \qquad (7.3.3)$$

$$\delta \psi_a^{\dagger \alpha A} = \sqrt{2} \left(\dot{x}_a + i \frac{g_a}{2x_a} \right) \epsilon^{\alpha A} + \frac{i}{\sqrt{2}} \frac{l_a}{x_a} \psi_a^{\dagger \alpha A}$$
(7.3.4)

where

$$g_a = kC_1(E_a) + \psi^{\dagger \alpha A a} \psi_{\alpha A a}.$$
(7.3.5)

The supersymmetry transformations (7.3.2)-(7.3.4) are generated by the supercharges

$$Q_{\alpha A} = \frac{i}{\sqrt{2}} \left(p^a - \frac{g_a}{x_a} \right) \psi_{\alpha A a}, \tag{7.3.6}$$

$$\tilde{Q}^{\alpha A} = \frac{i}{\sqrt{2}} \left(p^a + \frac{g_a}{x_a} \right) \psi^{\dagger \alpha A}.$$
(7.3.7)

Also the action (7.3.1) has the one-dimensional conformal invariance. The corresponding Noether charges are now expressed as

$$H = \sum_{a=1}^{N} \left[\frac{p_a^2}{4} + \frac{\left(kC_1(E_a) + \psi^{\dagger \alpha A a} \psi_{\alpha A a} \right)^2}{4x_a^2} \right],$$
(7.3.8)

$$D = -\frac{1}{4} \sum_{a=1}^{N} \{x_a, p^a\}, \qquad (7.3.9)$$

$$K = \sum_{a=1}^{N} x_a^2.$$
(7.3.10)

According to the Dirac brackets (7.2.14) and (7.2.15), quantum operators of the canonical coordinates and momenta obey the quantum brackets

$$[x_a, p^b] = i\delta_{ab}, \qquad \left\{\psi_{\alpha Aa}, \psi^{\dagger\beta Bb}\right\} = -\delta_{\alpha\beta}\delta_{AB}\delta_{ab}. \tag{7.3.11}$$

Combining the supercharges and the conformal generators, we find the superconformal boost generators

$$S_{\alpha A} = \sqrt{2}i \sum_{a} x_a \psi_{\alpha A a}, \qquad (7.3.12)$$

$$\tilde{S}^{\alpha A} = \sqrt{2}i \sum_{a} x_a \psi_a^{\dagger \alpha A}.$$
(7.3.13)

The R-symmetry generator is given by

$$(J_{\alpha\beta})_{AB} = i \sum_{a} \psi_a^{\dagger\beta B} \psi_{\alpha Aa}.$$
(7.3.14)

Note that (7.3.14) is a complex 6×6 matrix with α , $\beta = +, -$ and A, B = 1, 2, 3 and it contains 36 complex valued elements.

Under the canonical relations (7.3.11) and the Weyl ordering ¹, the generators form the following algebra

$$[H,D] = iH, \quad [K,D] = -iK, \quad [H,K] = 2iD,$$
 (7.3.15)

$$[(J_{\alpha\beta})_{AB}, H] = 0, \quad [(J_{\alpha\beta})_{AB}, D] = 0, \quad [(J_{\alpha\beta})_{AB}, K] = 0, \quad (7.3.16)$$

$$[(J_{\alpha\beta})_{AB}, (J_{\gamma\delta})_{CD}] = i(J_{\alpha\delta})_{AD}\delta_{\beta\gamma}\delta_{BC} - i(J_{\gamma\beta})_{CB}\delta_{\alpha\delta}\delta_{AD},$$
(7.3.17)

$$[H, Q_{\alpha A}] = 0, \quad [D, Q_{\alpha A}] = -\frac{i}{2}Q_{\alpha A}, \quad [K, Q_{\alpha A}] = iS_{\alpha A}, [H, \tilde{Q}^{\alpha A}] = 0, \quad [D, \tilde{Q}^{\alpha A}] = -\frac{i}{2}\tilde{Q}^{\alpha A}, \quad [K, \tilde{Q}^{\alpha A}] = i\tilde{S}^{\alpha A},$$
 (7.3.18)

$$[H, S_{\alpha A}] = -iQ_{\alpha A}, \quad [D, S_{\alpha A}] = \frac{i}{2}S_{\alpha A}, \quad [K, S_{\alpha A}] = 0, [H, \tilde{S}^{\alpha A}] = -i\tilde{Q}^{\alpha A}, \quad [D, \tilde{S}^{\alpha A}] = \frac{i}{2}\tilde{S}^{\alpha A}, \quad [K, \tilde{S}^{\alpha A}] = 0,$$

$$(7.3.19)$$

$$\{Q_{\alpha A}, \tilde{Q}^{\beta B}\} = 2H\delta_{\alpha\beta}\delta_{AB},$$

$$\{S_{\alpha A}, \tilde{S}^{\beta B}\} = 2K\delta_{\alpha\beta}\delta_{AB},$$

$$\{Q_{\alpha A}, \tilde{S}^{\beta B}\} = -2D\delta_{\alpha\beta}\delta_{AB} - 2(J_{\alpha\beta})_{AB},$$

$$\{\tilde{Q}^{\alpha A}, S_{\beta B}\} = -2D\delta_{\alpha\beta}\delta_{AB} - 2(J_{\alpha\beta}^{\dagger})_{AB},$$

(7.3.20)

$$[(J_{\alpha\beta})_{AB}, Q_{\gamma C}] = i Q_{\alpha A} \delta_{\beta\gamma} \delta_{BC}, \qquad [(J_{\alpha\beta})_{AB}, S_{\gamma,C}] = i S_{\alpha A} \delta_{\beta\gamma} \delta_{BC}, [(J_{\alpha\beta})_{AB}, \tilde{Q}^{\gamma C}] = -i \tilde{Q}^{\alpha A} \delta_{\beta\gamma} \delta_{BC}, \qquad [(J_{\alpha\beta})_{AB}, \tilde{S}^{\gamma,C}] = -i \tilde{S}^{\alpha A} \delta_{\beta\gamma} \delta_{BC}.$$
(7.3.21)

¹One needs to pick up constant shifts as a quantum effect.

The Hamiltonian H, the dilatation generator D and the conformal boost generator form the one-dimensional conformal algebra $\mathfrak{so}(1,2) = \mathfrak{sl}(2,\mathbb{R}) = \mathfrak{su}(1,1)$. As each of the supercharges $Q_{\alpha A}$ and $\tilde{Q}^{\alpha A} = -(Q_{\alpha A})^{\dagger}$ contain six real components, there exist twelve supercharges. They are the square roots of the Hamiltonian H. In addition, there are as many superconformal charges $S_{\alpha A}$ and $\tilde{S}^{\alpha A}$, which are the square roots of the conformal boost generator K. The anti-commutators of the fermionic charges generate an extra bosonic R-symmetry generators $(J_{\alpha\beta})_{AB}$. They form the $\mathfrak{u}(6)$ algebra (7.3.17). Thus the action (7.3.1) describes the SU(1,1|6) invariant $\mathcal{N} = 12$ superconformal mechanics. In fact this belongs to the list of the simple supergroup for superconformal quantum mechanics, which we have shown in Table 3.2.

Following the AdS_2/CFT_1 correspondence, we expect that the superconformal quantum mechanical models (6.1.13) and (7.1.17) may be related to $AdS_2 \times T^2$ solutions, the so-called magnetic brane solutions [288, 289]. It may be interesting to check those correspondences.

Chapter 8

Curved Branes and Topological Twisting

In this chapter we will investigate the topological twisting and its relevant application as the world-volume description of curved branes in string theory and M-theory, which was firstly pointed out in [28]. In section 8.1 we will discuss various topological twisting procedures. In section 8.2 we will explain that the topologically twisted theories may yield the world-volume theories of the curved branes.

8.1 Topological twisting

Topological twisting is a modification of the Euclidean rotational group of a supersymmetric theory through an embedding into a global symmetry of the theory. The resulting theory will be topological if the twisted supersymmetry generators include at least one space-time scalar. Equivalently one can regard the twisting procedure as a gauging of an internal symmetry group in which a global symmetry is promoted to a space-time symmetry. In many cases, gauging can be performed by coupling of the internal symmetry current to the spin connection of the underlying manifold to the Lagrangian. We will give many examples of the topological twisting in the following.

8.1.1 d = 4, $\mathcal{N} = 2$ **SYM** theories

Let us consider topological twisting of d = 4, $\mathcal{N} = 2$ super Yang-Mills (SYM) theories [290]. We take $M_4 = \mathbb{R}^4$ whose rotational symmetry group is $Spin(4)_E \cong$

 $SU(2)_l \times SU(2)_r$. The global symmetry of the theory is $U(2)_R \simeq SU(2) \times U(1)$ R-symmetry. The field content is

- complex scalar field ϕ
- 2 complex fermionic fields λ^i_{α} , $\overline{\lambda}_{\dot{\alpha}\dot{i}}$
- gauge field $A_{\alpha\dot{\alpha}}$

where α are indices of the fundamental representation of $SU(2)_l$ and $\dot{\alpha}$ are indices of the fundamental representation of $SU(2)_r$. *i* denotes the fundamental representation of $SU(2)_R$. These indices are raised and lowered with the antisymmetric tensor $\epsilon_{\alpha\beta}$, $\epsilon_{\dot{\alpha}\dot{\beta}}$, ϵ_{ij} such that $\epsilon_{12} = \epsilon^{12} = 1$. All fields are the adjoint representation of compact group *G*. The scaling dimensions are

$$[\phi] = 1, \quad [\psi] = [\lambda] = \frac{3}{2}, \quad [A] = 1, \quad [\epsilon] = -\frac{1}{2}$$
 (8.1.1)

where ϵ is a supersymmetry parameter.

The supersymmetry transformations are

$$\delta A_{\mu} = -i\overline{\lambda}_{i}^{\dot{\alpha}}\sigma_{\mu\alpha\dot{\alpha}}\epsilon^{\alpha i} + i\overline{\epsilon}_{i}^{\dot{\alpha}}\sigma_{\mu\alpha\dot{\alpha}}\lambda^{\alpha i}, \qquad (8.1.2)$$

$$\delta\lambda^{i}_{\alpha} = \sigma^{\mu\nu}_{\alpha\beta}\epsilon^{\beta i}F_{\mu\nu} + i\epsilon^{i}_{\alpha}[\phi,\overline{\phi}] + i\sqrt{2}\sigma^{\mu}_{\alpha\dot{\alpha}}D_{\mu}\phi\epsilon^{ij}\overline{\epsilon}^{\dot{\alpha}}_{j}, \qquad (8.1.3)$$

$$\delta\overline{\lambda}_{\dot{\alpha}i} = \sigma^{\mu\nu}_{\dot{\alpha}\dot{\beta}}\overline{\epsilon}^{\dot{\beta}}_{i}F_{\mu\nu} - i\overline{\epsilon}_{\dot{\alpha}i}[\phi,\overline{\phi}] + i\sqrt{2}D_{\mu}\sigma^{\mu}_{\dot{\alpha}\alpha}\overline{\phi}\epsilon_{ij}\epsilon^{\alpha j}, \qquad (8.1.4)$$

$$\delta\phi = \sqrt{2}\epsilon^{\alpha i}\lambda_{\alpha i},\tag{8.1.5}$$

$$\delta \overline{\phi} = \sqrt{2} \overline{\epsilon}^{\dot{\alpha}}_{i} \overline{\lambda}^{i}_{\dot{\alpha}} \tag{8.1.6}$$

where ϵ_{α}^{i} and $\overline{\epsilon}_{\dot{\alpha}i}$ are supersymmetry parameters that transform as (2, 1, 2) and (1, 2, 2) respectively.

The Lorentzian action is given by

$$\mathcal{L} = \frac{1}{e^2} \int_M d^4 x \operatorname{Tr} \left(-\frac{1}{4} F_{\mu\nu} F^{\mu\nu} - i\overline{\lambda}_i^{\dot{\alpha}} \sigma_{\alpha\dot{\alpha}}^{\mu\nu} D_\mu \lambda^{\alpha i} - D_\mu \overline{\phi} D^\mu \phi - \frac{1}{2} [\overline{\phi}, \phi]^2 - \frac{1}{\sqrt{2}} \overline{\phi} \epsilon_{ij} [\lambda^{\alpha i}, \lambda_\alpha^j] + \frac{i}{\sqrt{2}} \phi \epsilon^{ij} [\overline{\lambda}_{\dot{\alpha} i}, \overline{\lambda}_i^{\dot{\alpha}}] \right).$$
(8.1.7)

Here Tr is an invariant quadratic form on the Lie algebra.

The classical $\mathcal{N} = 2$ theory has a U(2) symmetry acting on the two fermion $(\lambda, \overline{\lambda})$. The center $U(1)_R \subset U(2)$ is anomalous. On a given 4-manifold M_4 and for

a given instanton number*k*, the total violation ΔU of the $U(1)_R$ charge is given by the dimension of the Yang-Mills instanton moduli space [290]. For SU(2) this is

$$\Delta U = \dim \mathcal{M} = 8k - \frac{3}{2}(\chi + \sigma) \tag{8.1.8}$$

where χ and σ are the Euler characteristic and signature of M_4^1 . This was first discussed in [290].

The fields and supersymmetry parameters transform under $SO(4)_E \times U(2)_R \simeq$ $SU(2)_l \times SU(2)_r \times SU(2)_R \times U(1)_R$ as

$$\phi:(1,1,1)_2 \oplus (1,1,1)_{-2} \tag{8.1.9}$$

$$\psi, \lambda: (2, 1, 2)_1 \oplus (1, 2, 2)_{-1}$$
 (8.1.10)

$$A_{\mu}:(\mathbf{2},\mathbf{2},\mathbf{1})_{0} \tag{8.1.11}$$

$$\epsilon: (\mathbf{2}, \mathbf{1}, \mathbf{2})_1 \oplus (\mathbf{1}, \mathbf{2}, \mathbf{2})_{-1}.$$
 (8.1.12)

To perform the topological twisting, we leave $SU(2)_l$ undisturbed and pick a homomorphism

$$\pi: SU(2)_r \to SU(2)_R,$$
 (8.1.13)

and replace $SU(2)_r$ by a diagonal subgroup $SU(2)'_r = (1 + \pi)(SU(2)) \subset SU(2)_r \times SU(2)_R$. Then under the new rotational symmetry $SO(4)'_E \simeq SU(2)_l \times SU(2)'_r$, the fields and supersymmetry parameters transform as

$$\phi \to (\mathbf{1}, \mathbf{1})_2 \oplus (\mathbf{1}, \mathbf{1})_{-2}$$
 (8.1.14)

$$\psi, \lambda \to (2, 2)_1 \oplus (1, 1)_{-1} \oplus (1, 3)_{-1}$$
 (8.1.15)

$$A_{\mu} \rightarrow (\mathbf{2}, \mathbf{2})_0 \tag{8.1.16}$$

$$\epsilon \to (2,2)_1 \oplus (1,1)_{-1} \oplus (1,3)_{-1}.$$
 (8.1.17)

Thus the bosonic field content is

- complex scalar field ϕ : $(\mathbf{1}, \mathbf{1})_2 \oplus (\mathbf{1}, \mathbf{1})_{-2}$
- gauge field *A*_µ: (**2**, **2**)₀

and the fermionic field content is

- scalar field η: (1, 1)₋₁
- 1-form ψ_{μ} : (2, 2)₁
- 2-form (self-dual antisymmetric 2-tensor) $\chi^+_{\mu\nu}$: (1,3)₋₁.

¹The quantity $\frac{\chi+\sigma}{2}$ is always integer.

From (8.1.17), one can see that there exists one BRST charge.

In d = 4, $\mathcal{N} = 2$ SYM theories, the possible anomalies are related to the global SU(2) anomaly [291], which only appear when the corresponding moduli space is not orientable [290]. In Donaldson-Witten theory the moduli space is given by anti-self-dual connections, which is orientable [292]. Thus the twisted theory is anomaly free.

The twisted Lagrangian is

$$\mathcal{L} = \operatorname{Tr}\left(\frac{1}{4}F_{\alpha\beta}F^{\alpha\beta} - \frac{1}{2}D_{\alpha}\phi D^{\alpha}\sigma - i\eta D_{\alpha}\psi^{\alpha} + iD_{\alpha}\psi_{\beta} \cdot \chi^{\alpha\beta} - \frac{i}{8}\phi[\chi_{\alpha\beta},\chi^{\alpha\beta}] - \frac{i}{2}\sigma[\psi_{\alpha},\psi^{\alpha}] - \frac{i}{2}\phi[\eta,\eta] - \frac{1}{8}[\phi,\sigma]^{2}\right).$$
(8.1.18)

For the closure of supersymmetry algebra, it it necessary to introduce an auxiliary field $T_{ij} = T_{ji}$. It has scaling dimension [T] = 2 and transform $(\mathbf{1}, \mathbf{1}, \mathbf{3})_0$ under $SU(2)_l \times SU(2)_r \times SU(2)_R \times U(1)_R$. After twisting they transform $(\mathbf{1}, \mathbf{3})_0$ and identified with a 2-form.

Twisted $\mathcal{N} = 2$ supersymmetric gauge theories have an off-shell formulation such that the action can be expressed as a *Q*-exact expression up to a θ -term², where *Q* is the BRST charge.

8.1.2 d = 4, $\mathcal{N} = 2$ SCFT on $C \times \Sigma$

We now consider a four-dimensional $\mathcal{N} = 2$ superconformal field theory (SCFT) on $M_4 = C \times \Sigma$ whose holonomy group is reduced to $U(1)_C \times U(1)_{\Sigma}$, where *C* and Σ are Riemann surfaces. This has been discussed in [293]. The global symmetry of the theory is $SU(2)_R \times U(1)_R$ R-symmetry and $U(1)_B$ symmetry. The field content is

- complex scalar field in the adjoint representation: $\varphi, \overline{\varphi}$
- 2 complex scalar fields in representation R, R^{\vee} of G (squarks): q, \tilde{q}
- gauge field A_{μ}
- 2 gauginos: ψ , λ
- 2 complex left-handed quarks in representation R^{\vee} , R of $G: \psi_q, \psi_{\tilde{q}}$

² Because of the chiral anomaly inherent to the R-symmetry of $\mathcal{N} = 2$ SYM theories, observables are independent of θ -term up to rescaling. Thus one can ignore θ -term.

	$U(1)_R^\prime$
φ	0
$\overline{\varphi}$	0
ψ	+
λ	—
q	+
q	—
ψ_q	0
$\psi_{ ilde q}$	0

Table 8.1: The $U(1)'_R \subset SU(2)_R$ charge assignments for d = 4, $\mathcal{N} = 2$ SCFT field content.

• 2 complex right-handed quarks in representation R, R^{\vee} of $G: \overline{\psi_q}, \overline{\psi_{\tilde{q}}}$.

Before topological twisting, fields transform under $U(1)_C \times U(1)_\Sigma \times SU(2)_R \times U(1)_R \times U(1)_B$ as

$$\varphi, \overline{\varphi}: \mathbf{1}_{0020} \oplus \mathbf{1}_{00-20} \tag{8.1.19}$$

$$q, \tilde{q}: \mathbf{2}_{000-} \oplus \mathbf{2}_{000+} \tag{8.1.20}$$

$$A_{\mu}: \mathbf{1}_{2000} \oplus \mathbf{1}_{-2000} \oplus \mathbf{1}_{0200} \oplus \mathbf{1}_{0-200}$$
(8.1.21)

$$\psi, \lambda : \mathbf{2}_{+-+0} \oplus \mathbf{2}_{-++0}$$
 (8.1.22)

$$\overline{\psi}, \overline{\lambda}: \mathbf{2}_{--0} \oplus \mathbf{2}_{++-0} \tag{8.1.23}$$

$$\psi_{q}, \psi_{\tilde{q}}: \mathbf{1}_{+---} \oplus \mathbf{1}_{+--+} \oplus \mathbf{1}_{-+--} \oplus \mathbf{1}_{-+-+}$$
(8.1.24)

$$\overline{\psi}_{q'}\overline{\psi}_{\tilde{q}}:\mathbf{1}_{--++}\oplus\mathbf{1}_{++++}\oplus\mathbf{1}_{--+-}\oplus\mathbf{1}_{+++-}$$
(8.1.25)

. To perform the topological twisting, we pick a homomorphism $\pi : U(1)_E \to SU(2)_R \times U(1)_R \times U(1)_B$ and replace $U(1)_E$ by $U(1)'_E = (1 + \pi)(U(1)_E) \subset U(1)_E \times SU(2)_R \times U(1)_R \times U(1)_B$.

To pick a homomorphism, we consider the maximal torus $U(1)_R'$ of $SU(2)_R$

$$SU(2)_R \supset U(1)'_R.$$
 (8.1.26)

We assign $U(1)'_R$ charge for each field as in Table 8.1³. Then all of the U(1) charges are summarized in Table 8.2. In Table 8.2, the subscripts \pm indicate the upper and lower components of spinors. If Σ is flat, we should twist only $U(1)_C$ and there

³Our assignment is different from that in [293] where the both charges for q, \tilde{q} are –.

	$U(1)_C$	$U(1)_{\Sigma}$	$U(1)_R'$	$U(1)_R$	$U(1)_B$	section
φ	0	0	0	2	0	$\mathcal{O}_C\otimes\mathcal{O}_\Sigma$
$\overline{\varphi}$	0	0	0	-2	0	$\mathcal{O}_{C}\otimes\mathcal{O}_{\Sigma}$
ψ_+	+	_	+	+	0	$K_C^{\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}}$
ψ_{-}	_	+	+	+	0	$K_{\mathcal{L}}^{-\frac{1}{2}} \otimes K_{\Sigma}^{\frac{1}{2}}$
λ_+	+	_	_	+	0	$K_{C_{1}}^{\frac{1}{2}} \otimes K_{\Sigma_{1}}^{-\frac{1}{2}}$
λ_{-}	_	+	_	+	0	$K_{C_1}^{-\frac{1}{2}} \otimes K_{\Sigma_1}^{\frac{1}{2}}$
$\overline{\psi}_+$	_	_	_	_	0	$K_C^{-\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}}$
$\overline{\psi}_{-}$	+	+	_	_	0	$K^{\frac{1}{2}} \otimes K^{\frac{1}{2}}_{\Sigma}$
$\overline{\lambda}_+$	_	_	+	_	0	$K_C^{-\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}}$
$\overline{\lambda}_{-}$	+	+	+	_	0	$K_C^{rac{1}{2}}\otimes K_\Sigma^{rac{1}{2}}$
q	0	0	+	0	_	$\mathcal{O}_C\otimes\mathcal{O}_\Sigma$
q	0	0	_	0	+	$\mathcal{O}_{C}\otimes\mathcal{O}_{\Sigma}$
ψ_{q+}	+	_	0	_	_	$K_{C_{1}}^{\frac{1}{2}} \otimes K_{\Sigma_{1}}^{-\frac{1}{2}}$
ψ_{q-}	_	+	0	_	_	$K_{C}^{-\frac{1}{2}} \otimes K_{\Sigma}^{\frac{1}{2}}$
$\psi_{ ilde q+}$	+	_	0	_	+	$K_C^{\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}}$
$\psi_{ ilde q-}$	_	+	0	_	+	$K_{C_1}^{-\frac{1}{2}} \otimes K_{\Sigma_1}^{\frac{1}{2}}$
$\overline{\psi}_{q+}$	_	_	0	+	+	$K_C^{-\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}}$
$\overline{\psi}_{q-}$	+	+	0	+	+	$K_C^{rac{1}{2}}\otimes K_\Sigma^{rac{1}{2}}$
$\left \ \overline{\psi}_{\widetilde{q}+} \right $	_	_	0	+	_	$K_C^{-\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}}$
$\overline{\psi}_{\tilde{q}-}$	+	+	0	+	_	$K_C^{rac{1}{2}}\otimes K_\Sigma^{rac{1}{2}}$

Table 8.2: $U(1)_R$ charge assignments for d = 4, $\mathcal{N} = 2$ SCFT field content. The subscripts \pm indicate the upper and lower components of spinors. We denote the trivial bundle as \mathcal{O} and the canonical bundle as K.

	A-tv	A-twist		B-twist		
fields	$U(1)_C^{\prime}$	$U(1)'_{\Sigma}$	$U(1)_C^{\prime}$	$U(1)_{\Sigma}$		
φ	0	0	2	0		
\overline{arphi}	0	0	-2	0		
ψ_+	2	—	2	—		
ψ	0	+	0	+		
λ_+	0	—	2	—		
λ_{-}	-2	+	0	+		
$\overline{\psi}_+$	-2	—	-2	—		
$\overline{\psi}_{-}$	0	+	0	+		
$\overline{\lambda}_+$	0	—	-2	—		
$\overline{\lambda}_{-}$	2	+	0	+		
9	+	0	0	0		
Ĩ	—	0	0	0		
ψ_{q+}	+	—	0	—		
ψ_{q-}	—	+	-2	+		
$\psi_{ ilde q+}$	+	—	0	—		
$\psi_{ ilde q-}$	—	+	2	+		
$\overline{\psi}_{q+}$	—	—	0	—		
$\overline{\psi}_{q-}$	+	+	2	+		
$\overline{\psi}_{\widetilde{q}+}$	—	—	0	—		
$\overline{\psi}_{ ilde{q}-}$	+	+	2	+		

Table 8.3: The spin of the fields for A-twisted and B-twisted d = 4, $\mathcal{N} = 2$ SCFT on $C \times \Sigma$.

are two types of twisting

A-twist :
$$U(1)_C \to U(1)'_R$$
 (8.1.27)

B-twist
$$:U(2)_C \to U(1)_R.$$
 (8.1.28)

The field content of A-twist and B-twist listed in Table 8.3. If both *C* and Σ are curved, we should also twist $U(1)_{\Sigma}$. Although there are many possibilities for

	AA-	twist	BA-twist		BB-twist		BA+-twist	
fields	$U(1)_C^{\prime}$	$U(1)'_{\Sigma}$	$U(1)_C^{\prime}$	$U(1)_{\Sigma}$	$U(1)_C^{\prime}$	$U(1)'_{\Sigma}$	$U(1)_C^{\prime}$	$U(1)'_{\Sigma}$
φ	0	0	2	0	2	2	2	0
$\overline{\varphi}$	0	0	-2	0	-2	-2	-2	0
ψ_+	2	0	2	0	2	0	2	0
ψ	0	2	0	2	0	2	0	2
λ_+	0	-2	2	-2	2	0	2	-2
λ_{-}	-2	0	0	0	0	2	0	0
$\overline{\psi}_+$	-2	-2	-2	-2	-2	-2	-2	-2
$\overline{\psi}_{-}$	0	0	0	0	0	0	0	0
$\overline{\lambda}_+$	0	0	-2	0	-2	-2	-2	0
$\overline{\lambda}_{-}$	2	2	0	2	0	0	0	2
9	_	_	0	_	0	0	0	0
q	_	—	0	_	0	0	0	0
ψ_{q+}	+	—	0	—	0	-2	0	-2
ψ_{q-}	_	+	-2	+	-2	0	-2	0
$\psi_{ ilde q+}$	+	—	0	_	0	-2	0	0
$\psi_{ ilde q-}$	_	+	-2	+	-2	0	-2	2
$\overline{\psi}_{q+}$	_	_	0	_	0	0	0	0
$\overline{\psi}_{q-}$	+	+	2	+	2	2	2	2
$\overline{\psi}_{\tilde{q}+}$	_	_	0	_	0	0	0	-2
$\overline{\psi}_{\widetilde{q}-}$	+	+	2	+	2	2	2	0

Table 8.4: The spin of the fields for AA, BA, BB-twisted 4d $\mathcal{N} = 2$ SCFT on $C \times \Sigma$.

twisting, we consider the following cases

AA-twist :
$$U(1)_C \to U(1)'_R, U(1)_\Sigma \to U(1)'_R$$
 (8.1.29)

BA-twist :
$$U(1)_C \to U(1)_R, U(1)_\Sigma \to U(1)'_R$$
 (8.1.30)

$$BB-twist: U(1)_C \to U(1)_R, U(1)_\Sigma \to U(1)_R$$
(8.1.31)

BA+-twist :
$$U(1)_C \to U(1)_R, U(1)_\Sigma \to U(1)'_R \times U(1)_B.$$
 (8.1.32)

The results of twisting are given in Table 8.4.

A-twist

After twisting, the bosonic scalar fields φ , $\overline{\varphi}$, q, \tilde{q} remain scalars and the left-handed quarks ψ_q , $\psi_{\tilde{q}}$ remain sections of

$$K_{C}^{\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}} + K_{C}^{-\frac{1}{2}} \otimes K_{\Sigma}^{\frac{1}{2}}$$
(8.1.33)

and the right-handed quarks $\overline{\psi}_q$, $\overline{\psi}_{\tilde{q}}$ are sections of

$$K_{C}^{-\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}} + K_{C}^{\frac{1}{2}} \otimes K_{\Sigma}^{\frac{1}{2}}.$$
 (8.1.34)

On the other hand squarks q, \tilde{q} become sections of $K_C^{\frac{1}{2}}$ and $K_C^{-\frac{1}{2}}$. The gauginos ψ reduces to sections of

$$K_C \otimes K_{\Sigma}^{-\frac{1}{2}} + \mathcal{O}_C \otimes K_{\Sigma}^{\frac{1}{2}}$$
(8.1.35)

and λ become sections of

$$\mathcal{O}_C \otimes K_{\Sigma}^{-\frac{1}{2}} + K_C^{-1} \otimes K_{\Sigma}^{\frac{1}{2}}.$$
 (8.1.36)

Their right-handed partners are sections of

$$K_{C}^{-1} \otimes K_{\Sigma}^{-\frac{1}{2}} + \mathcal{O}_{C} \otimes K_{\Sigma}^{\frac{1}{2}}$$
 (8.1.37)

$$\mathcal{O}_{\mathcal{C}} \otimes K_{\Sigma}^{-\frac{1}{2}} + K_{\mathcal{C}} \otimes K_{\Sigma}^{\frac{1}{2}}.$$
(8.1.38)

In the original theory, we have eight supercharges. Since the transformations of supercharges under R-symmetry are identical to those of gauginos and only scalars on C survive in the twisted theory, (8.1.35)-(8.1.38) shows that there remains four supercharges

$$\mathcal{O}_{C} \otimes K_{\Sigma}^{\frac{1}{2}}, \quad \mathcal{O}_{C} \otimes K_{\Sigma}^{-\frac{1}{2}}, \\ \mathcal{O}_{C} \otimes K_{\Sigma}^{\frac{1}{2}}, \quad \mathcal{O}_{C} \otimes K_{\Sigma}^{-\frac{1}{2}}.$$
(8.1.39)

Two of them transform as spinors of positive chirality on Σ and the other two transform as those of negative chirality on Σ . Therefore if one takes into account the dimensional reduction to Σ , the theory on Σ has (2,2) supersymmetry.

B-twist

After the twisting the bosonic scalars φ becomes section of K_C and squarks q, \tilde{q} are unchanged. The quarks ψ_q and $\psi_{\tilde{q}}$ become sections of

$$\mathcal{O}_C \otimes K_{\Sigma}^{-\frac{1}{2}} + K_C^{-1} \otimes K_{\Sigma}^{\frac{1}{2}}$$
 (8.1.40)

$$\mathcal{O}_C \otimes K_{\Sigma}^{-\frac{1}{2}} + K_C \otimes K_{\Sigma}^{\frac{1}{2}}.$$
(8.1.41)

The gauginos ψ and λ are sections of

$$K_C \otimes K_{\Sigma}^{-\frac{1}{2}} + \mathcal{O}_C \otimes K_{\Sigma}^{\frac{1}{2}}$$
(8.1.42)

$$K_C \otimes K_{\Sigma}^{-\frac{1}{2}} + \mathcal{O}_C \otimes K_{\Sigma}^{\frac{1}{2}}$$
(8.1.43)

and $\tilde{\psi}$ and $\tilde{\lambda}$ are sections of

$$K_{C}^{-1} \otimes K_{\Sigma}^{-\frac{1}{2}} + \mathcal{O}_{C} \otimes K_{\Sigma}^{\frac{1}{2}}$$
 (8.1.44)

$$K_{\rm C}^{-1} \otimes K_{\Sigma}^{-\frac{1}{2}} + \mathcal{O}_{\rm C} \otimes K_{\Sigma}^{\frac{1}{2}}.$$
 (8.1.45)

From (8.1.42)-(8.1.45), we see that there are four supercharges

$$\mathcal{O}_C \otimes K_{\Sigma'}^{\frac{1}{2}}, \quad \mathcal{O}_C \otimes K_{\Sigma}^{\frac{1}{2}}$$

$$(8.1.46)$$

$$\mathcal{O}_C \otimes K_{\Sigma}^{\frac{1}{2}}, \quad \mathcal{O}_C \otimes K_{\Sigma}^{\frac{1}{2}},$$

$$(8.1.47)$$

which transform as spinors of the positive chirality on Σ . Thus the theory on Σ can have (4,0) supersymmetry.

AA-twist

After twisting, we have

$$\varphi \in \Gamma(\mathcal{O}_{\mathsf{C}} \otimes \mathcal{O}_{\Sigma}) \tag{8.1.48}$$

$$q \in \Gamma(K_C^{\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}}), \quad \tilde{q} \in \Gamma(K_C^{-\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}})$$
(8.1.49)

$$\psi_{q} \in \Gamma(K_{C}^{\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}} + K_{C}^{-\frac{1}{2}} \otimes K_{\Sigma}^{\frac{1}{2}}), \quad \psi_{\tilde{q}} \in (K_{C}^{\frac{1}{2}} \otimes K_{\Sigma}^{-\frac{1}{2}} + K_{C}^{-\frac{1}{2}} \otimes K_{\Sigma}^{\frac{1}{2}})$$
(8.1.50)
$$= -\Gamma(W^{-\frac{1}{2}} - W^{-\frac{1}{2}} - W^{\frac{1}{2}} - W^{\frac{1}{2}}) = -\Gamma(W^{-\frac{1}{2}} - W^{-\frac{1}{2}} - W^{\frac{1}{2}})$$
(8.1.50)

$$\overline{\psi}_{q} \in \Gamma(K_{C}^{2} \otimes K_{\Sigma}^{2} + K_{C}^{2} \otimes K_{\Sigma}^{2}), \quad \overline{\psi}_{\tilde{q}} \in \Gamma(K_{C}^{2} \otimes K_{\Sigma}^{2} + K_{C}^{2} \otimes K_{\Sigma}^{2})$$

$$(8.1.51)$$

$$(8.1.51)$$

$$\psi \in \Gamma(K_C \otimes \mathcal{O}_{\Sigma} + \mathcal{O}_C \otimes K_{\Sigma}), \quad \lambda \in \Gamma(\mathcal{O}_C \otimes K_{\Sigma} + K_C \otimes \mathcal{O}_{\Sigma})$$
(8.1.52)
$$\overline{\psi} \in \Gamma(K_C^{-1} \otimes K_{\Sigma}^{-1} + \mathcal{O}_C \otimes \mathcal{O}_{\Sigma}), \quad \overline{\lambda} \in \Gamma(\mathcal{O}_C \otimes \mathcal{O}_{\Sigma} + K_C \otimes K_{\Sigma}).$$
(8.1.53)

In other word, the fields transform under $U(1)_C^\prime imes U(1)_\Sigma^\prime$ as

$$\varphi, \overline{\varphi} \to \mathbf{1}_{00} \oplus \mathbf{1}_{00} \tag{8.1.54}$$

$$\psi, \lambda \to \mathbf{1}_{20} \oplus \mathbf{1}_{02} \oplus \mathbf{1}_{0-2} \oplus \mathbf{1}_{-20} \tag{8.1.55}$$

$$\overline{\psi}, \overline{\lambda} \to \mathbf{1}_{-2-2} \oplus \mathbf{1}_{00} \oplus \mathbf{1}_{00} \oplus \mathbf{1}_{22}$$
 (8.1.56)

$$q, \tilde{q} \to \mathbf{1}_{++} \oplus \mathbf{1}_{--} \tag{8.1.57}$$

 $\psi_q, \psi_{\tilde{q}} \to \mathbf{1}_{+-} \oplus \mathbf{1}_{-+} \oplus \mathbf{1}_{+-} \oplus \mathbf{1}_{-+}$ (8.1.58)

$$\overline{\psi}_{q}, \overline{\psi}_{\tilde{q}} \to \mathbf{1}_{--} \oplus \mathbf{1}_{++} \oplus \mathbf{1}_{--} \oplus \mathbf{1}_{++}. \tag{8.1.59}$$

Therefore the bosonic field content is

- 2 scalar fields ϕ, σ : $\mathbf{1}_{00} \oplus \mathbf{1}_{00}$
- gauge fields A_z , A_w : $\mathbf{1}_{20} \oplus \mathbf{1}_{-20} \oplus \mathbf{1}_{02} \oplus \mathbf{1}_{0-2}$
- spinor fields $\tilde{q}, \tilde{q}: \mathbf{1}_{++} \oplus \mathbf{1}_{--}$

and the fermionic field content is

- scalar field η : $\mathbf{1}_{00}$
- 1-forms $\psi_z, \psi_w: \mathbf{1}_{20} \oplus \mathbf{1}_{02} \oplus \mathbf{1}_{-02} \oplus \mathbf{1}_{-20}$
- 2-form χ : $\mathbf{1}_{-2-2} \oplus \mathbf{1}_{22}$
- spinor fields $\psi_q, \psi_{\tilde{q}}, \overline{\psi}_q, \overline{\psi}_{\tilde{q}}$: 2 ($\mathbf{1}_{+-} \oplus \mathbf{1}_{-+} \oplus \mathbf{1}_{--} \oplus \mathbf{1}_{++}$)

Focusing on the vector multiplet, one can check that the field content is same as that of $\mathcal{N} = 2$ twisted Donaldson-Witten theory as expected.

In the twisted theory we have two supercharges. Both of them are right-handed in 4-dimensions. Noting the spin of $U(1)_C \times U(1)_{\Sigma}$ for $\overline{\psi}_-, \overline{\lambda}_+$

$$\overline{\psi}_{-}:(+,+) \xrightarrow{\text{AA-twist}} (0,0) \tag{8.1.60}$$

$$\overline{\lambda}_{+}: (-,-) \xrightarrow{\text{AA-twist}} (0,0), \qquad (8.1.61)$$

one can see that the two supercharges have the opposite chiralities on both *C* and Σ .

BA-twist

After twisting we obtain

$$\varphi \in \Gamma(K_C \otimes \mathcal{O}_{\Sigma}) \tag{8.1.62}$$

$$q \in \Gamma(\mathcal{O}_C \otimes K_{\Sigma}^{-1}), \quad \tilde{q} \in \Gamma(\mathcal{O}_C \otimes K_{\Sigma}^{-1})$$
(8.1.63)

$$\psi_{q}, \psi_{\tilde{q}} \in \Gamma(\mathcal{O}_{C} \otimes K_{\Sigma}^{-\frac{1}{2}} + K_{C}^{-1} \otimes K_{\Sigma}^{\frac{1}{2}})$$
(8.1.64)

$$\overline{\psi}_{q}, \overline{\psi}_{\tilde{q}} \in \Gamma(\mathcal{O}_{\mathcal{C}} \otimes K_{\Sigma}^{-\frac{1}{2}} + K_{\mathcal{C}} \otimes K_{\Sigma}^{\frac{1}{2}})$$
(8.1.65)

$$\psi \in \Gamma(K_C \otimes \mathcal{O}_{\Sigma} + \mathcal{O}_C \otimes K_{\Sigma}), \quad \lambda \in \Gamma(K_C \otimes K_{\Sigma}^{-1} + \mathcal{O}_C \otimes \mathcal{O}_{\Sigma})$$
(8.1.66)

$$\overline{\psi} \in \Gamma(K_{\mathcal{C}}^{-1} \otimes K_{\Sigma}^{-1} + \mathcal{O}_{\mathcal{C}} \otimes \mathcal{O}_{\Sigma}), \quad \overline{\lambda} \in \Gamma(K_{\mathcal{C}}^{-1} \otimes \mathcal{O}_{\Sigma} + \mathcal{O}_{\mathcal{C}} \otimes K_{\Sigma}).$$
(8.1.67)

Therefore there exists two supercharges. One is left-handed and the other is righthanded in 4-dimensions. From the fact

$$\lambda_{-}: (-, +) \to (0, 0)$$
 (8.1.68)

$$\overline{\psi}_{-}:(+,+)\to(0,0),$$
 (8.1.69)

it turns out that two supercharges have the same chirality on Σ and the opposite chirality on *C*.

8.1.3 d = 4, $\mathcal{N} = 4$ SYM theories

Let us start with d = 10, $\mathcal{N} = 1$ SYM theory. d = 4, $\mathcal{N} = 4$ SYM theory is most easily derived by dimensional reduction from ten dimensions ⁴.

The field content of d = 10, $\mathcal{N} = 1$ SYM theory is

- gauge field A
- 16 fermionic fields (gauginos) Ψ

The gauge field *A* is a connection on a *G*-bundle *E*. The fermionic field Ψ is a positive chirality spinor field with values in the adjoint representation of *G*, that is a section of $S^+ \otimes \operatorname{ad}(E)$. We should note that the ten-dimensional spin representations

$$\mathbf{16}_{s} \text{ and } \mathbf{16}_{c} \text{ are } \begin{cases} \text{real and dual to each other : Lorentz signature} \\ \text{complex conjugate : Euclidean signature} \end{cases}$$

$$(8.1.70)$$

The gaugino Ψ is a ten-dimensional positive chirality spinor field

$$\Gamma^{11}\Psi = \Psi \tag{8.1.71}$$

where

$$\Gamma^{11} := i\Gamma^{12\dots10}.$$
 (8.1.72)

The conjugate is given by

$$\overline{\Psi} := \Psi^T \mathcal{C} \tag{8.1.73}$$

where C is a ten-dimensional charge conjugation matrix satisfying⁵

$$\mathcal{C}^T = \mathcal{C}, \quad \mathcal{C}\Gamma^M \mathcal{C}^{-1} = \Gamma^M. \tag{8.1.75}$$

$$\mathcal{C}^{T} = -\mathcal{C}, \quad \mathcal{C}\Gamma^{M}\mathcal{C}^{-1} = -\Gamma^{M} \tag{8.1.74}$$

which corresponds to C_+ , we choose C_- in (8.1.74).

⁴ From Nahm's theorem [12], ten-dimensional is the maximum possible dimension for SYM theory.

⁵Although there is another definition given by

physics convention	mathmatics convention		
D = d - iA' (A':Hermitian)	D = d + A (A:anti-Hermitian),		
$F = i(D)^2 = dA' - iA' \wedge A'$	$F = D^2 = dA + A \wedge A$		

Table 8.5: The physical and mathematical definitions of connection and field strength. The relations are given by -iF' = F, -iA' = A. Although the anti-Hermitian fields A are unnatural for G = U(1), they may avoid unnatural factors i.

The Lagrangian of Euclidean d = 10, $\mathcal{N} = 1$ SYM theory is given by

$$\mathcal{L} = \frac{1}{e^2} \operatorname{Tr} \left(\frac{1}{4} F_{MN} F^{MN} + \frac{1}{2} \overline{\Psi} \Gamma^M D_M \Psi \right)$$
(8.1.76)

where $M, N, \dots = 1, 2, \dots, 10$ are indices of ten-dimensional space-time and we define⁶

$$D_M \Psi := \partial_M \Psi - i[A_M, \Psi], \qquad (8.1.77)$$

$$F_{MN} := \partial_M A_N - \partial_N A_M - i[A_M, A_N]. \tag{8.1.78}$$

The 16 supersymmetries are

$$\delta A_M = \overline{\Psi} \Gamma_M \epsilon = -\overline{\epsilon} \Gamma_M \Psi \tag{8.1.79}$$

$$\delta \Psi = \frac{1}{2} F_{MN} \Gamma^{MN} \epsilon. \tag{8.1.80}$$

To consider the dimensional reduction of d = 10, $\mathcal{N} = 1$ SYM theories to four dimensions, we decompose the gamma matrices under $SO(10) \supset SO(4)_E \times SO(6)$ as

$$\begin{cases} \Gamma^{\mu} = \gamma^{\mu} \otimes \hat{\Gamma}^{7} \\ \Gamma^{I} = \mathbb{I}_{4} \otimes \hat{\Gamma}^{I} \end{cases}$$

$$(8.1.81)$$

where $\mu = 1, 2, 3, 4$ and I = 5, 6, 7, 8, 9, 10. $\hat{\Gamma}^{I}$ are six-dimensional gamma matrices satisfying

$$\{\hat{\Gamma}^{I}, \hat{\Gamma}^{J}\} = 2\delta^{IJ}, \quad (\hat{\Gamma}^{I})^{\dagger} = \Gamma^{I}$$
(8.1.82)

$$\hat{\Gamma}^7 = i\hat{\Gamma}^{12\cdots 6} = \begin{pmatrix} \mathbb{I}_4 & 0\\ 0 & -\mathbb{I}_4 \end{pmatrix}$$
(8.1.83)

 $^{^{6}}$ This is preferred convention in physics (see Table 8.5).

and γ^{μ} are four-dimensional gamma matrices

$$\{\gamma^{\mu}, \gamma^{\nu}\} = 2\delta^{\mu\nu}, \quad (\gamma^{\mu})^{\dagger} = \gamma^{\mu}, \quad \gamma^{5} := \gamma^{1\cdots 4}.$$
 (8.1.84)

The charge conjugation matrix C and the chiral matrix are decomposed as

$$\mathcal{C} = C \otimes \hat{C}, \tag{8.1.85}$$

$$\Gamma^{11} = \gamma^5 \otimes \hat{\Gamma}^7 \tag{8.1.86}$$

where *C* is the four-dimensional charge conjugation matrix satisfying ⁷

$$C^{T} = -C, \quad C\gamma^{\mu}C^{-1} = (\gamma^{\mu})^{T}, \quad C\gamma^{5}C^{-1} = (\gamma^{5})^{T}$$
 (8.1.87)

and \hat{C} is the six-dimensional charge conjugation matrix

$$\hat{C}^{T} = -\hat{C}, \quad \hat{C}\hat{\Gamma}^{I}\hat{C}^{-1} = (\hat{\Gamma}^{I})^{T}, \qquad C\hat{\Gamma}^{7}C^{-1} = -(\hat{\Gamma}^{7})^{T}$$
 (8.1.88)

The global symmetry of the theory is $SU(4)_R$ R-symmetry. The field content is

- 6 real scalar fields $\phi^I(I = 5, 6, \cdots, 10)$
- 16 fermionic fields (gauginos) $\psi^A(A = 1, 2, 3, 4)$
- gauge field A_{μ}

where indices I, J, \dots and A, B, \dots are **6** and **4** of $SU(4)_R$ R-symmetry.

Performing the dimensional reduction of (8.1.76), we obtain the Lagrangian of d = 4, $\mathcal{N} = 4$ SYM theories

$$\mathcal{L} = \frac{1}{e^2} \operatorname{Tr} \left(\frac{1}{4} F_{\mu\nu} F^{\mu\nu} + \frac{1}{2} D_{\mu} \phi_I D^{\mu} \phi^I - \frac{1}{4} [\phi_I, \phi_I] [\phi^I, \phi^I] + \frac{1}{2} \overline{\psi} \Gamma^{\mu} D_{\mu} \psi - \frac{i}{2} \overline{\psi} \Gamma^I [\phi_I, \psi] \right)$$
(8.1.89)

where $\phi_I := A_I (5 \le I \le 10)$ and $\mu, \nu = 1, 2, 3, 4$. If *G* is simple and if we require that Lagrangian is quadratic in derivatives, the above Lagrangian is unique except for the change of parameter *e*. However, we may have θ -term that measures the topology of the *G*-bundle *E*

$$\mathcal{L}_{\theta} = \frac{i\theta}{16\pi^2} \operatorname{Tr}(F \wedge F) = \frac{i\theta}{16\pi^2} \operatorname{Tr}(*F_{\mu\nu}F^{\mu\nu}), \qquad (8.1.90)$$

⁷Here we choose *C* as C_{-} .

which is θ times the second Chern class or instanton number of the bundle⁸. The parameter *e* and θ combine into a complex parameter

$$\tau = \frac{\theta}{2\pi} + \frac{4\pi i}{e^2}.\tag{8.1.91}$$

The Lagrangian (8.1.89) is invariant under 16 supersymmetries

$$\delta\phi_I = \overline{\psi}\Gamma_I \epsilon = -\overline{\epsilon}\Gamma_I \psi, \qquad (8.1.92)$$

$$\delta A_{\mu} = \overline{\psi} \Gamma_{\mu} \epsilon = -\overline{\epsilon} \Gamma_{\mu} \psi, \qquad (8.1.93)$$

$$\delta \psi = \frac{1}{2} F_{\mu\nu} F^{\mu\nu} \epsilon + D_{\mu} \Phi_I \Gamma^{\mu} \Gamma^I \epsilon - \frac{i}{2} [\phi_I, \phi_J] \Gamma^{IJ} \epsilon.$$
(8.1.94)

Before topological twisting, fields transform under $SO(4)_E \times SU(4)_R \simeq SU(2)_l \times SU(2)_r \times SU(4)_R$ as

$$\phi:(1,1,6)$$
 (8.1.95)

$$\psi:(2,1,\overline{4})\oplus(1,2,4)$$
 (8.1.96)

$$A_{\mu}:(\mathbf{2},\mathbf{2},\mathbf{1}).$$
 (8.1.97)

To perform the fully topological twisting, we pick a homomorphism $\pi : SO(4)_E \rightarrow SU(4)_R$ and replace $SO(4)_E$ by $SO(4)'_E = (1 + \pi)(SO(4)_E) \subset SO(4)_E \times SO(6)_R$.

The choice of π amounts to embedding $SO(4)_E \simeq SU(2)_l \times SU(2)_r$ in $SU(4)_R$ as

$$\pi: SU(2)_l \times SU(2)_r \to \left(\begin{array}{cc} SU(2)_l & 0\\ 0 & SU(2)_r \end{array}\right), \tag{8.1.98}$$

which leads us to consider the decomposition

$$SU(4) \supset SU(2) \times SU(2) \times U(1). \tag{8.1.99}$$

Under (8.1.99), we still have several possible embedding determined by telling how the **4** of $SU(4)_R$ transforms under $SU(2)_l \times SU(2)_r$. Up to an exchange of left and right, there are three inequivalent transformations of **4** of $SU(4)_R$ under (8.1.99).

(i) GL twist
$$\mathbf{4} = (\mathbf{2}, \mathbf{1})_1 \oplus (\mathbf{1}, \mathbf{2})_{-1}$$

(ii) VW twist $\mathbf{4} = (\mathbf{1}, \mathbf{2})_1 \oplus (\mathbf{1}, \mathbf{2})_{-1}$ (8.1.100)
(iii) DW twist $\mathbf{4} = (\mathbf{1}, \mathbf{2})_0 \oplus (\mathbf{1}, \mathbf{1})_1 \oplus (\mathbf{1}, \mathbf{1})_{-1}$

⁸In $\mathcal{N} = 4$ SYM theories θ -terms are observable because there is no chiral anomaly and we cannot shift them. This situation is different from $\mathcal{N} = 2$ SYM.

Geometric Langlands (GL) twist

GL twist has the branching

$$\begin{aligned} \mathbf{4} &= (\mathbf{2}, \mathbf{1})_1 \oplus (\mathbf{1}, \mathbf{2})_{-1} \\ \overline{\mathbf{4}} &= (\mathbf{2}, \mathbf{1})_{-1} \oplus (\mathbf{1}, \mathbf{2})_1. \end{aligned} \tag{8.1.101}$$

1. fermionic fields

Noting that

$$(\mathbf{2},\mathbf{1})_0 \times ((\mathbf{2},\mathbf{1})_{-1} \oplus (\mathbf{1},\mathbf{2})_1) = (\mathbf{1},\mathbf{1})_{-1} \oplus (\mathbf{3},\mathbf{1})_{-1} \oplus (\mathbf{2},\mathbf{2})_1$$
 (8.1.102)

and

$$(\mathbf{1},\mathbf{2})_0 \times ((\mathbf{2},\mathbf{1})_1 \oplus (\mathbf{1},\mathbf{2}_{-1})) = (\mathbf{2},\mathbf{2})_1 \oplus (\mathbf{1},\mathbf{1})_{-1} \oplus (\mathbf{1},\mathbf{3})_{-1},$$
 (8.1.103)

one can see that the fermionic fields transform under $SU(2)'_l \times SU(2)'_r \times U(1)$ as

$$(\mathbf{1},\mathbf{1})_{-1} \oplus (\mathbf{3},\mathbf{1})_{-1} \oplus (\mathbf{2},\mathbf{2})_1 \oplus (\mathbf{2},\mathbf{2})_1 \oplus (\mathbf{1},\mathbf{1})_{-1} \oplus (\mathbf{1},\mathbf{3})_{-1}.$$
 (8.1.104)

Similarly one can obtain the transformations of supersymmetries. Thus, from (8.1.104) we see that GL twist leads to two unbroken BRST charges which have the same U(1) charge.

2. bosonic fields

The bosonic scalar field $\mathbf{6}_v$ of $SO(6)_R$ is produced by the product of SO(6) spinor $\mathbf{8} = \mathbf{4} + \overline{\mathbf{4}}$ as

$$8 \times 8 = (4 + \overline{4}) \times (4 + \overline{4})$$

=4 \times 4 + 4 \times \overline{4} + \overline{4} \times 4 + \overline{4} \times \overline{4}}
=([1] + [3]) + ([0] + [2]) + ([0] + [2]) + ([1] + [3])
=(6 + 10) + (1 + 15) + (1 + 15) + (6 + 10). (8.1.105)

Note that $\mathbf{6}_v$ is the antisymmetric product of $\mathbf{4}$

$$\begin{aligned} \mathbf{6}_{v} &= (\mathbf{4} + \mathbf{4})_{a} \\ &= (((\mathbf{2}, \mathbf{1})_{1} \oplus (\mathbf{1}, \mathbf{2})_{-1}) \otimes ((\mathbf{2}, \mathbf{1})_{1} \oplus (\mathbf{1}, \mathbf{2})_{-1}))_{a} \\ &= (((\mathbf{1}, \mathbf{1})_{2} \oplus (\mathbf{1}, \mathbf{1})_{-2} \oplus (\mathbf{2}, \mathbf{2})_{0}) \oplus ((\mathbf{3}, \mathbf{1})_{2} \oplus (\mathbf{2}, \mathbf{2})_{0} \oplus (\mathbf{1}, \mathbf{3})_{-2}))_{a} \\ &= (\mathbf{1}, \mathbf{1})_{2} \oplus (\mathbf{1}, \mathbf{1})_{-2} \oplus (\mathbf{2}, \mathbf{2})_{0}. \end{aligned}$$
(8.1.106)

Thus the bosonic field content consists of

- 2 scalar fields : $(1, 1)_2 \oplus (1, 1)_{-2}$
- 1-form : (2, 2)₀
- gauge field : (**2**, **2**)₀

and the fermionic field content is

- 2 scalar fields : $(1, 1)_{-1} \oplus (1, 1)_{-1}$
- 2 1-forms : $(2,2)_1 \oplus (2,2)_1$
- 2 2-forms ${}^9: (\mathbf{3},\mathbf{1})_{-1} \oplus (\mathbf{1},\mathbf{3})_{-1}$.

Under decomposition $SO(10) \supset SO(4)_E \times SO(4) \times SO(2)$, the decomposition of the ten-dimensional gamma matrices is given by

$$\begin{cases} \Gamma^{\mu} = \gamma^{\mu} \otimes \gamma^{5} \otimes -\sigma_{3} \\ \Gamma^{\mu+4} = \mathbb{I}_{4} \otimes \gamma^{\mu} \otimes \mathbb{I}_{2} \\ \Gamma^{i+8} = \mathbb{I}_{4} \otimes \gamma^{5} \otimes \sigma_{i} \end{cases}$$

$$(8.1.107)$$

where $\mu = 1, 2, 3, 4$ and i = 1, 2. σ_i are Pauli matrices and γ^{μ} are four-dimensional gamma matrices defined by (8.1.84). The charge conjugation matrix C is decomposed as

$$\mathcal{C} = \mathcal{C} \otimes \mathcal{C} \otimes \sigma_1 \tag{8.1.108}$$

Under the decomposition (8.1.107), ten-dimensional chirality matrix is expressed as

$$\Gamma^{11} = -\left(\gamma^5 \otimes \gamma^5 \otimes \sigma_3\right). \tag{8.1.109}$$

1. bosonic fields

For the bosonic fields we redefine

$$\Phi_{\mu} := \phi_{4+\mu}, \quad (\mu = 1, 2, 3, 4), \tag{8.1.110}$$

$$A_{\mu}^{\pm} := \frac{1}{\sqrt{2}} (A_{\mu} \pm i \Phi_{\mu}), \qquad (8.1.111)$$

$$\varphi := \frac{1}{\sqrt{2}}(\phi_9 + i\phi_{10}), \quad \overline{\varphi} := \frac{1}{\sqrt{2}}(\phi_9 - i\phi_{10}).$$
 (8.1.112)

⁹Note that this is not self-dual.

2. fermionic fields

Noting the fermionic field content of GL twist, one can decompose the 10dimensional fermionic field Ψ under the decomposition $SO(10) \supset SO(4)_E \times$ $SO(4) \times SO(2)$ as ¹⁰

$$\Psi_{pq\alpha} = \frac{1}{\sqrt{2}} \left(\eta_{\alpha} \mathbb{I}_{4pq} + \psi_{\mu\alpha} \gamma^{\mu}_{pq} + \frac{1}{2} \chi_{\mu\nu\alpha} \gamma^{\mu\nu}_{pq} + \omega_{\mu\alpha} \gamma^{\mu5}_{pq} + \zeta_{\alpha} \gamma^{5}_{pq} \right) C^{-1} \quad (8.1.113)$$

where p, q and α are indices of $SO(4)_E$, SO(4) and SO(2) respectively. $\eta_{\alpha}, \zeta_{\alpha}$ are scalars $(\mathbf{1}, \mathbf{1})_- \oplus (\mathbf{1}, \mathbf{1})_-$ and $\psi_{\mu\alpha}, \omega_{\mu\alpha}$ are 1-forms $(\mathbf{2}, \mathbf{2})_+ \oplus (\mathbf{2}, \mathbf{2})_+$ and $\chi_{\mu\nu} = -\chi_{\nu\mu}$ is a 2-form $(\mathbf{3}, \mathbf{1})_- \oplus (\mathbf{1}, \mathbf{3})_-$.

From the decompositions (8.1.107) and the chirality condition (8.1.71) for the femionic fields, we see that

$$\sigma_3 \eta = \eta, \quad \sigma_3 \psi_\mu = \psi_\mu, \tag{8.1.114}$$

$$\sigma_3 \chi_{\mu\nu} = -\chi_{\mu\nu}, \quad \sigma_3 \omega_\mu = \omega_\mu, \tag{8.1.115}$$

$$\sigma_3 \zeta = -\zeta \tag{8.1.116}$$

Therefore ψ_{μ} , ω_{μ} have U(1) ghost charge +1 and η , $\chi_{\mu\nu}$, ζ have U(1) charge -1. This is consistent to the results of the fermionic field content for GL twist.

3. supersymmetries

For the sypersymmetries we can also expand as

$$\epsilon_{pq\alpha} = \frac{1}{\sqrt{2}} \left(\epsilon_{\alpha} \mathbb{I}_{4pq} + \epsilon_{\mu\alpha} \gamma^{\mu}_{pq} + \frac{1}{2} \epsilon_{\mu\nu\alpha} \gamma^{\mu\nu}_{pq} + \tilde{\epsilon}_{\mu\alpha} \gamma^{\mu5} + \tilde{\epsilon}_{\alpha} \gamma^{5}_{pq} \right) C^{-1}. \quad (8.1.117)$$

From the decompositions (8.1.107) and the chirality condition for the supersymmetries, we see that

$$\sigma_3 \epsilon = -\epsilon, \quad \sigma_3 \tilde{\epsilon} = -\tilde{\epsilon} \tag{8.1.118}$$

Therefore both BRST charges ϵ and $\tilde{\epsilon}$ have U(1) charge -1 as expected.

¹⁰The inverse of charge conjugation matrices C^{-1} is included just because of the convenience of the calculation.

The BRST transformation is given by

$$\delta A_{\mu} = -2(\bar{\epsilon}\omega^{\mu}) - 2(\bar{\tilde{\epsilon}}\psi^{\mu}), \qquad (8.1.119)$$

$$\delta \Phi_{\mu} = -2i(\bar{\epsilon}\psi^{\mu}) - 2i(\bar{\tilde{\epsilon}}\omega^{\mu}), \qquad (8.1.120)$$

$$\delta \varphi = 2i\overline{\epsilon}\sigma_{+}\zeta + 2i\overline{\tilde{\epsilon}}\sigma_{+}\eta, \qquad (8.1.121)$$

$$\delta\tilde{\varphi} = 2i\bar{\varepsilon}\sigma_{-}\zeta + 2i\bar{\tilde{\varepsilon}}\sigma_{-}\eta, \qquad (8.1.122)$$

$$\delta\eta = iD_{\mu}\Phi^{\mu}\tilde{\epsilon} - \frac{1}{2}[\phi^{i},\phi^{j}](\epsilon_{ij}\epsilon), \qquad (8.1.123)$$

$$\delta\psi_{\mu} = -iD_{\mu}\phi_i(\sigma_i\epsilon) - i[\Phi_{\mu},\phi_i](\sigma_i\tilde{\epsilon}), \qquad (8.1.124)$$

$$\delta\chi_{\mu\nu} = -F_{\mu\nu}\epsilon + \epsilon_{\mu\nu\rho\sigma}F^{\rho\sigma}\tilde{\epsilon} + 2i\epsilon_{\mu\nu\rho\sigma}D^{\rho}\Phi^{\sigma}\epsilon + 2iD_{\mu}\Phi_{\nu}\tilde{\epsilon} + i[\Phi_{\mu},\Phi_{\nu}]\epsilon - i\epsilon_{\mu\nu\rho\sigma}[\Phi^{\rho},\Phi^{\sigma}]\tilde{\epsilon}, \qquad (8.1.125)$$

$$\delta\omega_{\mu} = -iD_{\mu}\phi_i(\sigma_i\tilde{\epsilon}) + i[\Phi_{\mu},\phi_i](\sigma_i\epsilon), \qquad (8.1.126)$$

$$\delta\zeta = -iD_{\mu}\Phi^{\mu}\epsilon - \frac{1}{2}(\epsilon_{ij}\tilde{\epsilon})[\phi_i,\phi_j]$$
(8.1.127)

where we introduce

$$\sigma_{+} := \frac{1}{\sqrt{2}} (\sigma_{1} + i\sigma_{2}), \qquad (8.1.128)$$

$$\sigma_{-} := \frac{1}{\sqrt{2}} (\sigma_{1} - i\sigma_{2}). \tag{8.1.129}$$

Vafa-Witten (VW) twist

VW twist corresponds to the following branching:

$$\begin{aligned} \mathbf{4} &= (\mathbf{1}, \mathbf{2})_1 \oplus (\mathbf{1}, \mathbf{2})_{-1} \\ \overline{\mathbf{4}} &= (\mathbf{1}, \mathbf{2})_{-1} \oplus (\mathbf{1}, \mathbf{2})_1. \end{aligned} \tag{8.1.130}$$

1. fermionic fields

Noting that

$$(\mathbf{2},\mathbf{1})_0 \times ((\mathbf{1},\mathbf{2})_{-1} \oplus (\mathbf{1},\mathbf{2})_1) = (\mathbf{2},\mathbf{2})_{-1} \oplus (\mathbf{2},\mathbf{2})_1$$
 (8.1.131)

and

$$(\mathbf{1},\mathbf{2})_0 \times ((\mathbf{1},\mathbf{2})_1 \oplus (\mathbf{1},\mathbf{2})_{-1}) = (\mathbf{1},\mathbf{1})_1 \oplus (\mathbf{1},\mathbf{3})_1 \oplus (\mathbf{1},\mathbf{1})_{-1} \oplus (\mathbf{1},\mathbf{3})_{-1}$$
, (8.1.132)

it turns out that the fermionic fields transform under $SU(2)'_l \times SU(2)'_r \times U(1)$ as

$$(\mathbf{2},\mathbf{2})_{-1} \oplus (\mathbf{2},\mathbf{2})_1 \oplus (\mathbf{1},\mathbf{1})_1 \oplus (\mathbf{1},\mathbf{3})_1 \oplus (\mathbf{1},\mathbf{1})_{-1} \oplus (\mathbf{1},\mathbf{3})_{-1}.$$
 (8.1.133)

Therefore VW twist gives rise to two unbroken BRST charges which have the opposite U(1) charge.

2. bosonic fields

The bosonic scalar field $\mathbf{6}_v$ of $SO(6)_R$ is given by the antisymmetric product of **4**

$$\begin{aligned} \mathbf{6}_{v} &= (\mathbf{4} \times \mathbf{4})_{a} \\ &= (((\mathbf{1}, \mathbf{2})_{1} \oplus (\mathbf{1}, \mathbf{2})_{-1}) \times ((\mathbf{1}, \mathbf{2})_{1} \oplus (\mathbf{1}, \mathbf{2})_{-1}))_{a} \\ &= (((\mathbf{1}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{3})_{0} \oplus (\mathbf{1}, \mathbf{1})_{2} \oplus (\mathbf{1}, \mathbf{1})_{-2}) \oplus ((\mathbf{1}, \mathbf{3})_{2} \oplus (\mathbf{1}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{3})_{0} \oplus (\mathbf{1}, \mathbf{3})_{-2}))_{a} \\ &= (\mathbf{1}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{3})_{0} \oplus (\mathbf{1}, \mathbf{1})_{2} \oplus (\mathbf{1}, \mathbf{1})_{-2}. \end{aligned}$$

$$(8.1.134)$$

Thus the bosonic field content consists of

- 3 scalar fields : $(1, 1)_{-2} \oplus (1, 1)_0 \oplus (1, 1)_2$
- 2-form : $(1,3)_0$
- gauge field : (**2**, **2**)₀

and the fermionic one is

- 2 scalar fields : $(1, 1)_+ \oplus (1, 1)_-$
- 1-form : $(2, 2)_{-} \oplus (2, 2)_{+}$
- 2 2-from : $(1,3)_- \oplus (1,3)_+$.

Donaldson-Witten (DW) twist

DW twist has the branching

$$\begin{aligned} \mathbf{4} &= (\mathbf{1}, \mathbf{2})_0 \oplus (\mathbf{1}, \mathbf{1})_1 \oplus (\mathbf{1}, \mathbf{1})_{-1} \\ \overline{\mathbf{4}} &= (\mathbf{1}, \mathbf{2})_0 \oplus (\mathbf{1}, \mathbf{1})_{-1} \oplus (\mathbf{1}, \mathbf{1})_1. \end{aligned} \tag{8.1.135}$$

1. fermionic fields

Noticing that

$$(\mathbf{2},\mathbf{1})_0 \times ((\mathbf{1},\mathbf{2})_0 \oplus (\mathbf{1},\mathbf{1})_{-1} \oplus (\mathbf{1},\mathbf{1})_1) = (\mathbf{2},\mathbf{2})_0 \oplus (\mathbf{2},\mathbf{1})_{-1} \oplus (\mathbf{2},\mathbf{1})_1$$
 (8.1.136)

and

$$(\mathbf{1},\mathbf{2})_0 \times ((\mathbf{1},\mathbf{2})_0 \oplus (\mathbf{1},\mathbf{1})_1 \oplus (\mathbf{1},\mathbf{1})_{-1}) = (\mathbf{1},\mathbf{1})_0 \oplus (\mathbf{1},\mathbf{3})_0 \oplus (\mathbf{1},\mathbf{2})_1 \oplus (\mathbf{1},\mathbf{2})_{-1},$$
(8.1.137)

one can see that fermionic fields transform under $SU(2)'_l \times SU(2)'_r \times U(1)$ as

$$(2,2)_0 \oplus (2,1)_{-1} \oplus (2,1)_1 \oplus (1,1)_0 \oplus (2,3)_0 \oplus (1,2)_1 \oplus (1,2)_{-1},$$
 (8.1.138)

which implies that DW twist allows one unbroken BRST charge.

2. bosonic fields

The bosonic scalar field $\mathbf{6}_v$ of $SO(6)_R$ is given by the antisymmetric product of **4**

$$\begin{aligned} \mathbf{6}_{v} &= (\mathbf{4} \times \mathbf{4})_{a} \\ &= (((\mathbf{1}, \mathbf{2})_{0} \oplus (\mathbf{1}, \mathbf{1})_{1} \oplus (\mathbf{1}, \mathbf{1})_{-1}) \times ((\mathbf{1}, \mathbf{2})_{0} \oplus (\mathbf{1}, \mathbf{1})_{1} \oplus (\mathbf{1}, \mathbf{1})_{-1}))_{a} \\ &= ((2(\mathbf{1}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{2})_{1} \oplus (\mathbf{1}, \mathbf{2})_{-1}) \oplus ((\mathbf{1}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{3})_{0} \oplus (\mathbf{1}, \mathbf{2})_{1} \oplus (\mathbf{1}, \mathbf{2})_{-1} \oplus (\mathbf{1}, \mathbf{1})_{2} \oplus (\mathbf{1}, \mathbf{1})_{-2}))_{a} \\ &= 2(\mathbf{1}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{2})_{1} \oplus (\mathbf{1}, \mathbf{2})_{-1}. \end{aligned}$$

$$(8.1.139)$$

Thus the bosonic field content consists of

- 2 scalar fields : $(1, 1)_0 \oplus (1, 1)_0$
- 2 spinor fields : $(1, 2)_1 \oplus (1, 2)_{-1}$
- gauge field : (**2**, **2**)₀

8.1.4 d = 4, $\mathcal{N} = 4$ SYM theories on $C \times \Sigma$

Now we discuss a d = 4, $\mathcal{N} = 4$ SYM theory on $M_4 = C \times \Sigma$. We consider the twisting by using the embedding

$$U(1)_C \to U(1)_R.$$
 (8.1.140)

The assignment of U(1) charges are given in Table 8.6 where +, - signs denote upper and lower components of spinors and right-handed fermions indicated with bars. Under $U(1)'_C \times U(1)'_\Sigma \times SU(2)_2$ the fields transform as

$$\phi \to \mathbf{1}_{00} \oplus \mathbf{3}_{00} \oplus \mathbf{1}_{20} \oplus \mathbf{1}_{-20} \tag{8.1.141}$$

$$\psi \to \mathbf{2}_{2-} \oplus \mathbf{2}_{0-} \oplus \mathbf{2}_{0+} \oplus \mathbf{2}_{-2+}$$
 (8.1.142)

$$\overline{\psi} \to \mathbf{2}_{0-} \oplus \mathbf{2}_{-2-} \oplus \mathbf{2}_{2+} \oplus \mathbf{2}_{0+} \tag{8.1.143}$$

$$A_{\mu} \to \mathbf{1}_{20} \oplus \mathbf{1}_{02} \oplus \mathbf{1}_{0-2}.$$
 (8.1.144)

The bosonic field content is

- 2 complex scalar fields ϕ , $\overline{\phi}$: $\mathbf{1}_{00} \oplus \mathbf{3}_{00}$
- 1-form $\Phi_w, \Phi_{\overline{w}} : \mathbf{1}_{20} \oplus \mathbf{1}_{-20}$
- gauge field $A_z, A_{\overline{z}}, A_w, A_{\overline{w}} : \mathbf{1}_{20} \oplus \mathbf{1}_{-20} \oplus \mathbf{1}_{02} \oplus \mathbf{1}_{0-2}$

and the fermionic field content is

	$U(1)_C$	$U(1)_{\Sigma}$	$U(1)_R$	$U(1)_C^{\prime}$	$U(1)'_{\Sigma}$
ϕ^1	0	0	0	0	0
ϕ^2	0	0	0	0	0
ϕ^3	0	0	0	0	0
ϕ^4	0	0	0	0	0
ϕ^5	0	0	0	2	2
ϕ^6	0	0	0	-2	-2
ψ^1_+	+	_	+	2	_
ψ_+^2	+	_	+	2	_
ψ^3_+	+	_	+	0	_
ψ^4_+	+	_	+	0	_
ψ^1	—	+	+	0	+
ψ_{-}^{2}	_	+	+	0	+
ψ_{-}^{3}	_	+	_	-2	+
ψ^4	—	+	_	-2	+
$ \overline{\psi}^1_+$	—	—	—	0	—
$ \overline{\psi}_{+}^{2} $	_	_	_	0	_
$ \overline{\psi}_{+}^{3} $	_	_	_	-2	_
$ \overline{\psi}_{+}^{4} $	_	_	_	-2	_
$ \overline{\psi}_{-}^{1} $	+	+	+	2	+
$ \overline{\psi}_{-}^{2} $	+	+	+	2	+
$\overline{\psi}_{-}^{3}$	+	_	+	0	+
$\overline{\psi}^4$	+	_	+	0	+

Table 8.6: U(1) charges for VW partial twisting.

- 8 scalar fields $\eta^a_-, \chi^a_+, \overline{\eta}^a_-, \overline{\chi}^a_+ : \mathbf{2}_{0-} \oplus \mathbf{2}_{0+} \oplus \mathbf{2}_{0-} \oplus \mathbf{2}_{0+}$
- 1-form $\lambda_{w-}^a, \chi_{\overline{w}+}^a, \overline{\lambda}_{\overline{w}}^a, \overline{\chi}_{w+}^a: \mathbf{2}_{2-} \oplus \mathbf{2}_{-2+} \oplus \mathbf{2}_{-2-} \oplus \mathbf{2}_{2+}$

where a = 1, 2 are the indices of the fundamental representations of the unbroken $SU(2)_2$ symmetry.

One can see that there are eight BRST charges. Four of them transform as spinors with positive chirality on Σ corresponding to χ^a_+ and $\overline{\chi}^a_+$ and the others, that is η^a_- and $\overline{\eta}^a_-$ transform as those with negative chirality on Σ . Therefore we can regard the theory on Σ has (4,4) supersymmetry.

8.1.5 d = 3, $\mathcal{N} = 4$ SYM theories

The global symmetry of the theory is $SO(4) \simeq SU(2)_1 \times SU(2)_2$ R-symmetry. In order to understand the symmetry, it is convenient to construct d = 3, $\mathcal{N} = 4$ theories by dimensional reduction from d = 6, $\mathcal{N} = 1$ supersymmetric gauge theories. $SU(2)_1$ is the double cover of rotational symmetry SO(3) in the three reduced coordinates and $SU(2)_2$ is the R-symmetry in six-dimensional $\mathcal{N} = 1$ SYM theories [294, 295, 296, 297, 298].

The field content is

- 3 scalar fields ϕ^i
- fermionic field ψ
- gauge fields A_{μ}

Before topological twisting, fields transform under $SU(2)_E \times SU(2)_1 \times SU(2)_2$ as

 $\phi:(\mathbf{1},\mathbf{3},\mathbf{1})$ (8.1.145)

$$\psi:(2,2,2)$$
 (8.1.146)

$$A_{\mu}:(\mathbf{3},\mathbf{1},\mathbf{1}). \tag{8.1.147}$$

To perform the fully topological twisting, we pick a homomorphism $\pi : SU(2)_E \rightarrow SU(2)_1 \times SU(2)_2$ and replace $SU(2)_E$ by $SU(2)'_E = (1 + \pi)(SU(2)_E) \subset SU(2)_E \times SU(2)_1 \times SU(2)_2$.

We many have two choices of π as

$$\begin{cases} (i) \text{A-twist} \quad SU(2)_E \to SU(2)_2 \\ (ii) \text{B-twist} \quad SU(2)_E \to SU(2)_1 \end{cases}$$

A-twist

After twisting, the fields transform under the new symmetry $SU(2)'_E \times SU(2)_1$ as

$$\phi \rightarrow (\mathbf{1}, \mathbf{3}) \tag{8.1.148}$$

$$\psi \rightarrow (\mathbf{1}, \mathbf{2}) \oplus (\mathbf{3}, \mathbf{2})$$
 (8.1.149)

$$A_{\mu} \to (3, 1).$$
 (8.1.150)

There are two BRST charges and this is just the dimensional reduction of Donaldson-Witten theory (twisted d = 4, $\mathcal{N} = 2$ theory). The field content is same as d = 3 super BF model [299, 300, 301, 302] associated with the Casson invariant.

B-twist

After twisting, the fields transform under the new symmetry $SU(2)'_E \times SU(2)_2$ as

$$\phi \rightarrow (\mathbf{3}, \mathbf{1}) \tag{8.1.151}$$

$$\psi \rightarrow (\mathbf{1}, \mathbf{2}) \oplus (\mathbf{3}, \mathbf{2})$$
 (8.1.152)

$$A_{\mu} \rightarrow (\mathbf{3}, \mathbf{1}). \tag{8.1.153}$$

In B-twist we also have two BRST charges. As B-twist is related A-twist under the exchange of $SU(2)_1$ and $SU(2)_2$, it is expected that B-twist may be regarded as a mirror discription of the Casson invariant because d = 3, $\mathcal{N} = 4$ mirror symmetry has mirror pair under this exchange.

8.1.6 d = 3, $\mathcal{N} = 8$ SYM theories

The global symmetry of the theory is $Spin(7)_R$ R-symmetry. If we construct the theories by the dimensional reduction of d = 10, $\mathcal{N} = 1$ SYM theories, this is recognized as double cover of rotational symmetry SO(7) in the seven reduced coordinates. The field content is

- 7 scalar fields ϕ^i
- fermionic field ψ
- gauge fields A_{μ}

Before topological twisting, fields transform under $SU(2)_E \times Spin(7)_R$ as

$$\phi:(1,7)$$
 (8.1.154)

$$\psi:(\mathbf{2},\mathbf{8})$$
 (8.1.155)

$$A_{\mu}:(\mathbf{3},\mathbf{1}). \tag{8.1.156}$$

To perform the fully topological twisting, we pick a homomorphism $\pi : SU(2)_E \rightarrow Spin(7)_R$ and replace $SU(2)_E$ by $SU(2)'_E = (1 + \pi)(SU(2)_E) \subset SU(2)_E \times Spin(7)_R$.

The homomorphism π is determined by the decomposition of Spin(7) under SU(2) and the embedding of SU(2) in Spin(7). Although there are many possible decompositions, we consider the following branchings

$$Spin(7) \supset SU(2)_1 \times SU(2)_2 \times SU(2)_3.$$
 (8.1.157)

Under (8.1.157), 7 and 8 of *Spin*(7) decomposed as [303]

$$7 = (2, 2, 1) \oplus (1, 1, 3) \tag{8.1.158}$$

$$\mathbf{8} = (\mathbf{2}, \mathbf{1}, \mathbf{2}) \oplus (\mathbf{1}, \mathbf{2}, \mathbf{2}). \tag{8.1.159}$$

Then one can consider two different types of embedding with the residual global symmetry ¹¹ $SU(2) \times SU(2)$

A-twist :
$$SU(2)_E \rightarrow SU(2)_3$$

B-twist : $SU(2)_E \rightarrow SU(2)_1$. (8.1.160)

A-twist

After twisting the fields transform under $SU(2)'_E \times SU(2)_1 \times SU(2)_2$ as

$$\phi \to (1, 2, 2) \oplus (3, 1, 1)$$
 (8.1.161)

$$\psi \to (1,2,1) \oplus (3,2,1) \oplus (1,1,2) \oplus (3,1,2).$$
 (8.1.162)

The bosonic field content consists of

- 4 scalar fields: (1, 2, 2)
- 1-form: (3,1,1)
- gauge field: (3, 1, 1)

and fermionic fields are

- 4 scalar fields: (1, 2, 1) \oplus (1, 1, 2)
- 4 vector fields: $(3, 2, 1) \oplus (3, 1, 2)$.

Thus in A-twist, there are four BRST charges transforming as two SU(2) doublets. It turns out that A-twist topological theories is the dimensional reduction of twisted d = 4, $\mathcal{N} = 4$ theories with GL twist and VW twist.

¹¹Note that $SU(2)_E \rightarrow SU(2)_2$ is same as B twist.

B-twist

After twisting, the fields transform under $SU(2)'_E \times SU(2)_2 \times SU(2)_3$ as

$$\phi \to (2,2,1) \oplus (1,1,3)$$

$$\psi \to (1,1,2) \oplus (3,1,2) \oplus (2,2,2).$$
(8.1.163)
(8.1.164)

The bosonic field content is

- 3 scalar fields: (1, 1, 3)
- 2 spinor fields: (2, 2, 1)
- gauge fields: (3,1,1)

and the fermionic field content is

- 2 scalar fields: (1, 1, 2)
- 2 vector fields: (3, 1, 2)
- 4 spinor fields: (2, 2, 2).

In B-twist, we have two BRST charges transforming as a SU(2) doublet. B-twist topological theories are the dimensional reduction of twisted d = 4, $\mathcal{N} = 4$ theories with DW twist.

8.1.7 d = 3, $\mathcal{N} = 8$ SYM theories on $\mathbb{R} \times \Sigma$

Now consider a three-dimensional $\mathcal{N} = 8$ SYM theories on $M_3 = \mathbb{R} \times \Sigma$.

Before twisting, fields transform under $SO(2)_E \times Spin(7)_R$ as

$$\phi: 7_0$$
 (8.1.165)

$$\psi: \mathbf{8}_+ \oplus \mathbf{8}_- \tag{8.1.166}$$

$$A_{\mu} : \mathbf{1}_{-2} \oplus \mathbf{1}_{0} \oplus \mathbf{1}_{2}. \tag{8.1.167}$$

To determine the homomorphism, we consider the decomposition of Spin(7) under SO(2) as

A-twist :
$$Spin(7) \supset SO(5) \times SO(2)$$
 (8.1.168)

B-twist :
$$Spin(7) \supset SO(3) \times SO(4) \supset SO(3) \times SO(2)_1 \times SO(2)_2$$
 (8.1.169)

$$C-\text{twist}: Spin(7) \supset SO(6) \supset SO(2)_1 \times SO(2)_2 \times SO(2)_3$$
(8.1.170)
A-twist

Under (8.1.168), 7 and 8 of $Spin(7)_R$ decomposed as

$$7 = 5_0 \oplus \mathbf{1}_{-2} \oplus \mathbf{1}_2 \tag{8.1.171}$$

$$\mathbf{8} = \mathbf{4}_+ \oplus \overline{\mathbf{4}}_-. \tag{8.1.172}$$

Then after twisting, fields transform under $SO(2)_E \times SO(5)_R$ as

$$7_0 \to 5_0 \oplus 1_{-2} \oplus 1_2$$
 (8.1.173)

$$\mathbf{8}_{+} \oplus \mathbf{8}_{-} \to \mathbf{4}_{2} \oplus \overline{\mathbf{4}}_{0} \oplus \mathbf{4}_{0} \oplus \overline{\mathbf{4}}_{-2} \tag{8.1.174}$$

$$\mathbf{1}_{-2} \oplus \mathbf{1}_0 \oplus \mathbf{1}_2 \to \mathbf{1}_{-2} \oplus \mathbf{1}_0 \oplus \mathbf{1}_2. \tag{8.1.175}$$

Thus there are eight BRST charges in A-twisted d = 3, $\mathcal{N} = 8$ SYM theory on $\mathbb{R} \times \Sigma$.

B-twist

Under (8.1.169), 7 and 8 of $Spin(7)_R$ decomposed as

$$7 = \mathbf{3}_{00} \oplus \mathbf{1}_{0-2} \oplus \mathbf{1}_{02} \oplus \mathbf{1}_{20} \oplus \mathbf{1}_{-20} \tag{8.1.176}$$

$$8 = 2_{++} \oplus 2_{+-} \oplus 2_{-+} \oplus 2_{--}. \tag{8.1.177}$$

We normalize $SO(2)_1$, $SO(2)_2$ charges by dividing by two and simply take a sum of all of the charges including the original rotational $SO(2)_E$ charges. Performing this twisting, fields transform under $SO(2)'_E \times SO(3)_R$ as

$$7_0 \to 3_0 \oplus 2(1_+) \oplus 2(1_-)$$
 (8.1.178)

$$\mathbf{8}_{+} \oplus \mathbf{8}_{-} \to \mathbf{2}_{2} \oplus 2(\mathbf{2}_{+}) \oplus 2(\mathbf{2}_{0}) \oplus 2(\mathbf{2}_{-}) \oplus \mathbf{2}_{-2}$$
(8.1.179)

$$\mathbf{1}_{-2} \oplus \mathbf{1}_0 \oplus \mathbf{1}_2 \to \mathbf{1}_{-2} \oplus \mathbf{1}_0 \oplus \mathbf{1}_2. \tag{8.1.180}$$

Thus there are four BRST charges in B-twisted d = 3, $\mathcal{N} = 8$ SYM theory on $\mathbb{R} \times \Sigma$.

C-twist

Under (8.1.170), 7 and 8 of $Spin(7)_R$ decomposed as

$$7 = \mathbf{1}_{200} \oplus \mathbf{1}_{-200} \oplus \mathbf{1}_{020} \oplus \mathbf{1}_{0-20} \oplus \mathbf{1}_{002} \oplus \mathbf{1}_{00-2} \oplus \mathbf{1}_{000}$$

$$(8.1.181)$$

$$8 = 1_{+++} \oplus 1_{++-} \oplus 1_{+-+} \oplus 1_{+--} \oplus 1_{-++} \oplus 1_{-+-} \oplus 1_{--+} \oplus 1_{---}.$$
 (8.1.182)

We normalize $SO(2)_1$, $SO(2)_2$, $SO(2)_3$ charges by dividing by three and simply take a sum of all of the charges including the original rotational $SO(2)_E$ charges. Performing this twisting, fields transform under $SO(2)'_E$ as

$$7_0 \to 3(\mathbf{1}_{\frac{2}{3}}) \oplus 3(\mathbf{1}_{-\frac{2}{3}}) \oplus \mathbf{1}_0$$
 (8.1.183)

$$\mathbf{8}_{+} \oplus \mathbf{8}_{-} \to 2(\mathbf{1}_{0}) \oplus 2(\mathbf{1}_{\frac{2}{3}}) \oplus 2(\mathbf{1}_{-\frac{2}{3}}) \oplus \mathbf{1}_{\frac{4}{3}} \oplus \mathbf{1}_{-\frac{4}{3}}$$
(8.1.184)

$$\mathbf{1}_{-2} \oplus \mathbf{1}_0 \oplus \mathbf{1}_2 \to \mathbf{1}_{-2} \oplus \mathbf{1}_0 \oplus \mathbf{1}_2. \tag{8.1.185}$$

Thus there are two BRST charges in B-twisted d = 3, $\mathcal{N} = 8$ SYM theory on $\mathbb{R} \times \Sigma$.

8.1.8 d = 2, $\mathcal{N} = 8$ SYM theories

The global symmetry of the theory is $Spin(8)_R$ R-symmetry. The field content is

- 8 scalar fields ϕ^i
- fermionic fields ψ
- gauge field A_{μ} .

Before topological twisting, fields transform under $SO(2)_E \times Spin(8)_R$ as

$$\boldsymbol{\phi}: \mathbf{8}_{v0} \tag{8.1.186}$$

$$\psi: \mathbf{8}_{c+} \oplus \mathbf{8}_{s-} \tag{8.1.187}$$

$$A_{\mu}: \mathbf{1}_{-2} \oplus \mathbf{1}_{2}. \tag{8.1.188}$$

To perform the topological twisting, we pick a homomorphism $\pi : SO(2)_E \rightarrow Spin(8)_R$ and replace $SO(2)_E$ by $SO(2)'_E = (1 + \pi)(SO(2)_E) \subset SO(2)_E \times Spin(8)_R$.

The homomorphism π is determined by the decomposition of Spin(8) under SO(2) and the embedding of SO(2) in Spin(8). Although there are many possible decompositions, we consider the following two types of branching

A-twist :
$$Spin(8)_R \supset SO(6)_R \times SO(2)_1$$
 (8.1.189)
B-twist : $Spin(8)_R \supset SO(4)_1 \times SO(4)_2 \supset SO(4)_1 \times SU(2)_1 \times SU(2)_2$
 $\supset SO(4)_1 \times SU(2)_1 \times SO(2)_2$ (8.1.190)

A-twist

Under (8.1.189), $\mathbf{8}_v$, $\mathbf{8}_s$ and $\mathbf{8}_c$ of $Spin(\mathbf{8})_R$ decomposed as

$$\mathbf{8}_{v} = \mathbf{6}_{0} \oplus \mathbf{1}_{2} \oplus \mathbf{1}_{-2} \tag{8.1.191}$$

$$\mathbf{8}_{s} = \mathbf{4}_{+} \oplus \mathbf{4}_{-}$$
 (8.1.192)

$$\mathbf{8}_{c} = \mathbf{4}_{-} \oplus \mathbf{4}_{+}.$$
 (8.1.193)

Then the choice of π amounts to $SO(2)'_E \to SO(2)_1$ and the fields transform under $SO(2)_E \times SO(6)_R$ as

$$\mathbf{8}_{v} \rightarrow \mathbf{6}_{0} \oplus \mathbf{1}_{2} \oplus \mathbf{1}_{-2} \tag{8.1.194}$$

$$\mathbf{8}_s \to \mathbf{4}_0 \oplus \overline{\mathbf{4}}_{-2} \tag{8.1.195}$$

$$\mathbf{8}_{c} \rightarrow \mathbf{4}_{0} \oplus \overline{\mathbf{4}}_{2} \tag{8.1.196}$$

There are eight supercharges transforming 4 of $SO(6)_R$. This is just the dimensional reduction of A-twisted d = 3, $\mathcal{N} = 8$ SYM theory.

To see this, let us consider the further decomposition

$$SO(6)_R \simeq SU(4)_R \supset SU(2)_1 \times SU(2)_2 \times U(1).$$
 (8.1.197)

Under (8.1.197), 6 and 4 decomposed as ¹²

$$\mathbf{6} = (\mathbf{2}, \mathbf{2})_0 \oplus (\mathbf{1}, \mathbf{1})_2 \oplus (\mathbf{1}, \mathbf{1})_{-2} \tag{8.1.198}$$

$$4 = (2, 1)_{+} \oplus (1, 2)_{-} \tag{8.1.199}$$

$$\overline{4} = (2,1)_{-} \oplus (1,2)_{+}.$$
 (8.1.200)

Then the fields transform under $SO(2)'_E \times SU(2)_1 \times SU(2)_2$ as

$$\mathbf{8}_{v} \to (\mathbf{2}, \mathbf{2})_{0} \oplus (\mathbf{1}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{1})_{2} \oplus (\mathbf{1}, \mathbf{1})_{-2}$$
(8.1.201)

$$\mathbf{8}_{s} \to (\mathbf{2}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{2})_{0} \oplus (\mathbf{2}, \mathbf{1})_{-2} \oplus (\mathbf{1}, \mathbf{2})_{-2}$$
(8.1.202)

$$\mathbf{8}_{c} \to (\mathbf{2}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{2})_{0} \oplus (\mathbf{2}, \mathbf{1})_{2} \oplus (\mathbf{1}, \mathbf{2})_{2}$$
(8.1.203)

Thus bosonic field content is

- 6 scalar fields: $2(1, 1)_0 \oplus (2, 2)_0$
- 1-form: $(1,1)_2 \oplus (1,1)_{-2}$

and the fermionic field content is

- 8 scalar fields: $2(2,1)_0 \oplus 2(1,2)_0$
- 4 1-forms: $(2, 1)_2 \oplus (2, 1)_{-2} \oplus (1, 2)_2 \oplus (1, 2)_{-2}$.

The two bosonic scalars $2(1, 1)_0$ correspond to the third components of the gauge field and 1-form in three dimensions. Also the four fermionic scalars $(2, 1)_0 \oplus (1, 2)_0$ are the third components of the 1-form in three dimensions.

¹²As in (8.1.100), we have other possible decompositions. (8.1.197) is same as GL twist.

B-twist

Under (8.1.190), $\mathbf{8}_v$, $\mathbf{8}_s$ and $\mathbf{8}_c$ of Spin(8) decomposed as

$$\begin{aligned} \mathbf{8}_{v} = (\mathbf{4}, \mathbf{1}) \oplus (\mathbf{1}, \mathbf{4}) \\ = (\mathbf{4}, \mathbf{1}, \mathbf{1}) \oplus (\mathbf{1}, \mathbf{2}, \mathbf{2}) = (\mathbf{4}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{2})_{+} \oplus (\mathbf{1}, \mathbf{2})_{-} \end{aligned} \tag{8.1.204} \\ \mathbf{8}_{s} = (\mathbf{2}, \mathbf{2}) \oplus (\mathbf{2}', \mathbf{2}') \end{aligned}$$

$$= (2,2,1) \oplus (2',1,2) = (2,2)_0 \oplus (2',1)_+ \oplus (2',1)_-$$
(8.1.205)

$$\begin{aligned} \mathbf{8}_{c} = (\mathbf{2}, \mathbf{2}') \oplus (\mathbf{2}', \mathbf{2}) \\ = (\mathbf{2}, \mathbf{1}, \mathbf{2}) \oplus (\mathbf{2}', \mathbf{2}, \mathbf{1}) = (\mathbf{2}, \mathbf{1})_{+} \oplus (\mathbf{2}, \mathbf{1})_{-} \oplus (\mathbf{2}', \mathbf{2})_{0}. \end{aligned}$$
 (8.1.206)

Then the choice of π amounts to $SO(2)_E \rightarrow SO(2)_2$ and the fields transform under $SO(2)'_E \times SO(4)_R \times SO(2)_1$ as

$$\mathbf{8}_{v} \to (\mathbf{4}, \mathbf{1})_{0} \oplus (\mathbf{1}, \mathbf{2})_{+} \oplus (\mathbf{1}, \mathbf{2})_{-}$$
 (8.1.207)

$$\mathbf{8}_{s} \to (\mathbf{2}, \mathbf{2})_{-} \oplus (\mathbf{2}', \mathbf{1})_{0} \oplus (\mathbf{2}', \mathbf{1})_{-2}$$
 (8.1.208)

$$\mathbf{8}_{c} \to (\mathbf{2},\mathbf{1})_{2} \oplus (\mathbf{2},\mathbf{1})_{0} \oplus (\mathbf{2}',\mathbf{2})_{+}.$$
 (8.1.209)

The bosonic field contnt is

- 4 scalar fields: (**4**, **1**)₀
- 4 spinors: $(1, 2)_+ \oplus (1, 2)_-$

and the fermionic field content is

- 4 scalar fields: $(2', 1)_0 \oplus (2, 1)_0$
- 8 spinors: (2, 2)_− ⊕ (2', 2)₊
- 2 1-forms: $(2', 1)_{-2} \oplus (2, 1)_2$.

There are four BRST charges in B-twist.

8.1.9 d = 3, $\mathcal{N} = 4$ Chern-Simons matter theories

Gaiotto and Witten gave a general prescription for coupling Chern-Simons theory to hypermultiplets, which allows for a new large class of three-dimensional $\mathcal{N} = 4$ supersymmetric gauge theories [304]. Gaiotto-Witten theory can be regarded as a three-dimensional $\mathcal{N} = 4$ gauged sigma-model with a hyperkähler target space X.

The field content is

• gauge field A_{μ}^{m}

- hypermultiplet boson q^A_{α}
- hypermultiplet fermion $\psi^A_{\dot{\alpha}}$
- twisted hypermultiplet $\tilde{q}^A_{\dot{\alpha}}$
- twisted hypermultiplet $\tilde{\psi}^A_{\alpha}$

where *m* is the adjoint indices raised by invariant quadratic form k^{mn} of the gauge group. The gauge group is a subgroup of Sp(2n) and we denote the anti-Hermitian generators of the gauge group by $(t^m)^A_B(A, B, \dots = 1, \dots, 2n)$, which satisfy

$$[t^m, t^n] = f^{mn}_{\ \ p} t^p, \quad t_{AB} = \omega_{AC} t^C_B \tag{8.1.210}$$

where ω_{AB} are the anti-symmetric invariant tensor.

The hyper-multiplet fields satisfy the reality condition

$$(q^A_{\alpha})^* = \epsilon^{\alpha\beta} \omega_{AB} q^B_{\beta}, \qquad (8.1.211)$$

$$(\psi^A_{\alpha})^* = \epsilon^{\dot{\alpha}\dot{\beta}}\omega_{AB}\psi^B_{\dot{\beta}} \tag{8.1.212}$$

where $(\alpha, \beta; \dot{\alpha}, \dot{\beta})$ are the indices of $SU(2) \times SU(2)$ R-symmetry.

For $\mathcal{N} = 4$ supersymmetry, t_{AB}^m satisfy the fundamental identity

$$k_{mn}t^m_{(AB}t^n_{C)D} = 0 (8.1.213)$$

where A, B, C, \cdots are symmetrized. (8.1.213) is nothing but the Jacobi identity for three fermionic generators of a Lie superalgebra

$$[M^m, M^n] = f^{mn}_{\ \ p} M^p, \qquad (8.1.214)$$

$$[M^m, Q_A] = Q_B(t^m)^B_A, (8.1.215)$$

$$\{Q_A, Q_B\} = t_{AB}^m M_m. \tag{8.1.216}$$

Lagrangian of the Gaiotto-Witten theory is given by ¹³

$$\mathcal{L} = \frac{1}{2} \epsilon^{\mu\nu\lambda} \left(k_{mn} A^m_{\mu} \partial_{\nu} A^n_{\lambda} + \frac{1}{3} f_{mnp} A^m_{\mu} A^n_{\nu} A^p_{\lambda} \right) + \omega_{AB} \left(-\epsilon^{\alpha\beta} D q^A_{\alpha} D q^B_{\beta} + i \epsilon^{\dot{\alpha}\dot{\beta}} \psi^A_{\dot{\alpha}} D_{\mu} \gamma^{\mu} \psi^B_{\dot{\beta}} \right) - i k_{mn} \epsilon^{\alpha\beta} \epsilon^{\dot{\gamma}\dot{\delta}} j^m_{\alpha\dot{\gamma}} j^n_{\beta\dot{\delta}} - \frac{1}{12} f_{mnp} (\mu^m)^{\alpha}_{\beta} (\mu^n)^{\beta}_{\gamma} (\mu^p)^{\gamma}_{\alpha}$$
(8.1.217)

¹³An overall coefficients of the Lagrangian should satisfy an integrality condition to make the quantum theory well-defined. But here we suppress them.

where

$$\mu^m_{\alpha\beta} := t^m_{AB} q^A_\alpha q^B_\beta, \tag{8.1.218}$$

$$j^m_{\alpha\dot{B}} := t^m_{AB} q^A_\alpha \psi^B_{\dot{\beta}}, \tag{8.1.219}$$

$$\rho^m_{\dot{\alpha}\dot{\beta}} := t^m_{AB} \psi^A_{\dot{\alpha}} \psi^B_{\dot{\beta}} \tag{8.1.220}$$

are the momentum map multiplet.

The Euclidean Lagrangian is

$$\mathcal{L} = -\frac{i}{2} \epsilon^{\mu\nu\lambda} \left(k_{mn} A^m_{\mu} \partial_{\nu} A^n_{\lambda} + \frac{1}{3} f_{mnp} A^m_{\mu} A^n_{\nu} A^p_{\lambda} \right) - \omega_{AB} \left(-\epsilon^{\alpha\beta} D q^A_{\alpha} D q^B_{\beta} + i \epsilon^{\dot{\alpha}\dot{\beta}} \psi^A_{\dot{\alpha}} D_{\mu} \gamma^{\mu} \psi^B_{\dot{\beta}} \right) + i k_{mn} \epsilon^{\alpha\beta} \epsilon^{\dot{\gamma}\dot{\delta}} j^m_{\alpha\dot{\gamma}} j^n_{\beta\dot{\delta}} + \frac{1}{12} f_{mnp} (\mu^m)^{\alpha}_{\beta} (\mu^n)^{\beta}_{\gamma} (\mu^p)^{\gamma}_{\alpha}.$$
(8.1.221)

Note that it differs from the Lorentzian Lagrangian (8.1.217) by the factor (-i) for Chern-Simons term and an overall sign for the matter terms. Now the fermionic fields do not obey the reality conditions.

The supersymmetry transformations are

$$\delta q^A_{\alpha} = i \epsilon^{\dot{\alpha}}_{\alpha} \psi^A_{\dot{\alpha}}, \qquad (8.1.222)$$

$$\delta\psi^A_{\dot{\alpha}} = \left(D^\mu\gamma_\mu q^A_\alpha + \frac{1}{3}k_{mn}(t^m)^A_B q^B_\beta(\mu^n)^\beta_\alpha\right)\epsilon^\alpha_{\dot{\alpha}},\tag{8.1.223}$$

$$\delta A^m_\mu = i \epsilon^{\alpha \dot{\alpha}} \gamma_\mu j^m_{\alpha \dot{\alpha}}. \tag{8.1.224}$$

The supersymmetry parameter ϵ transforms as (2, 2) in $SU(2) \times SU(2)$ R-symmetry and satisfies the reality condition

$$(\epsilon^{\dot{\alpha}}_{\alpha})^* = -\epsilon^{\alpha\beta}\epsilon_{\dot{\alpha}\dot{\beta}}\epsilon^{\dot{\beta}}_{\beta}.$$
(8.1.225)

The supersymmetry transformations are same as in the Lorentzian case.

Furthermore we can add twisted hyper-multiplets $(\tilde{q}^A_{\dot{\alpha}}, \tilde{\psi}^A_{\alpha})$ to Gaiotto-Witten theory. This is regarded as a non-linear sigma model [304]. The Lagrangian is

$$\mathcal{L} = \frac{1}{2} \epsilon^{\mu\nu\lambda} \left(k_{mn} A^m_{\mu} \partial_{\nu} A^n_{\lambda} + \frac{1}{3} f_{mnp} A^m_{\mu} A^n_{\nu} A^p_{\lambda} \right) + \omega_{AB} \left(-\epsilon^{\alpha\beta} D q^A_{\alpha} D q^B_{\beta} + i \epsilon^{\dot{\alpha}\dot{\beta}} \psi^A_{\dot{\alpha}} D_{\mu} \gamma^{\mu} \psi^B_{\dot{\beta}} \right) + \omega_{AB} \left(-\epsilon^{\dot{\alpha}\dot{\beta}} D \tilde{q}^B_{\dot{\alpha}} + i \epsilon^{\alpha\beta} \tilde{\psi}^A_{\alpha} D_{\mu} \gamma^{\mu} \tilde{\psi}^B_{\beta} \right) - i k_{mn} \left(\epsilon^{\alpha\beta} \epsilon^{\dot{\gamma}\dot{\delta}} j^m_{\alpha\dot{\gamma}} j^n_{\beta\dot{\delta}} + \epsilon^{\dot{\alpha}\dot{\beta}} \epsilon^{\gamma\delta} \tilde{j}^m_{\dot{\alpha}\dot{\gamma}} \tilde{j}^n_{\dot{\beta}\dot{\delta}} + 4 \epsilon^{\alpha\gamma} \epsilon^{\dot{\beta}\dot{\delta}} j^m_{\alpha\dot{\beta}} \tilde{j}^n_{\dot{\alpha}\dot{\beta}} - \epsilon^{\alpha\gamma} \epsilon^{\beta\delta} \mu^m_{\alpha\beta} \rho^{\gamma}_{\dot{\gamma}\dot{\delta}} - \epsilon^{\alpha\gamma} \epsilon^{\beta\delta} \mu^m_{\alpha\beta} \rho_{\gamma\delta} \right) - \frac{1}{12} f_{mnp} (\mu^m)^{\alpha}_{\beta} (\mu^n)^{\beta}_{\gamma} (\mu^p)^{\alpha}_{\alpha} + (\tilde{\mu}^m)^{\dot{\alpha}}_{\dot{\beta}} (\tilde{\mu}^n)^{\dot{\beta}}_{\dot{\gamma}} (\tilde{\mu}^\rho)^{\dot{\gamma}}_{\dot{\alpha}} - \frac{1}{2} \tilde{\mu}^{mn} (\mu_m)^{\alpha}_{\beta} (\mu_n)^{\beta}_{\alpha} - \frac{1}{2} \mu^{mn} (\tilde{\mu}_m)^{\dot{\alpha}}_{\dot{\beta}} (\tilde{\mu}_n)^{\dot{\beta}}_{\dot{\alpha}}$$
(8.1.226)

where the twisted moment map is defined by

$$\mu^{mn} = \epsilon^{\alpha\beta} (t^m t^n)_{AB} q^A_{\alpha} q^B_{\beta}, \qquad (8.1.227)$$

$$\tilde{\mu}^{mn} = \epsilon^{\dot{\alpha}\dot{\beta}} (\bar{t}^m \bar{t}^n)_{AB} \tilde{q}^A_{\dot{\alpha}} \tilde{q}^B_{\dot{\beta}}.$$
(8.1.228)

The supersymmetry transformation is

$$\delta q^A_{\alpha} = i \epsilon^{\dot{\alpha}}_{\alpha} \psi^A_{\dot{\alpha}}, \qquad (8.1.229)$$

$$\delta\psi^{A}_{\dot{\alpha}} = \left(D_{\mu}\gamma^{\mu}q^{A}_{\alpha} + \frac{1}{3}(t_{m})^{A}_{B}q^{B}_{\beta}(\mu^{m})^{\beta}_{\alpha}\right)\epsilon^{\alpha}_{\dot{\alpha}},\tag{8.1.230}$$

$$\delta \tilde{\psi}^{A}_{\alpha} = \left(D_{\mu} \gamma^{\mu} q^{A}_{\dot{\alpha}} + \frac{1}{3} (\tilde{t}_{m}) \tilde{q}^{B}_{\dot{\beta}} (\tilde{\mu}^{m})^{\dot{\beta}}_{\dot{\alpha}} \right) \epsilon^{\dot{\alpha}}_{\alpha} - (\tilde{t}_{m})^{A}_{B} \tilde{q}^{B}_{\dot{\beta}} (\mu^{m})^{\beta}_{\alpha} \epsilon^{\beta \dot{\beta}}, \tag{8.1.231}$$

$$\delta A^m_\mu = i\epsilon^{\alpha\dot{\alpha}}\gamma_\mu (j^m_{\alpha\dot{\alpha}} + \tilde{j}^m_{\dot{\alpha}\alpha}). \tag{8.1.232}$$

The topological twisting for Gaiotto-Witten theory was discussed in [305, 306].

1. flat target space X

If the target *X* is flat, Gaiotto-Witten theory has $SU(2) \times SU(2)$ R-symmetry. The topologically twisted theory is equivalent to the pure Chern-Simons theory whose gauge group is a supergroup [305]. In other words, the topologically twisted Gaiotto-Witten theory is obtained from the supergroup Chern-Simons theory by gauge fixing the odd part of the supergroup and the even part of the supergroup gives rise to gauge group *G*.

2. general target space *X*

For general target *X*, Gaiotto-Witten theory has SU(2) R-symmetry. The topologically twisted Gaiotto-Witten theory can be interpreted as a gauged Rozansky-Witten theory [307], that is a hybrid of Chern-Simons and Rozansky-Witten theory [305]. It is associated to a quadruple:

- (a) *G*: a compact Lie group
- (b) κ : invariant metric on the Lie algebra
- (c) X: hyperkähler manifold with a tri-holomorphic action of G
- (d) *I*: complex structure on *X* such that the complex moment map with respect to the complex symplectic form Ω_I is isotropic with respect to κ

Before twisting, fields and supercharges transform under $SU(2)_E \times SU(2)_1 \times$

 $SU(2)_2$ as

$$q:(\mathbf{1},\mathbf{2},\mathbf{1})$$
 (8.1.233)

$$\psi: (2, 1, 2)$$
 (8.1.234)

$$Q: (2, 2, 2).$$
 (8.1.235)

We may also have fields of twisted hypermultiplet, which transform as

$$\tilde{q}: (1,1,2)$$
 (8.1.236)

$$\tilde{\psi}$$
: (2, 2, 1). (8.1.237)

Depending on which SU(2) factor we use, we may think two types of twisting

A-twist :
$$SU(2)_E \to SU(2)_2$$
 (8.1.238)

B-twist :
$$SU(2)_E \to SU(2)_1$$
. (8.1.239)

However, if both hypermultiplet and twisted hypermultiplet are present, A-twist and B-twist are the same. We call it AB-twist.

A-twist

After A-twisting, the fields and supercharges transform under $SU(2)'_E \times SU(2)$ as

$$q \to (\mathbf{1}, \mathbf{2}) \tag{8.1.240}$$

$$\psi \rightarrow (\mathbf{1}, \mathbf{1}) \oplus (\mathbf{3}, \mathbf{1})$$
 (8.1.241)

$$Q \to (1,2) \oplus (3,1).$$
 (8.1.242)

Thus in the bosonic field content we have

• 2 scalar fields: (1,2)

and in the fermionic field content we include

- scalar field (1, 1)
- 1-form (3, 1).

There are two BRST charges in A-twisted theory.

1. bosonic fields

In A-twist all of the hypermultiplet scalars remain scalars.

2. fermionic fields

We decompose the fermionic fields under $SU(2)_E \times SU(2)_1 \times SU(2)_2$ as

$$(\psi_{\alpha'})^{A}_{\dot{\alpha}} = \frac{1}{\sqrt{2}} \left(\psi^{A} \mathbb{I}_{\alpha'\dot{\alpha}} + \Psi^{A}_{i}(\sigma_{i})_{\alpha'\dot{\alpha}} \right) \sigma_{2}^{-1}$$
(8.1.243)

where $\alpha', \dot{\alpha}$ are the indices of $SU(2)_E$, $SU(2)_2$ respectively. $A = 1, 2, \dots, 2n$ is again the index of Sp(2n).

3. supersymmetries

For the supersymmetries we expand as

$$(\epsilon^{\alpha'})^{\dot{\alpha}}_{\alpha} = \frac{1}{\sqrt{2}} \left(\epsilon_{\alpha} \mathbb{I}^{\alpha' \dot{\alpha}} + \epsilon_{i\alpha} (\sigma_i)^{\alpha' \dot{\alpha}} \right) (\sigma_2)^{-1}.$$
(8.1.244)

B-twist

After B-twisting, the fields tranform under $SU(2)'_E \times SU(2)$ as

$$q \to (\mathbf{2}, \mathbf{1}) \tag{8.1.245}$$

$$\psi \rightarrow (\mathbf{2}, \mathbf{2})$$
 (8.1.246)

$$Q \to (1, 2) \oplus (3, 2).$$
 (8.1.247)

Thus the bosonic field content is

• spinor field: (2, 1)

and the fermionic field content is

• 2 spinor fields (2, 2).

We have two BRST charges in B-twisted theory.

AB-twist

After twisting, the fields transform under $SU(2)'_E \times SU(2)_2$ as

$$q \to (\mathbf{1}, \mathbf{2}) \tag{8.1.248}$$

$$\psi \to (\mathbf{1}, \mathbf{1}) \oplus (\mathbf{3}, \mathbf{1}) \tag{8.1.249}$$

$$\tilde{q} \rightarrow (\mathbf{2}, \mathbf{1})$$
 (8.1.250)

$$ilde{\psi}
ightarrow (\mathbf{2},\mathbf{2})$$
 (8.1.251)

$$Q \to (\mathbf{1}, \mathbf{2}) \oplus (\mathbf{3}, \mathbf{2}). \tag{8.1.252}$$

Thus the bosonic field content is

- 2 scalar fields : (1, 2)
- spinor field: (2, 1)

and the fermionic field content is

- scalar fields : (1, 1)
- 1-form : (3, 1)
- 2 spinor fields: (2, 2).

Again we have two BRST charges in AB-twisted theory.

8.1.10 d = 3, $\mathcal{N} = 5$ Chern-Simons matter theories

Three-dimensional $N \ge 5$ theories can be understood in the Gaiotto-Witten framework by adding twisted hypermultiplets [308]. The target spaces of $N \ge 5$ theories are only flat spaces and their orbifolds.

We may consider the decomposition of $SO(5)_R$ R-symmetry under $SO(3) \simeq SU(2)$ as

$$SO(5) \supset SO(2) \times SO(3) \tag{8.1.253}$$

$$SO(5) \supset SO(4) \supset SU(2)_1 \times SU(2)_2.$$
 (8.1.254)

Under (8.1.253), **5** of $SO(5)_R$ decomposed as

$$\mathbf{5} = \mathbf{3}_0 \oplus \mathbf{1}_{-2} \oplus \mathbf{1}_2. \tag{8.1.255}$$

Noting that

$$\mathbf{2}_{0} \times (\mathbf{3}_{0} \oplus \mathbf{1}_{-2} \oplus \mathbf{1}_{2}) = \mathbf{2}_{0} \oplus \mathbf{4}_{0} \oplus \mathbf{2}_{2} \oplus \mathbf{2}_{-2}, \qquad (8.1.256)$$

we see that there are no BRST charges.

On the other hand, under (8.1.254), 5 and 4 of $SO(5)_R$ decomposed as

$$\mathbf{5} = \mathbf{4} + \mathbf{1} = (\mathbf{2}, \mathbf{2}) \oplus (\mathbf{1}, \mathbf{1}) \tag{8.1.257}$$

$$4 = 4 = (2, 1) \oplus (1, 2). \tag{8.1.258}$$

This is nothing but the AB-twist in d = 3, $\mathcal{N} = 4$ Chern-Simons matter theory.

8.1.11 d = 3, $\mathcal{N} = 6$ Chern-Simons matter theories

We may consider the decomposition of $SO(6)_R$ R-symmetry under $SO(3) \simeq SU(2)$ as

$$SO(6) \supset SO(3) \times SO(3)$$
 (8.1.259)

$$SO(6) \supset SO(2) \times SO(4) \supset SO(2) \times SU(2)_1 \times SU(2)_2. \tag{8.1.260}$$

Under (8.1.259), **5** of $SO(5)_R$ decomposed as

$$\mathbf{6} = (\mathbf{3}, \mathbf{1}) \oplus (\mathbf{1}, \mathbf{3}). \tag{8.1.261}$$

Noting that

$$\mathbf{2} \times \mathbf{3} = \mathbf{2} \oplus \mathbf{4} \oplus \mathbf{2} \oplus \mathbf{2},\tag{8.1.262}$$

we see that there are no BRST charges.

On the other hand, under (8.1.260), **6** of $SO(6)_R$ decomposed as

$$\mathbf{6} = \mathbf{4}_0 \oplus \mathbf{1}_{-2} \oplus \mathbf{1}_2 = (\mathbf{2}, \mathbf{2})_0 \oplus (\mathbf{1}, \mathbf{1})_{-2} \oplus (\mathbf{1}, \mathbf{1})_2. \tag{8.1.263}$$

As seen from the appearance of (2, 2), this is nothing but the AB-twist in d = 3, $\mathcal{N} = 4$ Chern-Simons matter theory.

8.1.12 d = 3, $\mathcal{N} = 8$ Chern-Simons matter theories

To perfom the topological twisting, we put the BLG theory on a three-dimensional Euclidean space. The fermionic fields and supersymmetry parameters are defined as eleven-dimensional fermions and their conjugate are given by

$$\overline{\Psi} := \Psi^T \mathcal{C} \tag{8.1.264}$$

where C is a eleven-dimensional matrix satisfying

$$\mathcal{C}^{T} = -\mathcal{C}, \quad \mathcal{C}\Gamma^{M}\mathcal{C}^{-1} = -(\Gamma^{M})^{T}.$$
(8.1.265)

Gamma matrix Γ^M is the representation of eleven-dimensional Clifford algebra

$$\left\{\Gamma^{M},\Gamma^{N}\right\} = 2g^{MN}, \quad \Gamma^{11} := i\Gamma^{12\cdots 10}.$$
 (8.1.266)

 Γ^M can be decomposed under $SO(11) \supset SO(3) \times SO(8)$ as

$$\begin{cases} \Gamma^{i} = \sigma_{i} \otimes \tilde{\Gamma}^{9} \\ \Gamma^{I+3} = \mathbb{I} \otimes \tilde{\Gamma}^{I} \end{cases}$$
(8.1.267)

where $\tilde{\Gamma}^9 := \tilde{\Gamma}^{1\cdots 8}$. Note that

$$\Gamma^{123} = i\Gamma^{45\dots 11} = i(\mathbb{I} \otimes \tilde{\Gamma}^9).$$
(8.1.268)

The fermionic fields Ψ are $\mathbf{8}_c$ of $SO(8)_R$, so they satisfy the chirality condition

$$\Gamma^{45\dots 11}\Psi = -\Psi, \tag{8.1.269}$$

$$\Gamma^{123}\Psi = -i\Psi. (8.1.270)$$

The Euclidean BLG Lagrangian is given by

$$\mathcal{L} = \frac{1}{2} (D_{\mu} X^{I}, D^{\mu} X^{I}) - \frac{i}{2} (\overline{\Psi}, \Gamma^{\mu} D_{\mu} \Psi) - \frac{i}{4} \left(\overline{\Psi} \Gamma^{IJ} [X^{I}, X^{J}, \Psi] \right) + \frac{1}{12} \left([X^{I}, X^{J}, X^{K}], [X^{I}, X^{J}, X^{K}] \right) - \frac{i}{2} \epsilon^{\mu\nu\lambda} \left[\operatorname{Tr} \left(A_{\mu a b} \partial_{\nu} \tilde{A}^{a b}_{\lambda} \right) + \frac{2}{3} \operatorname{Tr} \left(A_{\mu a b} \tilde{A}^{a}_{\nu g} \tilde{A}^{b}_{\lambda g} \right) \right],$$

$$(8.1.271)$$

which differs from Lorentzian case by the factor (-i) for the Chern-Simons terms and a overall sign factors for matter terms.

The supersymmetry transformations are

$$\delta X_a^I = i\overline{\epsilon}\Gamma^I \Psi_a, \tag{8.1.272}$$

$$\delta \Psi_a = D_\mu X_a^I \Gamma^\mu \Gamma^I \epsilon - \frac{1}{6} X_b^I X_c^J X_d^K f^{bcd}_{\ a} \Gamma^{IJK} \epsilon, \qquad (8.1.273)$$

$$\delta \tilde{A}^a_{\mu b} = i \bar{\epsilon} \Gamma_\mu \Gamma^I X^I_c \Psi_d f^{cda}{}_b. \tag{8.1.274}$$

This is exactly same as in the Lorentzian transformations (4.1.49)-(4.1.51). ϵ is the unbroken supersymmetry parameters obeying¹⁴

$$\Gamma^{34\dots11}\epsilon = \epsilon, \tag{8.1.275}$$

$$\Gamma^{123}\epsilon = i\epsilon \tag{8.1.276}$$

Before topological twisting, fields and supersymmetry parameter transform under $SU(2)_E \times SO(8)_R$ as

$$X: (\mathbf{1}, \mathbf{8}_{v}) \tag{8.1.277}$$

$$\Psi: (\mathbf{2}, \mathbf{8}_{c}) \tag{8.1.278}$$

$$\boldsymbol{\epsilon}: (\mathbf{2}, \mathbf{8}_s). \tag{8.1.279}$$

Although there are many possible ways of the twisting, we consider the following decomposition $SO(8)_R$ under SO(3) [303]

A-twist :
$$SO(8) \supset SO(5) \times SO(3)$$
 (8.1.280)

B-twist :
$$SO(8) \supset G_2 \supset SU(2)_1 \times SU(2)_2$$
. (8.1.281)

¹⁴ This convention is different from that in [306]. There are two choices for $\tilde{\Gamma}^9 = \pm \tilde{\Gamma}^{12\cdots 8}$. We take $\tilde{\Gamma}^9 = +\tilde{\Gamma}^{12\cdots 8}$.

A-twist

Under (8.1.280), we can obtain the decomposition of $\mathbf{8}_v$, $\mathbf{8}_s$ and $\mathbf{8}_c$ as

$$\mathbf{8}_v = (\mathbf{5}, \mathbf{1}) \oplus (\mathbf{1}, \mathbf{3})$$
 (8.1.282)

$$\mathbf{8}_{s} = (\mathbf{4}, \mathbf{2}) \tag{8.1.283}$$

$$\mathbf{8}_{c} = (\mathbf{4}, \mathbf{2}). \tag{8.1.284}$$

Then the transformations under $SO(3)'_E \times SO(5)_R$ of the fields are

$$X \to (\mathbf{1}, \mathbf{5}) \oplus (\mathbf{3}, \mathbf{1}) \tag{8.1.285}$$

$$\Psi \to (\mathbf{1}, \mathbf{4}) \oplus (\mathbf{3}, \mathbf{4}) \tag{8.1.286}$$

$$\epsilon \rightarrow (1,4) \oplus (3,4).$$
 (8.1.287)

The bosonic field content is

- 5 scalar fields : (1, 5)
- 1-form : (3, 1)

and the fermionic field content is

- 4 scalar fields : (1, 4)
- 4 1-form : (3, 4).

Therefore there exists four supercharges.

We decompose the gamma matrices under $SO(11) \supset SO(3)_E \times SO(5) \times SO(3)$ as

$$\begin{cases} \Gamma^{i} = \sigma_{i} \otimes \mathbb{I}_{4} \otimes \mathbb{I}_{2} \otimes \sigma_{1} \\ \Gamma^{\mu+3} = \mathbb{I}_{2} \otimes \gamma^{\mu} \otimes \mathbb{I}_{2} \otimes \sigma_{2} \\ \Gamma^{\mu+9} = \mathbb{I}_{2} \otimes \mathbb{I}_{4} \otimes \sigma_{i} \otimes \sigma_{3} \end{cases}$$

$$(8.1.288)$$

where σ_i are Pauli matrices and γ^{μ} are five-dimensional gamma matrices satisfying

$$\{\gamma^{\mu}, \gamma^{\nu}\} = 2\delta^{\mu\nu}\gamma^{5} := \gamma^{1234}.$$
(8.1.289)

The charge conjugation matrix can be expressed as

$$\mathcal{C} = \sigma_2 \otimes \mathcal{C} \otimes \sigma_2 \otimes \mathbb{I}_2 \tag{8.1.290}$$

where σ_2 is a three-dimensional charge conjugation matrix

$$(\sigma_2)^T = -\sigma_2 \sigma_2 \sigma_i \sigma_2^{-1} = -(\sigma_2)^T$$
 (8.1.291)

and *C* is a five-dimensional charge conjugation matrix

$$(C)^{T} = -CC\gamma^{\mu}C^{-1} = (\gamma^{\mu})^{T}.$$
 (8.1.292)

The chirality matrix is given by

$$\Gamma^{123} = -i\Gamma^{45\cdots 11}i(\mathbb{I}\otimes\mathbb{I}\otimes\mathbb{I}\otimes\sigma_1).$$
(8.1.293)

1. bosonic fields

For the bosonic fields we redefine

$$\phi^{I} := X^{I} (I = 4, 5, 6, 7, 8),$$

$$\Phi^{\mu} := X^{\mu+8} (\mu = 1, 2, 3)$$
(8.1.294)

2. fermionic fields

We expand the elven-dimensional fermionic fields Ψ under the decomposition $SO(11) \supset SO(3)_E \times SO(5) \times SO(3)$ as

$$\Psi_{p\alpha q} = \frac{1}{\sqrt{2}} \left(\psi_{\alpha} \mathbb{I}_{pq} + \Psi_{i\alpha} \sigma_{ipq} \right) \sigma_2^{-1}$$
(8.1.295)

where p, q, α are indices of $SO(3)_E, SO(5), SO(3)$ respectively. ψ_{α} and $\Psi_{i\alpha}$ are scalars (1, 4) and a 1-form (3, 4).

B-twist

Under (8.1.281), we can obtain the decomposition of $\mathbf{8}_v$, $\mathbf{8}_s$ and $\mathbf{8}_c$ as

$$\mathbf{8}_{v} = \mathbf{7} + \mathbf{1} = (\mathbf{1}, \mathbf{3}) \oplus (\mathbf{2}, \mathbf{2}) \oplus (\mathbf{1}, \mathbf{1})$$
(8.1.296)

$$\mathbf{8}_{s} = \mathbf{7} + \mathbf{1} = (\mathbf{1}, \mathbf{3}) \oplus (\mathbf{2}, \mathbf{2}) \oplus (\mathbf{1}, \mathbf{1})$$
 (8.1.297)

$$\mathbf{8}_{c} = \mathbf{7} + \mathbf{1} = (\mathbf{1}, \mathbf{3}) \oplus (\mathbf{2}, \mathbf{2}) \oplus (\mathbf{1}, \mathbf{1}). \tag{8.1.298}$$

Choosing the homomorphism as $SU(2)_E \rightarrow SU(2)_1$, the transformations under $SO(3)'_E \times SU(2)'_2$ of the fields are

$$X \to (1,3) \oplus (2,2) \oplus (1,1)$$
 (8.1.299)

$$\Psi \to (2,3) \oplus (1,2) \oplus (3,2) \oplus (2,1)$$
 (8.1.300)

$$\epsilon \rightarrow (\mathbf{2},\mathbf{3}) \oplus (\mathbf{1},\mathbf{2}) \oplus (\mathbf{3},\mathbf{2}) \oplus (\mathbf{2},\mathbf{1}).$$
 (8.1.301)

The bosonic field content is

• 4 scalar fields : (1,3) \oplus (1,1)

• 2 spinor fields : (2, 2)

and the fermionic field content is

- 2 scalar fields : (1, 2)
- 4 spinor fields : $(2,3) \oplus (2,1)$.
- 2 1-form : (3, 2).

Therefore there exists two supercharges.

8.1.13 d = 2, $\mathcal{N} = (2, 2)$ non-linear sigma-model

There are two types global symmetry in the theory, which are called $U(1)_V$ vector R-symmetry and $U(1)_A$ axial R-symmetry ¹⁵.

The field content is

- scalar fields $\phi^I(z, \overline{z})$
- fermionic fields $\psi_{+}^{I}(z,\overline{z}), \psi_{-}^{I}(z,\overline{z})$

The bosonic field $\phi^I(z, \overline{z})$ is a map from 2-dimensional genus *g* Riemann surface Σ to a target space *X* of metric *g*

$$\phi^{I}(z,\overline{z}): \Sigma \to X \tag{8.1.304}$$

where z, \overline{z} are the local coordinates on Σ . The fermionic field ψ_+^I is a section of $K^{\frac{1}{2}} \otimes \phi^*(TX)$ and ψ_-^I is a section of $K^{-\frac{1}{2}} \otimes \phi^*(TX)$

$$\psi^{I}_{\pm}(z,\overline{z}) \in \Gamma(K^{\pm\frac{1}{2}} \otimes \phi^{*}(TX))$$
(8.1.305)

where *TX* is the holomorphic tangent bundle to *X*. *K* and K^{-1} are the canonical and anti-canonical bundle on Σ (i.e. the bundle of (1,0) and (0,1) forms) and $K^{\frac{1}{2}}$ and $K^{-\frac{1}{2}}$ are square roots of these.

$$U(1)_{V}:\begin{cases} (\theta^{+},\overline{\theta}^{+}) & \to (e^{-i\alpha}\theta^{+},e^{i\alpha}\overline{\theta}^{+})\\ (\theta^{-},\overline{\theta}^{-}) & \to (e^{-i\alpha}\theta^{+},e^{i\alpha}\overline{\theta}^{+}) \end{cases}$$
(8.1.302)

$$U(1)_{A}: \begin{cases} (\theta^{+}, \overline{\theta}^{+}) & \to (e^{-i\alpha}\theta^{+}, e^{i\alpha}\overline{\theta}^{+}) \\ (\theta^{-}, \overline{\theta}^{-}) & \to (e^{i\alpha}\theta^{-}, e^{-i\alpha}\overline{\theta}^{-}) \end{cases}.$$

$$(8.1.303)$$

¹⁵Although $SO(2)_E$ rotational symmetry acts on the variables $z, \overline{z}, \theta^{\pm}$ and $\overline{\theta}$ simultaneously, one can construct two U(1) groups that act only on a subset of variables and leave the measure invariant and keep the chiral fields to be chiral

Here we consider the case where we have $\mathcal{N} = (2, 2)$ supersymmetry, which require that X is Kähler. We denote the local complex coordinates by ϕ^i and their complex conjugate by $\phi^{\overline{i}} = \overline{\phi}^{i}$. As the complexified tangent bundle TX has a decomposition as $TX = T^{1,0}X \oplus T^{0,1}X$, (8.1.305) becomes

$$\begin{split} \psi_{+}^{i} &\in \Gamma(K^{\frac{1}{2}} \otimes \phi^{*}(T^{1,0}X)), \\ \psi_{+}^{\bar{i}} &\in \Gamma(K^{\frac{1}{2}} \otimes \phi^{*}(T^{0,1}X)), \\ \psi_{-}^{i} &\in \Gamma(K^{-\frac{1}{2}} \otimes \phi^{*}(T^{1,0}X)), \\ \psi_{-}^{\bar{i}} &\in \Gamma(K^{-\frac{1}{2}} \otimes \phi^{*}(T^{0,1}X)) \end{split}$$

$$(8.1.306)$$

and ψ^i_{\pm} and $\psi^{\bar{i}}_{\pm}$ are left- and right-moving fermionic fields respectively.

The action is

$$S = 2t \int_{\Sigma} d^{2}z \left(\frac{1}{2} g_{IJ} \partial_{z} \phi^{I} \partial_{\bar{z}} \phi^{J} + i g_{i\bar{j}} \psi^{\bar{j}}_{-} D_{z} \psi^{i}_{-} + i g_{i\bar{j}} \psi^{\bar{j}}_{+} D_{\bar{z}} \psi^{i}_{+} + R_{i\bar{j}k\bar{l}} \psi^{i}_{+} \psi^{\bar{j}}_{+} \psi^{k}_{-} \psi^{\bar{l}}_{-} \right)$$
(8.1.307)

where *t* is a coupling constant or a string tension depending on the overall volume of *X* and $R_{i\bar{i}k\bar{l}}$ is the Riemann tensor of target space *X*.

Originally fields transform under $SO(2)_E \times U(1)_V \times U(1)_A$ as in Table 8.7¹⁶. Likewise supersymmetry generators transform as Table 8.8. Depending on which R-symmetry we use, there are two homomorphisms for the twisting

A-twist :
$$U(1)_E \to U(1)_V$$
 (8.1.314)

B-twist
$$:U(1)_E \to U(1)_A$$
. (8.1.315)

The results of the twisting are summarized in Table 8.9 and Table 8.10.

$$e^{i\alpha F_V}:\Phi(x^{\mu},\theta^{\pm},\overline{\theta}^{\pm})\mapsto e^{i\alpha q_V}\Phi(x^{\mu},e^{-i\alpha}\theta^{\pm},e^{i\alpha}\overline{\theta}^{\pm})$$
(8.1.308)

$$e^{i\beta F_A}: \Phi(x^{\mu}, \theta^{\pm}, \overline{\theta}^{\pm}) \mapsto e^{i\beta q_A} \Phi(x^{\mu}, e^{\pm i\beta} \theta^{\pm}, e^{\pm i\beta} \overline{\theta}^{\pm})$$
(8.1.309)

where F_V , F_A are the generators of the vector and the axial R-symmetry and q_V , q_A are the vector and the axial R-charges respectively. Therefore we see that

$$\psi_{+new}^{i} = e^{i\alpha(1-q_{V})}\psi_{+old}^{i}, \qquad \qquad \psi_{+new}^{i} = e^{i\beta(1-q_{A})}\psi_{+old}^{i} \qquad (8.1.310)$$

$$\psi_{-new}^{i} = e^{i\alpha(1-q_V)}\psi_{-old}^{i}, \qquad \qquad \psi_{-new}^{i} = e^{i\beta(-1-q_A)}\psi_{-old}^{i}$$
(8.1.311)

$$\psi^{i}_{-new} = e^{i\alpha(-1+q_V)}\psi^{i}_{+old}, \qquad \qquad \psi^{i}_{-new} = e^{i\alpha(1-q_A)}\psi^{i}_{-old}. \qquad (8.1.313)$$

Setting $q_V = q_A = 0$, we obtain the $U(1)_V$ and the $U(1)_A$ charges in Table 8.7.

¹⁶ From (8.1.302) and (8.1.303), the vector R-rotations and the axial R-rotations of superfield are given by

	$U(1)_E$	$U(1)_V$	$U(1)_A$	\mathcal{L}
φ	0	0	0	\mathcal{O}
ψ^i_+	+	_	+	$K^{\frac{1}{2}}$
ψ^i	_	—	—	$K^{-\frac{1}{2}}$
$\psi^{\overline{i}}_+$	+	+	—	$K^{\frac{1}{2}}$
$\psi^{\overline{i}}_{-}$	_	+	+	$K^{-\frac{1}{2}}$

Table 8.7: U(1) charges of the d = 2, $\mathcal{N} = (2,2)$ sigma model fields. \mathcal{L} is the complex line bundle on Σ in which the fields take values. \mathcal{O} is the trivial bundle and K is the canonical bundle.

	$U(1)_E$	$U(1)_V$	$U(1)_A$	\mathcal{L}
Q_+	+	_	+	$K^{\frac{1}{2}}$
<i>Q</i> _	_	_	_	$K^{-\frac{1}{2}}$
\overline{Q}_+	+	+	_	$K^{\frac{1}{2}}$
\overline{Q}_{-}	_	+	+	$K^{-\frac{1}{2}}$

Table 8.8: U(1) charges of the d = 2, $\mathcal{N} = (2,2)$ sigma model supersymmetry generators.

	A-twist		B-tw	rist
fields	$U(1)'_E$	\mathcal{L}	$U(1)'_E$	\mathcal{L}
φ	0	\mathcal{O}	0	\mathcal{O}
ψ^i_+	2	Κ	2	Κ
ψ^i	0	\mathcal{O}	-2	K^{-1}
$\psi^{\overline{i}}_+$	0	\mathcal{O}	0	\mathcal{O}
$\psi^{\overline{i}}_{-}$	-2	K^{-1}	0	\mathcal{O}

Table 8.9: The spin of field for A-twisted and B-twisted d = 2, $\mathcal{N} = (2, 2)$ sigma model.

	A-twist		B-tw	rist
fields	$U(1)'_E$	\mathcal{L}	$U(1)'_E$	\mathcal{L}
Q_+	2	Κ	2	Κ
<i>Q</i> _	0	\mathcal{O}	-2	K^{-1}
\overline{Q}_+	0	\mathcal{O}	0	\mathcal{O}
\overline{Q}_{-}	-2	K^{-1}	0	\mathcal{O}

Table 8.10: The spin of the supersymmetry generators for A-twisted and B-twisted d = 2, $\mathcal{N} = (2, 2)$ sigma model.

A-model

After performing A-twist, the bundles in which the fermionic fields take values are modified as

$$\begin{split} \psi_{z}^{i} &\in \Gamma(K \otimes \phi^{*}(T^{1,0}X)), \\ \psi^{i} &\in \Gamma(\phi^{*}(T^{0,1}X)), \\ \psi^{\overline{i}} &\in \Gamma(\phi^{*}(T^{1,0}X)), \\ \psi^{\overline{i}}_{\overline{z}} &\in \Gamma(K^{-1} \otimes \phi^{*}(T^{0,1}X)). \end{split}$$
(8.1.316)

B-model

B-twist changes the bundles in which fermionic fields take values as

$$\begin{split} \psi_{z}^{i} &\in \Gamma(K \otimes \phi^{*}(T^{1,0}X)), \\ \psi^{\bar{i}} &\in \Gamma(\phi^{*}(T^{0,1}X)), \\ \psi_{\overline{z}}^{i} &\in \Gamma(K^{-1} \otimes \phi^{*}(T^{1,0}X)), \\ \psi^{\bar{i}} &\in \Gamma(\phi^{*}(T^{0,1}X)). \end{split}$$
(8.1.317)

These topological twisted theories are known as A-model and B-model topological sigma-models, or topological string theories [309, 230, 310, 311] ¹⁷.

8.2 Curved branes and twisted theories

Let us consider the gauge theories arising from the dimensional reduction of tendimensional $\mathcal{N} = 1$ SYM theory to (p + 1) dimensions. It is known that these theories describe the low energy world-volume dynamics of flat D*p*-branes [20].

¹⁷See [312] for the detailed review on the topological string theory.

On the other hand, when one consider curved branes wrapping around a nontrivial cycle C in the ambient manifold X, the cycle has to be identified with a calibrated submanifold and satisfy some stringent conditions to preserve some fraction of the supersymmetries.

As discussed in [28], curved world-brane theories are obtained by topological twisting along the directions where the world-volume is curved. To see this, we remember that the bosonic scalar fields are associated with translations of the D-brane. Thus when D-brane wrap around curved cycle C in X, there are only $(10 - \dim X)$ actual scalar fields and the other translational modes are identified with the section of the normal bundle N_C to C in X. Therefore these modes should be twisted if the normal bundle is non-trivial and so are their superpartners.

From the above observations, for given supersymmetric cycles C and their ambient manifolds X, one can determine

- 1. the bosonic field content
- 2. the number of scalar supercharges

of the world-volume topological gauge theories of D-branes. On the contrary, one can check whether there exists supersymmetric cycles with the required properties for given topological gauge theories.

Noting that there is a global invariance under the rotational $SO(10 - \dim X)$ symmetry ¹⁸ of the uncompactified dimensions, the original R-symmetry SO(9 - p) should be decomposed as

$$SO(9-p) \subset SO(10 - \dim X) \times SO(\dim X - p - 1). \tag{8.2.1}$$

Then, under the branching (8.2.1) of R-symmetry, we try to perform topological twisting by using the second factor $SO(\dim X - p - 1)$ corresponding to the normal bundle N_C in X. A relevant information is given by Table 8.11.

The preserved fraction of supersymmetries are derived as follows. The holonomy group of K₃ surface is SU(2), so the spinor of SO(4) is decomposed under $SO(4) \supset SU(2)_H \times SU(2) \times U(1) \supset SU(2)_H$ as

$$4 = (2,1)_{-} \oplus (1,2)_{+} = 2 \oplus 1 \oplus 1.$$
(8.2.2)

Thus 2 of 4 supercharges are constant spinors and we have $\frac{1}{2}$ BPS background.

¹⁸ When one considers Euclidean D-branes, the invariant rotational symmetry is $SO(1, 9 - \dim X)$ because curved D-branes do not wrap the time direction.

ambient manifolds (dimensions)	holonomy	submanifolds	SUSY
Calabi-Yau 2-fold (4)	$SU(2) \subset SO(4)$	holomorphic curve (2)	$\frac{1}{2}$
Calabi-Yau 3-fold (6)	$SU(3) \subset SO(6)$	Lagrangian (3)	$\frac{1}{4}$
G ₂ manifold (7)	$G_2 \subset SO(7)$	coassociative (4)	$\frac{1}{8}$
		associative (3)	$\frac{1}{8}$
Spin(7) manifold (8)	$Spin(7) \subset SO(8)$	Cayley (4)	$\frac{1}{16}$
Calabi-Yau 4-fold (8)	$SU(4) \subset SO(8)$	Lagrangian (4)	$\frac{1}{8}$
Hyperkähler manifold (8)	$Sp(2) \subset SO(8)$		$\frac{3}{16}$
$CY_2 \times CY_2$ (8)	$SU(2) \times SU(2) \subset SO(8)$		$\frac{1}{4}$
Calabi-Yau 5-fold (10)	$SU(5) \subset SO(10)$	Lagrangian (5)	$\frac{1}{16}$

Table 8.11: The ambient manifolds and the examples of calibrated submanifolds that preserve the fraction of supersymmetry. Note that all of the Calabi-Yau manifolds include holomorphic submanifolds as calibrated submanifolds.

The holonomy group of Calabi-Yau 3-fold is SU(3), so the spinor of SO(6) is decomposed under $SO(6) \supset SU(3)$ as

$$\mathbf{4} = \mathbf{3} \oplus \mathbf{1}.\tag{8.2.3}$$

Thus 1 of 4 supercharges is constant spinor and we have $\frac{1}{4}$ BPS background.

The holonomy group of Calabi-Yau 4-fold is SU(4), so the spinor of SO(8) is decomposed under $SO(8) \supset SU(4)$ as

$$\mathbf{8}_{s} \oplus \mathbf{8}_{c} = \mathbf{6} \oplus \mathbf{1} \oplus \mathbf{1} \oplus \mathbf{4} \oplus \overline{\mathbf{4}}$$

$$(8.2.4)$$

Thus 2 of 16 supercharges is constant spinor and we have $\frac{1}{8}$ BPS background.

The holonomy group of G_2 manifold is G_2 , so the spinor of SO(7) is decomposed under $SO(7) \supset G_2$ as

$$\mathbf{8} = \mathbf{1} \oplus \mathbf{7} \tag{8.2.5}$$

Thus 1 of 8 supercharges is constant spinor and we have $\frac{1}{8}$ BPS background.

The holonomy group of Spin(7) manifold is Spin(7), so the spinor of SO(8) is decomposed under $SO(8) \supset Spin(7)$ as

$$\mathbf{8}_{s} \oplus \mathbf{8}_{c} = \mathbf{7} \oplus \mathbf{1} \oplus \mathbf{8} \tag{8.2.6}$$

Thus 1 of 16 supercharges is constant spinor and we have $\frac{1}{16}$ BPS background.

The holonomy group of Calabi-Yau 5-fold is SU(5), so the spinor of SO(10) is decomposed under $SO(10) \supset SU(5)$ as

$$\mathbf{16} \oplus \mathbf{16}' = \mathbf{1}_{-5} \oplus \overline{\mathbf{5}}_3 \oplus \mathbf{10}_{-} \oplus \mathbf{1}_5 \oplus \mathbf{5}_{-3} \oplus \mathbf{10}_{+} \tag{8.2.7}$$

Thus 2 of 32 supercharges is constant spinor and we have $\frac{1}{16}$ BPS background.

8.2.1 D3-branes and twisted d = 4, $\mathcal{N} = 4$ SYM theories

The first example is the low-energy effective field theories of the D₃-branes wrapped on curved four-manifold. A set of these descriptions can be obtained by the three distinct topologically twisted d = 4, $\mathcal{N} = 4$ SYM theories.

1. GL twist D-brane

The fact that dim C = 4 and that there are two scalar fields means that the theory describes 4-cycle C in 4 + (6 - 2) = 8-dimensional manifold X. The existence of two preserved BRST charges indicates that 8-manifold preserve $\frac{2}{16} = \frac{1}{8}$ of the supersymmetry. From the above facts and Table 8.11, X is a Calabi-Yau 4-fold and C is a special Lagrangian submanifold¹⁹.

Moreover it it known that in the case where special Lagrangian submanifold is embedded in Calabi-Yau 4-fold, the normal bundle N_C can be identified with the cotangent bundle T_C^* [313]. This is consistent to the fact that the remaining four scalar fields combine to form one 1-form on C.

Note that a global U(1) ghost number symmetry corresponds to the rotational symmetry of the two uncompactified dimensions. The two scalars having opposite U(1) charges are identified with the 2-dimensional vector and the 1-form is a U(1)-singlet.

2. VW twist D-brane

The fact that dim C = 4 and that there are three scalar fields means that the theory describes 4-cycle C in 4 + (6 - 3) = 7-dimensional manifold X. The existence of two preserved BRST charges indicates that 7-manifold preserve $\frac{2}{16} = \frac{1}{8}$ of the supersymmetry. From the above facts and Table 8.11, X is a G_2 manifold and C is a coassociative submanifold.

It is known that for a coassociative 4-submanifold in G_2 manifold, the normal bundle is $(\mathbf{1}, \mathbf{3})_0$ [313]. This is consistent to the results obtained by the twisting.

¹⁹Special Lagrangian submanifold is a submanifold for which the real part of the holomorphic form restricts to the volume form on the submanifold.

twist	submanifold (dimension)	ambient manifold (dimension)	SUSY
GL twist	Lagrangian (4)	Calabi-Yau 4-fold (8)	$\frac{16}{8} = 2$
VW twist	coassociative (4)	G_2 manifold (7)	$\frac{16}{8} = 2$
DW twist	Cayley (4)	Spin(7) manifold (8)	$\frac{16}{16} = 1$

Table 8.12: Three types of topological twists for d = 4, $\mathcal{N} = 4$ SYM theories, curved D₃-branes (submanifolds) and ambient manifolds.

3. DW twist D-brane

The fact that dim C = 4 and that there are two scalar fields means that the theory describes 4-cycle C in 4 + (6 - 2) = 8-dimensional manifold X. The existence of one preserved BRST charge indicates that 8-manifold preserve $\frac{1}{16} = \frac{1}{16}$ of the supersymmetry. From the above facts and Table 8.11, X is a Spin(7) manifold and C is a Cayley submanifold.

It it known that for the Spin(7) manifold the normal bundle is $S_+ \oplus V$ where S_+ is a spin bundle of a given chirality and V is a 2-dimensional bundle [28]. When V is trivial, this becomes $S_+ \oplus S_+$, that is $(\mathbf{1}, \mathbf{2})_+ \oplus (\mathbf{1}, \mathbf{2})_-$.

These results are summarized in Table 8.12

8.2.2 **D2-branes and twisted** d = 3, $\mathcal{N} = 8$ **SYM theories**

The D2-branes wrapped on three-manifold are given by the topologically twisted d = 3, $\mathcal{N} = 8$ SYM theories.

1. A-twist

The fact that dim C = 3 and that there are four scalar fields means that the theory describes 3-cycle C in 3 + (7 - 4) = 6-dimensional manifold X. The existence of the four preserved BRST charges indicates that 6-manifold preserve $\frac{4}{16} = \frac{1}{4}$ of the supersymmetry. From the above facts and Table 8.11, X is a Calabi-Yau 3-fold and C is a special Lagrangian submanifold.

Also it is known that the normal bundle N_C can be identified with the cotangent bundle T_C^* [313]. This is consistent to the fact that the remaining three scalar fields combine to form one 1-form on C.

A global $SU(2)_1 \times SU(2)_2 \simeq SU(4)$ ghost number symmetry corresponds to the rotational symmetry of the four uncompactified dimensions. The four scalars transform as a 4_v of SO(4) and the 1-form is an SO(4)-singlet.

twist	submanifold (dimension)	ambient manifold (dimension)	SUSY
A-twist	Lagrangian (3)	Calabi-Yau 3-fold (6)	$\frac{16}{4} = 4$
B-twist	associative (3)	G_2 manifold (7)	$\frac{16}{8} = 2$

Table 8.13: Two types of topological twists for d = 3, $\mathcal{N} = 8$ SYM theories, curved D2-branes (submanifolds) and ambient manifolds.

2. B-twist

The fact that dim C = 3 and that there are three scalar fields means that the theory describes 3-cycle C in 3 + (7 - 3) = 7-dimensional manifold X. The existence of the two preserved BRST charges indicates that 7-manifold preserve $\frac{2}{16} = \frac{1}{8}$ of the supersymmetry. From the above facts and Table 8.11, X is a G_2 manifold and C is an associative submanifold.

Also it is known that for an associative 3-submanifold in G_2 manifold, the normal bundle is $N_C = S \otimes V$ where *S* is a spinor bundle of *C* and *V* is a rank two SU(2)-bundle. This is consistent to the fact that the twisted bosonic spinors (**2**, **2**, **1**) are an SU(2)-doublet of spinors on *C*.

Again a global $SU(2)_3 \simeq SO(3)$ symmetry corresponds to the rotational symmetry of the four uncompactified dimensions. The three scalars transform as a $\mathbf{3}_v$ of SO(3) and the twisted bosonic spinors (**2**, **2**, **1**) are SO(3)-singlet.

These results are summarized in Table 8.13

8.2.3 **D2-branes and twisted** d = 3, $\mathcal{N} = 8$ **SYM theories on** $\mathbb{R} \times \Sigma$

The low-energy effective theories of the D2-branes wrapping on the holomorphic Riemann surface Σ are the partially twisted d = 3, $\mathcal{N} = 8$ SYM theories:

1. A-twist

The fact that dim $\Sigma = 2$ and that there are five scalar fields means that the theory describes 2-cycle Σ in 2 + (7 - 5) = 4-dimensional manifold X. The existence of the eight preserved BRST charges indicates that 4-manifold preserve $\frac{8}{16} = \frac{1}{2}$ of the supersymmetry. From the above facts and Table (8.11), X is a K3 surface and Σ is a holomorphic curve.

2. B-twist

The fact that dim $\Sigma = 2$ and that there are three scalar fields means that the theory describes 2-cycle Σ in 2 + (7 - 3) = 6-dimensional manifold *X*.

twist	submanifold (dimension)	ambient manifold (dimension)	SUSY
A-twist	holomorphic (2)	K3 surface (4)	$\frac{16}{2} = 8$
B-twist	holomorphic (2)	Calabi-Yau 3-fold (6)	$\frac{16}{4} = 4$
C-twist	holomorphic (2)	Calabi-Yau 4-fold (8)	$\frac{16}{8} = 2$

Table 8.14: Three types of topological twists for d = 3, $\mathcal{N} = 8$ SYM theories on $\mathbb{R} \times \Sigma$, curved D2-branes (submanifolds) and ambient manifolds.

The existence of the four preserved BRST charges indicates that 6-manifold preserve $\frac{4}{16} = \frac{1}{4}$ of the supersymmetry. From the above facts and Table (8.11), X is a Calabi-Yau 3-fold and Σ is a holomorphic curve.

3. C-twist

The fact that dim $\Sigma = 2$ and that there are one scalar field means that the theory describes 2-cycle Σ in 2 + (7 - 1) = 8-dimensional manifold X. The existence of the two preserved BRST charges indicates that 6-manifold preserve $\frac{2}{16} = \frac{1}{8}$ of the supersymmetry. From the above facts and Table (8.11), X is a Calabi-Yau 4-fold and Σ is a holomorphic curve.

These results are summarized in Table 8.14

8.2.4 Relationship between d = 4 and d = 3 twists

The d = 4 twisting and the d = 3 twisting are connected via dimensional reduction.

1. DW twist and B-twist

Let us define Cayley 4-form in local coordinates \mathbb{R}^8 as [314, 313]

$$\begin{split} \Omega_{Cayley} &:= dx^{0123} + \left(dx^{01} - dx^{23} \right) \wedge \left(dx^{45} - dx^{67} \right) \\ &+ \left(dx^{02} + dx^{13} \right) \wedge \left(dx^{46} + dx^{57} \right) \\ &+ \left(dx^{03} - dx^{12} \right) \wedge \left(dx^{47} - dx^{56} \right) + dx^{4567} \end{split}$$
(8.2.8)

where $dx^{ijk} := dx^i \wedge dx^j \wedge dx^k$, etc. Then one can define Spin(7) manifold to be the subgroup of GL(8) that preserve Ω_{Cayley} . Integrating over the fibre x^0 , we obtain

$$\pi_* \Omega_{Cayley} = dx^{123} + dx^1 \wedge \left(dx^{45} - dx^{67} \right) + dx^2 \wedge \left(dx^{46} + dx^{57} \right) + dx^3 \wedge \left(dx^{47} - dx^{56} \right)$$
(8.2.9)

On the other hand, the associative 3-form Ω_{ass} characterizing associative 3manifolds of G_2 manifolds is defined as [314]

$$\Omega_{ass} := dx^{456} + dx^4 \wedge \left(dx^{01} - dx^{23} \right) \\ + dx^5 \wedge \left(dx^{02} + dx^{13} \right) \\ + dx^6 \wedge \left(dx^{03} - dx^{12} \right)$$
(8.2.10)

Thus

$$\pi_* \Omega_{Cayley} = \Omega_{ass}.$$
 (8.2.11)

Therefore DW twist is related to B twist by the dimensional reduction.

2. VW twist and A-twist

VW twist theory corresponds to coassociative submanifolds of G_2 manifolds characterized by the Hodge dual 4-form $\Omega_{coass} = *\Omega_{ass}$, which is expressed as [313]

$$\Omega_{coass} = dx^{0123} - dx^{56} \wedge \left(dx^{01} - dx^{23} \right) + dx^{46} \wedge \left(dx^{02} + dx^{13} \right) - dx^{45} \wedge \left(dx^{03} - dx^{12} \right).$$
(8.2.12)

Integrating this over x^0 , one obtains

$$\pi_*\Omega_{coass} = dx^{123} - dx^{156} + dx^{246} - dx^{345}.$$
(8.2.13)

On the other hand, the holomorphic volume form of a Calabi-Yau 3-fold characterizing special Lagrangian submanifolds is

$$\Omega_{slag} = dz^{1} \wedge dz^{2} \wedge dz^{3}$$

= $\left(dx^{123} - dx^{453} - dx^{156} - dx^{426} \right) + i \left(dx^{423} + dx^{513} + dx^{612} - dx^{456} \right).$

Thus

$$\pi_* \Omega_{coass} = \operatorname{Re}\Omega_{slag}.$$
(8.2.14)

Therefore VW twist is associated with A twist by the dimensional reduction.

3. GL twist and A-twist

Suppose that the Calabi-Yau 4-fold is locally of the form

$$CY_4 = CY_3 \times T^2.$$
 (8.2.15)

Then special Lagrangian of CY_4 wrapping around one of the circles reduce to special Lagrangian submanifolds of CY_3 by the double dimensional reduction. Therefore GL twist is related to A-twist by the double dimensional reduction.

8.2.5 **D1-branes and twisted** d = 2, $\mathcal{N} = 8$ **SYM theories**

The world-volume theories of the D1-branes wrapped on holomorphic Riemann surfaces are topologically twisted d = 2, $\mathcal{N} = 8$ SYM theories.

1. A-twist

The fact that dim C = 2 and that there are six scalar fields means that the theory describes 2-cycle C in 2 + (8 - 6) = 4-dimensional manifold X. The existence of the eight preserved BRST charges indicates that 4-manifold preserve $\frac{8}{16} = \frac{1}{2}$ of the supersymmetry. From the above facts and Table 8.11, X is a K3 surface and C is a holomorphic curve.

Let us consider the normal bundle $N_{\mathcal{C}}$. Noting that

$$T_X = T_{\mathcal{C}} \oplus N_{\mathcal{C}}, \tag{8.2.16}$$

$$c_1(T_X) = 0 \tag{8.2.17}$$

for holomorphic genus g curve C in Calabi-Yau n-folds X, we see that

$$c_1(N_c) = -c_1(T_c) = 2g - 2.$$
 (8.2.18)

Alternatively as $\wedge^n T_X$ is trivial, one has

$$\wedge^n T_X = T_{\mathcal{C}} \wedge^{n-1} N_{\mathcal{C}} = 1, \qquad (8.2.19)$$

which gives the condition of the canonical bundle K_C on C

$$\wedge^{n-1}N_{\mathcal{C}} = K_{\mathcal{C}} \tag{8.2.20}$$

because $T_{\mathcal{C}} = K_{\mathcal{C}}^{-1}$.

If *X* is a K₃ surface and *C* is a holomorphic curve, then n = 2 and N_C has rank one and (8.2.20) becomes

$$N_{\mathcal{C}} = K_{\mathcal{C}}.\tag{8.2.21}$$

This is consistent to the fact that remaining two scalar fields combine to form a single one-form on C.

A global $SO(6)_R$ ghost number symmetry corresponds to the rotational symmetry of the six uncompactified dimensions. The six scalars transform as a $\mathbf{6}_v$ of SO(6) and the one-form is an SO(6)-singlet.

twist	submanifold (dimension)	ambient manifold (dimension)	SUSY
A-twist	holomorphic curve (2)	K3 surface (4)	$\frac{16}{2} = 8$
B-twist	holomorphic curve (2)	Calabi-Yau 3-fold (6)	$\frac{16}{4} = 4$

Table 8.15: Two types of topological twists for d = 2, $\mathcal{N} = 8$ SYM theories, curved D1-branes (submanifolds) and ambient manifolds.

2. B-twist

The fact that dim C = 2 and that there are four scalar fields means that the theory describes 2-cycle C in 2 + (8 - 4) = 6-dimensional manifold X. The existence of the four preserved BRST charges indicates that 6-manifold preserve $\frac{4}{16} = \frac{1}{4}$ of the supersymmetry. From the above facts and Table 8.11, X is a Calabi-Yau 3-fold and C is an holomorphic curve.

In this case (8.2.20) becomes

$$\wedge^2 N_{\mathcal{C}} = K_{\mathcal{C}} \tag{8.2.22}$$

and generally this is solved by

$$N_{\mathcal{C}} = K_{\mathcal{C}}^{\frac{1}{2}} \otimes V \tag{8.2.23}$$

where V is a rank two bundle with trivial determinant.

These results are summarized in Table 8.15

8.2.6 M2-branes and twisted BLG theory

The low-energy description of the two M2-branes wrapping curved three-fold are as follows:

1. A-twist

The fact that dim C = 3 and that there are five scalar fields means that the theory describes 3-cycle C in 3 + (8 - 5) = 6-dimensional manifold X. The existence of the four preserved BRST charges indicates that 6-manifold preserve $\frac{4}{16} = \frac{1}{4}$ of the supersymmetry. From the above facts and Table 8.11, X is a Calabi-Yau 3-fold and C is a special Lagrangian submanifold.

Also it is known that the normal bundle N_C can be identified with the cotangent bundle T_C^* [313]. This is consistent to the fact that the remaining three scalar fields combine to form one 1-form on C.

twist	submanifold (dimension)	ambient manifold (dimension)	SUSY
A-twist	Lagrangian (3)	Calabi-Yau 3-fold (6)	$\frac{16}{4} = 4$
B-twist	associative (3)	G_2 manifold (7)	$\frac{16}{8} = 2$

Table 8.16: Two types of topological twists for BLG model, curved M2-branes (submanifolds) and ambient manifolds.

A global SO(5) ghost number symmetry corresponds to the rotational symmetry of the four uncompactified dimensions. The five scalars transform as a 5_v of SO(5) and the 1-form is an SO(5)-singlet.

2. B-twist

The fact that dim C = 3 and that there are three scalar fields means that the theory describes 3-cycle C in 3 + (8 - 4) = 7-dimensional manifold X. The existence of the two preserved BRST charges indicates that 7-manifold preserve $\frac{2}{16} = \frac{1}{8}$ of the supersymmetry. From the above facts and Table 8.11, X is a G_2 manifold and C is an associative submanifold.

These results are summarized in Table 8.16 and same as that of D2-brane instantons (Table 8.13).

Chapter 9

Curved M2-branes and Topological Twisting

In this chapter we will return to the study of the M2-branes and discuss that the topologically twisted A_4 BLG-model may describe the two wrapped M2-branes around a holomorphic Riemann surface in Calabi-Yau manifold based on the work of [51]. We will study the preserved supersymmetry on the wrapped branes around a holomorphic Riemann surface inside a Calabi-Yau manifold in section 9.1, 9.2 and 9.3. In section 9.4 we will specify the appropriate twisting procedures for our wrapped M2-branes.

9.1 M2-branes wrapping a holomorphic curve

Now we are ready to discuss the M2-branes wrapping curved Riemann surface. Recall that the BLG action (4.1.31) and the ABJM action (5.1.1) may describe the dynamics of probe membranes propagating in a fixed background geometry with an SO(8) and an SU(4) holonomy respectively. For both cases, the world-volume M_3 is considered as a flat space-time $\mathbb{R}^{1,2}$ or $\mathbb{R} \times T^2$. Now let us consider more general situations where curved M2-branes reside in some fixed curved background geometries. If we naively put the theory on a general three dimensional manifold, all supersymmetries are broken. However, here we shall wrap the M2-branes on a Riemann surface Σ_g of genus g that can preserve supersymmetry (i.e. supersymmetric two-cycles) as the form

$$M_3 = \mathbb{R} \times (\Sigma_g \subset X) \tag{9.1.1}$$

where \mathbb{R} is viewed as a time direction and *X* is a real 2(n + 1)-dimensional space preserving supersymmetry with vanishing three-form gauge field. Thus far the

only known supersymmetric two-cycles, i.e. calibrated two-cycles, in special holonomy manifolds are holomorphic two-cycles in Calabi-Yau spaces. The corresponding two-form calibrations are Kähler calibrations. Accordingly we will take the ambient space X as an (n + 1)-dimensional Calabi-Yau space and the other space as flat. Namely the geometry of the M-theory is taken as

$$\mathbb{R}^{1,8-2n} \times CY_{n+1}.$$
(9.1.2)

9.2 Supersymmetry in Calabi-Yau space

As a first step to count the number of preserved supersymmetries in our setup, one should know the dimension of the vector space formed by the corresponding Killing spinor ϵ , that is the amount of supersymmetries in the background geometry. Since we are now considering the background geometries with vanishing four-form flux, the Killing spinor equation is given by

$$\nabla_M \epsilon = \left(\partial_M + \frac{1}{4}\omega_{MPQ}\Gamma^{PQ}\right)\epsilon = 0 \tag{9.2.1}$$

where ω_{MPQ} , M, N, P, $Q = 0, 1, \dots, 10$ is an eleven-dimensional Levi-Civita spin connection. This leads to the integrability condition

$$[\nabla_M, \nabla_N]\epsilon = \frac{1}{4} R_{MNPQ} \Gamma^{PQ} \epsilon = 0, \qquad (9.2.2)$$

which implies that a Killing spinor ϵ transforms as a singlet under the restricted holonomy group $H \subset Spin(1, 10)$ generated by $R_{MNPQ}\Gamma^{PQ}$. Therefore the amount of preserved supersymmetries in the special holonomy manifold is equivalent to the number of singlets in the decomposition of the spinor representation **32** of Spin(1, 10) into the representation of the holonomy group H. In our case the background geometries are taken as Calabi-Yau (n + 1)-folds with the holonomy H = SU(n + 1), n = 1, 2, 3, 4 and the decompositions are as follows.

1. *CY*₅

In this case the geometry is of the form $\mathbb{R} \times CY_5$. This splits the Spin(10) into SU(5) and the corresponding decomposition of the spinor representation is given by

$$16 = 10_{-} \oplus 5_{3} \oplus 1_{-5}$$

$$16' = 10_{+} \oplus 5_{-3} \oplus 1_{5}.$$
 (9.2.3)

Here the capital letters denote the representations of the SU(5) and the subscripts stand for the U(1) charges which appear after the decomposition $Spin(10) \rightarrow SU(5) \times U(1)$. The existence of two singlets implies that the space $\mathbb{R} \times CY_5$ preserves $\frac{2}{32} = \frac{1}{16}$ supersymmetries.

Let us define an explicit set of projections defining the Killing spinors. To this end we need to specify how the Calabi-Yau spaces live in the elevendimensional space-time. We shall consider the situations where the Calabi-Yau manifolds fill in the order (x^1, x^2) , (x^9, x^{10}) , (x^7, x^8) , (x^5, x^6) and (x^3, x^4) . Then the Killig spinors can be defined by the eigenvalues ± 1 for the following set of commuting matrices

$$\Gamma^{12910}, \quad \Gamma^{91078}, \quad \Gamma^{7856}, \quad \Gamma^{5634}.$$
 (9.2.4)

The corresponding Killing spinors for CY_5 can be defined by the projection

$$\Gamma^{12910}\epsilon = \Gamma^{91078}\epsilon = \Gamma^{7856}\epsilon = \Gamma^{5634}\epsilon = -\epsilon.$$
(9.2.5)

Note that this implies that $\Gamma^{012}\epsilon = \epsilon$.

2. CY₄

For this case the geometry is the product form $\mathbb{R}^{1,2} \times CY_4$. This leads to the decomposition of the *Spin*(8) into *SU*(4) and that of the spinor representation

$$\mathbf{8}_{s} = \mathbf{6}_{0} \oplus \mathbf{1}_{2} \oplus \mathbf{1}_{-2}$$
$$\mathbf{8}_{c} = \mathbf{4}_{-} \oplus \mathbf{4}_{+}.$$
(9.2.6)

We see that the decomposition provides two singlets from sixteen components. Thus the geometry $\mathbb{R}^{1,2} \times CY_4$ can preserve $\frac{2}{16} = \frac{1}{8}$ supersymmetries. In this case the projection for the Killing spinor is given by

$$\Gamma^{12910}\epsilon = \Gamma^{91078}\epsilon = \Gamma^{7856}\epsilon = -\epsilon.$$
(9.2.7)

3. *CY*₃

In this case the geometry is given by $\mathbb{R}^{1,4} \times CY_3$. This decomposes the *Spin*(6) into *SU*(3) and correspondingly spinor representation decomposes as

$$\begin{aligned} \mathbf{4} &= \mathbf{3}_{-} \oplus \mathbf{1}_{3} \\ \overline{\mathbf{4}} &= \mathbf{3}_{+} \oplus \mathbf{1}_{-3}. \end{aligned} \tag{9.2.8}$$

The appearance of two singlets from eight components means that there are $\frac{2}{8} = \frac{1}{4}$ supersymmetries in the product space $\mathbb{R}^{1,4} \times CY_3$. Therefore the Killing spinor can be defined by the projection

$$\Gamma^{12910}\epsilon = \Gamma^{91078}\epsilon = -\epsilon. \tag{9.2.9}$$

4. *CY*₂

For this case the geometry is the product space $\mathbb{R}^{1,6} \times CY_2$. The decomposition of Spin(4) into $SU(2) \times SU(2)$ gives rise to that of the spinor representation

$$2 = (2, 1)$$

 $2' = (1, 2).$ (9.2.10)

The presence of two singlets under one part of the SU(2) implies that there are $\frac{2}{4} = \frac{1}{2}$ supersymmetries in the geometry $\mathbb{R}^{1,6} \times CY_2$. The corresponding Killing spinors satisfy the projection

$$\Gamma^{12910}\epsilon = -\epsilon. \tag{9.2.11}$$

9.3 Calibration and supersymmetric cycle

As a next step we shall consider the situation where the M2-branes wrapping a Riemann surface Σ_g propagate in a Calabi-Yau space without back reaction. In order to hold supersymmetry on the world-volume, Σ_g turns out to be a calibrated two-cycle, i.e. holomorphic curve of a Calabi-Yau manifold. To see this let us briefly review the background material concerning a calibration. In general a calibration on a special holonomy manifold *X* is a differential *p*-form φ obeying [27]

$$d\varphi = 0, \tag{9.3.1}$$

$$\varphi|_{\mathcal{C}_p} \leq \operatorname{Vol}|_{\mathcal{C}_p}, \quad \forall \mathcal{C}_p$$

$$(9.3.2)$$

where C_p is any *p*-cycle in *X* and Vol is the volume form on the cycle induced from the metric on *X*. Here the inequality is defined locally, namely $\varphi|_{C_p} = a \cdot \text{Vol}|_{C_p}$ for some $a \in \mathbb{R}$, and $\varphi|_{C_p} \leq \text{Vol}|_{C_p}$ if $a \leq 1$. A *p*-cycle Σ is said to be calibrated by φ if it satisfies

$$\varphi|_{\Sigma} = \operatorname{Vol}|_{\Sigma}. \tag{9.3.3}$$

We remark that a calibrated submanifold is a minimal surface in their homology class because

$$\operatorname{Vol}(\Sigma) = \int_{\Sigma} \varphi = \int_{M_{p+1}} d\varphi + \int_{\Sigma'} \varphi = \int_{\Sigma'} \varphi \le \operatorname{Vol}(\Sigma')$$
(9.3.4)

where Σ' is another *p*-cycle in the same homology class such that $\partial M_{p+1} = \Sigma - \Sigma'$.

It is known that Calabi-Yau (n + 1)-folds admit two different types of calibrations; the Kähler form *J* and the real part of holomorphic (n + 1, 0)-form Ω . One can construct calibrations as bilinear forms of spinors [315, 316]

$$J_{MN} = i\epsilon^{\dagger}\Gamma_{MN}\epsilon, \qquad (9.3.5)$$

$$\Omega_{M_1\cdots M_{n+1}} = \epsilon^T \Gamma_{M_1\cdots M_{2(n+1)}} \epsilon.$$
(9.3.6)

Now we consider the condition so that a bosonic configuration of membranes is supersymmetric. Since one can always add a second probe brane without breaking supersymmetry if it is wrapped on the supersymmetric cycle which the original probe brane is wrapping, a simple way to find such condition is to analyze an effective world-volume action of a single membrane [317]. The action for a supermembrane coupled to d = 11 supergravity is given by [318]

$$S = \int d^{3}x \left[\frac{1}{2} \sqrt{-h} h^{\mu\nu} \partial_{\mu} X^{M} \partial_{\nu} X^{N} g_{MN} - \frac{1}{2} \sqrt{-h} - i \sqrt{-h} h^{\mu\nu} \overline{\Theta} \Gamma_{\mu} \nabla_{\nu} \Theta + \frac{1}{6} \epsilon^{\mu\nu\lambda} C_{MNP} \partial_{\mu} X^{M} \partial_{\nu} X^{N} \partial_{\lambda} X^{P} + \cdots \right]$$
(9.3.7)

where $h_{\mu\nu}$, μ , $\nu = 0, 1, 2$ is the metric of the world-volume, $h = \det(h_{\mu\nu})$, g_{MN} , $M = 0, 1, \dots, 10$ is the d = 11 space-time metric. X^M is a space-time coordinate and Θ is a fermionic space-time coordinate. C_{MNP} is a three-form gauge field, which is now taken to be zero in our background geometries. The action (9.3.7) is invariant under the rigid supersymmetry transformations

$$\delta_{\epsilon} X^{M} = i \overline{\epsilon} \Gamma^{M} \Theta, \qquad (9.3.8)$$

$$\delta_{\epsilon}\Theta = \epsilon \tag{9.3.9}$$

where ϵ is a constant anti-commuting eleven-dimensional spinor. Also the action (9.3.7) has a local fermionic symmetry, called κ -symmetry. The κ -symmetry transformation is given by

$$\delta_{\kappa} X^{M} = 2i\overline{\Theta} \Gamma^{M} P_{+} \kappa(x), \qquad (9.3.10)$$

$$\delta_{\kappa}\Theta = 2P_{+}\kappa(x) \tag{9.3.11}$$

where $\kappa(x)$ is a d = 11 spinor and the matrices

$$P_{\pm} = \frac{1}{2} \left(1 \pm \frac{1}{6\sqrt{-h}} \epsilon^{\mu\nu\lambda} \partial_{\mu} X^{M} \partial_{\nu} X^{N} \partial_{\lambda} X^{P} \Gamma_{MNP} \right)$$
(9.3.12)

are projection operators satisfying

$$P_{\pm}^2 = 1, \quad P_+P_- = 0, \quad P_+ + P_- = 1.$$
 (9.3.13)

To extract the physical degrees of freedom, we must choose the suitable gauge that fixes the local world-volume reparametrization and the local κ -symmetry. Let us fix the reparametrization by choosing $x^0 = X^0$. Then the projection operator (9.3.12) can be expressed as

$$P_{\pm} = \frac{1}{2} \left(1 \pm \Gamma \right) \tag{9.3.14}$$

where

$$\Gamma := \frac{1}{2\sqrt{\det(h_{\Sigma ij})}} \Gamma^0 \epsilon^{ij} \partial_i X^M \partial_j X^N \Gamma_{MN}.$$
(9.3.15)

Here $h_{\Sigma ij}$, i, j = 1, 2 is the metric of the Riemann surface wrapped by the M2brane and $\sqrt{\det(h_{\Sigma ij})}$ is the area of the surface. As a next step we want to fix the local κ -symmetry on the world-volume. In order for a bosonic world-volume configuration to be supersymmetric, the global supersymmetry transformations (9.3.9) need to be compensated for by the κ -symmetry transformations (9.3.11)

$$(\delta_{\epsilon} + \delta_{\kappa}) \Theta = \epsilon + 2P_{+}\kappa(x) = 0.$$
(9.3.16)

Acting P_{-} on both sides we find that

$$P_{-}\epsilon = \frac{1-\Gamma}{2}\epsilon = 0. \tag{9.3.17}$$

Therefore the supersymmetry preserved by the M2-branes is given by the Killing spinor ϵ which obeys the projection (9.3.16). Noting that $\Gamma^2 = 1$ and $\Gamma^{\dagger} = \Gamma$, we find that

$$\epsilon^{\dagger} \frac{1-\Gamma}{2} \epsilon = \epsilon^{\dagger} \frac{(1-\Gamma)(1-\Gamma)}{4} \epsilon = \left| \frac{1-\Gamma}{2} \epsilon \right|^2 \ge 0.$$
(9.3.18)

By normalizing the Killing spinors such that $\epsilon^{\dagger}\epsilon = 1$, the inequality (9.3.18) can be rewritten as

$$\operatorname{Vol}(\Sigma_g) \ge \varphi$$
 (9.3.19)

M3	supersymmetry
$\boxed{\mathbb{R} \times (\Sigma_g \subset \mathrm{K3})}$	8
$\mathbb{R} \times (\Sigma_g \subset CY_3)$	4
$\boxed{\mathbb{R} \times (\Sigma_g \subset CY_4)}$	2
$\boxed{\mathbb{R} \times (\Sigma_g \subset CY_5)}$	2

Table 9.1: The amounts of the preserved supersymmetries for the M2-branes wrapping holomorphic curves Σ_g in Calabi-Yau spaces. Note that the M2-branes can wrap a holomorphic curve in a *CY*₅ without loss of the supersymmetries.

where $\operatorname{Vol}(\Sigma_g) = \sqrt{\operatorname{det}(h_{\Sigma ij})}$ is the area of the Riemann surface and φ is the differential two-form defined by

$$\varphi = -\frac{1}{2} \left(\overline{\epsilon} \Gamma_{MN} \epsilon \right) dX^M \wedge dX^N.$$
(9.3.20)

Hence the two-form (9.3.20) satisfies the condition (9.3.2) for the calibration and has the bilinear expression for Kähler calibration J (see (9.3.5)). Moreover it can be shown that the two-form (9.3.20) obeys the other required condition (9.3.1) for the calibration by noting the explicit expression (9.3.20)¹. Therefore we can conclude that the two-form (9.3.20) is a Kähler calibration and that the supersymmetric twocycle Σ_g wrapped by the M2-branes is a calibrated two-cycle, i.e. a holomorphic curve. Notice that (9.3.16) is precisely the chirality condition $\Gamma^{012}\epsilon = \epsilon$ imposed on the supersymmetry parameters in the BLG-model (see (4.1.52)).

At this stage we are ready to count the number of preserved supersymmetries in our M2-brane configurations by combining the two different types of projections; the projections (9.2.5), (9.2.7), (9.2.9) and (9.2.11) for the background Calabi-Yau manifolds and the projection (9.3.16) (or (4.1.52)) for the membranes wrapped around a calibrated two-cycle Σ_g . In most of the cases wrapped branes break half of the supersymmetries preserved by the special holonomy manifolds according to the additional projection for the branes wrapping calibrated cycles. However, for the Calabi-Yau 5-fold the projection condition (9.3.16) for the M2-branes does not give rise to a further constraint on the surviving two Killing spinors. This implies that M2-branes can wrap a holomorphic curve in a Calabi-Yau 5-fold without breaking down the supersymmetry. The amounts of preserved supersymmetries by the M2-branes wrapping holomorphic curves Σ_g in Calabi-Yau spaces are summarized in Table 9.1. Upon the dimensional reduction to \mathbb{R} , the arising quantum mechanics on \mathbb{R} will have the same number of supersymmetries.

¹It can also be checked by using the supersymmetry algebra [319].

9.4 Topological twisting

In general a quantum field theory on the curved M2-branes interacts with gravity, however, it is also possible to get a supersymmetric quantum field theory on $\mathbb{R} \times \Sigma_g$ by taking the appropriate decoupling limit $l_p \to 0$ while keeping the volume of Σ_g and that of *X* fixed. In order to derive such low-energy effective theories on the curved world-volume, we recall how the BLG-model describes the dynamics of the flat M2-branes. In the BLG-model the fields and supercharges transform under $SO(2)_E \times SO(8)_R$ as

$$X_a^I : \mathbf{8}_{v0}$$

$$\Psi_a : \mathbf{8}_{c+} \oplus \mathbf{8}_{c-}$$

$$\epsilon : \mathbf{8}_{s+} \oplus \mathbf{8}_{s-}.$$
(9.4.1)

The eight scalar fields $X^{I'}$ s transform as the vector representations of the R-symmetry $SO(8)_R$ which represents the rotational group of the transverse space of the M2branes. In other words, they are sections of the normal bundle, which is trivial in this case. However, corresponding to the geometry given in (9.1.1), now the tangent bundle T_X of the ambient Calabi-Yau manifold X is decomposed as

$$T_X = T_\Sigma \oplus N_\Sigma \tag{9.4.2}$$

where T_{Σ} is the tangent bundle over the Riemann surface Σ_g and N_{Σ} is the normal bundle over the surface. Therefore we need to take into account the existence of the non-trivial normal bundle of calibrated cycles and to introduce new dynamical variables instead of the original scalar fields. These transitions from scalars, i.e. trivial normal bundle to the non-trivial normal bundles are intimately connected with the way in which the field theory on $\mathbb{R} \times \Sigma_g$ realizes supersymmetry. Along with the coupling to the curvature on the Riemann surface, there now exists a coupling to an external SO(2n) gauge group, the R-symmetry background. Thus one can preserve supersymmetry on the holomorphic Riemann surface by choosing the SO(2) Abelian background from the SO(2n) appropriately.

There is a beautiful observation that such an effective description for curved branes can be obtained by topological twisting [28]. Here we attempt to twist the BLG-model to obtain the low-energy descriptions for the curved M2-branes ². Schematically topological twisting procedure can be achieved by replacing the original Euclidean rotational group $SO(2)_E$ on the Riemann surface by a different

² For the ABJM-model the geometric meaning of the topological twisting is less clear because the classical $SU(4)_R$ R-symmetry reflects the orbifolds. In this paper we will focus on the BLG-model.
subgroup $SO(2)'_E$ of $SO(2)_E \times SO(8)_R$. Although there are many possible ways to pick such subgroups, here we will consider the following decomposition

$$SO(8) \supset SO(8-2n) \times SO(2n)$$

$$\supset SO(8-2n) \times SO(2)_1 \times \dots \times SO(2)_n.$$
(9.4.3)

The SO(8 - 2n) is a rotational group of the Euclidean space perpendicular to the Riemann surface, while the $SO(2)_i$ are diagonal subgroups of the external SO(2n) gauge group. The meaning of this decomposition is that the Calabi-Yau manifold X enjoys the decomposable line bundles as the form

$$X = \mathcal{L}_1 \oplus \cdots \oplus \mathcal{L}_n \to \Sigma_g. \tag{9.4.4}$$

Under the decomposition (9.4.3), the R-charges for $\mathbf{8}_v$, $\mathbf{8}_s$ and $\mathbf{8}_c$ are determined as follows:

1. $SO(8) \supset SO(6) \times SO(2)_1$

$$\begin{aligned} \mathbf{8}_{v} = \mathbf{6}_{0} \oplus \mathbf{1}_{2} \oplus \mathbf{1}_{-2} \\ \mathbf{8}_{s} = \mathbf{4}_{+} \oplus \overline{\mathbf{4}}_{-} \\ \mathbf{8}_{c} = \mathbf{4}_{-} \oplus \overline{\mathbf{4}}_{+}. \end{aligned}$$
(9.4.5)

2. $SO(8) \supset SO(4) \times SO(2)_1 \times SO(2)_2$

$$\begin{aligned}
\mathbf{8}_{v} = \mathbf{4}_{00} \oplus \mathbf{1}_{02} \oplus \mathbf{1}_{0-2} \oplus \mathbf{1}_{20} \oplus \mathbf{1}_{-20} \\
\mathbf{8}_{s} = \mathbf{2}_{++} \oplus \mathbf{2}'_{+-} \oplus \mathbf{2}_{--} \oplus \mathbf{2}'_{-+} \\
\mathbf{8}_{c} = \mathbf{2}_{-+} \oplus \mathbf{2}'_{--} \oplus \mathbf{2}_{+-} \oplus \mathbf{2}'_{++}.
\end{aligned}$$
(9.4.6)

3. $SO(8) \supset SO(2) \times SO(2)_1 \times SO(2)_2 \times SO(2)_3$

$$\begin{aligned} \mathbf{8}_{v} =& \mathbf{2}_{000} \oplus \mathbf{1}_{002} \oplus \mathbf{1}_{00-2} \oplus \mathbf{1}_{020} \oplus \mathbf{1}_{0-20} \oplus \mathbf{1}_{200} \oplus \mathbf{1}_{-200} \\ \mathbf{8}_{s} =& \mathbf{1}_{+++} \oplus \mathbf{1}_{++-} \oplus \mathbf{1}_{+-+} \oplus \mathbf{1}_{-++} \oplus \mathbf{1}_{--+} \oplus \mathbf{1}_{-+-} \oplus \mathbf{1}_{-++} \\ \mathbf{8}_{c} =& \mathbf{1}_{-++} \oplus \mathbf{1}_{-+-} \oplus \mathbf{1}_{---} \oplus \mathbf{1}_{--+} \oplus \mathbf{1}_{+-+} \oplus \mathbf{1}_{++--} \oplus \mathbf{1}_{+++-} \oplus \mathbf{1}_{++++}. \end{aligned}$$
(9.4.7)

With one of the decompositions (9.4.5)-(9.4.8), we can now define a new generator s', i.e. the $SO(2)'_E$ charge by

$$s' := s - \sum_{i=1}^{n} a_i T_i.$$
(9.4.9)

Here *s* denotes a generator of the original rotational group $SO(2)_E$, T_i represents a generator of the subgroup $SO(2)_i$ diagonally embedded in the external gauge group SO(2n) and a_i 's are the constant parameters characterizing the twisting procedures. From now on we normalize these charges *s*', *s* and T_i such that they are twice as the usual spin on the Riemann surface. Since a_i 's are related to the degrees of the line bundles \mathcal{L}_i 's as

$$\deg(\mathcal{L}_i) = \begin{cases} 2|g-1|a_i & \text{for } g \neq 1\\ a_i & \text{for } g = 1 \end{cases}$$
(9.4.10)

and the degrees correspond to the first Chern classes, the conditions that *X* is Calabi-Yau are given by

$$\sum_{i=1}^{n} a_i = \begin{cases} -1 & \text{for } g = 0\\ 0 & \text{for } g = 1\\ 1 & \text{for } g > 1 \end{cases}$$
(9.4.11)

Note that the Calabi-Yau conditions (9.4.11) simultaneously ensure the existence of the covariant constant spinors in the twisted theories. One can easily check that the topological twists underlying the decompositions (9.4.5), (9.4.6), (9.4.7) and (9.4.8) preserve 8, 4, 2 and 2 supersymmetries as we expect for K₃, *CY*₃, *CY*₄ and *CY*₅.

Therefore given the decomposable line bundle structures of the Calabi-Yau manifolds (9.4.4), we can determine the topological twisting procedure from the two conditions (9.4.10) and (9.4.11). For a K3 surface, i.e. for $a_2 = a_3 = a_4 = 0$, the local geometry is $T^*\Sigma_g$ and a single twisting parameter a_1 is uniquely determined by the Calabi-Yau condition. For other Calabi-Yau spaces the Calabi-Yau conditions are not so powerful and there are infinitely many ways of the twisting characterized by a_i , or the degrees of the line bundles.

Chapter 10

SCQM from M2-branes in a K3 surface

In this chapter we will give further detailed investigation on the wrapped M2branes on the holomorphic Riemann surface of genus g > 1 in a K3 surface. Firstly we will discuss the field content and the supersymmetry in the twisted theory and their consistency in section 10.1. Then we will derive the twisted theory in section 10.2. Finally we will compactify the twisted theory on the Riemann surface and find the IR quantum mechanics in section 10.3. The theory turns out to be the $\mathcal{N} = 8$ superconformal gauged quantum mechanics.

10.1 K3 twisting

In order to obtain the world-volume description for the membranes wrapping a curved Riemann surface of genus g > 1 embedded in a K3 surface, we should carry out the topological twisting utilizing the decomposition (9.4.5). Requiring the existence of covariant constant spinors, the twisting procedure can be uniquely determined since the external gauge field is nothing but an SO(2) Abelian background in this case. Note that the twisting for $\Sigma_g = \mathbb{P}^1$ can be realized just by the orientation reversal.

The decomposition $SO(2)_E \times SO(8)_R \rightarrow SO(2)'_E \times SO(6)_R$ yields the new field content and the supersymmetry parameters characterized by the following representations for the twisted field theory with g > 1:

$$X^{I}: \mathbf{8}_{v0} \to \mathbf{6}_{0} \oplus \mathbf{1}_{2} \oplus \mathbf{1}_{-2}$$

$$\epsilon: \mathbf{8}_{s+} \oplus \mathbf{8}_{s-} \to \mathbf{4}_{0} \oplus \overline{\mathbf{4}}_{2} \oplus \mathbf{4}_{-2} \oplus \overline{\mathbf{4}}_{0}$$

$$\Psi: \mathbf{8}_{c+} \oplus \mathbf{8}_{c-} \to \mathbf{4}_{2} \oplus \overline{\mathbf{4}}_{0} \oplus \mathbf{4}_{0} \oplus \overline{\mathbf{4}}_{-2}.$$
(10.1.1)

The results of the topological twisting for the components of fields and supersymmetry parameters are shown in Table 10.1 and 10.2 respectively. In the twisted theory the bosonic field content is six scalar fields ϕ^I transforming as $\mathbf{6}_0$ and one-forms Φ_z , $\Phi_{\overline{z}}$ transforming as $\mathbf{1}_2 \oplus \mathbf{1}_{-2}$. The fermionic field content is eight scalar fields ψ , $\tilde{\lambda}$ as $\mathbf{4}_0 \oplus \overline{\mathbf{4}}_0$ and one-forms Ψ_z , $\tilde{\Psi}_{\overline{z}}$ as $\mathbf{4}_2 \oplus \overline{\mathbf{4}}_{-2}$. The supersymmetry parameters are eight scalars ϵ , $\tilde{\epsilon}$ as $\mathbf{4}_0 \oplus \overline{\mathbf{4}}_0$ and one-forms $\tilde{\epsilon}_z$, $\epsilon_{\overline{z}}$ as $\overline{\mathbf{4}}_2 \oplus \mathbf{4}_{-2}$.

Here and hereafter we distinguish 4 and $\overline{4}$ in terms of tildes over the fermionic objects.

We should note that there are six bosonic scalar fields and eight fermionic scalar charges in the twisted theory. Since a Riemann surface is a real two-dimensional manifold and there are six scalar fields, the theory should describe the circumstance where the two-cycle lives in a 2 + (8 - 6) = 4-dimensional curved manifold *X*. The existence of eight scalar supercharges indicates that the four-manifold preserves $\frac{8}{16} = \frac{1}{2}$ of the supersymmetries. This is the case where a holomorphic Riemann surface Σ_g is embedded in a K₃ surface.

Locally the K₃ geometry is the cotangent bundle $T^*\Sigma_g$. The remaining two scalar fields combine to yield one-forms on the Riemann surface. They represent the motion of the M₂-branes along the non-trivial normal bundle N_{Σ} over the Riemann surface inside the K₃ surface. Under the *SO*(6) rotational group of the six uncompactified dimensions, the six scalars transform as vector representations $\mathbf{6}_v$ and the one-forms are just singlets. We take the eleven-dimensional space-time configuration as

where \circ denotes the direction in which the geometrical objects extend, while \times denotes the direction in which they localize. Note that the projection (9.2.11) for the K3 surface encodes the configuration (10.1.2). The world-volume of the M2-branes extend to a time direction x^0 and spacial directions x^1, x^2 . The spacial directions x^1, x^2 are tangent to the compact Riemann surface in the K3 surface. The normal geometry of the M2-branes is divided into two parts; one is the normal bundle N_{Σ} inside the K3 surface, extending to two directions x^9, x^{10} and the other is the flat Euclidean space transverse to the K3 surface, labeled by x^3, \dots, x^8 .

	$SO(2)_E$	$SO(2)_1$	$SO(2)'_E$	\mathcal{L}
ϕ^1	0	0	0	\mathcal{O}
ϕ^2	0	0	0	\mathcal{O}
ϕ^3	0	0	0	\mathcal{O}
ϕ^4	0	0	0	\mathcal{O}
ϕ^5	0	0	0	\mathcal{O}
ϕ^6	0	0	0	\mathcal{O}
$\Phi_{\overline{z}}$	0	2	-2	K^{-1}
Φ_z	0	-2	2	Κ
Ψ_{z1}	1	-1	2	K
Ψ_{z2}	1	-1	2	Κ
Ψ_{z3}	1	-1	2	Κ
Ψ_{z4}	1	-1	2	Κ
$ ilde{\lambda}_1$	1	1	0	\mathcal{O}
$\tilde{\lambda}_2$	1	1	0	\mathcal{O}
$\tilde{\lambda}_3$	1	1	0	\mathcal{O}
$ ilde{\lambda}_4$	1	1	0	\mathcal{O}
$ ilde{\Psi}_{\overline{z}1}$	-1	1	-2	K^{-1}
$ ilde{\Psi}_{\overline{z}2}$	-1	1	-2	K^{-1}
$ ilde{\Psi}_{\overline{z}3}$	-1	1	-2	K^{-1}
$ ilde{\Psi}_{\overline{z}4}$	-1	1	-2	K^{-1}
ψ_1	-1	-1	0	\mathcal{O}
ψ_2	-1	-1	0	\mathcal{O}
ψ_3	-1	-1	0	\mathcal{O}
ψ_4	-1	-1	0	\mathcal{O}

Table 10.1: The twisting for bosonic scalar fields X^{I} 's and fermionic fields Ψ 's of the BLG-model when the Riemann surface of genus g > 1 is embedded in a K3 surface. \mathcal{L} is the complex line bundle over Σ_g in which the fields take values. \mathcal{O} and K are the trivial bundle and the canonical bundle respectively.

	$SO(2)_E$	$SO(2)_1$	$SO(2)'_E$	\mathcal{L}
ϵ_1	1	1	0	\mathcal{O}
ϵ_2	1	1	0	\mathcal{O}
ϵ_3	1	1	0	\mathcal{O}
ϵ_4	1	1	0	\mathcal{O}
$ ilde{\epsilon}_{z1}$	1	-1	2	Κ
$ ilde{\epsilon}_{z2}$	1	-1	2	Κ
$ ilde{\epsilon}_{z3}$	1	-1	2	Κ
$ ilde{\epsilon}_{z4}$	1	-1	2	Κ
$ ilde{\epsilon}_1$	-1	-1	0	\mathcal{O}
$\tilde{\epsilon}_2$	-1	-1	0	\mathcal{O}
$ ilde{\epsilon}_3$	-1	-1	0	\mathcal{O}
$ ilde{\epsilon}_4$	-1	-1	0	\mathcal{O}
$\epsilon_{\overline{z}1}$	-1	1	-2	K^{-1}
$\epsilon_{\overline{z}2}$	-1	1	-2	K^{-1}
$\epsilon_{\overline{z}3}$	-1	1	-2	K^{-1}
$\epsilon_{\overline{z}4}$	-1	1	-2	K^{-1}

Table 10.2: The twisted supersymmetry parameters of the BLG-model probing a K3 surface. The eight covariant constant spinors play the role of BRST generators in the twisted theory. The result is consistent to the fact that a holomorphic curve inside a K3 surface can preserve a half of the supersymmetries (see Table 9.1).

10.2 Twisted theory

The space-time configuration (10.1.2) breaks down the space-time symmetry group SO(1, 10) to $SO(2)_E \times SO(6)_R \times SO(2)_1$. Then the SO(1, 10) gamma matrix can be decomposed as

$$\begin{cases} \Gamma^{\mu} = \gamma^{\mu} \otimes \hat{\Gamma}^{7} \otimes \sigma_{2} & \mu = 0, 1, 2 \\ \Gamma^{I+2} = \mathbb{I}_{2} \otimes \hat{\Gamma}^{I} \otimes \sigma_{2} & I = 1, \cdots, 6 \\ \Gamma^{i+8} = \mathbb{I}_{2} \otimes \mathbb{I}_{8} \otimes \gamma^{i} & i = 1, 2 \end{cases}$$
(10.2.1)

where ${}^{1}\hat{\Gamma}^{I}$ is the SO(6) gamma matrix obeying

$$\{\hat{\Gamma}^{I}, \hat{\Gamma}^{J}\} = 2\delta^{IJ}, \quad (\hat{\Gamma}^{I})^{\dagger} = \Gamma^{I}$$
(10.2.2)

$$\hat{\Gamma}^7 = -i\hat{\Gamma}^{12\cdots 6} = \begin{pmatrix} \mathbb{I}_4 & 0\\ 0 & -\mathbb{I}_4 \end{pmatrix}.$$
(10.2.3)

Similarly the SO(1, 10) charge conjugation matrix C is expressed as

$$\mathcal{C} = \epsilon \otimes \hat{\mathcal{C}} \otimes \epsilon \tag{10.2.4}$$

where $\epsilon := i\sigma_2$ is introduced as the charge conjugation matrix with the relations

$$\epsilon^T = -\epsilon, \quad \epsilon \gamma^\mu \epsilon^{-1} = -(\gamma^\mu)^T$$
 (10.2.5)

while \hat{C} is the SO(6) charge conjugation matrix satisfying ²

$$\hat{C}^{T} = -\hat{C}, \qquad \hat{C}\hat{\Gamma}^{I}\hat{C}^{-1} = (\hat{\Gamma}^{I})^{T}, \qquad \hat{C}\hat{\Gamma}^{7}\hat{C}^{-1} = -(\hat{\Gamma}^{7})^{T}.$$
 (10.2.7)

Under the decomposition (10.2.1), the SO(8) chiral matrix becomes

$$\Gamma^{012} = \Gamma^{34\dots10} = \mathbb{I}_2 \otimes \hat{\Gamma}^7 \otimes \sigma_2. \tag{10.2.8}$$

(d + 1)-th component of d = t + s dimensional gamma matrices can be defined by [320]

$$\Gamma^{d+1} := \sqrt{(-1)^{\frac{s-t}{2}}} \Gamma^{12 \cdots d}$$

where *s* and *t* are correspond to the dimension of space and time respectively. In the above case s = 6 and t = 0. Note that minus sign should be included in (10.2.3) since we are now considering the decomposition of (4.1.6).

 2 In even dimensional space-time, a charge conjugation matrix can be defined in two ways. Instead of (10.2.7), we may define

$$\hat{C}^T = \hat{C}, \quad \hat{C}\hat{\Gamma}^I\hat{C}^{-1} = -(\hat{\Gamma}^I)^T.$$
 (10.2.6)

However, Majorana spinors are only allowed for (10.2.7).

For the twisted bosonic fields we set

$$\phi^I := X^{I+2},\tag{10.2.9}$$

$$\Phi_z := \frac{1}{\sqrt{2}} (X^9 - iX^{10}), \qquad \Phi_{\overline{z}} := \frac{1}{\sqrt{2}} (X^9 + iX^{10}), \qquad (10.2.10)$$

$$A_{z} := \frac{1}{\sqrt{2}} (A_{1} - iA_{2}), \qquad \qquad A_{\overline{z}} := \frac{1}{\sqrt{2}} (A_{1} + iA_{2}) \qquad (10.2.11)$$

where the bosonic scalar fields ϕ^{I} 's transform as the vector representations $\mathbf{6}_{v}$ of the SO(6) global symmetry and the indices $I = 1, \dots, 6$ label the flat transverse directions. The bosonic one-froms, Φ_{z} and $\Phi_{\overline{z}}$ are the SO(6)-singlets and they describe the motion in the normal geometry N_{Σ} of the Riemann surface inside the K₃ surface. These Higgs fields ϕ^{I} , Φ_{z} and $\Phi_{\overline{z}}$ are the 3-algebra valued.

Next, consider the twisted fermionic objects. Originally the fermionic fields Ψ are $SL(2, \mathbb{R})$ spinors that transform as the spinor representations $\mathbf{8}_c$ of the $SO(8)_R$ R-symmetry. After the decompositions $Spin(1, 10) \rightarrow Spin(2) \times Spin(6) \times Spin(2)$, as seen from (10.1.1), the fermionic fields Ψ are split into the representations $\mathbf{4}_2$, $\overline{\mathbf{4}}_0$, $\mathbf{4}_0$ and $\overline{\mathbf{4}}_{-2}$, whose component fields are denoted by Ψ_z , $\tilde{\lambda}$, ψ and $\tilde{\Psi}_{\overline{z}}$ respectively. Accordingly they can be expanded as

$$\Psi_{A}^{\alpha\beta} = \frac{i}{\sqrt{2}} \psi_{A} (\gamma_{+}\epsilon^{-1})^{\alpha\beta} + i \tilde{\Psi}_{\bar{z}A} (\gamma^{\bar{z}}\epsilon^{-1})^{\alpha\beta} - \frac{i}{\sqrt{2}} \tilde{\lambda}_{A} (\gamma_{-}\epsilon^{-1})^{\alpha\beta} - i \Psi_{zA} (\gamma^{z}\epsilon^{-1})^{\alpha\beta}$$
(10.2.12)

where the three indices α , A and β denote the $SO(2)_E$ spinor, the $SO(6)_R$ spinor and the $SO(2)_1$ spinor respectively. Here we have introduced the matrices γ_{\pm} , γ^z and $\gamma^{\overline{z}}$ defined by

$$\gamma_{+} := \frac{1}{\sqrt{2}}(\mathbb{I}_{2} + \sigma_{2}), \quad \gamma_{-} := \frac{1}{\sqrt{2}}(\mathbb{I}_{2} - \sigma_{2}), \quad (10.2.13)$$

$$\gamma^{z} := \frac{1}{\sqrt{2}} (\gamma^{1} + i\gamma^{2}) = \frac{1}{\sqrt{2}} \begin{pmatrix} i & 1\\ 1 & -i \end{pmatrix},$$
(10.2.14)

$$\gamma^{\bar{z}} := \frac{1}{\sqrt{2}} (\gamma^1 - i\gamma^2) = \frac{1}{\sqrt{2}} \begin{pmatrix} -i & 1\\ 1 & i \end{pmatrix}.$$
 (10.2.15)

As seen from (10.2.12), the above matrices enable us to carry out the topological twisting, or in other words the identification of the index α with the index β . The matrices γ_+ and $\gamma^{\overline{z}}$ are associated with the conjugate spinor representations $\mathbf{8}_{c-}$ and yield $\mathbf{4}_0$ and $\overline{\mathbf{4}}_{-2}$, while the other pair of matrices γ_- and γ^z are associated with $\mathbf{8}_{c+}$ and give rise to $\mathbf{4}_2$ and $\overline{\mathbf{4}}_0$. Together with the decomposition (10.2.8) and the chirality condition (4.1.8) for Ψ , one can check that the expansion (10.2.12)

leads to the relations; $\hat{\Gamma}^{7}\psi = \psi$, $\hat{\Gamma}^{7}\tilde{\Psi}_{\overline{z}} = -\tilde{\Psi}_{\overline{z}}$, $\hat{\Gamma}^{7}\tilde{\lambda} = -\tilde{\lambda}$ and $\hat{\Gamma}^{7}\Psi_{z} = \Psi_{z}$. For the \mathcal{A}_{4} algebra all of these fermionic fields are the fundamental representations of the SO(4) gauge group. We define the conjugate of the SO(6) spinors as

$$\overline{\psi} := \psi^T \hat{C}, \quad \overline{\tilde{\lambda}} := \tilde{\lambda}^T \hat{C}, \quad \overline{\Psi}_z := \Psi_z^T \hat{C}, \quad \overline{\tilde{\Psi}}_{\overline{z}} := \tilde{\Psi}_{\overline{z}}^T \hat{C}. \tag{10.2.16}$$

Likewise, the supersymmetry parameters originally transform as the $SL(2, \mathbb{R})$ spinor representations of the rotational group of the world-volume and $\mathbf{8}_s$ of the SO(8) R-symmetry in the BLG-model, while in the twisted theory they reduce to the four distinct representations $\mathbf{4}_0$, $\overline{\mathbf{4}}_2$, $\mathbf{4}_{-2}$ and $\overline{\mathbf{4}}_0$. Thus we can write supersymmetry parameters as

$$\epsilon_A^{\alpha\beta} = \frac{i}{\sqrt{2}} \tilde{\epsilon}_A (\gamma_+ \epsilon^{-1})^{\alpha\beta} + i\epsilon_{\bar{z}A} (\gamma^{\bar{z}} \epsilon^{-1})^{\alpha\beta} - \frac{i}{\sqrt{2}} \epsilon_A (\gamma_- \epsilon^{-1})^{\alpha\beta} - i\tilde{\epsilon}_{zA} (\gamma^z \epsilon^{-1})^{\alpha\beta}.$$
(10.2.17)

Here again the indices α , A and β label $SO(2)_E$, $SO(6)_R$ and $SO(2)_1$ respectively. Since ϵ and $\tilde{\epsilon}$ are fermionic scalars on an arbitrary Riemann surface, they are identified with supercharges and hence the effective theory will be endowed with the corresponding eight supercharges.

Plugging the expressions (10.2.1), (10.2.9), (10.2.10), (10.2.11) and (10.2.12) into the original BLG Lagrangian (4.1.31), we find the topologically twisted BLG Lagrangian

$$\begin{aligned} \mathcal{L} &= \frac{1}{2} (D_0 \phi^I, D_0 \phi^I) - (D_z \phi^I, D_{\overline{z}} \phi^I) + (D_0 \Phi_z, D_0 \Phi_{\overline{z}}) - 2 (D_z \Phi_{\overline{w}}, D_{\overline{z}} \Phi_w) \\ &+ (\overline{\lambda}, D_0 \psi) + (\overline{\Psi}_z, D_0 \widetilde{\Psi}_{\overline{z}}) - (\overline{\Psi}_{\overline{z}}, D_0 \Psi_z) - 2i (\overline{\Psi}_{\overline{z}}, D_z \psi) + 2i (\overline{\lambda}, D_{\overline{z}} \Psi_z) \\ &+ \frac{i}{2} (\overline{\lambda} \widehat{\Gamma}^{IJ}, [\phi^I, \phi^J, \psi]) - i (\overline{\Psi}_{\overline{z}} \widehat{\Gamma}^{IJ}, [\phi^I, \phi^J, \Psi_z]) \\ &+ 2i (\overline{\psi} \widehat{\Gamma}^I, [\Phi_{\overline{z}}, \phi^I, \Psi_z]) - 2i (\overline{\lambda} \widehat{\Gamma}^I, [\Phi_z, \phi^I, \Psi_{\overline{z}}]) \\ &+ i (\overline{\lambda}, [\Phi_z, \Phi_{\overline{z}}, \psi]) - 2i (\overline{\Psi}_{\overline{w}}, [\Phi_z, \Phi_{\overline{z}}, \Psi_w]) \\ &- \frac{1}{12} \left([\phi^I, \phi^J, \phi^K], [\phi^I, \phi^J, \phi^K] \right) - \frac{1}{2} \left([\Phi_z, \phi^I, \phi^J], [\Phi_{\overline{z}}, \phi^I, \phi^J] \right) \\ &- \frac{1}{2} \left([\Phi_z, \Phi_w, \phi^I], [\Phi_{\overline{z}}, \Phi_{\overline{w}}, \phi^I] \right) - \frac{1}{2} \left([\Phi_z, \Phi_w, \Phi_{\overline{v}}], [\Phi_{\overline{z}}, \Phi_w, \phi^I] \right) \\ &+ \frac{1}{6} \left([\Phi_z, \Phi_w, \Phi_v], [\Phi_{\overline{z}}, \Phi_{\overline{w}}, \Phi_{\overline{v}}] \right) + \frac{1}{2} \left([\Phi_z, \Phi_w, \Phi_{\overline{v}}], [\Phi_{\overline{z}}, \Phi_w, \Phi_v] \right) + \mathcal{L}_{TCS}. \end{aligned}$$

$$(10.2.18)$$

Here we have introduced (,) as the trace form on the 3-algebra introduced in (4.1.13) and we have defined the covariant derivatives $D_z := \frac{1}{\sqrt{2}}(D_1 - iD_2)$ and $D_{\overline{z}} := \frac{1}{\sqrt{2}}(D_1 + iD_2)$.

Substituting the expressions (10.2.1), (10.2.9), (10.2.10), (10.2.11), (10.2.12) and (10.2.17) into the supersymmetry transformations (4.1.49)-(4.1.51) for BLG theory, we can read off the following BRST transformations

$$\delta \phi_a^I = i \overline{\tilde{\epsilon}} \widehat{\Gamma}^I \tilde{\lambda}_a - i \overline{\epsilon} \widehat{\Gamma}^I \psi_a, \tag{10.2.19}$$

$$\delta \Phi_{za} = -i\bar{\tilde{\epsilon}}\Psi_{za},\tag{10.2.20}$$

$$\delta \Phi_{\bar{z}a} = -i\bar{\epsilon}\tilde{\Psi}_{\bar{z}a},\tag{10.2.21}$$

$$\delta\psi_a = iD_0\phi_a^I\hat{\Gamma}\tilde{\epsilon} - 2D_{\overline{z}}\Phi_{za}\epsilon + \frac{1}{6}[\phi^I,\phi^J,\phi^K]_a\hat{\Gamma}^{IJK}\tilde{\epsilon} + [\Phi_z,\Phi_{\overline{z}},\phi^I]_a\hat{\Gamma}^I\tilde{\epsilon}, \qquad (10.2.22)$$

$$\delta\tilde{\lambda}_{a} = iD_{0}\phi_{a}^{I}\hat{\Gamma}^{I}\epsilon - 2D_{z}\Phi_{\overline{z}a}\tilde{\epsilon} - \frac{1}{6}[\phi^{I},\phi^{J},\phi^{K}]_{a}\hat{\Gamma}^{IJK}\epsilon + [\Phi_{z},\Phi_{\overline{z}},\phi^{I}]_{a}\hat{\Gamma}^{I}\epsilon, \qquad (10.2.23)$$

$$\delta \Psi_{za} = -D_z \phi^I \hat{\Gamma}^I \tilde{\epsilon} - i D_0 \Phi_z \epsilon + \frac{1}{2} [\Phi_z, \phi^I, \phi^J]_a \hat{\Gamma}^{IJ} \epsilon + \frac{1}{3} [\Phi_w, \Phi_{\overline{w}}, \Phi_z]_a \epsilon, \qquad (10.2.24)$$

$$\delta \tilde{\Psi}_{\bar{z}a} = D_{\bar{z}} \phi_a^I \hat{\Gamma}^I \epsilon + i D_0 \Phi_{\bar{z}a} \tilde{\epsilon} + \frac{1}{2} [\Phi_{\bar{z}}, \phi^I, \phi^J]_a \hat{\Gamma}^{IJ} \tilde{\epsilon} + \frac{1}{3} [\Phi_{\overline{w}}, \Phi_w, \Phi_{\bar{z}}]_a \tilde{\epsilon}, \qquad (10.2.25)$$

$$\delta \tilde{A}_{0a}^{b} = -\bar{\epsilon} \hat{\Gamma}^{I} \phi_{c}^{I} \psi_{d} f^{cdb}{}_{a} - \bar{\tilde{\epsilon}} \hat{\Gamma}^{I} \phi_{c}^{I} \tilde{\lambda}_{d} f^{cdb}{}_{a} - 2\bar{\epsilon} \Phi_{zc} \tilde{\Psi}_{\bar{z}d} f^{cdb}{}_{a} + 2\bar{\tilde{\epsilon}} \Phi_{\bar{z}c} \Psi_{zd} f^{cdb}{}_{a}, \quad (10.2.26)$$

$$\delta \hat{A}^{b}_{za} = 2i\overline{\epsilon} \hat{\Gamma}^{1} \phi^{T}_{c} \Psi_{zd} f^{cdb}_{\ a} + 2i\overline{\epsilon} \Phi_{zc} \hat{\lambda}_{d} f^{cdb}_{\ a'}$$
(10.2.27)

$$\delta \tilde{A}^{b}_{\bar{z}a} = -2i\bar{\bar{e}}\hat{\Gamma}^{I}\phi^{I}_{c}\tilde{\Psi}_{\bar{z}d}f^{cdb}_{\ a} + 2i\bar{\bar{e}}\Phi_{\bar{z}c}\psi_{d}f^{cdb}_{\ a}.$$
(10.2.28)

10.3 Derivation of quantum mechanics

In the previous section we have derived the topologically twisted BLG-model as the low-energy effective field theories on the curved M2-branes. Now we attempt to reduce the theory further to a low-energy effective one-dimensional field theory on \mathbb{R} . As mentioned in the analysis for the membranes wrapped around a torus, when the size of the Riemann surface shrinks, only the light degrees of freedom are relevant. To keep track of them we have to find the static configurations that minimize the energy, that is the zero-energy conditions. We can replace the zeroenergy conditions by a set of BPS equations. In addition, we set all the fermionic fields to zero because we are interested in bosonic BPS configurations. Then the BPS equations, which correspond to the vanishing conditions of the BRST transformations (10.2.22)-(10.2.25) for the fermionic fields, are

$$D_z \phi^I = 0, \quad D_{\overline{z}} \phi^I = 0, \tag{10.3.1}$$

$$D_z \Phi_{\overline{z}} = 0, \quad D_{\overline{z}} \Phi_z = 0, \tag{10.3.2}$$

$$[\phi^{I}, \phi^{J}, \phi^{K}] = 0, \tag{10.3.3}$$

$$[\Phi_z, \Phi_{\overline{z}}, \phi^I] = 0, \quad [\Phi_z, \phi^I, \phi^J] = 0, \quad [\Phi_{\overline{z}}, \phi^I, \phi^J] = 0,$$
 (10.3.4)

$$[\Phi_w, \Phi_{\overline{w}}, \Phi_z] = 0, \quad [\Phi_{\overline{w}}, \Phi_w, \Phi_{\overline{z}}] = 0.$$
(10.3.5)

Note that due to the algebraic equations (10.3.3), (10.3.4) and (10.3.5), all the bosonic Higgs fields have to lie in the same plane in the SO(4) gauge group. Thus we can write them as

$$\phi^{I} = (\phi^{I1}, \phi^{I2}, 0, 0)^{T}, \quad \Phi_{z} = (\Phi_{z}^{1}, \Phi_{z}^{2}, 0, 0)^{T}, \quad \Phi_{\overline{z}} = (\Phi_{\overline{z}}^{1}, \Phi_{\overline{z}}^{2}, 0, 0)^{T}.$$
 (10.3.6)

From the supersymmetry we can write the corresponding fermionic partners as

$$\psi = (\psi^1, \psi^2, 0, 0)^T, \qquad \tilde{\lambda} = (\tilde{\lambda}^1, \tilde{\lambda}^2, 0, 0)^T, \qquad (10.3.7)$$

$$\Psi_{z} = (\Psi_{z}^{1}, \Psi_{z}^{2}, 0, 0)^{T}, \qquad \qquad \tilde{\Psi}_{\overline{z}} = (\tilde{\Psi}_{\overline{z}}^{1}, \tilde{\Psi}_{\overline{z}}^{2}, 0, 0)^{T}.$$
(10.3.8)

The configurations (10.3.6)-(10.3.8) generically break the original SO(4) gauge group down to $U(1) \times U(1)$. Taking into account these solutions and the BPS equations (10.3.1), (10.3.2) we find that $\tilde{A}_{z3}^1 = \tilde{A}_{z3}^2 = \tilde{A}_{z4}^1 = \tilde{A}_{z4}^2 = 0$. This implies that these components of the gauge field now become massive by the Higgs mechanism. Then we should follow the time evolution for remaining degrees of freedom in the low-energy effective theory.

To achieve this consistently we further need to impose the Gauss law constraint. This requires that the gauge field is flat; $\tilde{F}_{z\bar{z}} = 0$. Recall that we are now considering the case where the genus of the Riemann surface is greater than one. In that case the generic flat connections are irreducible. As long as we only consider irreducible flat connections, the Laplacian has no zero modes. Accordingly it is not allowed for scalar fields to have non-trivial values and it is required that $\phi^I = 0^3$. To sum up, the above set of equations over the compact Riemann surface of genus g > 1 reduces to

$$\tilde{F}_{z\overline{z}2}^1 = 0,$$
 (10.3.9)

$$\partial_{\bar{z}} \Phi_{z1} + \tilde{A}^1_{\bar{z}2} \Phi_{z2} = 0, \tag{10.3.10}$$

$$\partial_{\bar{z}} \Phi_{z2} - \tilde{A}_{\bar{z}2}^1 \Phi_{z1} = 0. \tag{10.3.11}$$

We now want to determine the generic BPS configuration obeying (10.3.9)-(10.3.11). Since we are now considering a compact Riemann surface of genus g, there are g holomorphic (1,0)-forms ω_i , $i = 1, \dots, g$ and g anti-holomorphic (0,1)-forms $\overline{\omega}_i$. Let us normalize them as

$$\int_{a_i} \omega_j = \delta_{ij}, \quad \int_{b_i} \omega_j = \Omega_{ij} \tag{10.3.12}$$

with a_i , b_i being canonical homology basis for $H_1(\Sigma_g)$ (see Figure 10.1). The matrix



Figure 10.1: A Riemann surface Σ_g of genus g. a_i and b_i generate $H_1(\Sigma_g)$.

 Ω is the period matrix of the Riemann surface. It is a $g \times g$ complex symmetric matrix with positive imaginary part. The equation (10.3.9) imposes the flatness condition for the U(1) gauge field \tilde{A}_{z2}^1 . The space of the U(1) flat connection on a compact Riemann surface is the torus known as the Jacobi variety denoted by $Jac(\Sigma_g)$. The flat gauge fields can be expressed in the form [284]

$$\tilde{A}_{z2}^{1} = -2\pi \sum_{i,j=1}^{g} \left(\Omega - \overline{\Omega}\right)_{ij}^{-1} \Theta^{i} \omega_{j}, \qquad \tilde{A}_{\overline{z}2}^{1} = 2\pi \sum_{i,j=1}^{g} \left(\Omega - \overline{\Omega}\right)_{ij}^{-1} \overline{\Theta}^{i} \overline{\omega}_{j} \qquad (10.3.13)$$

where $\Theta^i := \zeta^i + \overline{\Omega}_{ij} \xi^j$ represents the complex coordinate of $\operatorname{Jac}(\Sigma_g)$ which characterizes the twists $e^{2\pi i \xi^i}$ and $e^{-2\pi i \zeta^i}$ around the *i*-th homology cycles a_i and b_i . Notice that $\xi^i \to \xi^i + m^i$, $\zeta^i \to \zeta^i + n^i$ for $n^i, m^i \in \mathbb{Z}$ gives rise to the same point on $\operatorname{Jac}(\Sigma_g)$. This implies that $\operatorname{Jac}(\Sigma_g) = \mathbb{C}^g / L_\Omega$ where L_Ω is the lattice generated by $\mathbb{Z}^g + \Omega \mathbb{Z}^g$. We define a function

$$\varphi := -2\pi \sum_{i,j=1}^{g} \left(\Omega - \overline{\Omega}\right)_{ij}^{-1} \left(\Theta^{i} f_{j}(z) - \overline{\Theta}^{i} \overline{f}_{j}(\overline{z})\right)$$
(10.3.14)

where $f_i(z) := \int^z \omega_i$ is the holomorphic function of z that obeys the relations $f_i|_{a_j} = \delta_{ij}$ and $f_i|_{b_j} = \Omega_{ij}$. We then can express the flat gauge fields as

$$\tilde{A}_{z2}^1 = \partial_z \varphi, \qquad \tilde{A}_{\overline{z}2}^1 = \partial_{\overline{z}} \varphi.$$
 (10.3.15)

The above expressions (10.3.13) for the U(1) flat connection allows us to write the

³ Such BPS solutions with the irreducible connections have been considered in the fourdimensional topologically twisted Yang-Mills theories defined on the product of two Riemann surfaces [321, 36, 293] and the corresponding decoupling limit for the brane description has been argued in [40].

generic solutions to the equation (10.3.10) and (10.3.11) in the following forms:

$$\Phi_{z1}(z,\bar{z}) - i\Phi_{z2}(z,\bar{z}) = e^{-i\varphi(z,\bar{z})} \sum_{i=1}^{g} x_A^i \omega_i,$$

$$\Phi_{z1}(z,\bar{z}) + i\Phi_{z2}(z,\bar{z}) = e^{i\varphi(z,\bar{z})} \sum_{i=1}^{g} x_B^i \omega_i$$
(10.3.16)

where x_A^i , $x_B^i \in \mathbb{C}$ are constant on the Riemann surface. Since we take the limit where the Riemann surface Σ_g shrinks to zero size, the space-time configurations of the membranes should be expressed as single-valued functions of z and \overline{z} in the low-energy effective quantum mechanics. In other words, ξ^i and ζ^i can only be integers and therefore the U(1) flat gauge fields \tilde{A}_{z2}^1 and $\tilde{A}_{\overline{z2}}^1$ are quantized. The single-valuedness condition requires that the point of the Jac(Σ_g) is fixed.

Putting all together, the general bosonic BPS configurations are given by

$$\begin{split} \phi^{I} &= 0 \\ \Phi_{z} &= \sum_{i=1}^{g} \begin{pmatrix} \frac{1}{2} \left(e^{-i\varphi} x_{A}^{i} + e^{i\varphi} x_{B}^{i} \right) \\ \frac{i}{2} \left(e^{-i\varphi} x_{A}^{i} - e^{i\varphi} x_{B}^{i} \right) \\ 0 \\ 0 \end{pmatrix} \omega_{i}, \quad \Phi_{\overline{z}} &= \sum_{i=1}^{g} \begin{pmatrix} \frac{1}{2} \left(e^{i\varphi} \overline{x}_{A}^{i} + e^{-i\varphi} \overline{x}_{B}^{i} \right) \\ -\frac{i}{2} \left(e^{i\varphi} \overline{x}_{A}^{i} - e^{-i\varphi} \overline{x}_{B}^{i} \right) \\ 0 \\ 0 \end{pmatrix} \overline{\omega}_{i}, \\ \tilde{A}_{z} &= \begin{pmatrix} 0 & \partial_{z} \varphi(z, \overline{z}) & 0 & 0 \\ -\partial_{z} \varphi(z, \overline{z}) & 0 & 0 & 0 \\ 0 & 0 & 0 & \tilde{A}_{z4}^{3}(z, \overline{z}) \\ 0 & 0 & -\tilde{A}_{z4}^{3}(z, \overline{z}) & 0 \end{pmatrix}$$
(10.3.17)

where \tilde{A}_{z4}^3 and $\tilde{A}_{\bar{z}4}^3$ are the Abelian gauge fields associated with preserved U(1) symmetry and they do not receive any constraints from the BPS conditions.

By virtue of the supersymmetry we can write the corresponding fermionic fields from the bosonic configurations (10.3.17) as

$$\begin{split} \psi &= 0, \qquad \qquad \tilde{\lambda} = 0, \\ \Psi_z &= \sum_{i=1}^g \begin{pmatrix} \frac{1}{2} \left(\Psi_A^i + \Psi_B^i \right) \\ \frac{i}{2} \left(\Psi_A^i - \Psi_B^i \right) \\ 0 \\ 0 \end{pmatrix} \omega_i, \qquad \tilde{\Psi}_{\overline{z}} = \sum_{i=1}^g \begin{pmatrix} \frac{1}{2} \left(\tilde{\Psi}_A^i + \tilde{\Psi}_B^i \right) \\ -\frac{i}{2} \left(\tilde{\Psi}_A^i - \tilde{\Psi}_B^i \right) \\ 0 \\ 0 \end{pmatrix} \overline{\omega}_i. \quad (10.3.18) \end{split}$$

By inserting the BPS configuration (10.3.17) and (10.3.18) into the twisted action

(10.2.18), we find

$$S = \int_{\mathbb{R}} dt \int_{\Sigma_{g}} d^{2}z \left[(D_{0}\Phi_{z}^{a}, D_{0}\Phi_{\bar{z}a}) + \left(\overline{\Psi}_{z}^{a}, D_{0}\tilde{\Psi}_{\bar{z}a}\right) - \left(\overline{\tilde{\Psi}}_{\bar{z}}^{a}, D_{0}\Psi_{za}\right) - \frac{k}{2\pi}\tilde{A}_{02}^{1}\tilde{F}_{z\bar{z}4}^{3} - \frac{k}{4\pi}\left(\tilde{A}_{z2}^{1}\dot{A}_{\bar{z}4}^{3} - \tilde{A}_{\bar{z}2}^{1}\dot{A}_{z4}^{3}\right) \right].$$
(10.3.19)

Since the gauge fields \tilde{A}_{z2}^1 , $\tilde{A}_{\bar{z}2}^1$ are quantized and there are no their time derivatives in the effective action, we can integrate them out as the auxiliary fields. They give rise to the constraints $\dot{A}_{z4}^3 = \dot{A}_{\bar{z}4}^3 = 0$.

In order to perform the integration over the Riemann surface, we use the Riemann bilinear relation [322]

$$\int_{\Sigma_g} \omega \wedge \eta = \sum_{i=1}^g \left[\int_{a_i} \omega \int_{b_i} \eta - \int_{b_i} \omega \int_{a_i} \eta \right].$$
(10.3.20)

By carrying out the integration over the Riemann surface Σ_g we find the gauged quantum mechanical action

$$S = \int_{\mathbb{R}} dt \left[\sum_{i,j} \left(\operatorname{Im} \Omega \right)_{ij} \left(D_0 x^{ia} D_0 \overline{x}_a^j + \overline{\Psi}^{ia} D_0 \overline{\Psi}_a^j - \overline{\widetilde{\Psi}}^{ia} D_0 \Psi_a^j \right) - k C_1(E) \widetilde{A}_{02}^1 \right].$$
(10.3.21)

Here the indices a = A, B stand for the two internal degrees of freedom for the two M2-branes. The covariant derivatives are defined by

$$D_0 x_A^i = \dot{x}_A^i + i \tilde{A}_{02}^1 x_A^i, \qquad D_0 x_B^i = \dot{x}_B^i - i \tilde{A}_{02}^1 x_B^i, \qquad (10.3.22)$$

$$D_0 \Psi_A^i = \dot{\Psi}_A^i + i \tilde{A}_{02}^1 \Psi_A^i, \qquad D_0 \Psi_B^i = \dot{\Psi}_B^i - i \tilde{A}_{02}^1 \Psi_B^i, \qquad (10.3.23)$$

$$D_0 \tilde{\Psi}_A^i = \dot{\tilde{\Psi}}_A^i - i \tilde{A}_{02}^1 \tilde{\Psi}_A^i, \qquad D_0 \Psi_B^i = \dot{\tilde{\Psi}}_B^i + i \tilde{A}_{02}^1 \tilde{\Psi}_B^i \qquad (10.3.24)$$

and the Chern number $C_1(E) \in \mathbb{Z}$ is associated to the U(1) principal bundle $E \to \Sigma_g$ over the Riemann surface

$$C_1(E) = \int_{\Sigma_g} c_1(E) = \frac{1}{2\pi} \int_{\Sigma_g} d^2 z \tilde{F}_{z\bar{z}4}^3.$$
 (10.3.25)

The action (10.3.21) has the invariance under the one-dimensional $SL(2, \mathbb{R})$ conformal transformations

$$\delta t = f(t) = a + bt + ct^2, \qquad \qquad \delta \partial_0 = -\dot{f} \partial_0, \qquad (10.3.26)$$

$$\delta x_a^i = \frac{1}{2} \dot{f} x_a^i, \qquad \qquad \delta \tilde{A}_{02}^1 = -\dot{f} \tilde{A}_{02}^1, \qquad (10.3.27)$$

$$\delta \Psi_a^i = 0, \qquad \qquad \delta \tilde{\Psi}_a^i = 0. \tag{10.3.28}$$

The action (10.3.21) is also invariant under the $\mathcal{N} = 8$ supersymmetry transformation laws

$$\delta x_a^i = 2i\bar{\tilde{\epsilon}}\Psi_a^i, \qquad \qquad \delta \bar{x}_a^i = 2i\bar{\epsilon}\tilde{\Psi}_a^i, \qquad (10.3.29)$$

$$\delta \Psi_a^i = -i D_0 x_a^i \epsilon, \qquad \qquad \delta \tilde{\Psi}_a^i = i D_0 \overline{x}_a^i \tilde{\epsilon}, \qquad (10.3.30)$$

$$\delta \tilde{A}_{02}^1 = 0. \tag{10.3.31}$$

We thus conclude that the $\mathcal{N} = 8$ superconformal gauged quantum mechanics (10.3.21) may describe the low-energy effective motion of the two wrapped M2branes around Σ_g probing a K3 surface.

We see from the action (10.3.21) that the U(1) gauge field \tilde{A}_{02}^1 , due to the absence of the kinetic term, is regarded as an auxiliary field. In consequence the gauge field has no contribution to the Hamiltonian. Hence the corresponding gauge symmetry yields an integral of motion as a moment map $\mu : \mathcal{M} \to \mathfrak{u}(1)^*$ and we can reduce the phase space \mathcal{M} to $\mathcal{M}_c = \mu^{-1}(c)$ by fixing the inverse of the moment map at a point $c \in \mathfrak{u}(1)^*$. Choosing a temporal gauge $\tilde{A}_{02}^1 = 0$, we find the action

$$S = \int_{\mathbb{R}} dt \sum_{i,j} \left(\operatorname{Im}\Omega \right)_{ij} \left(\dot{x}^{ia} \dot{\overline{x}}^{j}_{a} + \overline{\Psi}^{ia} \dot{\overline{\Psi}}^{j}_{a} - \overline{\widetilde{\Psi}}^{ia} \dot{\Psi}^{j}_{a} \right)$$
(10.3.32)

and the Gauss law constraint

$$\phi_0 := kC_1(E) + i\sum_{i,j} \left(\operatorname{Im}\Omega \right)_{ij} \left[K_{ij} + 2\left(\overline{\Psi}^i_A \tilde{\Psi}^j_A - \overline{\Psi}^i_B \tilde{\Psi}^j_B \right) \right] = 0$$
(10.3.33)

where

$$K_{ij} := \left(\dot{x}_A^i \overline{x}_A^j - x_A^i \dot{\overline{x}}_A^j\right) - \left(\dot{x}_B^i \overline{x}_B^j - x_B^i \dot{\overline{x}}_B^j\right).$$
(10.3.34)

The constraint equation (10.3.33) requires that all states in the Hilbert space are gauge invariant. In this case the symmetry of the system is not so large as in the previous superconformal gauged quantum mechanical models (6.1.13) and (7.1.17). It is curious to know whether the superconformal gauged quantum mechanics (10.3.21) (or (10.3.32) together with (10.3.33)) have a reduced Lagrangian description with an inverse-square type potential. However, our result may drop a hint on the obstructed construction of SCQM that a large class of SCQM could be formulated as "gauged quantum mechanics" with the help of auxiliary gauge fields as in [150, 151, 152].

It might be helpful to determine the corresponding supermultiplet for our $\mathcal{N} = 8$ superconformal quantum mechanics (10.3.21). We, however, do not fully

understand it because our derivation does not rely on the superfield formulation and the reduced quantum mechanical description has not been acquired so far. Judging from the representations (10.1.1) of the physical variables under the remaining R-symmetry SO(6), the corresponding supermultiplet may be inferred as the *g* sets of (**2**, **8**, **6**) multiplet. However, after integrating out the single auxiliary gauge field \tilde{A}_{02}^1 , the physical degrees of freedom may be reduced and thus the supermultiplets may be modified.

Chapter 11

Conclusion and Discussion

11.1 Conclusion

In this thesis we have established the new connection between two subjects; the superconformal quantum mechanics and the M2-branes by examining the IR superconformal quantum mechanics resulting from the multiple M2-branes wrapped around a compact Riemann surface Σ_g after shrinking the size of the Riemann surface.

We have seen that conformal symmetry and supersymmetry in quantum mechanics, i.e. one-dimensional field theory are rather out of the way in that they contain numerous unfamiliar features which are not observed in higher dimensional field theories.

Instead of the morbid Hamiltonian, one can label the state in terms of the eigenstate of the compact operator $L_0 = \frac{1}{2}(H + K)$ and the second Casimir operator of the $SL(2, \mathbb{R})$ conformal symmetry group. Although one cannot assume the existence of both normalizable conformally invariant states and invariant primary operators due to the fact that the quantum mechanics is based on the Hilbert space not on the Fock space, the 2-point, 3-point and 4-point functions which satisfy the conformal constraints can be constructed by using those two defects [86, 87]. We have also discussed the interesting observations [65, 66] that the motion of the particle near the horizon of the extreme Reissner-Nordström black hole is described by the (super)conformal mechanics. This indicates that (super)conformal quantum mechanics may caputure the information of the dual AdS₂ gravity. Obviously further surveys are needed to understand AdS₂/CFT₁ correspondence.

Due to the reduced Poincaré symmetry, one-dimensional supersymmetry has the special properties that (i) the number of the component fields in the supermultiplet is larger than the number \mathcal{N} of supersymmetry if \mathcal{N} is greater that eight and that (ii) the number *n* of physical bosonic component fields is not necessarily same as that of the fermions. These facts allow us to construct various supermultiplets $(n, \mathcal{N}, \mathcal{N} - n)$ only for $\mathcal{N} = 1, 2, 4$ and 8 supersymmetric quantum mechanics. Indeed we have argued that for such supersymmetric quantum mechanics there have been continuous attempts to construct superconformal mechanical models by appealing the superspace and superfield formalism.

We have shown that the IR quantum mechanics arising from the BLG-model and the ABJM-model wrapped on a torus are the $\mathcal{N} = 16$ and $\mathcal{N} = 12$ superconformal gauged quantum mechanical models respectively. Furthermore after the integration of the auxiliary gauge fields, we found that the OSp(16|2) quantum mechanics (6.3.1) and SU(1,1|6) quantum mechanics (7.3.1) emerge from the reduced theories. Both of them are $\mathcal{N} > 8$ superconformal quantum mechanical models which have not been available by the superspace and superfield formalism so far. It is interesting to investigate their spectrums, wavefunctions and correlation functions for those new superconformal mechanical models.

We have also surveyed the membranes wrapped around a genus $g \neq 1$ Riemann surface. In this case the surface is singled out as a calibrated holomorphic curve in a Calabi-Yau manifold to preserve supersymmetry. We have found that the IR quantum mechanical models have $\mathcal{N} = 8$, 4, 2 and 2 supersymmetries for K₃, CY_3 , CY_4 and CY_5 respectively. Especially when the Calabi-Yau manifolds are constructed via decomposable line bundles over the Riemann surface, the K₃ surface essentially allows for a unique topological twist while for the other Calabi-Yau manifolds there are infinitely many topological twists which are specified by the degrees of the line bundles.

We have especially analyzed the two membranes wrapping a holomorphic genus g > 1 curve embedded in a K₃ surface based on the topologically twisted BLG-model. We have found the new $\mathcal{N} = 8$ superconformal gauged quantum mechanics (10.3.21) that may describe the low-energy dynamics of the wrapped M2-branes in a K₃ surface. It is known that [150, 151, 152] there are the connections of the gauged quantum mechanics to the conformal mechanical models, the Calogero model and their generalizations. An interesting question is what type of interaction potential, if it exists, may characterize our superconformal "gauged" quantum mechanics (10.3.21). The structure of the resulting theory may indicate that generic SCQM takes the form of superconformal gauged quantum mechanics along with auxiliary gauge fields.

11.2 Future directions

There may be a number of future aspects of the present work. In the following we will briefly discuss the possible three applications.

11.2.1 AdS_2/CFT_1 correspondence

 AdS_{d+1}/CFT_d correspondence [17] is an important example of the holographic principle [323].

For d = 2 it has been shown [324] that the Hilbert space of the any quantum gravity on an asymptotically AdS₃ space-time is a representation of the twodimensional conformal group and that the central charge of the d = 2 CFT is given by

$$c = \frac{3l}{2G} \tag{11.2.1}$$

where *l* is the AdS₃ radius and *G* is Newton constant. The relationship between the BTZ black hole and the state in the two-dimensional CFT indicates that the entropy of the black may be defined as the logarism of the degeneracy of the corresponding states in the CFT. In this perspective the entropy of the *d* = 3 Baãdos-Teitelboim-Zanelli (BTZ) black hole is computed by counting the states of the *d* = 2 conformal field theory on the boudnary of AdS₃ [325]

$$S = 2\pi \sqrt{\frac{cn_R}{6}} + 2\pi \sqrt{\frac{cn_L}{6}}$$
(11.2.2)

where n_R and n_L are the eigenvalues of the Virasoro generators L_0 and L_0 respectively. For large L_0 one can use the Cardy formula to evaluate the degeneracy of the states and it has been shown [326, 327, 328, 329] that the result agrees with the one obtained by Wald's formula [330, 331, 332, 333].

The case of d = 1, i.e. AdS_2/CFT_1 correspondence [78, 334, 80, 79, 81, 83, 82, 335, 336, 337, 75, 84, 85, 338, 339, 340, 341, 76, 342, 343, 344] is less understood, however, it is extremely significant case of AdS_{d+1}/CFT_d correspondence in that all known extremal black holes contain the AdS_2 factor in their near horizon geometries [345, 346]. The two candidates for the CFT_1 have been proposed

(i) conformal quantum mechanics(ii) a chiral half of a *d* = 2 CFT.

For the former only the global $SL(2, \mathbb{R})$ acts nontrivially on the Hilbert space, while in the in latter case one copy of the Virasoro generators acts nontrivially on

the Hilbert space. In [341] the central charge for the CFT_1 which corresponds to the quantum gravity with a U(1) gauge field on AdS_2 has been given by

$$c = \frac{3kE^2l^4}{4} \tag{11.2.3}$$

where *l* is the AdS₂ radius, *E* is the electric field and *k* is the level of the U(1) current. The expression is similar to (11.2.1) for AdS₃/CFT₂ correspondence. It has been discussed [78, 341] that the latter idea of the non-trivial action of the Virasoro could be consistent and AdS_2/CFT_1 correspondence reduces to the CFT_2/CFT_2 duality on the strip. As discussed in [78, 341], this idea could be true when AdS₂ is generated as a S^1 compactification of AdS₃, however, there may be other types of the AdS_2 which do not arise as a S^1 compactification of AdS_3 and therefore the former possibility could still be a good candidate of the CFT₁. In the former perspective, it has been proposed [343] that the logarithm of the ground state degeneracy in a conformal quantum mechanics living on the boundary of AdS_2 yields the definition of the entropy of the extremal black hole in the quantum theory. Furthermore it has been pointed out in [86, 87] that the correlation functions of the conformal quantum mechanics [54] have the expected scaling behaviors from AdS_2/CFT_1 correpondence although one cannot assume the existence of the normalized and conformal invariant vacuum states in conformal quantum mechanics as in other higher dimensional conformal field theories. It is interesting to investigate whether our superconformal quantum mechanics resulting from the wrapped M2-branes around a compact Riemann surface in M-theory could provide some examples of the AdS_2/CFT_1 correspondence in the former perspective.

11.2.2 Indices and the reduced Gromov-Witten invariants

Another topic is the computation of the indices and their applications. For instance, the BPS partition function which gives rise to the counding of the BPS states may be related to the number of the supersymmetric two-cycles of genus *g* in our setup. Indeed, in the setup where the curved D₃-branes wrapping supersymmetric two-cycles embedded in K₃ surface, the formula for the numbers of rational curves with *g* double points on a K₃ surface, the so-called reduced Gromov-Witten invariants [347] has been conjectured by Yau and Zaslow [39] in the computation of the BPS partition function by appealing the string duality. Closely related to their setup, our $\mathcal{N} = 8$ superconformal gauged quantum mechanics (10.3.21) appears from the wrapped M₂-branes instead of the D₃-brane. It would be interesting to compute the indices and to extract enumerative information and structure from our model. In order to compute the indices we take a trace over the eigenstates. As discussed in section 2.2, it is difficult to calculate a trace over the eigenstate of the Hamiltonian H for the superconformal quantum mechanics because there is no normalizable ground state and its spectrum is continuous. As proposed in [73], the indices in superconformal quantum mechanics can be defined by taking a trace over the eigenstates of the compact operator $L_0 = \frac{1}{2}(H + K)$ which has a normalizable ground state and the discrete eigenvalues with equal spacing as

$$\mathcal{I}(\mathcal{O}) = \text{Tr}_{L_0}(-1)^{2J} \mathcal{O}e^{-\beta(L_0-J)}$$
(11.2.4)

where *J* is the R-symmetry generator and O is some operator in the theory. It is an open problem to evaluate indices and understand their physical and mathematical implication for our superconformal quantum mechanics.

11.2.3 1d-2d relation

Finally we want to comment on the "1d-2d relation", which is motivated by the fascinating stories arising from the compactification of M5-branes, for example, the AGT-relation [348], the DGG-relation [349] and the 2d-4d relation [350]. It has been argued that the world-volume theories of multiple M5-branes can be described by the six-dimensional superconformal field theories labeled by a simply-laced Lie algebra g, the so-called (2,0) theories. Via compactification, such theories leads to a family $T[M_{6-d}, g]$ of *d*-dimensional superconformal field theories which can be labeled by a choice of a specific manifold M_{6-d} and a Lie algebra g. From this perspective the AGT-relation, the DGG-relation and 2d-4d relation are regarded as the decomposition of the six-dimensional world-volume of M5-branes as 6 = 4 + 2, 3 + 3 and 2 + 4 respectively.

On the other hand, the world-volume theories of multiple M2-branes can be described by the three-dimensional superconformal field theories. Unlike the M5branes we know the explicit Lagrangian for such world-volume theories as the BLG-model and the ABJ(M) model. It would be attractive to find the new relationship between the superconformal field theories and the geometries or relevant dualities from M2-branes, i.e. "1d-2d relation" arising from the decomposition of the three-dimensional world-volume of M2-branes as 3 = 1 + 2. As an exchange of the order we may have two ways of the compactification

$$\begin{array}{c} \mbox{3d SCFT on } \mathbb{R} \times \Sigma_g \\ \swarrow & \searrow & (11.2.5) \\ \mbox{1d SCQM on } \mathbb{R} & \mbox{2d TQFT on } \Sigma_g, \end{array}$$

which suggests a new set of dualities in the sense that the partition functions or indices on both sides yield the same result. As we discussed in section 3.5, the WDVV equation [230, 231] and the twisted periods [232, 233] which are relevant to two-dimensional geometries and topological field theories appear from the constraint conditions for the constructions of $\mathcal{N} = 4$ superconformal mechanics. It would be interesting to investigate whether our M-theoretical construction of superconformal quantum mechanics could help to understand and generalize such relations as the "1d-2d relation".

Bibliography

- J. Scherk and J. H. Schwarz, "Dual Models for Nonhadrons," Nucl.Phys. B81 (1974) 118–144.
- [2] T. Yoneya, "Connection of Dual Models to Electrodynamics and Gravidynamics," *Prog.Theor.Phys.* **51** (1974) 1907–1920.
- [3] C. Hull and P. Townsend, "Unity of superstring dualities," Nucl. Phys. B438 (1995) 109–137, hep-th/9410167.
- [4] E. Witten, "String theory dynamics in various dimensions," Nucl. Phys. B443 (1995) 85–126, hep-th/9503124.
- [5] M. B. Green and J. H. Schwarz, "Supersymmetrical String Theories," *Phys.Lett.* B109 (1982) 444–448.
- [6] J. H. Schwarz and P. C. West, "Symmetries and Transformations of Chiral N=2 D=10 Supergravity," *Phys.Lett.* B126 (1983) 301.
- [7] P. S. Howe and P. C. West, "The Complete N=2, D=10 Supergravity," Nucl. Phys. B238 (1984) 181.
- [8] F. Giani and M. Pernici, "N=2 SUPERGRAVITY IN TEN-DIMENSIONS," *Phys.Rev.* D30 (1984) 325–333.
- [9] M. Huq and M. Namazie, "Kaluza-Klein Supergravity in Ten-dimensions," *Class.Quant.Grav.* **2** (1985) 293.
- [10] I. Campbell and P. C. West, "N=2 D=10 Nonchiral Supergravity and Its Spontaneous Compactification," Nucl. Phys. B243 (1984) 112.
- [11] E. Cremmer, B. Julia, and J. Scherk, "Supergravity Theory in Eleven-Dimensions," *Phys.Lett.* **B76** (1978) 409–412.
- [12] W. Nahm, "Supersymmetries and their Representations," Nucl. Phys. B135 (1978) 149.

- [13] P. Townsend, "The eleven-dimensional supermembrane revisited," *Phys.Lett.* B350 (1995) 184–187, hep-th/9501068.
- [14] J. H. Schwarz, "The power of M theory," Phys.Lett. B367 (1996) 97–103, hep-th/9510086.
- [15] M. Duff, P. S. Howe, T. Inami, and K. Stelle, "Superstrings in D=10 from Supermembranes in D=11," *Phys.Lett.* B191 (1987) 70.
- [16] J. Polchinski, "Dirichlet Branes and Ramond-Ramond charges," *Phys.Rev.Lett.* **75** (1995) 4724–4727, hep-th/9510017.
- [17] J. M. Maldacena, "The Large N limit of superconformal field theories and supergravity," *Adv.Theor.Math.Phys.* 2 (1998) 231–252, hep-th/9711200.
- [18] S. Gubser, I. R. Klebanov, and A. M. Polyakov, "Gauge theory correlators from noncritical string theory," *Phys.Lett.* B428 (1998) 105–114, hep-th/9802109.
- [19] E. Witten, "Anti-de Sitter space and holography," Adv.Theor.Math.Phys. 2 (1998) 253–291, hep-th/9802150.
- [20] E. Witten, "Bound states of strings and p-branes," Nucl. Phys. B460 (1996) 335–350, hep-th/9510135.
- [21] J. Bagger and N. Lambert, "Modeling Multiple M2's," *Phys.Rev.* D75 (2007) 045020, hep-th/0611108.
- [22] J. Bagger and N. Lambert, "Gauge symmetry and supersymmetry of multiple M2-branes," *Phys.Rev.* D77 (2008) 065008, 0711.0955.
- [23] J. Bagger and N. Lambert, "Comments on multiple M2-branes," JHEP 0802 (2008) 105, 0712.3738.
- [24] A. Gustavsson, "Algebraic structures on parallel M2-branes," *Nucl.Phys.* B811 (2009) 66–76, 0709.1260.
- [25] A. Gustavsson, "Selfdual strings and loop space Nahm equations," JHEP 0804 (2008) 083, 0802.3456.
- [26] O. Aharony, O. Bergman, D. L. Jafferis, and J. Maldacena, "N=6 superconformal Chern-Simons-matter theories, M2-branes and their gravity duals," *JHEP* 0810 (2008) 091, 0806.1218.

- [27] R. Harvey and H. B. Lawson, Jr., "Calibrated geometries," Acta Math. 148 (1982) 47–157.
- [28] M. Bershadsky, C. Vafa, and V. Sadov, "D-branes and topological field theories," Nucl. Phys. B463 (1996) 420–434, hep-th/9511222.
- [29] C. Lozano, "Duality in topological quantum field theories," hep-th/9907123.
- [30] J. Kato, N. Kawamoto, and A. Miyake, "N=4 twisted superspace from Dirac-Kahler twist and off-shell SUSY invariant actions in four dimensions," *Nucl.Phys.* B721 (2005) 229–286, hep-th/0502119.
- [31] V. Mikhaylov, "On the Solutions of Generalized Bogomolny Equations," *JHEP* **1205** (2012) 112, 1202.4848.
- [32] K. Ito, H. Nakajima, and S. Sasaki, "Twisted $\mathcal{N} = 4$ Super Yang-Mills Theory in Ω -background," *JHEP* **1310** (2013) 080, 1307.7565.
- [33] N. Marcus, "The Other topological twisting of N=4 Yang-Mills," *Nucl.Phys.* B452 (1995) 331–345, hep-th/9506002.
- [34] M. Blau and G. Thompson, "Aspects of N(T) ≥ two topological gauge theories and D-branes," Nucl.Phys. B492 (1997) 545–590, hep-th/9612143.
- [35] J. Labastida and C. Lozano, "Mathai-Quillen formulation of twisted N=4 supersymmetric gauge theories in four-dimensions," *Nucl.Phys.* B502 (1997) 741–790, hep-th/9702106.
- [36] A. Kapustin and E. Witten, "Electric-Magnetic Duality And The Geometric Langlands Program," Commun.Num.Theor.Phys. 1 (2007) 1–236, hep-th/0604151.
- [37] C. Vafa and E. Witten, "A Strong coupling test of S duality," *Nucl.Phys.* B431 (1994) 3–77, hep-th/9408074.
- [38] J. P. Yamron, "Topological Actions From Twisted Supersymmetric Theories," *Phys.Lett.* **B213** (1988) 325.
- [39] S.-T. Yau and E. Zaslow, "BPS states, string duality, and nodal curves on K3," Nucl.Phys. B471 (1996) 503-512, hep-th/9512121.

- [40] J. M. Maldacena and C. Nunez, "Supergravity description of field theories on curved manifolds and a no go theorem," *Int.J.Mod.Phys.* A16 (2001) 822–855, hep-th/0007018.
- [41] F. Benini and N. Bobev, "Exact two-dimensional superconformal R-symmetry and c-extremization," *Phys.Rev.Lett.* **110** (2013), no. 6 061601, 1211.4030.
- [42] F. Benini and N. Bobev, "Two-dimensional SCFTs from wrapped branes and c-extremization," *JHEP* **1306** (2013) 005, 1302.4451.
- [43] P. Karndumri and E. O Colgain, "Supergravity dual of *c*-extremization," *Phys.Rev.* D87 (2013), no. 10 101902, 1302.6532.
- [44] K. Nagasaki and S. Yamaguchi, "Two-dimensional superconfomal field theories from Riemann surfaces with boundary," 1412.8302.
- [45] K. Lee, S. Lee, and J.-H. Park, "Topological Twisting of Multiple M2-brane Theory," *JHEP* 0811 (2008) 014, 0809.2924.
- [46] J. P. Gauntlett, N. Kim, S. Pakis, and D. Waldram, "Membranes wrapped on holomorphic curves," *Phys.Rev.* D65 (2002) 026003, hep-th/0105250.
- [47] J. P. Gauntlett, "Branes, calibrations and supergravity," hep-th/0305074.
- [48] N. Kim, "Comments on AdS_2 solutions from M2-branes on complex curves and the backreacted Kähler geometry," 1311.7372.
- [49] B. de Wit, H. Nicolai, and N. Warner, "The Embedding of Gauged N = 8Supergravity Into d = 11 Supergravity," *Nucl.Phys.* **B255** (1985) 29.
- [50] B. de Wit and H. Nicolai, "The Consistency of the S⁷ Truncation in D=11 Supergravity," *Nucl.Phys.* B281 (1987) 211.
- [51] T. Okazaki, "Membrane Quantum Mechanics," Nucl. Phys. B890 (2015) 400-441, 1410.8180.
- [52] E. Witten, "Dynamical Breaking of Supersymmetry," Nucl. Phys. B188 (1981) 513.
- [53] F. Cooper and B. Freedman, "Aspects of Supersymmetric Quantum Mechanics," Annals Phys. 146 (1983) 262.

- [54] V. de Alfaro, S. Fubini, and G. Furlan, "Conformal Invariance in Quantum Mechanics," *Nuovo Cim.* A34 (1976) 569.
- [55] G. H. Shortley, "The inverse-cube central force field in quantum mechanics," *Physical Review* **38** (1931), no. 1 120.
- [56] K. Case, "Singular potentials," Phys.Rev. 80 (1950) 797–806.
- [57] L. D. Landau, E. M. Lifshitz, J. Sykes, J. Bell, and M. Rose, "Quantum mechanics, non-relativistic theory: Vol. 3 of course of theoretical physics," *Physics Today* 11 (2009), no. 12 56–59.
- [58] T. Tietz, "Discrete states for non-singular and singular potential problems," *Il Nuovo Cimento* (1955-1965) **11** (1959), no. 1 126–130.
- [59] A. Bastai, L. Bertocchi, S. Fubini, G. Furlan, and M. Tonin, "On the treatment of singular bethe-salpeter equations," *Il Nuovo Cimento Series 10* 30 (1963), no. 6 1512–1531.
- [60] K. Meetz, "Singular potentials in nonrelativistic quantum mechanics," Il Nuovo Cimento Series 10 34 (1964), no. 3 690–708.
- [61] M. Englefield, "Scattering by potentials with an inverse-square attraction," *Proceedings of the Physical Society* 83 (1964), no. 4 519.
- [62] E. Guggenheim, "The inverse square potential field," *Proceedings of the Physical Society* **89** (1966), no. 3 491.
- [63] E. Ferreira, J. Sesma, and P. Torres, "On scattering and bound states for a singular potential," *Prog.Theor.Phys.* 43 (1970) 1–9.
- [64] R. Jackiw, "Introducing scale symmetry," Phys. Today 25N1 (1972) 23–27.
- [65] P. Claus, M. Derix, R. Kallosh, J. Kumar, P. K. Townsend, et. al., "Black holes and superconformal mechanics," *Phys.Rev.Lett.* 81 (1998) 4553–4556, hep-th/9804177.
- [66] G. Gibbons and P. Townsend, "Black holes and Calogero models," *Phys.Lett.* B454 (1999) 187–192, hep-th/9812034.
- [67] J. de Azcarraga, J. Izquierdo, J. Perez Bueno, and P. Townsend,
 "Superconformal mechanics and nonlinear realizations," *Phys.Rev.* D59 (1999) 084015, hep-th/9810230.

- [68] J. Michelson and A. Strominger, "Superconformal multiblack hole quantum mechanics," JHEP 9909 (1999) 005, hep-th/9908044.
- [69] J. Michelson and A. Strominger, "The Geometry of (super)conformal quantum mechanics," Commun.Math.Phys. 213 (2000) 1–17, hep-th/9907191.
- [70] A. Maloney, M. Spradlin, and A. Strominger, "Superconformal multiblack hole moduli spaces in four-dimensions," JHEP 0204 (2002) 003, hep-th/9911001.
- [71] R. Kallosh, "Black holes and quantum mechanics," hep-th/9902007.
- [72] G. Papadopoulos, "Conformal and superconformal mechanics," *Class.Quant.Grav.* 17 (2000) 3715–3742, hep-th/0002007.
- [73] R. Britto-Pacumio, A. Strominger, and A. Volovich, "Two black hole bound states," *JHEP* **0103** (2001) 050, hep-th/0004017.
- [74] E. Ivanov, S. Krivonos, and J. Niederle, "Conformal and superconformal mechanics revisited," Nucl.Phys. B677 (2004) 485–500, hep-th/0210196.
- [75] S. Bellucci, A. Galajinsky, E. Ivanov, and S. Krivonos, "AdS(2) / CFT(1), canonical transformations and superconformal mechanics," *Phys.Lett.* B555 (2003) 99–106, hep-th/0212204.
- [76] A. Galajinsky, "Particle dynamics on AdS(2) x S**2 background with two-form flux," *Phys.Rev.* D78 (2008) 044014, 0806.1629.
- [77] S. Bellucci, S. Krivonos, A. Shcherbakov, and A. Sutulin, "N=4 Superconformal Mechanics and Black Holes," *Phys.Rev.* D78 (2008) 125001, 0807.1785.
- [78] A. Strominger, "AdS(2) quantum gravity and string theory," JHEP 9901 (1999) 007, hep-th/9809027.
- [79] P. Townsend, "The M(atrix) model / AdS(2) correspondence," hep-th/9903043.
- [80] T. Nakatsu and N. Yokoi, "Comments on Hamiltonian formalism of AdS / CFT correspondence," Mod.Phys.Lett. A14 (1999) 147–160, hep-th/9812047.
- [81] M. Spradlin and A. Strominger, "Vacuum states for AdS(2) black holes," *JHEP* 9911 (1999) 021, hep-th/9904143.

- [82] J. D. Blum, "Supersymmetric quantum mechanical description of four-dimensional black holes," JHEP 0001 (2000) 006, hep-th/9907101.
- [83] M. Cadoni and S. Mignemi, "Asymptotic symmetries of AdS(2) and conformal group in d = 1," Nucl.Phys. B557 (1999) 165–180, hep-th/9902040.
- [84] A. Strominger, "A Matrix model for AdS(2)," JHEP 0403 (2004) 066, hep-th/0312194.
- [85] C. Leiva and M. S. Plyushchay, "Conformal symmetry of relativistic and nonrelativistic systems and Ads / CFT correspondence," *Annals Phys.* 307 (2003) 372–391, hep-th/0301244.
- [86] C. Chamon, R. Jackiw, S.-Y. Pi, and L. Santos, "Conformal quantum mechanics as the CFT₁ dual to AdS₂," *Phys.Lett.* B701 (2011) 503–507, 1106.0726.
- [87] R. Jackiw and S.-Y. Pi, "Conformal Blocks for the 4-Point Function in Conformal Quantum Mechanics," *Phys.Rev.* D86 (2012) 045017, 1205.0443.
- [88] H. G. Dosch, S. J. Brodsky, and G. F. de Teramond, "Modified Anti-de-Sitter Metric, Light-Front Quantized QCD, and Conformal Quantum Mechanics," 1401.7710.
- [89] G. F. de Teramond, H. G. Dosch, and S. J. Brodsky, "Baryon Spectrum from Superconformal Quantum Mechanics and its Light-Front Holographic Embedding," 1411.5243.
- [90] H. Azuma and S. Iso, "Explicit relation of quantum hall effect and Calogero-Sutherland model," *Phys.Lett.* B331 (1994) 107–113, hep-th/9312001.
- [91] C. Kane, S. Kivelson, D. Lee, and S. Zhang, "General validity of jastrow-laughlin wave functions," *Physical Review B* 43 (1991), no. 4 3255.
- [92] N. Kawakami and S.-K. Yang, "Finite size scaling in one-dimensional quantum liquid with long range interaction," *Phys.Rev.Lett.* 67 (1991) 2493–2496.
- [93] A. Dabholkar, "Fermions and nonperturbative supersymmetry breaking in the one-dimensional superstring," *Nucl.Phys.* **B368** (1992) 293–310.

- [94] J. McGreevy, S. Murthy, and H. L. Verlinde, "Two-dimensional superstrings and the supersymmetric matrix model," *JHEP* 0404 (2004) 015, hep-th/0308105.
- [95] H. L. Verlinde, "Superstrings on AdS(2) and superconformal matrix quantum mechanics," hep-th/0403024.
- [96] J. Gates, S. James, A. Jellal, E. H. Saidi, and M. Schreiber, "Supersymmetric embedding of the quantum Hall matrix model," *JHEP* 0411 (2004) 075, hep-th/0410070.
- [97] A. Agarwal and A. P. Polychronakos, "BPS operators in N=4 SYM: Calogero models and 2-D fermions," *JHEP* 0608 (2006) 034, hep-th/0602049.
- [98] W.-Y. Wen, "AdS/NRCFT for the (super) Calogero model," 0807.0633.
- [99] F. Haldane, "Exact Jastrow-Gutzwiller resonating valence bond ground state of the spin 1/2 antiferromagnetic Heisenberg chain with 1/r**2 exchange," *Phys.Rev.Lett.* **60** (1988) 635.
- [100] B. Sriram Shastry, "Exact solution of an S = 1/2 Heisenberg antiferromagnetic chain with long ranged interactions," *Phys.Rev.Lett.* 60 (1988) 639.
- [101] J. A. Minahan and A. P. Polychronakos, "Integrable systems for particles with internal degrees of freedom," *Phys.Lett.* B302 (1993) 265–270, hep-th/9206046.
- [102] C.-r. Ahn and W. M. Koo, "gl(n m) color Calogero-Sutherland models and super Yangian algebra," *Phys.Lett.* B365 (1996) 105–112, hep-th/9505060.
- [103] K. Hikami and B. Basu-Mallick, "Supersymmetric polychronakos spin chain: motif, distribution function, and character," *Nuclear Physics B* 566 (2000), no. 3 511–528.
- [104] B. Basu-Mallick, N. Bondyopadhaya, and D. Sen, "Low energy properties of the SU(m|n) supersymmetric Haldane-Shastry spin chain," *Nucl.Phys.* B795 (2008) 596–622, 0710.0452.
- [105] J. Barba, F. Finkel, A. González-López, and M. Rodríguez,
 "Polychronakos-frahm spin chain of b c n type and the berry-tabor conjecture," *Physical Review B* 77 (2008), no. 21 214422.

- [106] B. Basu-Mallick, F. Finkel, and A. González-López, "Exactly solvable dn-type quantum spin models with long-range interaction," *Nuclear Physics B* 812 (2009), no. 3 402–423.
- [107] M. Bawin and S. Coon, "The Singular inverse square potential, limit cycles and selfadjoint extensions," *Phys.Rev.* A67 (2003) 042712, quant-ph/0302199.
- [108] H. E. Camblong and C. R. Ordonez, "Renormalization in conformal quantum mechanics," *Phys.Lett.* A345 (2005) 22–30, hep-th/0305035.
- [109] E. Braaten and D. Phillips, "The Renormalization group limit cycle for the 1/r**2 potential," *Phys.Rev.* A70 (2004) 052111, hep-th/0403168.
- [110] T. Barford and M. C. Birse, "Effective theories of scattering with an attractive inverse-square potential and the three-body problem," *J.Phys.* A38 (2005) 697–720, nucl-th/0406008.
- [111] C. Beenakker and B. Rajaei, "Exact solution for the distribution of transmission eigenvalues in a disordered wire and comparison with random-matrix theory," *Physical Review B* 49 (1994), no. 11 7499.
- [112] M. Caselle, "Distribution of transmission eigenvalues in disordered wires," *Physical review letters* 74 (1995), no. 14 2776.
- [113] G. Date, M. Murthy, and R. Vathsan, "Classical many-particle clusters in two dimensions," *Journal of Physics: Condensed Matter* 10 (1998), no. 26 5875.
- [114] B. Simons, P. Lee, and B. Altshuler, "Matrix models, one-dimensional fermions, and quantum chaos," *Physical review letters* 72 (1994), no. 1 64.
- [115] M. Murthy and R. Shankar, "Thermodynamics of a one-dimensional ideal gas with fractional exclusion statistics," *Physical review letters* 73 (1994), no. 25 3331.
- [116] Z. N. C. Ha, Quantum many-body systems in one dimension. Adv. Stat. Mech. World Scientific, Singapore, 1996.
- [117] M. Feigel'Man and M. Skvortsov, "Supersymmetric model of a 2d long-range bose liquid," *Nuclear Physics B* **506** (1997), no. 3 665–684.
- [118] A. Khare and K. Ray, "A quantum many-body problem in two dimensions: ground state," *Physics Letters A* **230** (1997), no. 3 139–143.

- [119] G. Oas, "Universal cubic eigenvalue repulsion for random normal matrices," *Physical Review E* 55 (1997), no. 1 205.
- [120] V. Bardek, J. Feinberg, and S. Meljanac, "Fluctuations around Periodic BPS-Density Waves in the Calogero Model," *JHEP* 1008 (2010) 018, 1004.4283.
- [121] E. D'Hoker and D. Phong, "Lectures on supersymmetric Yang-Mills theory and integrable systems," hep-th/9912271.
- [122] A. Gorsky and A. Mironov, "Integrable many body systems and gauge theories," hep-th/0011197.
- [123] P. Desrosiers, L. Lapointe, and P. Mathieu, "Supersymmetric Calogero-Moser-Sutherland models and Jack superpolynomials," *Nucl.Phys.* B606 (2001) 547–582, hep-th/0103178.
- [124] P. Desrosiers, L. Lapointe, and P. Mathieu, "Supersymmetric Calogero-Moser-Sutherland models: Superintegrability structure and eigenfunctions," hep-th/0210190.
- [125] P. Desrosiers, L. Lapointe, and P. Mathieu, "Jack polynomials in superspace," *Commun.Math.Phys.* 242 (2003) 331–360, hep-th/0209074.
- [126] P. Desrosiers, L. Lapointe, and P. Mathieu, "Generalized Hermite polynomials in superspace as eigenfunctions of the supersymmetric rational CMS model," *Nucl. Phys.* B674 (2003) 615–633, hep-th/0305038.
- [127] P. Desrosiers, L. Lapointe, and P. Mathieu, "Classical symmetric functions in superspace," J. Algebraic Combin. 24 (2006), no. 2 209–238.
- [128] A. Okounkov and R. Pandharipande, "Quantum cohomology of the Hilbert scheme of points in the plane," *Invent. Math.* **179** (2010), no. 3 523–557.
- [129] A. Okounkov and R. Pandharipande, "The quantum differential equation of the Hilbert scheme of points in the plane," *Transform. Groups* 15 (2010), no. 4 965–982.
- [130] K. Hikami and M. Wadati, "Integrable systems with long range interactions, w-infinity algebra and energy spectrum," *Phys.Rev.Lett.* **73** (1994) 1191–1194.
- [131] H. Awata, Y. Matsuo, S. Odake, and J. Shiraishi, "Collective field theory, Calogero-Sutherland model and generalized matrix models," *Phys.Lett.* B347 (1995) 49–55, hep-th/9411053.

- [132] A. Bordner, E. Corrigan, and R. Sasaki, "Generalized Calogero-Moser models and universal LAX pair operators," *Prog.Theor.Phys.* 102 (1999) 499–529, hep-th/9905011.
- [133] S. Khastgir, A. Pocklington, and R. Sasaki, "Quantum Calogero-Moser models: Integrability for all root systems," J.Phys. A33 (2000) 9033–9064, hep-th/0005277.
- [134] F. Calogero, "Solution of a three-body problem in one-dimension," *J.Math.Phys.* 10 (1969) 2191–2196.
- [135] F. Calogero, "Ground state of one-dimensional N body system," J.Math.Phys. 10 (1969) 2197–2200.
- [136] R. Britto-Pacumio, J. Michelson, A. Strominger, and A. Volovich, "Lectures on superconformal quantum mechanics and multiblack hole moduli spaces," hep-th/9911066.
- [137] S. Fedoruk, E. Ivanov, and O. Lechtenfeld, "Superconformal Mechanics," *J.Phys.* A45 (2012) 173001, 1112.1947.
- [138] P. K. Ghosh, "Supersymmetric Many-particle Quantum Systems with Inverse-square Interactions," *J.Phys.* **A45** (2012) 183001, 1111.6255.
- [139] M. Olshanetsky and A. Perelomov, "Classical integrable finite dimensional systems related to Lie algebras," *Phys.Rept.* **71** (1981) 313.
- [140] M. Olshanetsky and A. Perelomov, "Quantum Integrable Systems Related to Lie Algebras," *Phys.Rept.* 94 (1983) 313–404.
- [141] A. M. Perelomov and A. Reyman, *Integrable systems of classical mechanics and Lie algebras*. Springer, 1990.
- [142] V. Pasquier, "A Lecture on the Calogero-Sutherland models," hep-th/9405104.
- [143] E. D'Hoker and D. Phong, "Calogero-Moser Lax pairs with spectral parameter for general Lie algebras," Nucl.Phys. B530 (1998) 537–610, hep-th/9804124.
- [144] A. P. Polychronakos, "Generalized statistics in one-dimension," hep-th/9902157.

- [145] A. P. Polychronakos, "Physics and Mathematics of Calogero particles," J.Phys. A39 (2006) 12793–12846, hep-th/0607033.
- [146] M. Milekovic, S. Meljanac, and A. Samsarov, "Calogero model(s) and deformed oscillators," SIGMA 2 (2006) 035, hep-th/0603140.
- [147] Y. Kuramoto and Y. Kato, *Dynamics of one-dimensional quantum systems: inverse-square interaction models*. Cambridge University Press, 2009.
- [148] G. Barucchi and T. Regge, "Conformal Properties of a Class of Exactly Solvable n Body Problems in Space Dimension One," J.Math.Phys. 18 (1977) 1149.
- [149] S. Wojciechowski, "Representations of the Algebra SL(2,r) and Integration of Hamiltonian Systems," *Phys.Lett.* A64 (1977) 273–275.
- [150] D. Kazhdan, B. Kostant, and S. Sternberg, "Hamiltonian group actions and dynamical systems of Calogero type," *Comm. Pure Appl. Math.* **31** (1978), no. 4 481–507.
- [151] A. P. Polychronakos, "Integrable systems from gauged matrix models," *Phys.Lett.* B266 (1991) 29–34.
- [152] S. Fedoruk, E. Ivanov, and O. Lechtenfeld, "Supersymmetric Calogero models by gauging," *Phys.Rev.* D79 (2009) 105015, 0812.4276.
- [153] V. Akulov and A. Pashnev, "QUANTUM SUPERCONFORMAL MODEL IN (1,2) SPACE," *Theor.Math.Phys.* 56 (1983) 862–866.
- [154] S. Fubini and E. Rabinovici, "SUPERCONFORMAL QUANTUM MECHANICS," Nucl. Phys. B245 (1984) 17.
- [155] S. R. Coleman, J. Wess, and B. Zumino, "Structure of phenomenological Lagrangians. 1.," *Phys.Rev.* 177 (1969) 2239–2247.
- [156] J. Callan, Curtis G., S. R. Coleman, J. Wess, and B. Zumino, "Structure of phenomenological Lagrangians. 2.," *Phys.Rev.* 177 (1969) 2247–2250.
- [157] D. V. Volkov, "Phenomenological Lagrangians," *Fiz.Elem.Chast.Atom.Yadra* 4 (1973) 3–41.
- [158] N. Mukunda, "Unitary representations of the group O(2, 1) in an O(1, 1) basis," *J. Mathematical Phys.* **8** (1967) 2210–2220.

- [159] J. G. Kuriyan, N. Mukunda, and E. C. G. Sudarshan, "Master analytic representation: Reduction of 0(2, 1) in an 0(1, 1) basis," J. Mathematical Phys. 9 (1968) 2100–2108.
- [160] A. O. Barut and E. C. Phillips, "Matrix elements of representations of non-compact groups in a continuous basis," *Comm. Math. Phys.* 8 (1968) 52–65.
- [161] G. Lindblad and B. Nagel, "Continuous bases for unitary irreducible representations of SU(1, 1)," Ann. Inst. H. Poincaré Sect. A (N.S.) 13 (1970) 27–56.
- [162] R. Nakayama, "The World-Line Quantum Mechanics Model at Finite Temperature which is Dual to the Static Patch Observer in de Sitter Space," *Prog.Theor.Phys.* 127 (2012) 393–408, 1112.1267.
- [163] B. Freivogel, J. McGreevy, and S. J. Suh, "Exactly Stable Collective Oscillations in Conformal Field Theory," *Phys.Rev.* D85 (2012) 105002, 1109.6013.
- [164] D. Anninos, S. A. Hartnoll, and D. M. Hofman, "Static Patch Solipsism: Conformal Symmetry of the de Sitter Worldline," *Class.Quant.Grav.* 29 (2012) 075002, 1109.4942.
- [165] F. Dolan and H. Osborn, "Conformal Partial Waves: Further Mathematical Results," 1108.6194.
- [166] V. I. Arnold, Mathematical methods of classical mechanics. Springer-Verlag, New York-Heidelberg, 1978. Translated from the Russian by K. Vogtmann and A. Weinstein, Graduate Texts in Mathematics, 60.
- [167] J. Marsden and A. Weinstein, "Reduction of symplectic manifolds with symmetry," *Rep. Mathematical Phys.* **5** (1974), no. 1 121–130.
- [168] R. Abraham, J. E. Marsden, T. S. RaÈŻiu, and R. Cushman, Foundations of mechanics. Benjamin/Cummings Publishing Company Reading, Massachusetts, 1978.
- [169] J. Moser, "Various aspects of integrable hamiltonian systems," in *Dynamical* systems, pp. 137–195. Springer, 2011.

- [170] B. A. Dubrovin, S. P. Novikov, and A. T. Fomenko, *Sovremennaya geometriya*. "Nauka", Moscow, 1984. Metody teorii gomologii. [Methods of homology theory].
- [171] E. J. Routh, *The advanced part of a treatise on the dynamics of a system of rigid bodies. Being part II of a treatise on the whole subject.* Dover Publications, Inc., New York, 1955. 6th ed.
- [172] B. Zumino, "Nonlinear Realization of Supersymmetry in de Sitter Space," Nucl.Phys. B127 (1977) 189.
- [173] E. Ivanov and V. Ogievetsky, "The Inverse Higgs Phenomenon in Nonlinear Realizations," *Teor.Mat.Fiz.* 25 (1975) 164–177.
- [174] E. Ivanov, S. Krivonos, and V. Leviant, "Geometry of Conformal Mechanics," J.Phys. A22 (1989) 345.
- [175] C. B. Collins and J. M. Lang, "Singularities in self-similar spacetimes," *Classical Quantum Gravity* 3 (1986), no. 6 1143–1150.
- [176] G. S. Hall, "Singularities and homothety groups in spacetime," *Classical Quantum Gravity* **5** (1988), no. 5 L77–L80.
- [177] G. S. Hall and J. D. Steele, "Homothety groups in space-time," Gen. Relativity Gravitation 22 (1990), no. 4 457–468.
- [178] V. G. Kac, "Lie superalgebras," Advances in Math. 26 (1977), no. 1 8-96.
- [179] F. A. Berezin and G. I. Kac, "Lie groups with commuting and anticommuting parameters," Mat. Sb. (N.S.) 82 (124) (1970) 343–359.
- [180] M. Scheunert, W. Nahm, and V. Rittenberg, "Classification of All Simple Graded Lie Algebras Whose Lie Algebra Is Reductive. 1.," J.Math.Phys. 17 (1976) 1626.
- [181] M. Scheunert, W. Nahm, and V. Rittenberg, "Classification of All Simple Graded Lie Algebras Whose Lie Algebra Is Reductive. 2. Construction of the Exceptional Algebras," J.Math.Phys. 17 (1976) 1640.
- [182] V. Kac, "A Sketch of Lie Superalgebra Theory," Commun.Math.Phys. 53 (1977) 31–64.
- [183] L. Frappat, P. Sorba, and A. Sciarrino, "Dictionary on Lie superalgebras," hep-th/9607161.
- [184] P. Claus, R. Kallosh, and A. Van Proeyen, "Conformal symmetry on world volumes of branes," hep-th/9812066.
- [185] S. J. Gates and L. Rana, "A Theory of spinning particles for large N extended supersymmetry," *Phys.Lett.* B352 (1995) 50–58, hep-th/9504025.
- [186] J. Gates, S. James and L. Rana, "A Theory of spinning particles for large n extended supersymmetry. 2.," *Phys.Lett.* B369 (1996) 262–268, hep-th/9510151.
- [187] J. Gates, S. James, I. Linch, William Divine, and J. Phillips, "When superspace is not enough," hep-th/0211034.
- [188] B. de Wit, A. Tollsten, and H. Nicolai, "Locally supersymmetric D = 3 nonlinear sigma models," *Nucl.Phys.* B392 (1993) 3–38, hep-th/9208074.
- [189] A. Pashnev and F. Toppan, "On the classification of N extended supersymmetric quantum mechanical systems," J.Math.Phys. 42 (2001) 5257–5271, hep-th/0010135.
- [190] Z. Kuznetsova, M. Rojas, and F. Toppan, "Classification of irreps and invariants of the N-extended supersymmetric quantum mechanics," *JHEP* 0603 (2006) 098, hep-th/0511274.
- [191] M. Faux and J. Gates, S.J., "Adinkras: A Graphical technology for supersymmetric representation theory," *Phys.Rev.* D71 (2005) 065002, hep-th/0408004.
- [192] A. Hurwitz, "Über die Komposition der quadratischen Formen," *Math. Ann.* **88** (1922), no. 1-2 1–25.
- [193] J. Radon, "Lineare scharen orthogonaler matrizen," *Abh. Math. Sem. Univ. Hamburg* 1 (1922), no. 1 1–14.
- [194] J. F. Adams, "Vector fields on spheres," Ann. of Math. (2) 75 (1962) 603-632.
- [195] J. F. Adams, P. D. Lax, and R. S. Phillips, "Correction to "On matrices whose real linear combinations are nonsingular"," *Proc. Amer. Math. Soc.* 17 (1966) 945–947.
- [196] A. R. Rajwade, *Squares*, vol. 171 of *London Mathematical Society Lecture Note Series*. Cambridge University Press, Cambridge, 1993.

- [197] D. B. Shapiro, *Compositions of quadratic forms*, vol. 33 of *de Gruyter Expositions in Mathematics*. Walter de Gruyter & Co., Berlin, 2000.
- [198] A. Lenzhen, S. Morier-Genoud, and V. Ovsienko, "New solutions to the Hurwitz problem on square identities," J. Pure Appl. Algebra 215 (2011), no. 12 2903–2911.
- [199] R. Coles and G. Papadopoulos, "The Geometry of the one-dimensional supersymmetric nonlinear sigma models," *Class.Quant.Grav.* 7 (1990) 427–438.
- [200] G. Gibbons, G. Papadopoulos, and K. Stelle, "HKT and OKT geometries on soliton black hole moduli spaces," *Nucl.Phys.* B508 (1997) 623–658, hep-th/9706207.
- [201] P. S. Howe and G. Papadopoulos, "Further Remarks on the Geometry of Two-dimensional Nonlinear σ Models," *Class.Quant.Grav.* **5** (1988) 1647–1661.
- [202] J. Iliopoulos and B. Zumino, "Broken Supergauge Symmetry and Renormalization," *Nucl.Phys.* **B76** (1974) 310.
- [203] F. De Jonghe, K. Peeters, and K. Sfetsos, "Killing-Yano supersymmetry in string theory," *Class.Quant.Grav.* 14 (1997) 35–46, hep-th/9607203.
- [204] G. Gibbons, R. Rietdijk, and J. van Holten, "SUSY in the sky," *Nucl.Phys.* B404 (1993) 42–64, hep-th/9303112.
- [205] A. Galajinsky and O. Lechtenfeld, "Harmonic N=2 Mechanics," *Phys.Rev.* D80 (2009) 065012, 0907.2242.
- [206] S. Bellucci, A. Galajinsky, and S. Krivonos, "New many-body superconformal models as reductions of simple composite systems," *Phys.Rev.* D68 (2003) 064010, hep-th/0304087.
- [207] D. Z. Freedman and P. F. Mende, "An Exactly Solvable N Particle System in Supersymmetric Quantum Mechanics," Nucl. Phys. B344 (1990) 317–343.
- [208] A. Galajinsky, O. Lechtenfeld, and K. Polovnikov, "Calogero models and nonlocal conformal transformations," *Phys.Lett.* B643 (2006) 221–227, hep-th/0607215.

- [209] S. Fedoruk, "Superconformal Calogero Models as a Gauged Matrix Mechanics," *Acta Polytech.* **50** (2010) 23, 1002.2920.
- [210] E. Ivanov, S. Krivonos, and V. Leviant, "GEOMETRIC SUPERFIELD APPROACH TO SUPERCONFORMAL MECHANICS," J.Phys. A22 (1989) 4201.
- [211] F. Delduc and E. Ivanov, "Gauging N=4 Supersymmetric Mechanics," Nucl.Phys. B753 (2006) 211–241, hep-th/0605211.
- [212] E. Ivanov, S. Krivonos, and O. Lechtenfeld, "N=4, d = 1 supermultiplets from nonlinear realizations of D(2,1: alpha)," *Class.Quant.Grav.* 21 (2004) 1031–1050, hep-th/0310299.
- [213] E. Ivanov, S. Krivonos, and O. Lechtenfeld, "New variant of N=4 superconformal mechanics," *JHEP* 0303 (2003) 014, hep-th/0212303.
- [214] E. Ivanov and O. Lechtenfeld, "N=4 supersymmetric mechanics in harmonic superspace," JHEP 0309 (2003) 073, hep-th/0307111.
- [215] S. Hellerman and J. Polchinski, "Supersymmetric quantum mechanics from light cone quantization," hep-th/9908202.
- [216] C. Hull, "The Geometry of supersymmetric quantum mechanics," hep-th/9910028.
- [217] S. Bellucci, S. Krivonos, A. Marrani, and E. Orazi, "Root' action for N=4 supersymmetric mechanics theories," *Phys.Rev.* D73 (2006) 025011, hep-th/0511249.
- [218] F. Delduc and E. Ivanov, "Gauging N=4 supersymmetric mechanics II: (1,4,3) models from the (4,4,0) ones," *Nucl.Phys.* B770 (2007) 179–205, hep-th/0611247.
- [219] F. Delduc and E. Ivanov, "The Common origin of linear and nonlinear chiral multiplets in N=4 mechanics," *Nucl.Phys.* B787 (2007) 176–197, 0706.0706.
- [220] M. de Crombrugghe and V. Rittenberg, "Supersymmetric Quantum Mechanics," *Annals Phys.* **151** (1983) 99.
- [221] E. Ivanov and A. V. Smilga, "Supersymmetric gauge quantum mechanics: Superfield description," *Phys.Lett.* **B257** (1991) 79–82.

- [222] V. Berezovoj and A. Pashnev, "Three-dimensional N=4 extended supersymmetrical quantum mechanics," *Class.Quant.Grav.* 8 (1991) 2141–2147.
- [223] J. Wess, "Supersymmetry and Internal Symmetry," Acta Phys. Austriaca 41 (1975) 409–414.
- [224] N. Wyllard, "(Super)conformal many body quantum mechanics with extended supersymmetry," J.Math.Phys. 41 (2000) 2826–2838, hep-th/9910160.
- [225] F. Delduc and E. Ivanov, "N=4 superKdV equation," Phys.Lett. B309 (1993) 312-319, hep-th/9301024.
- [226] A. Galperin, E. Ivanov, V. Ogievetsky, and E. Sokatchev, "Harmonic superspace,".
- [227] A. Galajinsky, "N=4 superconformal mechanics from su(2) perspective," 1412.4467.
- [228] S. Krivonos, O. Lechtenfeld, and K. Polovnikov, "N=4 superconformal n-particle mechanics via superspace," *Nucl.Phys.* B817 (2009) 265–283, 0812.5062.
- [229] S. Bellucci, A. Galajinsky, and E. Latini, "New insight into WDVV equation," *Phys.Rev.* D71 (2005) 044023, hep-th/0411232.
- [230] E. Witten, "On the Structure of the Topological Phase of Two-dimensional Gravity," *Nucl.Phys.* **B340** (1990) 281–332.
- [231] R. Dijkgraaf, H. L. Verlinde, and E. P. Verlinde, "Topological strings in d < 1," *Nucl.Phys.* **B352** (1991) 59–86.
- [232] B. Dubrovin, "On almost duality for Frobenius manifolds," in *Geometry*, *topology, and mathematical physics*, vol. 212 of *Amer. Math. Soc. Transl. Ser.* 2, pp. 75–132. Amer. Math. Soc., Providence, RI, 2004.
- [233] M. Feigin and A. Silantyev, "Singular polynomials from orbit spaces," *Compos. Math.* 148 (2012), no. 6 1867–1879.
- [234] R. Martini and P. K. H. Gragert, "Solutions of WDVV equations in Seiberg-Witten theory from root systems," J. Nonlinear Math. Phys. 6 (1999), no. 1 1–4.

- [235] A. Veselov, "Deformations of the root systems and new solutions to generalized WDVV equations," *Phys.Lett.* A261 (1999) 297–302, hep-th/9902142.
- [236] O. A. Chalykh and A. P. Veselov, "Locus configurations and ∨-systems," *Phys. Lett. A* **285** (2001), no. 5-6 339–349.
- [237] M. V. Feigin and A. P. Veselov, "Logarithmic Frobenius structures and Coxeter discriminants," *Adv. Math.* 212 (2007), no. 1 143–162.
- [238] M. Feigin and A. P. Veselov, "On the geometry of ∨-systems," in Geometry, topology, and mathematical physics, vol. 224 of Amer. Math. Soc. Transl. Ser. 2, pp. 111–123. Amer. Math. Soc., Providence, RI, 2008.
- [239] O. Lechtenfeld and K. Polovnikov, "A New class of solutions to the WDVV equation," *Phys.Lett.* **A374** (2010) 504–506, 0907.2244.
- [240] O. Lechtenfeld, K. Schwerdtfeger, and J. Thurigen, "N=4 Multi-Particle Mechanics, WDVV Equation and Roots," SIGMA 7 (2011) 023, 1011.2207.
- [241] A. N. Sergeev and A. P. Veselov, "Deformed quantum Calogero-Moser problems and Lie superalgebras," *Comm. Math. Phys.* 245 (2004), no. 2 249–278.
- [242] L. K. Hoevenaars and R. Martini, "On the WDVV equations in five-dimensional gauge theories," *Phys. Lett. B* 557 (2003), no. 1-2 94–104.
- [243] M. V. Feigin, "Trigonometric solutions of WDVV equations and generalized Calogero-Moser-Sutherland systems," SIGMA Symmetry Integrability Geom. Methods Appl. 5 (2009) Paper 088, 10.
- [244] R. Martini and L. K. Hoevenaars, "Trigonometric solutions of the WDVV equations from root systems," *Lett. Math. Phys.* **65** (2003), no. 1 15–18.
- [245] I. A. B. Strachan, "Weyl groups and elliptic solutions of the WDVV equations," *Adv. Math.* **224** (2010), no. 5 1801–1838.
- [246] A. Veselov, "On geometry of a special class of solutions to generalized WDVV equations," hep-th/0105020.
- [247] S. Krivonos and O. Lechtenfeld, "Many-particle mechanics with D(2,1:alpha) superconformal symmetry," *JHEP* **1102** (2011) 042, 1012.4639.

- [248] A. Frölicher and A. Nijenhuis, "Theory of vector-valued differential forms.
 I. Derivations of the graded ring of differential forms," *Nederl. Akad. Wetensch. Proc. Ser. A.* **59** = *Indag. Math.* **18** (1956) 338–359.
- [249] A. Nijenhuis, "Jacobi-type identities for bilinear differential concomitants of certain tensor fields. I, II," *Nederl. Akad. Wetensch. Proc. Ser. A.* 58 = Indag. *Math.* 17 (1955) 390–397, 398–403.
- [250] P. S. Howe and G. Papadopoulos, "Twistor spaces for HKT manifolds," *Phys.Lett.* B379 (1996) 80–86, hep-th/9602108.
- [251] Z. Kuznetsova and F. Toppan, "D-module Representations of N=2,4,8 Superconformal Algebras and Their Superconformal Mechanics," *J.Math.Phys.* 53 (2012) 043513, 1112.0995.
- [252] S. Bellucci, E. Ivanov, S. Krivonos, and O. Lechtenfeld, "N = 8 superconformal mechanics," *Nucl.Phys.* B684 (2004) 321–350, hep-th/0312322.
- [253] S. Bellucci, E. Ivanov, S. Krivonos, and O. Lechtenfeld, "ABC of N = 8, d = 1 supermultiplets," *Nucl.Phys.* B699 (2004) 226–252, hep-th/0406015.
- [254] F. Delduc and E. Ivanov, "New Model of N=8 Superconformal Mechanics," *Phys.Lett.* B654 (2007) 200–205, 0706.2472.
- [255] V. Akulov and M. Kudinov, "Extended supersymmetric quantum mechanics," *Phys.Lett.* B460 (1999) 365–370, hep-th/9905070.
- [256] E. Ivanov, O. Lechtenfeld, and A. Sutulin, "Hierarchy of N=8 Mechanics Models," *Nucl.Phys.* B790 (2008) 493–523, 0705.3064.
- [257] A. Galperin, E. Ivanov, S. Kalitsyn, V. Ogievetsky, and E. Sokatchev, "Unconstrained Off-Shell N=3 Supersymmetric Yang-Mills Theory," *Class.Quant.Grav.* 2 (1985) 155.
- [258] A. Galperin, E. Ivanov, S. Kalitsyn, V. Ogievetsky, and E. Sokatchev, "Unconstrained N=2 Matter, Yang-Mills and Supergravity Theories in Harmonic Superspace," *Class.Quant.Grav.* 1 (1984) 469–498.
- [259] S. Gates, M. T. Grisaru, M. Rocek, and W. Siegel, "Superspace Or One Thousand and One Lessons in Supersymmetry," hep-th/0108200.

- [260] A. Galperin, E. Ivanov, and V. Ogievetsky, "Superspace Actions and Duality Transformations for N = 2 Tensor Multiplets," *Sov.J.Nucl.Phys.* **45** (1987) 157.
- [261] J. Gates, S. James and L. Rana, "Ultramultiplets: A New representation of rigid 2-d, N=8 supersymmetry," *Phys.Lett.* B342 (1995) 132–137, hep-th/9410150.
- [262] D.-E. Diaconescu and R. Entin, "A Nonrenormalization theorem for the d = 1, N=8 vector multiplet," *Phys.Rev.* D56 (1997) 8045–8052, hep-th/9706059.
- [263] J. Bagger, N. Lambert, S. Mukhi, and C. Papageorgakis, "Multiple Membranes in M-theory," *Phys.Rept.* 527 (2013) 1–100, 1203.3546.
- [264] M. A. Bandres, A. E. Lipstein, and J. H. Schwarz, "N = 8 Superconformal Chern-Simons Theories," *JHEP* 0805 (2008) 025, 0803.3242.
- [265] U. Gran, B. E. Nilsson, and C. Petersson, "On relating multiple M2 and D2-branes," JHEP 0810 (2008) 067, 0804.1784.
- [266] J. A. de Azcárraga and J. M. Izquierdo, "TOPICAL REVIEW: n-ary algebras: a review with applications," *Journal of Physics A Mathematical General* 43 (July, 2010) C3001, 1005.1028.
- [267] F. Passerini, "M2-Brane Superalgebra from Bagger-Lambert Theory," JHEP 0808 (2008) 062, 0806.0363.
- [268] K. Furuuchi, S.-Y. D. Shih, and T. Takimi, "M-Theory Superalgebra From Multiple Membranes," *JHEP* 0808 (2008) 072, 0806.4044.
- [269] E. Bergshoeff, J. Gomis, and P. K. Townsend, "M-brane intersections from world volume superalgebras," *Phys.Lett.* B421 (1998) 109–118, hep-th/9711043.
- [270] G. Papadopoulos and P. Townsend, "Intersecting M-branes," Phys.Lett. B380 (1996) 273–279, hep-th/9603087.
- [271] E. Bergshoeff, M. de Roo, E. Eyras, B. Janssen, and J. van der Schaar,
 "Intersections involving monopoles and waves in eleven-dimensions," *Class.Quant.Grav.* 14 (1997) 2757–2769, hep-th/9704120.
- [272] T. Banks, W. Fischler, S. Shenker, and L. Susskind, "M theory as a matrix model: A Conjecture," *Phys.Rev.* D55 (1997) 5112–5128, hep-th/9610043.

- [273] G. Papadopoulos, "M2-branes, 3-Lie Algebras and Plucker relations," *JHEP* **0805** (2008) 054, 0804.2662.
- [274] J. P. Gauntlett and J. B. Gutowski, "Constraining Maximally Supersymmetric Membrane Actions," JHEP 0806 (2008) 053, 0804.3078.
- [275] J. Distler, S. Mukhi, C. Papageorgakis, and M. Van Raamsdonk, "M2-branes on M-folds," JHEP 0805 (2008) 038, 0804.1256.
- [276] N. Lambert and C. Papageorgakis, "Relating U(N)xU(N) to SU(N)xSU(N) Chern-Simons Membrane theories," JHEP 1004 (2010) 104, 1001.4779.
- [277] M. Van Raamsdonk, "Comments on the Bagger-Lambert theory and multiple M2-branes," JHEP 0805 (2008) 105, 0803.3803.
- [278] N. Lambert and D. Tong, "Membranes on an Orbifold," *Phys.Rev.Lett.* 101 (2008) 041602, 0804.1114.
- [279] C. Montonen and D. I. Olive, "Magnetic Monopoles as Gauge Particles?," *Phys.Lett.* **B72** (1977) 117.
- [280] A. Gustavsson and S.-J. Rey, "Enhanced N=8 Supersymmetry of ABJM Theory on R**8 and R**8/Z(2)," 0906.3568.
- [281] O.-K. Kwon, P. Oh, and J. Sohn, "Notes on Supersymmetry Enhancement of ABJM Theory," *JHEP* **0908** (2009) 093, 0906.4333.
- [282] M. Benna, I. Klebanov, T. Klose, and M. Smedback, "Superconformal Chern-Simons Theories and AdS(4)/CFT(3) Correspondence," JHEP 0809 (2008) 072, 0806.1519.
- [283] D. Bashkirov and A. Kapustin, "Dualities between N = 8 superconformal field theories in three dimensions," *JHEP* 1105 (2011) 074, 1103.3548.
- [284] L. Alvarez-Gaume, G. W. Moore, and C. Vafa, "Theta Functions, Modular Invariance and Strings," *Commun.Math.Phys.* **106** (1986) 1–40.
- [285] B. de Wit and H. Nicolai, "Hidden Symmetry in d = 11 Supergravity," *Phys.Lett.* **B155** (1985) 47.
- [286] H. Nicolai, "D = 11 Supergravity With Local SO(16) Invariance," *Phys.Lett.* B187 (1987) 316.

- [287] B. de Wit and H. Nicolai, "d = 11 Supergravity With Local SU(8) Invariance," Nucl.Phys. B274 (1986) 363.
- [288] A. Almheiri, "Magnetic AdS2 × R2 at Weak and Strong Coupling," 1112.4820.
- [289] A. Donos, J. P. Gauntlett, and C. Pantelidou, "Magnetic and Electric AdS Solutions in String- and M-Theory," *Class.Quant.Grav.* 29 (2012) 194006, 1112.4195.
- [290] E. Witten, "Topological Quantum Field Theory," *Commun.Math.Phys.* **117** (1988) 353.
- [291] E. Witten, "An SU(2) Anomaly," Phys.Lett. B117 (1982) 324–328.
- [292] S. K. Donaldson and P. B. Kronheimer, *The geometry of four-manifolds*. Oxford Mathematical Monographs. The Clarendon Press Oxford University Press, New York, 1990. Oxford Science Publications.
- [293] A. Kapustin, "Holomorphic reduction of N=2 gauge theories, Wilson-'t Hooft operators, and S-duality," hep-th/0612119.
- [294] N. Seiberg and E. Witten, "Gauge dynamics and compactification to three-dimensions," hep-th/9607163.
- [295] A. Hanany and E. Witten, "Type IIB superstrings, BPS monopoles, and three-dimensional gauge dynamics," Nucl.Phys. B492 (1997) 152–190, hep-th/9611230.
- [296] J. de Boer, K. Hori, H. Ooguri, and Y. Oz, "Mirror symmetry in three-dimensional gauge theories, quivers and D-branes," *Nucl. Phys.* B493 (1997) 101–147, hep-th/9611063.
- [297] M. Porrati and A. Zaffaroni, "M theory origin of mirror symmetry in three-dimensional gauge theories," *Nucl.Phys.* B490 (1997) 107–120, hep-th/9611201.
- [298] G. Chalmers and A. Hanany, "Three-dimensional gauge theories and monopoles," Nucl.Phys. B489 (1997) 223–244, hep-th/9608105.
- [299] E. Witten, "Topology Changing Amplitudes in (2+1)-Dimensional Gravity," Nucl.Phys. B323 (1989) 113.

- [300] D. Birmingham, M. Blau, and G. Thompson, "GEOMETRY AND QUANTIZATION OF TOPOLOGICAL GAUGE THEORIES," *Int.J.Mod.Phys.* A5 (1990) 4721–4752.
- [301] D. Birmingham, M. Blau, M. Rakowski, and G. Thompson, "Topological field theory," *Phys.Rept.* **209** (1991) 129–340.
- [302] M. Blau and G. Thompson, "N=2 topological gauge theory, the Euler characteristic of moduli spaces, and the Casson invariant," *Commun.Math.Phys.* **152** (1993) 41–72, hep-th/9112012.
- [303] R. Slansky, "Group Theory for Unified Model Building," *Phys.Rept.* **79** (1981) 1–128.
- [304] D. Gaiotto and E. Witten, "Janus Configurations, Chern-Simons Couplings, And The theta-Angle in N=4 Super Yang-Mills Theory," JHEP 1006 (2010) 097, 0804.2907.
- [305] A. Kapustin and N. Saulina, "Chern-Simons-Rozansky-Witten topological field theory," *Nucl.Phys.* **B823** (2009) 403–427, 0904.1447.
- [306] E. Koh, S. Lee, and S. Lee, "Topological Chern-Simons Sigma Model," *JHEP* **0909** (2009) 122, 0907.1641.
- [307] L. Rozansky and E. Witten, "HyperKahler geometry and invariants of three manifolds," *Selecta Math.* **3** (1997) 401–458, hep-th/9612216.
- [308] K. Hosomichi, K.-M. Lee, S. Lee, S. Lee, and J. Park, "N=4 Superconformal Chern-Simons Theories with Hyper and Twisted Hyper Multiplets," *JHEP* 0807 (2008) 091, 0805.3662.
- [309] E. Witten, "Topological Sigma Models," Commun.Math.Phys. 118 (1988) 411.
- [310] E. Witten, "Two-dimensional gravity and intersection theory on moduli space," *Surveys Diff.Geom.* **1** (1991) 243–310.
- [311] E. Witten, "Mirror manifolds and topological field theory," hep-th/9112056.
- [312] K. Hori, S. Katz, A. Klemm, R. Pandharipande, R. Thomas, *et. al.*, "Mirror symmetry,".
- [313] R. C. McLean, "Deformations of calibrated submanifolds," *Comm. Anal. Geom.* 6 (1998), no. 4 705–747.

- [314] R. L. Bryant, "Metrics with exceptional holonomy," *Ann. of Math.* (2) **126** (1987), no. 3 525–576.
- [315] J. Dadok and F. R. Harvey, "Calibrations and spinors," *Acta Math.* **170** (1993), no. 1 83–120.
- [316] F. R. Harvey, *Spinors and calibrations*, vol. 9 of *Perspectives in Mathematics*. Academic Press, Inc., Boston, MA, 1990.
- [317] K. Becker, M. Becker, and A. Strominger, "Five-branes, membranes and nonperturbative string theory," Nucl.Phys. B456 (1995) 130–152, hep-th/9507158.
- [318] E. Bergshoeff, E. Sezgin, and P. Townsend, "Supermembranes and Eleven-Dimensional Supergravity," *Phys.Lett.* **B189** (1987) 75–78.
- [319] J. Gutowski, G. Papadopoulos, and P. Townsend, "Supersymmetry and generalized calibrations," *Phys.Rev.* D60 (1999) 106006, hep-th/9905156.
- [320] T. Kugo and P. K. Townsend, "Supersymmetry and the Division Algebras," *Nucl.Phys.* **B221** (1983) 357.
- [321] M. Bershadsky, A. Johansen, V. Sadov, and C. Vafa, "Topological reduction of 4-d SYM to 2-d sigma models," *Nucl.Phys.* B448 (1995) 166–186, hep-th/9501096.
- [322] H. M. Farkas and I. Kra, *Riemann surfaces*, vol. 71 of *Graduate Texts in Mathematics*. Springer-Verlag, New York, second ed., 1992.
- [323] L. Susskind, "The World as a hologram," J.Math.Phys. 36 (1995) 6377–6396, hep-th/9409089.
- [324] J. D. Brown and M. Henneaux, "Central Charges in the Canonical Realization of Asymptotic Symmetries: An Example from Three-Dimensional Gravity," *Commun.Math.Phys.* 104 (1986) 207–226.
- [325] A. Strominger, "Black hole entropy from near horizon microstates," JHEP 9802 (1998) 009, hep-th/9712251.
- [326] H. Saida and J. Soda, "Statistical entropy of BTZ black hole in higher curvature gravity," *Phys.Lett.* **B471** (2000) 358–366, gr-qc/9909061.
- [327] P. Kraus and F. Larsen, "Microscopic black hole entropy in theories with higher derivatives," *JHEP* **0509** (2005) 034, hep-th/0506176.

- [328] B. Sahoo and A. Sen, "BTZ black hole with Chern-Simons and higher derivative terms," *JHEP* 0607 (2006) 008, hep-th/0601228.
- [329] S. Nampuri, P. K. Tripathy, and S. P. Trivedi, "Duality Symmetry and the Cardy Limit," *JHEP* 0807 (2008) 072, 0711.4671.
- [330] R. M. Wald, "Black hole entropy is the Noether charge," *Phys.Rev.* D48 (1993) 3427–3431, gr-qc/9307038.
- [331] T. Jacobson, G. Kang, and R. C. Myers, "On black hole entropy," *Phys.Rev.* D49 (1994) 6587–6598, gr-qc/9312023.
- [332] V. Iyer and R. M. Wald, "Some properties of Noether charge and a proposal for dynamical black hole entropy," *Phys.Rev.* D50 (1994) 846–864, gr-qc/9403028.
- [333] T. Jacobson, G. Kang, and R. C. Myers, "Black hole entropy in higher curvature gravity," gr-qc/9502009.
- [334] J. M. Maldacena, J. Michelson, and A. Strominger, "Anti-de Sitter fragmentation," *JHEP* **9902** (1999) 011, hep-th/9812073.
- [335] J. Navarro-Salas and P. Navarro, "AdS(2) / CFT(1) correspondence and near extremal black hole entropy," Nucl.Phys. B579 (2000) 250–266, hep-th/9910076.
- [336] M. Caldarelli, G. Catelani, and L. Vanzo, "Action, Hamiltonian and CFT for 2-D black holes," JHEP 0010 (2000) 005, hep-th/0008058.
- [337] M. Cadoni, P. Carta, D. Klemm, and S. Mignemi, "AdS(2) gravity as conformally invariant mechanical system," *Phys.Rev.* D63 (2001) 125021, hep-th/0009185.
- [338] A. Giveon and A. Sever, "Strings in a 2-d extremal black hole," JHEP 0502 (2005) 065, hep-th/0412294.
- [339] T. Azeyanagi, T. Nishioka, and T. Takayanagi, "Near Extremal Black Hole Entropy as Entanglement Entropy via AdS(2)/CFT(1)," *Phys.Rev.* D77 (2008) 064005, 0710.2956.
- [340] R. K. Gupta and A. Sen, "Ads(3)/CFT(2) to Ads(2)/CFT(1)," *JHEP* **0904** (2009) 034, 0806.0053.

- [341] T. Hartman and A. Strominger, "Central Charge for AdS(2) Quantum Gravity," *JHEP* **0904** (2009) 026, 0803.3621.
- [342] A. Sen, "Quantum Entropy Function from AdS(2)/CFT(1) Correspondence," *Int.J.Mod.Phys.* A24 (2009) 4225–4244, 0809.3304.
- [343] A. Sen, "Entropy Function and AdS(2) / CFT(1) Correspondence," JHEP 0811 (2008) 075, 0805.0095.
- [344] A. Sen, "State Operator Correspondence and Entanglement in AdS_2/CFT_1 ," *Entropy* **13** (2011) 1305–1323, 1101.4254.
- [345] H. K. Kunduri, J. Lucietti, and H. S. Reall, "Near-horizon symmetries of extremal black holes," *Class.Quant.Grav.* **24** (2007) 4169–4190, 0705.4214.
- [346] P. Figueras, H. K. Kunduri, J. Lucietti, and M. Rangamani, "Extremal vacuum black holes in higher dimensions," *Phys.Rev.* D78 (2008) 044042, 0803.2998.
- [347] J. Bryan and N. C. Leung, "The enumerative geometry of *K*3 surfaces and modular forms," *J. Amer. Math. Soc.* **13** (2000), no. 2 371–410 (electronic).
- [348] L. F. Alday, D. Gaiotto, and Y. Tachikawa, "Liouville Correlation Functions from Four-dimensional Gauge Theories," *Lett.Math.Phys.* **91** (2010) 167–197, 0906.3219.
- [349] T. Dimofte, D. Gaiotto, and S. Gukov, "Gauge Theories Labelled by Three-Manifolds," *Commun.Math.Phys.* **325** (2014) 367–419, 1108.4389.
- [350] A. Gadde, S. Gukov, and P. Putrov, "Fivebranes and 4-manifolds," 1306.4320.